ERC/FEIS-0051





Filed: 10/188 HYDROELECTRIC DEVELOPMENT IN THE

UPPER OHIO RIVER BASIN FERC Docket No. EL85-19-114

Ohio, Pennsylvania, West Virginia FINAL

Environmental Impact Statement



Federal Energy Regulatory Commission Office of Hydropower Licensing

September 1988

FERC – FEIS

Hydroelectric Development In the Upper Ohio River Basin under:

....

EL85-19-114 P-7914-003 P-7909-002 P-4474-003 P-4017-002 P-7307-000 P-7399-000 P-8990-000 P-8654-001 P-7660-000 P-8908-000 P-4675-002 P-7041-001 P-7568-001 P-2971-002 P-3490-003 P-6901-001 P-10332-000 P-3218-001 P-6902-003 P-9999-000 P-6939-001 P-9042-000 P-10098-000 P-6998-0001

APPENDIX A

.

.

.

.

.

. ..

A REPORT OF A REPORT OF A REPORT OF A

LIST OF APPLICANT NAMES AND ADDRESSES

.

APPENDIX A

Project name <u>l</u> /	FERC No.	Applicant's name and address
Allegheny River L&D No. 7	7914-003	Allegheny Hydropower, Inc. 109 Union Street P.O. Box 45 Manchester, VT 05254
Allegheny River L&D No. 4	7909-002	County of Allegheny 119 Courthouse Grant Street Pittsburgh, PA 15219
Allegheny River L&D No. 3	4474-003	Borough of Cheswick 1410 Spruce Street Cheswick, PA 15024
		and
		Allegheny Valley North Council of Governments Springdale Borough Building 325 School Street Springdale, PA 14144
Allegheny River L&D No. 2	4017-002	The City of Pittsburgh 301 City-County Building Pittsburgh, PA 15219
Tygart Dam	7307-000	The City of Grafton City Building 1 West Main Street Grafton, WV 26354
Tygart Dam	7399-001	Noah Corporation 120 Calumet Court Aiken, SC 29801
Opekiska L&D	8990-000	Noah Corporation P.O. Drawer 640 Aiken, SC 29802
Hildebrand L&D	8654-001	Noah Corporation P.O. Drawer 640 Aiken, SC 29802
Point Marion L&D	7660-000	The Borough of Point Marion Point Marion Borough Building Point Marion, PA 15474
		and
		Noah Corporation 120 Calumet Court

•

Table A-1. Names and addresses of the applicants for 24 proposed hydropower projects in the Upper Ohio River Basin.

A-1

Aiken, SC 29801

Table A-1. (continued).

Project name	FERC No.	Applicant's name and address
Maxwell L&D	8908-000	The Borough of Brownsville Municipal Hall Brownsville, PA
		Washington County Board of Commissioners Courthouse Square 100 West Beau Street Washington, PA 15301
		and
		Pennsylvania Renewable Resources, Inc. Gulf and Western Building 15 Columbus Circle, Suite 906 New York, NY 10023
Monongahela L&D No. 4	4675-000	The Borough of Charleroi Municipal Hall Charleroi, Pennsylvania
		Washington County Board of Commissioners Courthouse Square 100 West Beau Street Washington, PA 15301
		and
		Pennsylvania Renewable Resources, Inc. Gulf and Western Building 15 Columbus Circle, Suite 906 New York, NY 10023
Emsworth L&D	7041-001	Potter Township 206 Mowry Road Monaca, PA 15061
Dashields L&D	7568-001	County of Allegheny 119 Courthouse Grant Street Pittsburgh, PA 15219
Montgomery L&D	2971-002	Allegheny Electric Cooperative. Inc. 212 Locust Street P.O. Box 1266 Harrisburg, PA 17108-1266
Montgomery	3490-003	Potter Township 206 Mowry Road Monaca, PA 15061
New Cumberland L&D	6901-001	The City of New Martinsville. 203 Main Street New Martinsville, WV 26155

A-2

Table A-1. (concluded).

Project name	FERC No.	Applicant's name and address
New Cumberland Hydroelectric Development	10332-000	WV Hydro, Inc. 120 Calumet Court Aiken, SC 29801
Pike Island	3218-001	The City of Orrville, Ohio Department of Public Works P.O. Box 126 Orrville, OH 44567
Willow Island L&D	6902-003	The City of New Martinsville 203 Main Street New Martinsville, WV 26155
Willow Island	9999-000	The City of St. Marys, St. Marys, WV 26170
Belleville	6939-001	The City of Jackson Memorial Building Broadway Street Jackson, OH 45640
Gallipolis L&D	9042-000	Gallia Hydro Partners c/o Mitex, Inc. 91 Newbury Street Boston, MA 02116
Gallipolis Development	10098-000	The City of Point Pleasant, West Virigina 400 Viand Street Point Pleasant, WV 25550
		and
		WV Hydro, Inc. 120 Calumet Court Aiken, SC 29801
Muskingum L&D No. 3	6998-001	The Upper Mississippi Water Company, Inc. c/o Mitex, Inc. 91 Newbury Street Boston, MA 02116

1/ L&D = Lock and Dam

APPENDIX B

WATER QUALITY IMPACTS MODELING METHODS AND DOCUMENTATION

ı.

.

8.1 INTRODUCTION

The water quality modeling conducted for the Ohio River Basin environmental impact statement is designed to provide information for the licensing decisions that FERC must make on proposed hydropower projects in the upper Ohio River Basin. These decisions include the acceptance or rejection of pending license applications, and the design or operation constraints to be placed on licensees. Several steps are involved in model development: (1) data acquisition to describe the system of interest, (2) formulation of an appropriate model to represent the system. (3) calibration of the model, (4) experimentation with the model, and (5) interpretation of model results for decisionmaking. Models are tools for problem solving, and the tool must fit the problem. We have attempted to keep our model building within the limitations of available data and the needs of the FERC licensing decisions.

The following sections discuss the conceptual formulations of the models used for impact assessment. The sources of data used for modeling and calibration, and the results of calibration are presented. A detailed description of the dissolved oxygen model is presented in Section B.4.

B.2 WATER QUALITY

B.2.1 COMPONENTS OF THE DO BUDGET

The dynamics of DO in large rivers can be represented by several sources and sinks: (1) bacterial respiration and biodegradation of carbonaceous and nitrogenous compounds in the water column and in the sediments, (2) algal respiration, (3) reaeration at the water surface, (4) primary production by algae, and (5) aeration of water spilled over dams.

8.2.1.1 Biochemical Oxygen Demand

Oxygen is removed from solution by respiration of microorganisms as they decay organic and nitrogen-containing materials. Higher concentrations of such materials, such as wastewater treatment plant effluents and ammonia, increase the populations of microorganisms and the rate at which they remove oxygen from the water. The biological oxidation (decay) of organic waste (measured as biochemical oxygen demand, or BOD) can be the major sink of DO where high BOD concentrations occur. Biological oxidation is typically modeled as a first-order decay process, as originally described by Streeter and Phelps (1925). The first-order model assumes the rate at which oxygen is removed from the water (milligrams per Liter per day) is equal to a constant (k_1) times the concentration of BOD. The Streeter-Phelps first-order model was used in this study.

BOD comes from both point sources such as wastewater discharges and tributaries and from nonpoint sources such as runoff, decay of algae, and decay of benthic matter. Because of the high flow rates of the study rivers, only a few large dischargers contribute enough BOD to directly increase concentrations significantly. The DO model includes BOD contributed from major wastewater dischargers where sufficient information was available to quantify the loading (sources of waste loading data are discussed in Section 2.3.2). Nitrogenous BOD (NOD) has not been modeled separately from carbonaceous BOD. NOD typically decays slower than carbonaceous BOD, but in a system as large and complex as the upper Ohio River basin, where the BOD concentration at any given point is the sum of contributions from a number of sources, the errors caused by using a single rate coefficient for all BOD are probably minor.

Benthic oxygen demand can be an important nonpoint source of BOD where organic matter accumulates in sediments (USEPA, 1985). Studies by the Corps Waterways Experiment Station in the lower Ohio River have shown that benthic oxygen demand can be a large fraction of the total BOD (personal communication, Mark Dortch, Corps Waterways Experiment Station, January 21, 1988). However, benthic oxygen demand rates are very difficult to measure or estimate, and little or no information is available for the upper Ohio River basin. No separate decay rate constant for benthic BOD was included in the model because inadequate information on benthic BOD rates exist. The model essentially assumes that benthic BOD and dissolved and suspended BOD can be modeled together using one rate coefficient and one concentration value.

Little information on other nonpoint sources of BOD is available. Calibration of the water quality model and comparison of measured instream BOD concentrations to point-source Toadings indicate that non point-source BOD is significant. Loadings of non point-source BOD were estimated by using them to calibrate the DO model; BOD loads were added at approximately

evenly spaced intervals along the rivers until the simulated oxygen demand was sufficient to make the modeled DO concentrations match measured ones (Section 8.2.4). The objective of the modeling study is to assess the impacts in changes in a DO source, dam aeration, and not to assess changes in any of of the BOD sources. Therefore, it is not crucial to the accuracy of the assessment to identify the individual sources of BOD.

B.2.1.2 Aeration at the Water Surface

Reaeration via gas transfer at the air-water interface (not at dams) is commonly modeled by assuming that the gas transfer rate is equal to a constant (k_2) times the DO deficit in the water. There are a number of equations to estimate k_2 (USEPA 1985, p. 102). For the Ohio River model, the model of O'Connor and Dobbins (1958) was selected to estimate k_2 because (1) it was originally verified using data from the Ohio River and, unlike most other models, is designed for use in deep channels; (2) it has been widely used and accepted; and (3) it was recommended by Ohio River Valley Water Sanitation Commission (ORSANCO) staff. The equation is:

$$k_2(20^{\circ}C) = 12.9 (U^{1/2}) / (H^{3/2}),$$

where k_2 is the reaeration rate coefficient at 20°C in units of days⁻¹. U is the average velocity in feet per second, and H is the depth in feet. The value of k_2 is then adjusted for the actual river temperature using the equation (USEPA 1985, p.125):

$$k_2(T) = k_2(20^{\circ}C) (1.024^{1+20})$$

B.2.1.3 Aeration at Dams

To model changes in BO in the basin caused by hydropower development, the aeration provided by each dam has been modeled. Each dam in the study has unique hydraulic characteristics that can affect aeration. There are two main types of navigation dams in the system, fixed-crest and gated dams.

The aeration capacity of the fixed-crest dams is probably highly influenced by the design of the apron at the base of the dam. There is generally good entrainment of air bubbles into the water as it crests the dam, and at some fixed-crest dams the water plunges deeply, which encourages dissolution of air from the bubbles. Other fixed-crest dams have a nearly horizontal apron that prevents a deep plunge, but some of these aprons have energy-dissipating structures that cause additional turbulence and aeration. There are 14 fixed-crest dams in the study area on the Allegheny, Monongahela, Muskingum, and Ohio rivers.

Several design characteristics influence aeration at the gated structures. Some of the gated dams have submerged outflows; flow beneath the opened gate exits below the surface of the downstream pool, so there is little entrainment of air as the water passes the dam. Other gated dams release water well above the downstream pool, so air is entrained and there is ample mixing and plunge for good aeration. Some of the gated dams also include small fixed-crest weirs that can discharge much of the total river flow at low flows. There are 15 gated dams in the study area on the Monongahela and Ohio rivers.

B.2.1.3.1 Development of a Statistical Model of Dam Aeration

The method used to model aeration at the navigation dams was to perform a statistical analysis of field data collected at each dam. Aeration data for each dam were provided by applicants for hydropower licenses and/or were available from historic data collected by the Pittsburgh District of the Corps (Section 2.3.5). At each dam, data were available from a number of measurements of the DO concentration above the dam (C_a) , the DO concentration below the dam (C_b) , the flow rate, and the temperature. These field data were used to determine a best-fit relation between the DO deficit (difference between the measured concentration and the saturation concentration) above the dam (D_a) and the deficit below the dam (D_b) , and to evaluate the influence of flow rate and temperature on aeration. The saturation concentration was used to correct table values for the elevation and water temperature of the field samples.

The field data for dam aeration are presented in Table B-1.

Theoretically, the rate at which oxygen is transferred from air into water is related to the DO deficit; the higher the DO deficit, the higher the rate of oxygen transfer into the

		A	legheny No	b. 7		
Temp <u>l</u> /	C <u>s2</u> /	C _a 3/	C <u>b4</u> /	D <u>a</u> 5/	D <u>b</u> 6/	Flow <u>7</u> /
20.00 19.10 19.10 19.40 19.60 19.80 17.30 17.30 17.00 24.9 25 23.7 21.8 25.4 27.3 27.9 26.6 24.3 24.1	8.82 8.98 8.99 8.89 8.89 9.32 9.32 9.32 9.32 9.32 8.1 8.1 8.1 8.1 8.7 8.1 7.8 7.7 7.9 8.2 8.3	9.26 9.35 9.30 9.21 9.25 9.38 10.25 10.00 9.99 8.1 7.5 8 8.3 7.3 7.8 7.4 7.6 8.1 8.9	9.23 9.41 9.30 9.21 9.29 9.93 9.75 9.81 7.8 7.5 7.9 8.2 7.8 7.6 7.7 8.4	0.44 0.37 0.32 0.28 0.36 0.52 0.93 0.68 0.61 0.0 0.6 0.3 0.4 0.8 -0.0 0.3 0.3 0.3 0.1 -0.6	0.41 0.43 0.32 0.28 0.33 0.43 0.43 0.43 0.43 0.43 0.43 0.43	18,900. 17,700. 17,700. 17,700. 17,700. 17,700. 16,500. 16,500. 16,500.
		A11	egheny No.	6		
Тетр	Cs	¢a	Cb	Da	Dþ	
24.90 24.80 24.00 22.10 25.00 27.30 27.10 25.60 23.30	8.03 8.04 8.17 8.47 8.01 7.68 7.71 7.78 8.28	8.10 7.00 7.00 8.10 6.70 7.40 7.00 6.70 8.20	8.10 7.60 7.60 8.30 6.90 7.60 7.00 7.00 8.20	0.07 1.04 1.17 0.37 1.31 0.28 0.71 1.08 0.08	0.07 0.44 0.57 0.17 1.11 0.08 0.71 0.78 0.08	
		ATT	egheny No.	5		
Temp	۲s	Ca	с _b	Da	Db	
24.5 24 24 24 23.5 23.5 18 17.5 17 17 17 17 17 17 24.8	8.2 8.3 8.3 8.3 8.4 9.5 9.6 9.6 9.6 9.6 9.6	8.7 8.4 8.5 8.5 7.4 8.2 9.3 9.2 9.2 9.2 9.2 8.5	8.1 9 8.6 8.3 7.8 7.2 9.1 9.2 9.1 9.5 9.2 9.5 9.5 9.5 9.5	-0.49 -0.11 -0.21 -0.21 -0.89 -0.43 2.17 0.08 0.58 0.58 0.38 0.58 0.38 0.38 0.38	0.11 -0.71 -0.31 -0.01 0.49 1.17 0.17 0.28 0.28 0.28 0.28 0.28 0.28 0.28 0.28	

Table B-1. Dam aeration data.

		Allegi	heny No. 5	(continued))	
Тетр	Cs	۲a	Сb	Da	Ðb	
24.6 24.3 22.2 25.6 27.1 27.5 27 23.3	8.2 8.6 8.0 7.8 7.7 7.8 8.4	7.3 7.3 8.3 6.5 7.2 6.7 6.7 8.1	7.4 7.5 8.4 6.9 7.5 7.1 6.5 8.1	0.89 0.94 0.29 1.53 0.60 1.04 1.11 0.34	0.79 0.74 0.19 1.13 0.30 0.64 1.31 0.34	
		FA	legheny No.	4		
Temp	۵s	C _a	с _b	Da	Db	
24.10 24.30 24.60 22.30 24.50 27.10 27.10 26.60 22.70	8.16 8.13 8.08 8.44 8.10 7.72 7.72 7.79 8.38	8.00 7.40 7.40 8.00 6.90 7.40 6.80 6.90 7.80	8.00 7.90 7.50 8.30 7.50 7.50 7.00 7.40 8.00	0.16 0.73 0.68 0.44 1.20 0.32 0.92 0.89 0.58	0.16 0.23 0.58 0.14 0.60 0.22 0.72 0.39 0.38	
	· · · · · · · · · · · · · · · · · · ·	Al	legheny Na.	3		
Temp	¢s	Ca	С _b	D _a '	Db	Flow
27.90 28.00 27.90 27.00 28.50 28.90 13.80 13.80 13.80 13.80 25.5 24.5 24.5 24.6 25.5 24.6 25.5 24.6 25.5 27.9 28 28.2 23.6	7.61 7.60 7.61 7.74 7.74 7.53 7.48 10.07 10.07 10.07 10.07 8.0 8.1 8.2 8.2 8.2 8.2 8.2 8.2 8.2 8.4	7.20 8.80 8.80 7.00 7.30 7.00 7.60 10.60 10.20 10.60 7.7 7.8 7.1 8.1 6.8 6.6 6.9 7 7.7	8.20 9.90 9.20 7.60 8.10 8.00 8.50 10.90 10.90 10.90 10.90 8.3 8.5 7.9 8.6 7.6 7.5 7.1 7.7 8.5	0.41 -1.20 -1.19 0.74 0.53 -0.12 -0.53 -0.13 -0.53 0.3 1.1 0.1 1.2 1.1 0.8 0.6 0.7	-0.59 -2.30 -1.59 0.14 -0.36 -0.47 -1.02 ~0.83 -0.83 -0.83 -0.83 -0.83 -0.3 -0.4 0.3 -0.4 0.3 -0.4 0.2 0.4 0.2 0.6 -0.1 ~0.1	7,000. 7,000. 7,000. 3,100. 3,100. 3,100. 3,100. 3,100. 3,100. 3,100. 3,100.

Table B-1. Dam aeration data (continued)

· · · ·

Allegheny No. 2								
Temp	Cs	Ca	с _b	Da	D _b	Flow		
16.10	9.58	7.80	8.90	1.78	0.68	6,500.		
25.30	7.98	7.30	8.40	0.68	-0.42	9,900.		
28.00	7.50	7.10	7.70	0.50	-0.10	6,600.		
20.00	7.00	7.80	8.00	-0.20	-0.40	5,580.		
27 61	7.02	7.00	7.98	0.10	-0.16	3.800.		
27.86	7.75	8 39	8 75	-0.15	-0.18	3,644. 3,688		
28.21	7.69	9.22	8,28	-1 54	-1.00	3,000.		
28.29	7.68	8.94	8.54	-1.29	-0.86	3,976.		
28.18	7.69	8.76	8.24	-1.11	-0.55	4.019		
28.20	7.69	8.47	8.07	-0.80	-0.38	4,063.		
28.10	7.71	8.13	7.78	-0,42	-0.07	3,107.		
21.50	8.75	8.83	10.19	-0.06	-1.44	11,900.		
21.46	8.75	8.90	10.41	-0.13	-1.65	11,663.		
21.41	8.77	8.88	10.31	-0.10	-1.54	11,425.		
21.34 91 EE	8.75 8 7A	8.61	10.10	0.16	-1.35	11,188.		
21.33	0./4	9.10	10.35	-0.30	-1.61	10,950.		
21.30	8.75 8.82	9.27 8.89	10.25	-0.48	-1.40	10,/13.		
20.95	8 85	9 19	10.12	-0.05	-1.50	10,475.		
20.85	8.87	9.21	10.29	-0.34	-1.20	10.230.		
20.79	8.87	8.72	9.35	0.12	-0.48	22,200		
20.52	8.92	8.58	9.38	0.35	-0.46	22,613.		
20.50	8.93	8.57	9.35	0.39	-0.42	23.025.		
20.44	8.94	8.84	10.14	0.12	-1.20	23,438.		
20.39	8.95	8.83	9.65	0.12	-0.70	23,850.		
20.37	8.95	8.98	9,57	-0.02	-0.62	24,263.		
20.23	8.98	9.04	9.64	-0.06	-0.66	24,675.		
21.13	0.52	8.09 9.53	9.40	0.12	-0.54	25,200.		
21.13	8 80	0-01	9.00	0.18	-0.88	24,725.		
21.58	8.74	8 35	9.20	0.43	-0.40	24,200.		
21.52	8.75	8 65	Q 42	0.35	-0.57	23,775.		
21.23	8.80	8.68	9 36	0 11	-0.56	22,300.		
21.20	8.81	8.74	9.34	0.07	-0.53	22.350.		
21.09	8.82	8.93	9.66	-0.13	-0,84	21,875.		
21.30	8.79	8.63	9.41	0.16	-0.62	21,400.		
	. - .		Opekiska					
Тетр	٤	Ca	Cb	Da	Db	Flow		
25.8	8.00	8.5	8.2	-0.50	-0.20	448		
25.8	8.00	9.2	9	-1.20	-1.00	675		
26.2	7.94	9.4	9.2	-1.46	-1.26	855		
25.4	8.05	8.1	7.9	-0.04	0.16	630		
20,5 25 5	7.98 P.or	1.5	7.4	0.48	0.58	490		
25.5	8 05	د.م د ع	0.2	1.75	1.85	467		
25.4	8.03 8.06	0.3 6 Q	0.4 7	1./3	1.05	440		
25.7	8.01	77	, 8 4	1.10	1.00	404 1 AEA		
25.3	8.08	7.7	7.6	0.31	-0.37 A 4R	1,050		
23.5	8.37	6	6.3	2.37	2.07	822		
23.3	8.40	7.6	8.1	0.80	0.30	730		
22.7	8.51	8.5	8.6	0.01	-0.09	1,040		

Table B-1. Dam aeration data (continued)

Opekiska (continued)								
Temp	C _s	Ca	Сђ	Da	Db	Flow		
22.7 22.7 22.1 22.2 21 20.5 17.8 14.3 15.1	B.51 8.51 8.61 8.59 8.81 8.90 9.42 10.18 9.99	8.8 8.6 8.2 9.2 7.3 7.5 8.3 8.4 8.9	8.4 9 9.3 7.9 8.5 9.3 8.8 9.6	-0.29 -0.09 0.41 -0.61 1.51 1.40 1.12 1.78 1.09	0.11 -0.49 -0.29 -0.71 0.91 0.40 0.12 1.38 0.39	1,040. 1,040. 998. 950. 899. 2,918. 1,417. 1,620. 744.		
			Hildebran		· , , , ,			
Temp	C _s	C _a	СЪ	Da	Db	Flow		
26.20 26.00 26.10 25.90 26.00 24.90 23.40 23.90 22.30 22.30 22.00 21.80 17.00 14.40 14.20	7.82 7.85 7.84 7.86 7.85 8.01 8.24 8.16 8.41 8.46 8.50 9.36 9.90 9.94	6.80 6.30 6.40 5.80 5.70 6.90 6.90 7.20 7.90 6.90 7.90 8.70 8.70 8.70 8.70	7.40 7.20 7.20 7.30 7.30 7.30 7.40 7.70 7.80 8.00 7.20 8.00 9.60 9.60 9.60 9.90	1.02 1.55 1.44 2.04 2.16 0.95 1.61 2.04 0.96 0.51 1.56 0.60 0.66 1.20 1.24	0.42 0.65 0.64 0.84 0.56 0.55 0.61 0.54 0.36 0.41 1.26 0.50 -0.24 0.30 0.04	478 535 468 432 465 700 836 717 1,071 1,071 1,030 970 4,974 2,242 1,230 909		
			Morgantow	'n				
Temp	¢ _s	Ca	¢b	D _a	Db	Flow		
24.20 24.10 24.00 24.00 24.00 24.00 24.00 24.20	8.12 8.14 8.15 8.15 8.15 8.15 8.15 8.12	6.40 6.30 6.70 6.50 7.00 7.10 6.80	7.10 7.00 7.20 7.00 7.50 7.40 7.60	1.72 1.84 1.45 1.65 1.15 1.05 1.32	1.02 1.14 0.95 1.15 0.65 0.75 0.52	700		
24.50 24.00 24.00 24.00 23.90 23.90 23.90 23.90 24.10 16.50	8.08 8.15 8.15 8.15 8.17 8.17 8.17 8.17 8.14 9.47	7.30 7.10 6.80 6.90 7.00 7.00 6.80 7.00 7.10 8.50	7.90 7.70 7.40 7.40 7.70 7.70 7.90 7.90 8.10 9.00	0.78 1.05 1.35 1.25 1.15 1.17 1.37 1.17 1.04 0.97	0.18 0.45 0.75 0.45 0.45 0.47 0.27 0.27 0.04 0.04 0.47	670 955 1,190 1,330		

Table B-1. Dam aeration data (continued)

		More	gantown (cor	tinued)		
Temp	¢ _s	Ca	¢۶	Da	Db	Flow
16.80 16.80 16.70 16.50 15.00 15.00 15.00 15.30 15.30 15.30 15.30 15.30	9.41 9.41 9.43 9.47 9.78 9.78 9.78 9.78 9.78 9.78 9.78 9.7	8.90 9.00 8.90 8.50 8.70 9.40 9.50 9.40 9.10 9.40 9.40 9.40	9.40 9.30 8.80 9.40 9.60 9.60 9.60 9.60 9.60 9.60 9.80 9.70 9.70	0.51 0.41 0.51 0.93 0.77 0.38 0.28 0.28 0.28 0.38 0.62 0.32 0.27 0.32	0.01 0.11 0.53 0.07 -0.02 0.18 0.18 0.18 0.18 0.12 -0.08 -0.03 0.02	1,300 1,240
			Point Mari	08		
Тетр	cs	Ca	с _b	D _a	Db	Flow
24.30 24.20 24.20 24.20 24.10 24.20 24.00 24.10 24.10 24.20 24.20 24.20 24.00 24.00 24.00	8.11 8.13 8.13 8.13 8.14 8.13 8.16 8.14 8.14 8.13 8.13 8.13 8.16 8.16 8.16	5.80 7.00 7.20 6.90 6.80 6.70 6.50 6.50 6.90 6.70 6.40 6.50 6.50 6.50 6.30	8.30 8.10 8.40 8.10 7.40 7.20 7.00 7.40 8.00 7.90 8.20 8.20 8.20 8.30 8.10	1.31 1.13 0.93 1.23 1.34 1.43 1.66 1.34 1.24 1.43 1.73 1.66 1.66 1.86	-0.19 0.03 -0.27 0.03 0.74 0.93 1.16 0.74 0.14 0.23 -0.07 -0.04 -0.14 0.06	640 690 1,190
24.00 24.00 21.30 21.20 21.00 21.00 21.00 20.80 20.70	8.16 8.16 8.59 8.61 8.64 8.64 8.64 8.68 8.68 8.70	6.30 6.00 8.20 8.30 8.30 8.10 7.60 7.50	8.20 8.10 9.00 9.20 9.20 9.20 9.20 9.00 8.00	1.86 2.16 0.59 0.41 0.34 0.34 0.54 1.08	-0.04 0.06 -0.31 -0.39 -0.56 -0.56 -0.36 0.68 0.70	1,530 1,100
21.00 14.90 15.00 15.00 15.00 14.90 14.90 14.70	8,64 9.81 9.79 9.79 9.79 9.81 9.81 9.85	7.90 9.40 9.60 9.50 9.90 9.60 9.60 9.60	8.30 10.20 10.20 10.50 10.50 10.50 10.20 10.90	0.74 0.41 0.19 0.29 -0.11 0.21 0.21 0.25	0.34 -0.39 -0.41 -0.41 -0.71 -0.69 -0.39 -1.05	1,100 1,070
14.70	9.80	9.70	11.00	0.15	-1.15	700

Table B-1. Dam aeration data (continued)

.

.

		Мо	nongahela N	o. 7		
Temp	Cs	Ca	Сb	D _a	0 _b	Flow
23.93 23.89 23.94 23.44 23.05 23.78 23.02 23.50 23.29 23.00 22.83 22.90 22.71	8.24 8.17 8.18 8.32 8.35 7.78 7.98 7.98 7.93 7.96 7.98 7.93 7.98	7.34 7.30 7.02 6.77 5.84 7.03 6.99 7.15 6.76 6.56 6.48 6.14 6.85	7.91 8.09 8.02 7.65 7.52 7.86 7.82 7.92 7.85 7.72 7.85 7.72 7.82 7.77 7.90	0.91 0.90 1.24 1.49 1.50 0.78 0.86 0.71 1.19 1.34 1.42 1.83 1.12	0.33 0.08 0.16 0.67 0.84 -0.08 0.16 -0.08 0.24 0.16 0.16 0.16 0.16 0.08	715 1,382 2,125 1,468 685 3,575 6,900 7,800 9,380 7,050 3,370 4,375 6,050
			Maxwell			
Temp	۵s	Ca	Съ	Da	Db	Flow
23.52 22.53 22.50 22.20 22.04 22.00 23.00 23.00 23.00 23.00 23.00 23.00 23.00 23.00 23.00 21.00 21.02 21.20 21.10 20.89 20.65	7.31 7.49 7.46 7.53 7.60 7.56 7.83 7.88 7.85 7.74 7.85 7.85 7.82 7.82 7.82 7.82 7.83 7.81 7.75 7.81 7.85	7.20 6.97 6.64 6.40 6.17 6.81 6.85 6.85 7.05 7.10 7.07 6.99 6.98 7.79 7.52 7.31 7.54 7.70 7.88	6.80 6.97 6.64 6.40 7.22 7.26 7.36 7.17 7.74 7.74 7.74 7.74 7.72 7.37 8.22 7.81 7.98 7.52 8.12 8.40	0.15 0.52 0.82 1.13 1.35 0.67 1.02 1.16 0.93 0.70 0.70 0.70 0.79 0.86 0.95 0.00 0.31 0.47 0.23 0.08 0.00	0.51 0.52 0.82 1.13 0.38 0.30 0.47 0.71 0.16 0.00 0.08 0.08 0.16 0.47 -0.39 0.00 -0.23 0.23 -0.31 -0.55	780 594 727 856 971 978 2,994 2,991 3,979 6,278 7,394 8,853 6,588 3,501 3,712 3,076 2,519 2,252 1,927 2,708
		Мо	nongahela N	0.4		
Temp	Cs	C _a	с _ь	0 _a	Db	
23.50 27.00 27.00 25.00 21.00 27.50 23.00 22.30 22.30 22.50 27.00	8.25 7.73 7.73 8.02 8.66 7.66 8.33 8.44 8.41 7.73	7.60 7.00 7.10 7.70 8.70 7.70 8.80 9.00 6.80 6.80	8.00 7.40 7.50 8.00 8.70 7.80 8.80 9.00 8.00 7.30	0.65 0.73 0.63 0.32 -0.04 -0.04 -0.47 -0.56 1.61 0.93	0.25 0.33 0.23 0.02 -0.04 -0.14 -0.47 -0.55 0.41 0.43	

•

Table 8-1. Dam aeration data (continued)

Monongahela No. 4 (continued)						
Temp	cs	C _a	с _b	Da	Db	
23.00	8.33	7.80	8.30	0.53	0.03	
24.30	8.13	7.30	7.80	0.83	0.33	
25.40	7.82	7.50	8.00	0.32	-0.18	
		Mo	nongahela N	lo. 3		
Temp	¢s	Ca	٤b	Da	Db	Flow
24.86	7.50	7.35	7.05	0.31	0.45	1,10
24.50	7.60	7.06	5.76	0.61	0.84	3,63
23.00	7.87	6.53	6.53	1.34	1.34	3,75
23.50	7.74	6.59	6.58	1.07	1.16	3,68
23.00	8.23	7.71	7.74	0.49	0.49	5,32
23.00	8.22	7.41	7.56	0.82	0.66	5,3
24.00	8.06	8.29	8.46	-0.24	-0.40	3,79
24.00	8.04	7.82	7.80	0.24	0.24	4,3)
23.00	8.25	7.96	7.92	0.25	0.33	5,59
23.00	8.25	7.78	8.00	0.41	0.25	3,92
21	8.81	7.50	7.80	1.31	1.01	
32	7.08	5.30	6.50	0.78	0.58	
20	7.97	8.80 8.50	8.10	-0.63	-0.13	
20.0	7.07 9.05	0.00	7.90	-0.03	-0.03	
2J.J 31 B	7 10	7 20	7.40	-0.10	0.03	
26 6	7 87	7.00	7 10	0.97	0.00	
29	7.51	7.90	7.70	-0.39	-0.19	
	·- <u>····</u>	Mc	onongahela t	lo. 2	··· · · ·	
Temp	Cs	Ca	с _b	Da	Db	Flow
24.14	7.66	6.76	7.51	1.01	0 15	3 7:
24.15	7.60	6.77	6.61	0.92	0.99	4.3
23.61	7.76	6.25	6.44	1.56	1.32	6.29
23.00	8.26	7.18	7.60	1.07	0.66	1,1
22.80	8.24	7.88	8.08	0.50	0.16	8,6
22.20	8.34	7.65	8.01	0.67	0.33	11,20
30	7.4	6	6.2	1.4	1.2	
26.2	7.9	7.1	7.3	0.8	0.6	
26	7.9	7.4	7.4	0.5	0.5	
24.3	8.3	7.1	7.8	1.2	0.5	
28.6	7.5	6.3	6.5	1.2	1	
25	8.1	7	7.2	1.1	0.9	
26.4	7.9	7.2	7.4	0.7	0.5	

Table B-1. Dam aeration data (continued).

			Emsworth			
Тетр	Cs	Ca	с _ь	Da	Db	Flow
13.10 13.17 12.84 13.24 13.24 13.39 13.39 13.39 13.45 12.44 12.49 12.45 12.54 12.54 12.54 12.54 12.54 12.54 12.33 12.34 9.37 9.39 9.37 9.45 9.60 9.58 7.7 25.4 25.7	10.24 10.22 10.30 10.06 10.08 10.17 10.16 10.39 10.39 10.37 10.37 10.37 10.37 10.37 10.37 10.42 10.41 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.15 11.16 11.16 11.15 11.16 11.10 11.10 11.10 11.10 11.10 11.10 11.10 11.00 10.000	$10.26 \\ 10.27 \\ 12.24 \\ 11.80 \\ 11.64 \\ 9.58 \\ 11.37 \\ 10.86 \\ 10.72 \\ 10.74 \\ 10.67 \\ 10.95 \\ 10.93 \\ 10.68 \\ 10.06 \\ 9.87 \\ 10.59 \\ 10.50 \\ 10.51 \\ 9.87 \\ 10.59 \\ 10.50 \\ 10.51 \\ 8.4 \\ 11.4 \\ 9.1 \\ 6.8 \\ 6.8 \\ 7.5 \\ 7.0 \\ 7.9 \\ 6.8 \\ 7.7 \\ 7.$	10.51 10.76 13.44 11.12 10.98 8.71 11.17 10.81 10.70 10.69 10.96 10.96 10.96 10.96 10.96 10.96 10.96 10.69 10.05 10.32 10.78 10.80 10.77 10.39 10.35 10.25 10.18 10.08 10.77 10.39 10.35 10.25 10.18 10.08 10.77 10.39 10.35 10.25 10.18 10.08 10.77 10.39 10.35 10.25 10.18 10.08 10.77 10.39 10.35 10.25 10.18 10.08 10.77 10.39 10.35 10.25 10.18 10.70 10.77 10.39 10.35 10.25 10.18 10.77 10.39 10.35 10.25 10.18 10.77 10.39 10.35 10.78 10.77 10.39 10.35 10.78 10.77 10.39 10.35 10.78 10.77 10.39 10.35 10.25 10.78 10.77 10.70 10.78 10.77 10.70 10.69 10.78 10.77 10.78 10.77 10.78 10.77 10.78 10.77 10.78 10.77 10.78 10.77 10.39 10.35 10.78 10.70 10.78 10.77 10.39 10.35 10.25 10.78 10.78 10.78 10.77 10.39 10.78 7.6 7.8 7.8 7.8 7.8 7.8 8.7 8.1	-0.02 -0.05 -1.94 -1.60 -1.58 0.50 -1.20 -0.70 -0.33 -0.36 -0.28 -0.58 -0.58 -0.56 -0.29 0.36 0.54 0.57 0.65 1.05 1.30 1.33 1.48 1.49 1.61 -0.0 1.7 0.9 -0.1 0.3 1.5 1.0 0.7 1.1 0.3 0.7	$\begin{array}{c} -0.27\\ -0.54\\ -3.14\\ -0.92\\ -0.92\\ 1.37\\ -1.00\\ -0.65\\ -0.32\\ -0.30\\ -0.59\\ -0.30\\ -0.59\\ -0.59\\ -0.30\\ 0.37\\ 0.09\\ 0.38\\ 0.35\\ 0.39\\ 0.75\\ 0.35\\ 0.39\\ 0.75\\ 0.75\\ 0.88\\ 0.6\\ -0.1\\ 0.2\\ 0.6\\ 1.3\\ 0.0\\ -0.1\\ -0.6\\ -0.1\\ -0.6\\ -0.1\\ \end{array}$	33,700 33,700 33,700 33,700 31,200 31,200 25,300 25,300 25,300 25,300 25,300 25,300 25,300 25,300 13,800 13,800 13,800 13,800 13,800 13,800 13,800
			Dashields	i		
Temp	Cs	Ca	с _р	D _a	Db	
18.00 26.30 19.00 9.40 18.00 23.30 26.80 24.80 25.50 25.50 27.50 25.60	9.22 7.85 9.03 11.16 9.22 8.30 7.78 8.07 7.96 8.04 7.68 7.93 7.95	9.10 5.70 8.00 11.40 9.10 6.90 7.10 7.30 6.80 7.60 7.30 8.50 7.90	9.60 7.30 8.70 12,00 9.70 8.20 8.40 7.65 7.30 8.10 8.60 9.30 8.60	0.12 2.15 1.03 -0.24 0.12 1.40 0.68 0.77 1.16 0.44 0.38 -0.57 0.05	-0.38 0.55 0.33 -0.84 -0.48 0.10 -0.62 0.42 0.66 -0.06 -0.92 -1.37 -0.65	

Table B-1. Dam aeration data (continued)

			Montgomer	y		<u>u</u> _
Temp	۲ ²	Ca	¢۵	Da	Db	Flow
27.50 28.00	7.68 7.61	7.90 9.80	8.10 9.70	-0,22 -2,19	-0.42 -2.09	5,270 5,990
27.70	7.65 7.64	9.50	8.40 8.50	-1.85 -0.86	-0.75	5,250
27.30	7.71	8.30	8.30	-0.53	-0.63	6,000 5,990
27.30	7.67	8.10 8,50	8.40 8.60	-0.39 -0.83	-0.69 -0.93	5,190 5,990
24.00 23.80	8.19 8.22	7.70 7.60	9,10 9,20	0.49 0.62	-D.91 -0.98	11,370 11,470
23.80 23.50	8.22 8.27	7.70 7.70	8.90 8.70	0.52 0.57	-0.68 -0.43	10,040
23.50	8.27	8,00 8,10	9.20	0.27	-0.93	10,400
23.50	8.27	7.70	9.20	0.57	-0.93	10,500
19.05	9.03	9.84	8.71	-0.81	0.32	32,800
18.55 18.65	9.12 9.10	9.83 9.86	8.91 9.58	-0.71 -0.76	0.21 -0.48	32,800 32,800
18.64 18.75	9.10 9.08	9.78 9.85	9.64 9.71	-0.68 -0.77	-0.54 -0.63	32,800 32,800
18.57	9.11	9.61	9.42	-0.50	-0.31	32,800
18.46	9.13	8.62	8.70	0.51	0.43	26,500
9.30	11.19	10.55	10.89	0.63	0.33	10,100
9.35 9.38	$11.17 \\ 11.17$	10.60 10.09	10.89 10.67	0.57 1.08	0,28 0,50	8,840 8,840
9.41 9.39	11.16 11.16	9.89 9.81	10.36 10.20	1.27 1.35	0.80 0.96	8,840 10,100
9.35 9.33	11.17 11.18	10.02 9.89	10.32	1.15 1.29	0.85 1.09	10,100 10,100
		<u> </u>	New Cumber!	and	<u></u>	
Temp	¢ _s	C _a	Сb	Da	Db	Flow
25.90	7.91	7.93	8.32	-0.02	-0.41	6,000
25.60 25.90	7.96 7.91	8.04 7.98	8.51 8.54	-0.08 -0.07	-0.55 -0.63	7,500 6,000
26.20 26.70	7.87 7.80	7.89 7.42	8.31 8.00	-0.02 0.38	-0,44 -0,20	3,000 4,500
27.70	7.65	7.01	7.96	0.65	-0.30	6,000 14,000
25.80	7.93	6.81	7.94	1.12	-0.01	9,000
25.10	8.03	6.95	8.27	1.08	-0.24	7,500
24.30	7.70 8.15	7.41 7.64	8.63	0.29	-0.93	6.000 7,500
24.60 24.70	8.11 8.09	7.97 7.90	8.65 8.66	0.14 0.19	-0.54 -0.57	4,500 4,500
24.20 24.70	8.17 8.09	8.71 8.76	8.76 8.82	-0,54 -0,67	-0.59 -0.73	4,500
25.40	7.99	8.71	8.74	-0.72	-0.75	4,500

Table B-1. Dam aeration data (continued)

B-	1	2	
Ð-	1	2	

		New Cu	umberland (continued)		
Temp	Cs	Ca	с _р	Da	Db	Flow
24.30	8.15	8.86	9.01	-0.71	-0.86	13,000
23.50	8.28	8.85	9.18	-0.57	-0.90	7,500
23.80	8.23	8.85	9.01	-0.62	-0.78	9,000
25.40	7.99	8.40	8.65	-0.41	-0.66	12,000
25.50	/.9/	8.10	8.50	-0.13	-0.53	17,000
20.00	7.90	8.35	8.70	-0.45	-0.80	14,000
20.50	7.83	8.40	8.70	~0.5/	-0.8/	15,500
25.50	7.97	0.0U 0.10	0.30	-0.53	-0.33	7,500
27 20	7 72	0.10 9.30	8.50 9.50	-0.08 -0.57	-0.77	9,000
27.20	7.53	8 40	8.50	-0.37	-0.77	7 500
	7.03	0.40	0,40	-0.77	-0.//	/1000
			Pike Isla	nđ		
Temp	۲s	Ca	СЪ	Da	Db	
20.00	8.87	8.70	9.00	0.17	-0.13	
28.00	7.62	6.10	6.20	1.52	1.42	
20.40	8.80	8.10	8.40	D.70	0.40	
13.10	10.26	9.80	9.90	0.46	0.36	
17.80	9.27	9.00	9.60	0.27	-0.33	
24.50	8.13	6.50	7.80	1.63	0.33	
23.30	8.31	6.60	7.40	1.71	0.91	
27.70	7.67	6.70	7.40	0.97	0.27	
25.80	7.93	7.90	8.40	0.03	-0.47	
27.80	7.65	6.80	6.80	0.85	0.85	
26.10	7.89	7.60	7.80	0.29	0.09	
29.10	7.48	7.50	7.70	-0.02	-0.22	
20.80	1.19	7.80	8.00	-0.01	-9.21	
27.40	1.11	00.8	8.00	-0.89	-0.69	
			Hannibal			
Temp	¢۶	Ca	с _b	Da	Db	Flow
27.4	7.75	8.19	8.11	-0.44	-0.36	14,900
27.2	7.78	7.89	8.82	-0.11	-1.04	9,400
27.2	7.78	8.14	8.14	-0.36	-0.36	5,700
26.9	7.83	7.89	7.96	-0.06	-0.13	5,700
26.7	7.86	7,51	8.26	0.35	-0.40	9,400
26.3	7.92	7.82	7.B	0.10	0.12	5,700
26.4	7.90	7.62	7.75	0.28	0.15	5,700
27	7.81	6.64	7.06	1.17	0.75	7,400
27.2	7.78	6.47	6.9	1.31	0.88	7,400
27.5	7.74	6.63	5.99	1.11	0.75	7,400
27.9	7.68	6.81	6.94	0.87	0.74	11,400
28	7.66	6.62	6.83	1.04	0.83	18,800
28	/.00	1.21	/.33	U.45	U.33	11,400
28	/.00	0.72	0.98	U.94 1 #F	U.DO 1 16	1,400
21.9	7.08	0.23	0.52	1.45	01.1	15,150
27.4	1.10	0.3/ 6 10	1.31	1.30	1 22	5,400 7 ANN
2/-1	7.20	0.19 5 5	0.50 6 67	1.01	1.44	12 150
L 1	1.01	0.0	0.0/	1.21	1.14	12,130

Table B-1. Dam aeration data (continued).

		Han	nibal (cont	inued)		
Temp	C _S	Ca	¢b	Da	D _b	Flow
27 27.1 27 26.7 26.6 26.4	7.81 7.80 7.81 7.86 7.87 7.90	6.71 6.82 7.17 7.32 7.25 7.18	6.86 7.13 7.35 8.48 8.02 7.55	1.10 0.98 0.64 0.54 0.62 0.72	0.95 0.67 0.46 -0.62 -0.15 0.35	7,400 18,800 14,900 9,400 9,400 7,400
···	<u></u>		Willow Isla	ind	<u>, , , , , , , , , , , , , , , , , , , </u>	
Temp	cs	Ca	с _b	Da	Db	Flow
24.50 24.00 24.50 25.00 26.00 26.00 26.50 26.50 26.50 26.00 25.50 26.00	8.14 8.22 8.14 8.06 7.92 7.92 7.77 7.85 7.85 7.92 8.06 7.99 7.92 7.92 7.92 7.92 7.92 7.92 7.92	9.00 9.00 8.70 9.00 8.25 8.00 7.35 7.30 7.35 7.30 7.65 6.80 7.05 7.40 7.80 8.00 8.00 8.20 8.30 8.20 8.30 8.20 8.15 7.90 7.70 8.10	9.10 9.40 9.20 9.00 8.40 7.40 7.40 7.70 7.10 7.30 7.70 8.00 8.20 8.05 8.30 8.50 8.30 8.50 8.40 8.30 8.50 8.40 8.30 8.10 8.10 8.20	-0.86 -0.78 -0.56 -0.94 -0.33 -0.08 0.42 0.55 0.20 1.12 1.01 0.59 0.12 -0.08 -0.28 -0.28 -0.28 -0.28 -0.23 -0.23 -0.15 -0.33	-0.96 -1.18 -1.06 -0.94 -0.48 -0.08 -0.37 -0.45 -0.15 -0.45 -0.15 -0.29 -0.08 -0.28 -0.28 -0.28 -0.13 -0.38 -0.48 -0.38 -0.13 -0.25 -0.43	10,500 9,060 10,500 14,700 13,300 47,500 68,000 92,000 74,300 45,300 62,400 38,300 51,800 24,400 17,800 22,900 14,700 17,700 13,300 10,500 9,000
<u> </u>	·····	<u> </u>	Bellevill	e	<u> </u>	- <u></u>
Temp	۲s	c _a	с _ь	Da	Db	Flow
29.00 29.00 29.45 29.50 29.73 29.78 28.84 28.85 27.47 26.43 25.49 26.56 26.46 26.61	7.51 7.51 7.45 7.44 7.41 7.53 7.53 7.53 7.53 7.71 7.86 7.85 7.84 7.85 7.84 7.86	6.57 6.55 7.12 7.61 7.01 6.60 6.61 5.58 5.77 5.80 5.98 6.67 6.54 5.93	5.78 7.04 7.31 6.91 7.02 6.82 6.85 6.85 5.92 6.11 5.87 6.43 6.14 6.04	0.94 0.96 0.33 -0.17 0.40 0.81 0.92 0.95 1.94 2.06 1.87 1.17 1.32 1.91	0.73 0.47 0.14 0.53 0.39 0.59 0.68 0.68 1.79 1.75 1.98 1.41 1.72 1.80	13,500 13,500 13,500 13,500 13,500 13,500 13,500 13,500 12,000 15,000 18,000 18,000 18,000 18,000 18,000

Table B-1. Dam aeration data (continued)

B-	14	
----	----	--

		Bel	leville (co	ntinued)		
Temp	۲s	¢ã	С _b	Da	٥ _b	Flow
26.74	7.82	5.83	6.36	1.99	1.46	18.000
16.20	9.61	9.22	9.58	0.39	0.03	31,500
16.60	9.53	9.11	9.23	0.42	0.30	29,750
17.00	9.45	8.80	8.32	0.65	1.13	29,750
16.40	9.57	8.32	9.17	1.25	0.40	29,750
16.40	9.57	9.41	9,93	0.16	-0.36	29,750
16.00	9.65	9.34	9.50	0.31	0.15	29.750
15.50	9.76	9.27	5.41	0.49	3.35	29,750
			Gallipoli	s		
Temp	¢s	Ca	сÞ	D _a	Db	Flow
20.00		7 20	7 40	1 70	1 50	01 500
18.50	9 18	6 80	7.40	2 28	1.00	91,900
23.00	8.39	5 60	7 20	2,30	1.50	120 000
23.50	8 31	6 70	6 40	1.61	1.17	130,900
25 00	8 08	7 30	7 50	0,01	1.91	30,700
25 00	8 08	7.50	7.50	0.76	U.58	35,400
26 00	7 04	7.50	7.90	0.38	0.16	4/,400
22 00	9 56	9 00	7.50	0.44	0.14	73,500
25 00	0.00	0.90 7 E0	8.00	-0.34	-0.04	/4,100
25.00	0.VC 7 D4	7.50	7.80	0.58	0.28	/2,500
20.00	1.34	7.40	7.60	U.54	U.34	55,400
23.00	8.08	7.90	1.70	0.18	0.38	62,200
24.00	8.24	7.30	7.60	0.94	0.64	52,800
24.00	8.24	/.10	7.30	1.14	0.94	51,900
27.00	1.19	5.80	7.20	1.99	0.59	66,800
27.00	7.79	6.50	7.20	1.29	0.59	69,500
27.00	1.79	7.10	7.30	0.69	0.49	67,200
27.00	7.79	7.00	7.10	0.79	0.69	64,800
28.00	7.65	7.60	7.90	0.05	-0.25	59,500
28.00	7.65	7.40	8.00	0.25	-0.35	50,100
27.50	7.72	7.70	7.10	0.02	0.62	41,900
28.00	7.65	7.20	7.00	0.45	0.65	39,100
28.00	7.65	7.70	00.8	-0.05	-0.35	37,800
28.00	7.65	7.70	8.00	-0.05	-0.35	27,200
29.00	7.52	7.90	8.20	-0.38	-0.68	24,400
29.00	7.52	8.80	9.20	-1.28	-1.68	20,400
28.00	7.65	9.10	8.10	-1.45	-0.45	20,000
28.00	7.65	8.20	10.00	~0.55	-2.35	22,200
28.00	7.65	8.00	8.00	-0.35	-0.35	24,200
28.00	7.65	7.30	7.50	0.35	0.15	31,700
27.00	7.79	7.40	7.70	0.39	0.09	25.600
27.50	7.72	7.30	7.50	0.42	0.22	28.300
28.00	7.65	7.90	7.90	-0.25	-0.25	23,200
28.00	7.65	7.60	7.70	0.05	-0.05	16.300
28.00	7.65	5.80	6.80	0.85	0.85	20,200
27.50	7.72	6.60	6,70	1.12	1.02	16.700
27.50	7.72	6.50	6.70	1.22	1.02	20,800
28.00	7.65	6.50	6,40	1.15	1.25	19,400
27.50	7.72	6.70	6.70	1.02	1.02	17.600
27.00	7.79	6.40	6.40	1.39	1.39	25.500
25.50	7.86	5,60	6.90	1,26	0.96	20,800
26.00	7.94	6,60	6.70	1.34	1.24	16,400

Table B-1. Dam aeration data (continued).

		Gall	ipolis (con	tinued)		
Temp	¢ _s	C _a	с _ь	Da	Db	Flow
25.00	7.94	6.50	7.00	1.44	0.94	14,200
26.00	7.94	6.40	6.80	1.54	1.14	13,400
26.00	7.94	6.50	6.30	1.44	1.64	16,000
26.00	7.94	6.40	6.50	1.54	1.44	22,500
26.00	/.94	6.50	6.50	1.44	1.44	30,200
24.50 24.00	8.10 8.24	7.10	6 90	1.00	1.00	12 200
24 00	8 24	7.20	7.40	1.04	0.84	17 200
24.00	8.24	6.90	7.10	1.34	1.14	20,100
24.50	8.16	7.20	7.40	0.96	0.76	18,000
24.00	8.24	7.30	7.50	0.94	0.74	15,600
24.00	8.24	6.80	7.30	1.44	0.94	14,200
24.00	8.24	7.10	7.40	1.14	0.84	13,800
24.00	0.24	7.10	7.40	1.14	0.89	15,900
24.00	8.24	7.20	7.30	1 04	0.94	14 500
24.00	8.24	7.10	7.20	1.14	1.04	15,600
23.50	8.31	6.60	7.30	1.71	1.01	18,000
25.00	8.08	6.60	7.20	1.48	0.88	35,000
25.00	8.08	6.70	6.80	1.38	1.28	29,200
23.00	8.39	5.60	6.80	1.79	1.59	119,400
19.00	9.08	8.00	8.10	1.08	0.98	56,100
17.00	9 46	9 30	9 40	0.70	0.76	29 700
<u> </u>	<u></u>		······································			
7	<i>c</i>	Mus	kingum No.	3	-	-
lemp	رج	رع 	ь			WOI1
29.30	7.47	10.30	7.40	-2.83	0.07	1,400
29.40	7.45	6.98	6.90	0.47	0.55	1,700
29.00 20 ED	7.51	5.8U P.50	7.30	U./1	0.21	1,700
29.00	7.44	7 50	7.00	-1.00	-0.00	1,700
29.50	7.44	5.70	5.90	1.74	1.54	1,700
29,30	7.47	5.30	6.50	2.17	0.97	1,900
26.50	7.85	6.90	7.70	0.95	0.15	2,800
27.00	7.78	6.60	7.60	1.18	0.18	2,800
26.80	7.81	6.60	7.10	1.21	0.71	2,600
27.10 25 60	/./b 7 eo	D./U 5 PO	1.00	1.05	U.10	2,600
26.00	7.03	6 30	7 30	1.05	1.53 A 62	2,000
26.40	7.86	5.70	7,30	2,16	0.56	2,300
25.70	7.96	9.80	8.40	-1.84	-0.44	1.800
24.80	8.10	7.30	7.50	0.80	0.60	2,100
26.30	7.88	11.00	8.80	-3.12	-0.92	2,100
26.40	7.86	12.10	8.60	-4.24	-0.74	2,100
26.50	7.85	7.60	8.20	0.25	-0.35	1,800
(3,80 26,20	7.95	1.30	9.10 5 10	0.05	~1.15	1,800
26 50	7.05	Q 30	6 40	-2.01	1.79	1,000
26.00	7.92	5,80	7.90	2.12	0.02	1.560
26.80	7.81	7.90	6.00	-0.09	1.81	1,560
26.70	7.82	9.10	8.10	-1.28	-0.28	1,560
26.50	7.85	7.70	6.90	0.15	0.95	1,360

٠

Table B-1. Dam aeration data (continued).

•

		Muskin	gum No. 3 (continued)		
Temp	¢s	Ca	с _b	Da	Db	Flow
26.80 27.00	7.81 7.78	7.60 7.20	8.00 6.80	0.21 0.58	-0.19 0.98	1,360 1,360
		Mu	iskingum No.	. 2		· · · · · · · · · · · · · · · · · · ·
Тетр	Cs	Ca	с _ь	Da	Db	Flow
26.00 26.00 26.00 26.10 26.20 26.20 26.50 24.30 23.90 23.80 24.00 23.90 23.80 24.00 24.00 24.00 24.00 23.40 23.40 23.40 23.20 22.50 22.50 22.50 22.00 22.00 22.00 22.00 22.10	7.92 7.92 7.92 7.92 7.90 7.89 7.89 7.85 8.17 8.23 8.25 8.25 8.22 8.22 8.22 8.22 8.22 8.22	7.50 7.20 7.00 7.80 8.70 6.90 6.70 6.60 8.60 7.80 7.30 6.80 7.30 6.80 7.20 6.80 6.50 6.70 7.40 6.90 6.80 7.70 8.10 8.90 7.70 6.30 6.50 6.40 6.80 6.80 6.80	7.00 7.20 7.20 7.20 7.20 6.30 6.60 6.60 8.50 7.60 7.20 6.70 6.70 7.30 7.30 7.30 7.30 7.30 7.00 7.70 7.50 7.40 8.10 8.20 8.70 8.20 8.70 8.20 7.30 7.30 7.30 7.30 7.30 7.40 8.10 8.20 7.20 7.20 7.20 7.20 7.20 7.20 7.20 7	0.42 0.72 0.92 0.10 -0.81 1.02 1.19 1.25 -0.43 0.43 0.95 1.45 1.62 1.03 1.03 1.42 1.72 1.52 0.91 1.47 1.54 0.75 0.35 -0.41 0.79 2.15 1.94 2.14 1.74 1.72	0.92 0.72 0.72 0.70 0.69 1.62 1.29 1.25 -0.33 0.63 1.05 1.55 1.55 1.55 1.52 0.93 0.93 1.32 1.22 1.22 0.61 0.87 0.94 0.35 0.25 -0.21 0.49 1.75 1.24 1.44 0.74 0.82	2,391 2,391 2,391 2,391 2,272 2,272 2,272 1,350 1,350 1,425 1,944 1,878 1,878 1,878 1,878 1,610 1,500 1,500 1,500 1,500 1,500 1,500 1,500 1,500

Table B-1. Dam aeration data (concluded)

1/ Water temperature, degrees Celsius.

2/ Oxygen saturation concentration, mg/L.

3/ Dissolved oxygen concentration above the dam, mg/L.

4/ Dissolved oxygen concentration below the dam, mg/L.

5/ Dissolved oxygen deficit above the dam, mg/L.

 $\underline{6}$ / Dissolved oxygen deficit below the dam, mg/L.

<u>7</u>/ River flow rate, cfs.

water is expected to be. Linear regression was used to determine the relation between D_a and D_b for each dam. The value of D_a was plotted against D_b for each pair of field-measured values; the data generally fell near straight lines, although scatter occurs, which is expected because of field measurement errors, errors in estimated DO saturation concentrations, and small variations in head (the distance the water falls at a dam) that normally occur.

Graphs of D_b vs D_a at some dams appeared to approximate straight lines that do not meet the common assumption that when D_a is zero, then D_b is zero (in other words, that no aeration occurs when the DO above the dam is at saturation). The dam aeration model is therefore:

 $D_b = M D_a + b$,

where M is referred to as the dam aeration coefficient and b is the dam aeration constant. Figure B-l is an example aeration graph where b is zero, and Figures B-2 and B-3 are example plots of D_b vs. D_a at dams where the assumption that when D_a is zero then D_b is zero does not appear to be true.

There are three explanations that could account for the apparent aeration $(D_b$ less than zero) when the above-dam deficit (D_a) is zero: field measurement errors, erroneous estimates of the saturation concentration, and the occurrence of supersaturation. Errors in field measurements of DO concentration are not believed to be the only cause of nonzero values of b, because the value of b was consistently negative at dams where the statistical fit of the aeration equation was good.

The actual saturation concentration (C_S) in the river may be significantly different from the literature values used in the analyses. This error would cause there to actually be a nonnegative value of D_A when the data show that D_A is zero. Variability in actual C_S values could be documented by measuring C_S in the field as Butts and Adkins (1987) did. Some field measurements of the DO saturation concentration were made by applicants, and significant variation from book values sometimes occurred. However, because of the few data collected and the unknown accuracy of the data, the results are not conclusive.

The third explanation for apparent mass transfer when D_a is zero is that supersaturation occurs. According to Henry's law, the value of C_s in water in equilibrium with air is proportional to the air pressure. Because air bubbles plunged beneath the surface below a dam are under higher than atmospheric pressure (the pressure on bubbles is twice atmospheric at a depth of approximately 32 feet), the local value of C_s in water surrounding submerged bubbles would be higher than C_s at the surface of the water column. Theoretically, supersaturation could occur if bubbles are submerged for a long enough time for sufficient mass transfer to occur. Measurement of supersaturation below navigation dams was not investigated as part of this study. However, a comparison between those dams where visual observations indicated that deep plunging of bubbles occurred and those dams where the linear regression model indicated that significant aeration occurred when D_a was zero indicates that aeration when D_a is zero is more likely to occur at dams where deep plunging occurs.

The linear regression coefficients for dam aeration used in the water quality models are listed in Table B-2. This table can be used to rank dams by their aeration capacity; in general, dams with lower aeration coefficients (M) and more negative values of b aerate better.

The regression models of below-dam DO deficit as a function of the above-dam deficit generally are adequate predictors of D_b , as shown by the root mean square errors in Table B-2 that are generally within the range of accuracy that can be expected with DO models. At several dams, the linear model does not seem to describe aeration well. At Muskingum 3 and Allegheny 2, the aeration data are extremely scattered. The data indicate that the best model for aeration at these two dams may be that saturation is reached or exceeded for all upstream deficits; this situation is approximated by the low value of the aeration coefficient M determined by the regression analysis.

B.2.1.3.2 Incorporation of Turbine and Lockage Flows

D0 below dams during hydropower generation is modeled by determining how much of the river flow passes through the turbines, with the remaining flow passing over or through the dam and becoming aerated, then: (1) calculating how much D0 the water passing the dam picks up using the dam aeration model described above, (2) assuming that generating flows are not aerated, and



Figure B-1. Aeration graph for Allegheny L&D No. 5.



Figure B-2. Aeration graph for Hannibal Dam.



Figure B-3. Aeration graph for Allegheny L&D No. 3.

Model:	Below-dam	deficit =	(Above-dam	deficit	x M} + b,	in mg/L	
Dam		b <u>1</u> /	M <u>2</u> /	Corr. coef. <u>3</u> / r ²	No. obs. <u>4</u> /	RMS error, <u>5</u> / mg/L	Data 'source <u>6</u> /
Muskingu Muskingu	um 3 um 2	0.38 0	0.13 0.72	0.08 0.55	28 30	0.77 0.34	app] app]
Gallipol Bellevil Willow I Hannibal Pike Isl New Cumt Montgome Dashield Emsworth	lis le [s]and land perland ery is	-0.1 0 -0.17 -0.28 -0.23 -0.5 -0.61 -0.67 -0.19	0.84 0.89 0.97 0.89 0.72 0.38 0.78 0.72 0.77	0.74 0.74 0.97 0.72 0.76 0.71 0.44 0.71 0.76	66 23 21 24 14 28 34 13 37	0.39 0.72 0.14 0.34 0.32 0.15 0.64 0.36 0.36 0.42	app] app] app] app] COE app] COE COE comb
Allegher Allegher Allegher Allegher Allegher Allegher Allegher Allegher	ay 2 ay 3 ay 4 ay 5 ay 6 ay 7 ay 8 ay 9	-0.92 -0.67 0 0 0.13 -0.62 0	0.12 0.92 0.56 0.57 0.82 0.90 0.61 0.58	0.01 0.86 0.59 0.37 0.91 0.79 0.88 0.58	34 19 24 15 19 10 1	0.48 0.26 0.15 0.34 0.28 0.2	app1 comb COE comb comb comb COE COE
Monongal Monongal Monongal Maxwell Monongal Point Ma Morganto Hildebra Opekiska	nela 2 nela 3 nela 7 arinn Dwn and	-0.2 0.14 -0.18 -0.22 -0.1 -0.64 -0.21 0.1 -0.15	0.93 0.81 0.61 0.36 0.36 0.40 0.65 0.32 0.8	0.72 0.88 0.92 0.45 0.50 0.55 0.69 0.26 0.83	13 18 12 20 14 24 32 15 22	0.26 0.22 0.13 0.32 0.15 0.21 0.3 0.38	comb comb comb app] app] app] app] app] app]

B-21

Table B-2. Linear regression model parameters.

1/ Dam aeration constant.

2/ Dam aeration coefficient.

- 3/ Coefficient of correlation between above-dam deficit and below-dam deficit.
- 4/ Number of measured observations.
- $\underline{5}/$ Root mean squared error, an estimate of the average difference between predicted and measured values of the observed below-dam deficit.
- 6/ appl: data from aeration studies conducted by applicants. COE: data from Pittsburgh District water quality surveys. comb: data from Pittsburgh District and applicants combined.

(3) calculating the final DO concentration after all the flow is mixed together again below the dam. Step (3) is done by calculating the average, weighted by flow, of the DO concentrations in the aerated and unaerated portions. The hydropower plants would withdraw water from all depths, so it is assumed that any stratification occurring above the plant would be eliminated.

B.2.1.3.3 Influence of Temperature and Flow on Aeration

A univariate linear regression analysis was also performed on data from each dam to determine whether temperature and the river flow rate had significant effects on aeration. The below-dam deficit D_b was used as the dependent variable, and D_a , the temperature, and the flow rate were attempted as independent variables. Results from these analyses are in Table B-3 for dams where sufficient data were available. The value of the correlation coefficient for each independent variable indicates the apparent ability of the variable to explain variation in below-dam deficits.

From the values in Table B-3, the conclusion can be drawn that neither temperature nor flow rate has a consistently significant effect on aeration. Some fairly high correlation coefficients for temperature and flow occur, but because these coefficients are not consistently positive or negative, it appears that correlations between below-dam deficit and temperature or flow may be spurious. The correlation coefficient for D_a , however, is always positive and generally high.

Indonandont	Coefficient of correlation (r) <u>1</u> /						
variable:	Above-dam deficit	Temperature	Flow				
Muskingum 3 Muskingum 2	0.74	0.16	0.02				
Gallipolis Belleville Willow Island Hannibal Pike Island New Gutherland	0.55	0.07	-0.07				
Montgomery Dashiolds	0.51	0.63	-0.65				
Emsworth	0.89	-0.58	~0. 6 7				
Allegheny 2 Allegheny 3 Allegheny 4 Allegheny 5	0.11 0.91	0.47 -0.07	0.03 -0.64				
Allegheny 6 Allegheny 7	0.91	0.56	0.48				
Monongahela 2 Nonongahela 3 Monongahela 4	0.87 0.97	0.61 0.02	-0.31 -0.28				
Maxwell Monongahela 7 Point Marion Morgantown Hildebrand Opekiska	0.67 0.71 0.67 0.83	0.21 -0.77 0.64 0.66	-0.57 0.10 0.00 -0.22				

Table B-3. Linear regression results for dam aeration.

I/ The magnitude of the correlation coefficient indicates how much of the variation in below-dam deficit is apparently explained by the independent variables above-dam deficit, temperature, and flow rate. The analyses were separate for each independent variable. The independence of D_b from temperature should not be confused with the dependence of the value of C_s (which is used to calculate the DO deficits) on temperature. The temperature is required to determine C_s but appears to have no other consistent effect on D_b . The independence of D_b from the flow rate is not surprising because the only factor controlling mass transfer of oxygen that flow would affect is the degree of turbulence; apparently, either turbulence is not an important factor controlling aeration at these navigation dams or flow does not affect turbulence sufficiently to affect aeration. Both of these situations are possible at navigation dams; from visual observations, it appears that there is high turbulence at almost all flows. At gated dams there is high turbulence but sometimes low air entrainment, so the rate of air entrainment probably controls aeration rates; at some fixed-crest dams there is high turbulence and high air entrainment but little plunge below the dam, which may limit aeration. It must also be considered that the data used in this study were deliberately collected at relatively low flows and that measurements made over a wider range of flows may show a higher influence of flow on aeration.

B.2.1.4 Algal Production

Algal production is a potentially important component of the overall DO budgets of the Ohio River. Gross production rates estimated by Odum (1956) in rivers similar to the main stem of the Ohio River ranged up to 40 $g/m^2/d$ and more. However, very little information is available to quantify this oxygen source for this study. Diurnal variation in DO concentrations measured by hydropower applicants in the summer of 1987 was quite variable, ranging from 2.5 mg/L on the Muskingum River to essentially zero at several locations on the Ohio River. Insufficient data were collected in the 1987 studies to quantify DO production by algae. The high summer DO concentrations and algae blooms that sometimes occur in the Ohio River system indicate that under certain conditions primary productivity contributes a substantial amount of DO.

The 1987 data and historic data collected by ORSANCO indicate that diurnal variations in DO concentration are usually small, indicating that primary productivity is usually a minor part of the DO budget. Research conducted by the University of Cincinnati in the Meldahl pool indicated that the navigation dams have increased depths sufficiently that algal production is limited by light penetration of the water, so the DO contributed by algae is relatively minor (personnal communication, Dr. Michael C. Miller, Department of Biological Sciences, University of Cincinnati, June 1, 1987).

Algal production of DO may be important at times, but because modeling analysis is interested in periods of low DO concentrations when algal production is low and because insufficient information is available to estimate or model algal production, it was not modeled as a DO source term separate from water surface aeration.

B.2.2 FORMULATION OF THE BASIN-LEVEL MODEL

The equations describing the various components of the D0 budget (water surface aeration, dam aeration, and B0D decay) for the Ohio River system were implemented in a LOTUS 1-2-3 spreadsheet on a microcomputer. The spreadsheet model is a simple implementation of the equations describing the important components of the D0 budget, including the Streeter-Phelps equations for B0D decay and surface aeration, the dam aeration models, and a simple hydraulic routing scheme. The model assumes that flows and water quality are steady state over time and that plug flow (no longitudinal mixing) occurs. The spreadsheet solves the equations analytically and determines the D0 concentration at the beginning, end, and sag point (if any) of each reach of a river and can graph output. A new reach begins at each point where a wastewater discharge or tributary enters the river or where a dam is located. River velocities are estimated by assuming that the navigation dams maintain a constant pool elevation and cross-sectional area over the range of flows being modeled, so the velocity is equal to the flow rate divided by the cross-sectional area.

The spreadsheet model has several important advantages over other potential implementations. The number of input parameters is low because of the model's simplicity, and input is generally restricted to parameters for which measured or reliably estimated values are available. Under the low-flow conditions that are of most interest, the assumptions of plug flow and constant cross-sectional area are probably valid. The spreadsheet model is documented in Section B.4.

8.2.3 DATA SOURCES

B.2.3.1 River Flows

Flow data were obtained from U.S. Geological Survey stream gages throughout the upper Ohio River basin. Recent and simulated historic flows on the Ohio River were obtained from the Ohio River Division and the Pittsburgh District of the Corps. Flow duration curves and estimates of lockage and leakage flows at dams were also obtained from the Corps.

B.2.3.2 Waste Loadings

Major point-source dischargers were identified using the retrieval program MSP on EPA's STORET water quality data base. Actual BOD discharge rates for the major dischargers were obtained from Discharge Monitoring Reports in EPA's Permit Compliance System data base, and in the files of the Ohio EPA, the Pennsylvania Department of Environmental Resources, and the West Virginia Department of Natural Resources. Point-source dischargers included in the model are those that (1) discharge enough BOD to cause a significant depression in DO and (2) also monitor the BOD of their discharge. There are a number of major industrial dischargers throughout the basin where BOD is not monitored, so the waste loads from such dischargers are unknown and were not modeled. Where available, waste loading data compiled by the states for waste allocation studies and basin management plans were also used to estimate loadings from major dischargers. BOD concentrations in tributaries were estimated as approximately equal to those of the main stem. Calibration of the model generally required the addition of higher waste loads at some point sources and BOD loads representing nonpoint sources (Section 2.4). The point-source discharges used in the model are shown in Figures B-4 and B-5.

B.2.3.3 Historic Water Quality

Data from the ORSANCO electronic monitors were analyzed to estimate the historic distribution of temperatures and DO concentrations. These distributions were used to estimate the frequency with which certain DO and temperature conditions occur. Data from the Pittsburgh District summer surveys (Section 2.3.6) were also used to determine typical summer DO profiles along the study rivers.

B.2.3.4 Channel Geometry

Channel cross sections were obtained in computer files from the Corps. These cross sections are generally a mile or less apart and cover the entire study system, including the Muskingum. A program was written to use these cross sections to determine average crosssectional areas, depths, and widths for any river reach at a given pool elevation. This program was used to estimate channel geometry parameters for the models.

B.2.3.5 Dam Aeration Rates

To evaluate aeration, hydropower applicants were requested by FERC to collect twodimensional transect measurements of DO and temperature above and below each of the dams in the study basin during the summer and early fall of 1987, except where applicants had previously made such measurements. Results were obtained at all but two dams. Data from the Pittsburgh District summer water quality surveys were also used to estimate dam aeration rates. The district annually measures DO and temperature in vertical profiles above and below each dam in the district. These Corps data were used to (1) fit the aeration models at the sites where applicants did not collect data, (2) extend the range of DO deficits measured, and (3) demonstrate that the dam aeration models developed are constant over the time period (13 years) the district has collected data.

B.2.3.6 Water Quality Data for Model Calibration

The D0 models were calibrated to field data collected in July and August of 1983 by the Water Quality section of the Pittsburgh District of the Corps, supplemented with ORSANCO data. The Pittsburgh District's annual summer water quality surveys provide the most comprehensive data on D0 concentrations collected throughout most of the study area during a limited time period, and the data are of high quality. The data from 1983 were selected for calibration because (1) D0 deficits were high, (2) the Pittsburgh District's survey covered the entire study system down to the Hannibal dam pool, and (3) waste load data were not available from the state of Pennsylvania for years prior to 1983. D0 and temperature measurements were made by



Point-source dischargers and major tributaries modeled on the mainstem of the Ohio River.

Figure B-4. Point-Source dischargers and major tributaries modeled on the main stem of the Ohio River.

B-25



Figure B-5. Point-Source dischargers and major tributaries modeled on the Allegheny, Monongahela Rivers.

the Pittsburgh District at 108 sites on the Allegheny, Monongahela, Tygart, and Ohio rivers, between July 26 and August 26. ORSANCO electronic monitor data were used for calibration below the Hannibal pool; unfortunately ORSANCO data are not available on the Allegheny and Monongahela rivers except near their confluence.

When the 1983 calibration data were collected, flows in the Allegheny and Ohio were at approximately the 70 percent exceedance level on the annual flow duration curves (that is, the flows were at a rate equalled or exceeded 70 percent of the time during the year). On the Monongahela, flows were at about the 90 percent annual exceedance level. On the Allegheny, temperatures were at a level exceeded only about 5 percent of the time in July and August. On the Monongahela, temperatures were at a level exceeded about 25 percent of the time in July and August. On the Ohio, temperatures were at a level exceeded about 20 percent of the time in July and August (data from the DRSANCO monitors were used as a historic baseline for water temperatures).

B.2.4 MODEL CALIBRATION

The following measures were required for calibration of the DO models for each river. In each river, the addition of substantial amounts of BOD was required to simulate DO deficits as large as measured deficits. The additional BOD loadings are required to account for BOD sources for which data are not available, including (1) major wastewater dischargers that are not required to monitor BOD, so no loading information is available; (2) the many minor wastewater dischargers in the system; (3) benthic oxygen demand; and (4) other non point sources of BOD such as rural and urban runoff.

B.2.4.1 Allegheny River

Calibration to 1983 conditions was achieved by assuming a starting BOD of 4 mg/L at Allegheny dam 9 and adding additional BOD at lower pools. A BOD decay rate (k_1) of 0.1 was used, except in the dam 7, 6, and 5 pools where a value of 0.2 was used. Figure B-6 is a comparison of measured and modeled DOs.

B.2.4.2 Monongahela River

The assumption was made that flows coming from Tygart Dam will be saturated with DO; however, DO conditions in the Opekiska pool and below appear not to be controlled by Tygart River conditions. Calibration was obtained by adding BOD loads at the head of most pools. and by setting water surface aeration to nearly zero in the Opekiska and Hildebrand pools. The reduced aeration is justified to simulate the stratified conditions caused by temperature differences between the Tygart and West Fork rivers and a thermal discharge above Opekiska. Stratification is passed downstream to the Hildebrand pool by the bottom discharge at Opekiska dam. Aeration at Opekiska was set to zero. A BOD decay rate (k_1) of 0.1 was used. Figure B-7 is a comparison of measured and modeled DOs.

B.2.4.3 Ohio River

Calibration was obtained by tripling the BOD discharge at ALCOSAN to simulate other Pittsburgh discharges and non-point BOD sources, and by adding BOD loads at the heads of the lower pools. During the 1983 calibration period, there was higher aeration at Emsworth and Dashields than the dam aeration models predict. Because the dam models are based on data collected under many different conditions, they were not adjusted to match the single set of 1983 conditions. Therefore the model predicts lower DO peaks than the 1983 data show. A BOD decay rate (k_1) of 0.18 was used. Figure B-8 is a comparison of measured and modeled DOs.

B.2.5 APPLICATION OF MODELS TO IMPACT ASSESSMENT

There is a fundamental dilemma in modeling natural systems for decisionmaking. An accurate model of a natural system must account for the variability in time and space of all the model parameters and outputs, but the range of model results must be limited enough to allow decision making. To reduce the variability in the decision-making process, the following set of design river conditions, based upon 'typical' and 'worst-case' conditions, were used as a basis for decisions that will protect water quality during most natural conditions:



Figure B-6. Allegheny River DO model calibration results.

B-28



D0, mg/l

Figure B-7. Monongahela River DD model calibration results.


D0, mg/1

Figure B-8. Ohio River DO model calibration results.

8-30

- (1) BOD loadings are those obtained from calibration.
- (2) Water temperatures are those exceeded 10 percent of the time in August at the ORSANCO robot monitors.
- (3) Flows are from two sets of conditions: low flows, where 7010 flows are used; and moderate flows, approximately the minimum flow at which all of the proposed projects would operate. The low flow conditions are those when DO concentrations are more likely to be low. The moderate flow conditions are those when the projects would have the greatest impacts on DO concentrations.
- (4) Rate coefficients for reaeration and BOD decay are those obtained from calibration.
 (5) Projects that have already been licensed (at Allegheny dams 5. 6, 8, and 9; and Racine and Hannibal dams on the Ohio River) are modeled as operating, with the spill flows required by their licenses (or spill flows expected to be required by operating agreements with the Corps at Allegheny 8 and 9).

Results of modeling under these conditions are presented in Section 4 of the EIS. Additional model runs were made using mean monthly flows and water temperatures for the months of April through November. Results of these runs (Figures 8-9 through 8-17) show how DOs and project impacts would change seasonally and were used to model impacts of the proposed projects on fish growth.

Although not documented as thoroughly as the above conditions were, experimentation with different temperatures, flow rates, and BOD loads has been conducted to ensure that recommendations based on the model are valid over a wide range of environmental conditions.

B.3 OPTIMIZATION MODELING

B.3.1 DECISIONS AND ALTERNATIVES TO BE CONSIDERED

The systems-level decision problem in licensing multiple hydroelectric projects on the upper Ohio River basin involves potential trade-offs between energy production and water quality. At each project, energy production is inversely related to the amount of water spilled over the dam because water that is spilled does not pass through the turbine and does not generate hydroelectricity, and reducing the spillage at dams adversely effects downstream DO concentrations (Section B.2.1.3.2). An objective of the FERC licensing process is to find the best combination of sites or operating schedules that can be licensed without causing unacceptable degradation of water quality. This objective can be formalized in words and a system of equations as follows:

MAXIMIZE ANNUAL HYDROELECTRIC PRODUCTION IN THE BASIN

SUBJECT TO PHYSICAL CONSTRAINTS AND MINIMUM WATER QUALITY CONDITIONS

or

MAX: $P_N = \Sigma p_i$

 $\begin{array}{l} \text{Subject to: } c_{ik} \geq c_S \text{ for all } i,k\\ \text{and } g_i + l_i + s_i = q_i \text{ for all } i \quad, \end{array}$

where:

P_N is the total annual energy production for all N sites; p_i is the annual energy production for the ith site, a function of g_i ; c_{ik} is the DO concentration at the kth location downstream from the ith site (defined by the Streeter-Phelps equation or modifications thereof); C₅ is the water quality standard for DO; g_1 is the generating flow through the turbine at the ith site; 1; is the combined lockage and leakage flows at the ith site; s; is the spillage at the ith site; and

q; is the total flow of the river at the ith site.

In this system of equations, the decision variable is the spillage flow, si, which determines both the energy production and downstream DO concentrations. At any of the sites of interest, the minimum spillage has been proposed by the license applicant and may be subject to



River Miles Above Pittsburgh

Figure B-9. Allegheny River DO model results from April to June, mean monthly flows and temperatures.

B-32

•



River Miles Above Pittsburgh

Figure B-10. Allegheny River DO model results from July to September, mean monthly flows and temperatures.

B#33



Figure B-11. Allegheny River DO model results from October to November, mean monthly flows and temperatures.



Figure B-12. Monongahela River DO model results from April to June, mean monthly flows and temperatures.



ο ε

Dissolved Oxygen concentration,

River Miles Above Pittsburgh

Figure B-13. Monongahela River DO model results from July to September, mean monthly flows and temperatures.



Figure B-14. Monongahela River DO model results from October to November, mean monthly flows and temperatures.



River Miles Below Pittsburgh

Figure 8-15. Ohio River DO model results from April to June, mean monthly flows and temperatures.



Dissolved Cxygen concentration, mg/f

River Miles Below Pittsburgh

Figure B-16. Dhio River DO model results from July to September, mean monthly flows and temperatures.



River Miles Below Pittsburgh

Figure B-17. Obio River DO model results from October to November, mean monthly flows and temperatures.

modification by FERC. Spillage values may range between zero (all water through the turbines) and $q_i - l_i$ (no water through the turbines). These equations can be solved to find the optimum combination of energy production and spillage values that will satisfy water quality standards. If optimal spillage is found to be high at some sites, rejection of the license application is a strong possibility. The equations can also be analyzed in a trade-off analysis to evaluate any number of alternative licensing scenarios. There are also a number of alternative objective formulations that can be used, but at this time we will not go beyond a simple formulation of the problem.

Sets of equations such as those listed above are called mathematical programs. They can be solved in a number of different ways, including linear programming, nonlinear programming, dynamic programming, and implicit enumeration. There are many applications of all four of these approaches in the water resources literature where math programming has been used to solve water quality problems. In the situation of interested on the Ohio River, multiple hydroelectric projects arranged in a branched series, the value of the objective function at any particular site is a function of the values of the decision variables at all sites upstream. That is, the energy production and the spillage required at a site is dependent on the energy production and spillage at all upstream sites. If there are upstream branches in the river network, as there are on the Ohio, then specific downstream outcomes can be determined from more than one unique combination of upstream decisions. This structure of the system complicates the solution technique that can be used. We believe that implicit.

B.3.2 BOUNDED IMPLICIT ENUMERATION

Enumerative methods are one approach to solving math programs. Explicit enumeration is the exhaustive solution of all possible combinations of the decision variables. Implicit enumeration methods are nonexhaustive algorithms that use some structured bookkeeping scheme to eliminate a priori certain combinations of variables that can be shown to yield non-optimal solutions. Bounded implicit enumeration (BIE) is one of these more efficient enumerative methods that uses an upper and lower bound to exclude undesirable solutions (Chang and Liaw 1987).

In the Ohio River system, each dam is a stage or node where decisions must be made on the spillage requirements. The first step in BIE is to define a number of discrete values for spillage at each site, ranging between zero and all the flow. These spillage options are indexed by the subscript j. The subscript i indexing each site increases from 1 to N from upstream to downstream. Then the BIE algorithm can be defined as the following steps, starting at the most upstream site and moving downstream:

(1) Find the minimal production value at each site and calculate the lower bound (P_1^L) of the objective function at that site (this lower bound is the minimum production at sites below the current site):

$$P_{i}^{L} = \sum_{k=i+1}^{N} P_{k};$$

- (2) Iterate through all the spillage options at the current site, setting a new upper bound on the objective function for each successively higher production level. For each spillage option that yields a production level between the current upper and lower bound, a complete DO curve is calculated downstream and a check on the DO constraint is made.
- (3) Move to the next downstream node, and iterate through the spillage options, trying to find a combination of feasible spillage and production levels that will exceed the current upper bound. For any combination of spillage options that do exceed the upper bound, the DO curve is calculated again and DO constraints are checked.

The calculation of DO curves within these iterations uses the same DO modeling equations for BOD decay, surface aeration, and dam aeration described in Section B-2 above. The BIE series of calculations proceeds through all hydropower sites in the basin until the last node (Gallipolis) is reached. During the calculations, combinations of the spillage options that have been enumerated and were feasible are saved, along with the DO profile that was calculated for them. To examine the trade-offs of generating power vs maintaining successively higher DO standards, the BIE algorithm is then restarted with a higher value for Cg. These secondary iterations can be repeated with DO values up to the current conditions. After all iterations are complete, the set of optimal and nonoptimal spillage combinations can be evaluated.

B.4 DISSOLVED OXYGEN MODEL DOCUMENTATION

The DO model is implemented at Lotus 1-2-3 spreadsheets and as a FORTRAN program. Complete documentation of both implementations is provided in Railsback and Jager (1988). The following is a list of the input and output parameters for the model. Each parameter is input or calculated for each of the rivers.

- 1. Reach name (input).
- 2. River mile (input).
- 3. Reach length (calculated).
- 4. Tributary flow (input).
- 5. Flow in reach (calculated).
- Cross-sectional area (input). The average cross-sectional area of the river in the reach (in square feet).
- 7. Depth (input). Average depth of the reach (in feet).
- 8. Velocity (calculated). The average velocity of the reach (in feet per second).
- 9. Travel time (calculated). The average time it takes water to travel the length of the reach (in days).
- 10. Tributary DO (input).
- 11. Tributary BOD concentration (input).
- 12. BOD loading (input). The point source BOD loading that takes place at the head of the reach, if any.
- 13. Dam aeration coefficient (input).
- 14. Dam aeration constant (input).
- 15. Flow not aerated (input). The flow rate in cubic feet per second (between zero and the total river flow) that does not pass over the dam or through the gates. This flow is used for lockage and hydropower generation.
- 16. DO above dam (calculated). The DO at the head of the reach (in milligrams per Liter), not including dam aeration if the reach starts at a dam.
- 17. Starting BOD (calculated). The BOD at the head of the reach (in milligrams per Liter).
- 18. Reach temperature (input).
- 19. DO saturation concentration (calculated).
- 20. kl at 20 degrees (input).
- 21. k2 at 20 degrees (calculated).
- 22. kl(T) (calculated). The BOD decay rate at stream temperature.
- 23. k2(T) (calculated). The stream reaeration rate at stream temperature.
- 24. Initial deficit (calculated). The DO deficit at the upstream end of the reach (in milligrams per Liter).

- 25. Final deficit (calculated). The DO deficit at the downstream end of the reach (in milligrams per Liter).
- 26. Starting DO (calculated). The DO concentration at the upstream end of the reach, following dam aeration.
- 27. Final DO (calculated). The DO concentration at the downstream end of the reach (in milligrams per Liter).
- 28. Final BOD (calculated). The BOD concentration remaining at the downstream end of the reach (in milligrams per Liter).
- 29. Critical time, final (calculated). The travel time from the top of the reach to the point of the critical DO concentration (in days), corrected to equal the travel time of the reach if the raw critical time is greater than the travel time of the reach and to equal zero if the raw critical time is negative.
- 30. Critical deficit, final (calculated). The critical DO deficit, corrected to equal the DO deficit at the downstream end of the reach if the raw critical time is greater than the reach travel time. This value is the highest deficit that occurs in the reach, whether it occurs at the beginning, end, or within the reach.
- 31. Critical DO concentration (calculated). The lowest DO concentration in the reach.
- 32. Critical point, in river miles (calculated). The river mile at which the lowest DO concentration (highest deficit) in the reach occurs.
- 33. D0 index (calculated). The D0 index is the integral of the curve of D0 vs distance for the reach. This parameter can be used as an indicator of impacts of changes in aeration or discharge on DO, because it combines both changes in DO concentration and the distance affected.

B.5 SENSITIVITY AND UNCERTAINTY OF THE MODEL

B.5.1 INTRODUCTION

The sensitivity and uncertainty of the Ohio River basin DD model, with the parameters used for the EIS, were analyzed. The sensitivity analysis investigates which parameters the model results are most sensitive to (i.e., which parameters, when varied, cause the greatest change in the modeled DD concentrations). The sensitivity analysis identifies processes (such as dam aeration, water surface aeration, and BOD decay) that have the greatest effect on DO concentrations at different locations. The uncertainty analysis is an investigation of variability in the results predicted by the model. This procedure is performed by including the estimated uncertainty in the model parameters into the model results to determine the uncertainty in the results. The uncertainty analysis essentially creates a stochastic DO model by treating model parameters as means of probability distributions instead of as constants.

B.5.2 SENSITIVITY ANALYSIS

8.5.2.1 Methods

The sensitivity analysis was performed using the Gradient Enhanced Software System (GRESS), developed at Oak Ridge National Laboratory (Oblow, 1983a, 1983b). GRESS enhances FORTRAN code by giving it the ability to determine partial derivatives of any selected output variable with respect to any selected input variable. GRESS also calculates a normalized sensitivity index that can be used to compare the model sensitivity among parameters having different units. GRESS was used to determine the partial derivatives and sensitivity indexes of the critical (Towest) DD concentration in each reach with respect to the following variables:

- 1. the initial DO concentrations in the Allegheny and Monongahela rivers,
- 2. the initial BOD concentrations in the Allegheny and Monongahela rivers,
- 3. the initial flows in the Allegheny and Monongahela rivers, 4. $k_{\rm I}$ in the Allegheny and Monongahela rivers, 5. $k_{\rm 2}$ in each reach,

- 6. the tributary flow in each reach,
- 7. the tributary DO concentration in each reach,
- 8. the tributary BOB concentration in each reach,
- 9. the point-source BOD loading in each reach,
- 10. the dam aeration constant in each reach with a dam,
- 11. the dam aeration coefficient in each reach with a dam,
- 12. the aeration rate (the increase in DO concentration in the spill flow at a dam) in each reach with a dam.
- 13. the flow rate used for generation (flow not aerated) in each reach with a dam, and
- 14. the water temperature in each reach.

The GRESS sensitivity analyses were conducted on the model with the parameters used for the design conditions (Section B.2.5) with the spill flows recommended for Alternative 3 in the EIS, the scenario upon which the staff recommendations in the EIS were based. The GRESS analyses were also conducted on the model with the parameters for the design conditions with none of the proposed new hydropower projects in operation (but with existing and licensed projects in operation). Each of these analyses produces over 27,700 partial derivatives as output; a small fraction of these values was analyzed graphically to develop an overall understanding of model sensitivities at important locations along the rivers.

B.5.2.2 Results

The GRESS sensitivity analysis shows that DO concentrations are generally most sensitive to water temperature (for example, see the values in Table B-4). This result is not surprising because of the direct dependency of DO saturation and the rate constants k_1 and k_2 on temperature. It should be noted that the values in Table B-4 for sensitivity to water temperature are related to the water temperature in the same reach that the output was calculated for; the sensitivity to changes in water temperature in the upstream reaches is not included. The sensitivity analysis also shows that model results are highly sensitive to the flow rate in most reaches (compare values in Table B-4 to sensitivity indexes in Figures B-18 through B-25). This result means that significant changes in predicted DO concentrations can be expected when different water temperatures and flows are modeled. The following analyses emphasize the sensitivity of the model to parameters other than temperature and flow.

The sensitivity to initial conditions (starting D0 and B0D concentrations in the Allegheny and Monongahela rivers) were compared to sensitivities to the rate coefficients k_1 and k_2 to determine the extent (over distance downstream) over which the assumed initial conditions are important. The sensitivities to the rate coefficients k_1 and k_2 were used for comparison to the sensitivity to initial conditions because they represent B0D decay and water surface aeration, which control D0 concentrations when initial conditions and dam aeration are not important, and because they are relatively constant throughout the rivers. Figures B-18 and B-19 show the results of these analyses. In the Allegheny, the model is not particularly sensitive to the initial B0D concentration. The predicted D0 concentrations are more sensitive to the initial D0 concentration than to k_1 and k_2 from Allegheny dam 9 to about Allegheny dam 5. This result is not surprising, because the licensed hydropower plants at Allegheny dams 9, 8, 6 and 5, combined with the low aeration efficiency of dam 7, provide little dam aeration in these reaches. If dam aeration were higher, the effects of initial D0 concentrations would be overwhelmed by the effects of dam aeration. This result indicates that the D0 concentration at Allegheny dam 9 will have a strong influence on D0 concentrations as far downstream as dam 5 wh all the licensed hydropower projects are in operation.

The predicted DO concentrations in the Monongahela River are relatively sensitive to initia and BOD concentrations as far downstream as Hildebrand Dam. Hildebrand is the first efficient aerator on the river, and apparently DO concentrations below Hildebrand are controlled more by aeration, water surface aeration, and BOD loads than by assumed initial conditions. This resul means that predicted DO concentrations in the Monongahela downstream of Hildebrand are insensit to assumed conditions at Tygart Dam.

The sensitivities of predicted DD concentrations to a number of parameters were determined critical locations on each river. The critical locations are those where the proposed hydropow would reduce DO concentrations the most, according to model analyses presented in the EIS. The critical locations are river mile (RM) 0 on the Allegheny, RM 65 (the sag point below Monongahe dam 7) on the Monongahela, and RM 100 on the Ohio River.



Figure B-18. Sensitivities of Allegheny River critical DO concentrations to initial conditions (Alternative 3, moderate flows).



Figure B-19. Sensitivities of Monongahela River critical DO concentrations to initial conditions (Alternative 3, moderate flows).



Sensitivities river mile 0. of the critical DO concentration





Sensitivity index



sensitivity to water surface aeration coefficient k2

sensitivity to aeration at dam at beginning of reach

Figure B-23. Sensitivities of Allegheny River critical DO concentrations to $k_{\rm Z}$ and dam aeration without the proposed projects.





Figure B-24. Sensitivities of Monongahela River critical DO concentrations to k_2 and dam aeration without the proposed projects.





sensitivity to aeration at dam at beginning of reach

Figure B-25. Sensitivities of Ohio River critical DO concentrations to k_2 and dam aeration without the proposed projects.

Location 1/	dC/dQ <u>2</u> / (mg/L)/cfs	dC/dT <u>3</u> / (mg/L)/ ⁰ C	Sensitivity index for Q <u>4</u> /	Sensitivity index for T <u>5</u> /
Allegheny RM 30	0.00032	-0.015	0.30	0.066
Allegheny RM O	C.000028	-0.145	0.02	0.50
Monongahela RM O	0.00011	-0.04	0.029	0.17
Monongahela RM 65	0.00033	-0.132	0.089	0.54
Ohio RM 54 <u>5</u> /		-0.084		0.08
Ohio RM 100		-0.052		0.21
Ohio RM 250		-0.022		0.11

Table B-4. Example sensitivities to flow and temperature at critical locations under design conditions and Alternative 3 spill flows

1/ RM-river mile.

2/ Partial derivative of the critical DO concentration with respect to river flow.

- $\underline{3}$ / Partial derivative of the critical DO concentration with respect to the water temperature in the same reach.
- 4/ GRESS sensitivity index for flow, which can be compared to values in Figures 12-19.
- 5/ GRESS sensitivity index for the water temperature in the same reach, which can be compared to values in Figures 12-19.
- 5/ The sensitivity of Ohio River DD concentrations to river flow was not estimated because it is complicated by the effects of many tributary inflows.

Figure B-20 shows the sensitivity of predicted DO concentrations at Allegheny RM O to initial conditions, the water surface aeration rate k_2 , the BOD decay rate k_1 , the flow rates used for generation (equal to the sensitivity to the spill flow rate), and the dam aeration rates (the increase in DO concentration, in milligrams per liter, that occurs in the spill flow) at the first four dams upstream. The figure shows that DO concentrations at Allegheny RM O are highly sensitive to aeration at Allegheny dam 2 under the conditions simulated for Alternative 3. Under this alternative, dam 2 has a high spill flow, which controls DO concentrations because this dam is a very efficient aerator. The DO concentrations at Allegheny RM O are relatively insensitive to flow, though DO concentrations upstream of dam 2 are sensitive to flow.

The sensitivity of predicted DO concentrations at Monongahela RM 65 is shown in Figure B-21. The DO concentrations at this location are most sensitive to the flow rate, the water surface aeration rate coefficient k_2 , and the aeration rate at Monongahela dam 7 (where no hydropower is proposed). Other parameters of importance to DO concentrations are the BOD loadings at several upstream reaches.

The sensitivity of predicted DO concentrations at Ohio RM 100 is shown on Figure B-22. This figure shows that DO concentrations are relatively sensitive to flow, BOD decay rate k_1 , and aeration at the dams upstream of RM 100. The sensitivity to the value of k_1 used in the Allegheny and Monongahela rivers indicates that DO concentrations in these rivers still affect DO concentrations at Ohio RM 100.

Figure B-23 compares the relative sensitivity of the modeled critical DO concentrations in each reach of the Allegheny River that starts with a dam to the surface aeration rate coefficient k_2 and to the amount of aeration (the milligram-per-liter increase in DO in the spill flow) taking place at the dam. The sensitivity indexes are for conditions without the proposed new hydropower projects but with the licensed projects at dam 9, 8, 6, and 5. These sensitivity indexes indicate the relative importance of surface aeration vs dam aeration in these reaches. The figure shows that below Allegheny dams 7, 6, and 5, water surface aeration is more important for maintaining DO concentrations, which is expected because dam 7 is a poor aerator and because of the licensed projects with low spill flows at dams 5 and 6. However, below dams 4, 3, and 2, the model becomes much more sensitive to dam aeration. This sensitivity indicates that below dam 4, dam aeration is important for maintaining DO concentrations in the Allegheny.

Figure B-24 shows the relative sensitivity of the model to k_2 and dam aeration in the Monongahela River without the proposed new hydropower. In the reach below Opekiska Dam, which provides little aeration, the model is not sensitive to dam aeration. Below Hildebrand and Point Marion dams, the model is more sensitive to dam aeration; and for the rest of the river, the model seems to be about equally sensitive to water surface aeration and dam aeration. These results indicate that dam aeration is especially important for maintaining D0 concentrations below Hildebrand and Point Marion dams and remains of importance in the reaches further down the Monongahela River.

The relative sensitivity of the model to k_2 and dam aeration in the Ohio River, without the proposed new hydropower, is shown in Figure B-25. The figure shows that predicted DO concentrations below the first five dams on the Ohio River are more sensitive to dam aeration than to water surface aeration. Below about RM 100, the model becomes much more sensitive to water surface aeration. This sensitivity, indicates that aeration at the first five dams of the Ohio is more important for maintaining DO concentrations than is aeration at the rest of the dams in the study. This result is expected because of the more efficient aeration at the upper five dams.

B.5.2.3 Conclusions

In general, the D0 model is most sensitive to water temperature and flow rate. The values of these parameters should be selected carefully in future modeling studies. The effects of variation in these parameters should be at least qualitatively investigated in any new studies because since they strongly influence predicted D0 concentrations.

From Allegheny dam 9 downstream to dam 5, the modeled DD concentrations are sensitive to the starting DD concentration at Allegheny dam 9, which is an input parameter. This starting DD concentration should be selected carefully, and the effects of variation in it should be investigated in any additional modeling studies. The model is not especially sensitive to the initial BOD concentration in the Allegheny nor to the initial DO and BOD concentrations in the Monongahela.

The DO concentrations at RM O of the Allegheny River and in the upper reaches of the Ohio River are very sensitive to aeration at Allegheny dam 2 when this dam is spilling water. This dam is very important for maintenance of DO concentrations in these reaches.

In the upper 100 river miles of the Ohio River, dam aeration is important for maintaining DO concentrations. The model is sensitive to the decay rate, k_1 , of BOD below Pittsburgh; consequently, obtaining measured values of this parameter would be useful to improve the model. Processes controlling DO in the Allegheny and Monongahela rivers have an important effect on DO in the Ohio River at least as far downstream as RM 100.

There are reaches in each river where dam aeration is and is not relatively important for maintaining DO concentrations (i.e., where DO concentrations are and are not sensitive to dam aeration). The reaches where the model is most sensitive to dam aeration are below Allegheny dam 4, below Hildebrand and Point Marion dams, and below the first five dams on the Ohio River. These reaches are generally where the most dam aeration occurs, so it appears that the model is more sensitive to dam aeration where the dam aeration rate is high. This fact implies that dam aeration has a greater than linear effect on critical DO concentrations; i.e., that as dam aeration increases, the DO concentrations rise at an increasing rate.

It should be noted that the sensitivities determined in this analysis change when the model parameters change. The sensitivities of the model to various parameters could change significantly when different scenarios or conditions are modeled. The results presented here describe the sensitivities of the model as it represents the design conditions and recommended spill flows in the EIS.

B.5.3 UNCERTAINTY ANALYSIS

B.5.3.1 Methods

The uncertainty analysis incorporates the estimated uncertainty in model parameters into an estimated uncertainty in the model results and provides confidence bounds for the model's predictions of DO concentrations. This method does not address uncertainties in how the model is formulated but assumes that the structure of the model (i.e., the equations used) is correct and addresses the uncertainty in the values of the model parameters.

The model parameters for the design conditions used in the EIS were determined by using the following steps (Section B.2):

- River flows, water temperatures, and the initial Allegheny River DD concentration were selected to represent conditions where DD concentrations are expected to be low. The values were selected after examining the range of measured historic values for these parameters. The initial DD concentration on the Monongahela River, at the outlet from Tygart Dam, was assumed to be at saturation because of aeration at the dam.
- Dam aeration parameters (b and M) were estimated from field data. The k₂ values were estimated using the O'Connor-Dobbins equation. Tributary DO and BOD concentrations were estimated.
- 3. The values for BOD loadings, k_1 , and, in one case, tributary DD concentrations were determined by calibrating the model to measured data.

The uncertainty analysis was performed by estimating the uncertainty in all the input parameters that were either estimated or determined from calibration. No uncertainty was assigned to the parameters (flow, temperature, and initial DO concentrations) that were selected as design conditions.

The uncertainty analysis was performed for the design conditions used in the EIS (Sect. B.2.5) with the proposed hydropower plants operating with the spill flows recommended under Alternatives 3 and 4 of the EIS because this is the model run on which the recommendations in the EIS were based. The analysis was also performed for the model with the assumption that none of the proposed new hydropower projects were in operation. The software used for the uncertainty analysis is the PRISM system developed at Oak Ridge National Laboratory (Gardner Rojder, and Bergstrom 1983; Gardner 1984).

The expected uncertainty in model parameters is represented by a frequency distribution that actual values of the parameter are expected to follow. The analysis therefore requires a description of the frequency distribution each parameter follows and a description of any important correlations between parameters. Each distribution is described by (1) the type of distribution followed, such as normal (Gaussian), uniform, or lognormal; (2) a mean value and a variance for normal and lognormal distributions; and (3) minimum and maximum values for uniform distributions. PRISM allows the use of bivariate distributions that describe the joint frequency distribution of two parameters whose values are correlated. The frequency distributions for model parameters were determined as follows.

The uncertainty in the dam aeration coefficients (M's) and constants (b's) was estimated from the linear regression analyses that were used to estimate these parameters. For each dam, a bivariate normal distribution was assigned to describe the joint frequency distribution of b and M. Regression analysis for the linear dam aeration model (Section B.2.1.3) using field data for each dam provided a full description of the bivariate normal distribution of b and M. The least-squares regression estimates of b and M (the values used in the model; Table B-2) are the means, and the variance-covariance matrix of the parameters provided by the Statistical Analysis System program complete the description of the bivariate normal distribution. The uncertainty in the k_2 estimates from the O'Connor-Dobbins equation was estimated from measured and calculated values of k_2 in the Obio River presented by O'Connor and Dobbins (1958). The equation for k_2 is

$$k_2 = Z(V^{0.5})/depth^{1.5}$$

where Z is a constant with a value of 12.9. The uncertainty in k_2 was assigned to the constant Z. Values of Z that reproduced the measured k_2 values for 22 field measurements were calculated. These values of Z were approximately lognormally distributed with a mean of 2.76 and a standard deviation of 0.633, and this lognormal distribution was assigned as the uncertainty in k_2 .

Uncertainties in tributary DO and BOD concentrations were assumed to be uniformly distributed within a range of $\pm 2 \text{ mg/L}$ of the mean (the mean being the value used in the model).

The uncertainty in the point-source BOD loadings were assumed to be normally distributed with a standard deviation of 25 percent of the value used in the model.

The uncertainty in k_1 was estimated from data published in USEPA (1985, p. 147). This document presents values of k_1 from a variety of rivers, including sediment oxygen demand, as does the k_1 used in the Ohio River model. The measured values of k_1 in rivers with approximately the same depths as those in this study were approximately uniformly distributed over a range of 0.08 to 0.5. A uniform distribution with this range was used for k_1 .

After frequency distributions were assigned to the parameters that reflected uncertainty, the model was executed 2000 times. A Latin-hypercube method (Rose and Schwartzman, 1981) was used to systematically assign parameter values for each execution that, over the 2000 executions, fit the frequency distributions assigned to each parameter. Statistics on the critical (lowest) DO concentration in each reach of the model was stored for each of the 2000 executions. The mean, maximum, minimum, and standard deviation of the 2000 values of critical DO concentration for each were determined.

B.5.3.2 Results

The mean critical DO concentrations and the 95 percent confidence intervals (CIs) for each reach of the Allegheny, Monongahela, and Ohio rivers for the model with the proposed hydropower projects are plotted in Figures. B-25, B-26, and B-27. If the structure of the model is correct (i.e., the uncertainty lies in the parameter values) and the critical DO concentrations generated in the uncertainty analysis are normally distributed for each reach, there is a 95 percent probability that true value of the standard deviation of the 2000 critical DO values for each reach.) In most cases the 95 percent CIs calculated in this way are close to the observed minimum and maximum DO concentrations generated by the uncertainty analysis.



Figure B-26. Uncertainty analysis for the Allegheny River with the proposed projects (CI = confidence intervals, DO = dissolved oxygen).





Figure B-27. Uncertainty analysis for the Monongahela River with the proposed projects ([[- confidence intervale DD - discolved avvgen]

There is relatively high uncertainty (95 percent CIs within about 1.5 mg/L of the means) in the Allegheny River model results from dam 9 as far downstream as the dam 4 reach (Figure B-26). The uncertainty decreases until RM 0, where the value is very small.

There is also relatively high uncertainty in the Monongahela River model results from Tygart Dam to the Hildebrand Dam reach, as Figure B-27 indicates. For the rest of the Monongahela, the uncertainty is relatively low, with the 95 percent CI within about 1 mg/L of the mean.

The model uncertainty on the Ohio River is low until it gradually increases below RM 100 (Figure B-28). In the reach between RM 0 and RM 100 where dam aeration is especially important, the 95 percent CI is within about 0.5 mg/L of the mean. At the reach below Gallipolis Dam, the 95 percent CI has expanded to about 1.5 mg/L from the mean.

Figures B-29, B-30, and B-31 show the uncertainty analysis results for the Allegheny, Monongahela, and Ohio rivers without the proposed new hydropower development. Also shown on these three figures are the ranges of measured DD concentrations (the mean and the mean \pm 2 standard deviations) at the ORSANCO water quality monitoring stations. These ORSANCO data were collected in the months of July, August, and September between 1980 and 1988, at times when the water temperature was between 26 and 30°C.

The D0 model uncertainties without the proposed new hydropower projects are similar to those for the model with Alternative 3 spill flows. At the ORSANCO monitoring stations on the Allegheny and Monongahela rivers and at the first three stations on the Ohio River, the mean measured D0 concentration is very close to the mean concentration modeled in the uncertainty analysis. The similarity between the modeled and measured mean concentrations indicates that the parameter frequency distributions used in the uncertainty analysis accurately model actual D0 concentrations. There is much more variability in the measured data than there is uncertainty in the model. The variability in the measured data is probably caused by variability in flow rate and in other processes such as primary productivity that are not incorporated in the model.

In all three rivers, the uncertainty in the model is lowest in the reaches that are most influenced by dam aeration. The clearest example of this is the Allegheny dam 2 reach, where the DO concentration is highly controlled by the aeration at dam 2 (Section B.5.2.2). Even though the aeration parameters for dam 2 have more uncertainty than those for most other dams, the resulting uncertainty in model results is essentially negligible. In contrast, below Ohio RM 200 where dam aeration has very little effect on DO concentrations, the uncertainty is relatively high. Below Ohio RM 200, DO concentrations are controlled more by the rate coefficients k_1 and k_2 than in the upper end of the river, and uncertainties in the values of these coefficients increase the uncertainty in the model.

B.5.3.3 Conclusions

The 95 percent CIs in the Ohio River basin DO model range from less than \pm 5 percent of the mean to about \pm 25 percent of the mean. The uncertainty analysis shows that the dam aeration models give stability to the model results because the uncertainty is much lower in reaches where DO concentrations are dominated by dam aeration. There is apparently less uncertainty in the linear regression dam aeration parameters (Secion B.2.1.3) than in the other parameters controlling DO concentrations. This conclusion is important because the purpose of the model is to evaluate impacts of changes in dam aeration and to select spill flows that provide adequate DO concentrations. The uncertainty in the model is lowest in the reaches where the decisions based on the model are most important.





Figure B-28. Uncertainty analysis for the Ohio River with the proposed projects (CI = confidence intervals. DO = dissolved oxygen).



Figure B-29. Uncertainty analysis for the Allegheny River without the proposed projects (CI = confidence intervals, DO = dissolved oxygen).



Figure B-30. Uncertainty analysis for the Monongahela River without the proposed projects (CI = confidence intervals, DO = dissolved oxygen).

- Modeled Upper Cl
- Modeled Lower Cl
- Measured Mean
- Measured Upper Cl
- Measured Lower Cl



Modeled Mean

- + Modeled Upper Cl
- Modeled Lower Cl
- A Measured Mean
- ✓ Measured Upper Cl
- × Measured Lower Cl

Figure B-31. Uncertainty analysis for the Ohio River without the proposed projects (CI = confidence intervals, DO = dissolved oxygen).

B.6. LITERATURE CITED

- Butts, T. A., and H. R. Adkins, 1987. Aeration Characteristics of Starved Rock Dam Tainter Gate Flow Controls, Illinois State Water Survey contract report 423, Champaign, Ill.
- Chang, S. Y., and S. L. Liaw, 1987. An efficient implicit enumeration algorithm for multistage systems, Paper presented at the TIMS/ORSA Joint National Meeting, New Orleans, Louisiana, May 4-6, 1987.
- Colt, J., 1984. Computation of Dissolved Gas Concentrations in Water as Functions of Temperature, Salinity, and Pressure, American Fisheries Society Special Publication 14.
- Gardner, R. H., B. Rojder, and U. Bergstrom, 1983. PRISM: A Systematic Method for Determining the Effect of Parameter Uncertainties on Model Predictions, Studsvik Energiteknik AB report/NW-83/555, Nykoping, Sweden.
- Gardner, R. H., 1984. A Unified Approach to Sensitivity and Uncertainty Analysis. in <u>Applied</u> <u>Simulation and Modeling</u>, ed. M. H. Hamza, Proceedings of the IASTED International Symposium, San Francisco, June 4-6, 1984, ACTA Press, Anaheim, Calif., pp. 155-157.
- Oblow, E. M., 1983a. An Automated Procedure for Sensitivity Analysis Using Computer Calculus. ORNL/TM-8776, Oak Ridge National Laboratory, Oak Ridge, Tenn.
- Oblow, E. M., 1983b. GRESS: Gradient-Enhanced Software System, Version D. User's Guide, ORNL/TM-8339, Oak Ridge National Laboratory, Oak Ridge, Tenn.
- O'Connor, D. J., and W. E. Dobbins, 1958. Mechanism of Reaeration in Natural Streams, Transactions, American Society of Civil Engineers, Paper 2934.
- Odum, H. T., 1956. Primary Production in Flowing Waters. Limnology and Oceanography <u>1</u>, (103) 107.
- Railsback, S. F., and H. I. Jager. 1988. Modeling of Hydropower Impacts on Dissolved Oxygen in the Upper Ohio River Basin. ORNL/TM-10953. Oak Ridge National Laboratory, Oak Ridge, Tenn.
- Rose, K. A., and G. L. Schwartzman, 1981. A Review of Parameter Sensitivity Methods Applicable to Ecosystem Models, U.S. Nuclear Regulatory Commission report NUREG/CR-2016.
- Streeter, H. W., and E. B. Phelps, 1925. A Study of the Pollution and Natural Purification of the Ohio River, Public Health Bulletin 146, U. S. Public Health Service, Washington, D. C.
- USEPA (U.S. Environmental Protection Agency), 1985. Rates, Constants, and Kinetics Formulations in Surface Water Quality Modeling, 2nd ed., Report EPA/600/3-85/040, Environmental Research Laboratory, Athens, Ga.

APPENDIX C

AERIAL PHOTOGRAPHS OF EXISTING LOCKS AND DAMS $\underline{1}^{\prime}$

.

2

 $[\]underline{1}/$ Source of photographs is recreational boat dock surveys conducted by the Pittsburgh and Huntington Districts of the Corps.


Figure C-1. Recent aerial photograph of Allegheny L&D No. 7 (FERC No. 7914).

£



Figure C-2. Recent aerial photograph of Allegheny L&D No. 4 (FERC No. 7909).



Figure C-3 Recent aerial photograph of Alleghny L&D No. 3 (FERCNo. 4474).



Figure C-4. Recent aerial photograph of Allegheny L&D No. 2 (FERC No. 4017).



Figure C-5. Recent aerial photograph of Opekiska L&D (FERC No. 8990).



Figure C-6. Recent aeria) photograph of Hildebrand L&D (FERC No. 8654)



Figure C-7. Recent aerial photograph of Point Marion L&D (FERC No. 7660).

C-7



Figure C-8. Recent aerial photograph of Maxwell L&D (FERC No. 8908).







Figure C-10. Recent aerial photograph of Emsworth L&D (FERC No. 7041).



Figure C-11. Recent aerial photograph of Dashields L&D (FERC No. 7568).



Figure C-12. Recent aerial photograph of Montgomery L&D (Competing applications - FERC Nos. 2971 and 3490).



Figure C-13. Recent aerial photograph of New Cumberland L&D (Competing applications - FERC Nos. 6901 and 10332).



Figure C-14. Recent aerial photograph of Pike Island L&D (FERC No. 3218).



Figure C-15. Recent aerial photograph of Willow Island L&D (Competing applications - FERC Nos. 6902 and 9999).



Figure C-16. Recent aerial photograph of Belleville L&D (FERC No. 6939).



Figure C-17. Recent aerial photograph of Gallipolis L&D (Competing applications - FERC Nos. 9042 and 10098).



Figure C-18. Recent aerial photograph of Muskingum L&D No. 3 (FERC No. 6998).

..



Figure C-19. Recent aerial photograph of Tygart Dam (Competing applications FERC No.7307 and 7399).

APPENDIX D

ACCESS FACILITIES INVENTORIED IN THE OHIO RIVER ACCESS STUDY

Navigational pool	Launch ramp fact1ftties	Launch Tanes	Unuseable launch ramp facilities	Launch 1anes	Total/useable launch lanes	Car/trailer parking spaces	Total/useable car trailer spaces	Car parking spaces	Boat dock (wet berths)		
ew Cumberland (Chio)											
Public	5	5			5	45	4 5	132	143		
Private	1	1	1	1	*	25	L.L.L.L.L.L.L.L.L.L.L.L.L.L.L.L.L.L.L.	20	1 - 5 n		
Total	6	5	Í	Î	5	70	45	167	143		
New Cumberland (Wes	t Virginia)			•			10	100	145		
Public	3	4	1	1	3	40	60	60	60		
Private	2	ź	-	•	2	25	25	87	97		
Total	5	6	1	1	5	65	65	147	147		
Pike Island (Ohio)									······································		
Public	2	3	1	Ŧ	2	135	60	110	76		
Private	3	3	F	-	3	60	60	70	96		
Total	5	6]	1	5	195	120	190	162		
Pike Island (West	Virginia)		-	-	•	270	110	100	102		
Public	ຶ 3໋	4			4	95	95				
Private	1	1			i	40	40	50	50		
Total	4	5			5	135	135	50	50		
Hannibal (Ohio)	·····				· · · · · · · · · · · · · · · · · · ·						
Public	3	4			4	135	135	6 1			
Private						30	30	45	71		
Total	3	4			4	165	165	105	. 71		
Hannibal (West Virg	inia)						200	100	71		
Public	1	2			2	15	15	46			
Private	6	6	1	1	5	25	10	50	03		
Total	7	8	1	1	7	40	25	96 96	9 3		
Willow Island (Ohio)								<u></u> ,,			
Public Deducto	5	8	2	2	6	106	105	310	5		
Tatal		~	•	•	-						
IULDI William Teland Alast	3 Mi	Ø	Z	Z	b	106	106	310	5		
WEST WEST	virginiaj	10				4 1 -					
PUDITC	ъ	18			18	170	170	6 5	51		
171VATE T-+-1	0										
lotat	8	18			18	170	170	65	51		

.

Table D. Access facilities inventoried in the Ohio River access study.

-

Table D. (Continued) _____ , .<u>___</u>

-

Navigationa] pool	Launch ramp facilities	Launch Tanes	Unuseable launch ramp facilities	Launch Janes	Total/useable launch lanes	Car/trailer parking spaces	Total/useable car trailer spaces	Car parking spaces	Boat dock (wet berths)
Belleville (Ohio)				·····					
Public	5 .	10	2	2	8	287	237	320	50
Private	8	8	4	4	4	8	8	120	140
Total	13	18	6	6	12	295	245	440	190
Belleville (West V	irginia)						F 19		150
Public	3	10			10	136	136	30	40
Private						30	20	45	90
Iotal	3	10			10	166	166	75	138
Racine (Ohio)		-+				<u></u>	·····		
Public		1			1	10	10	115	
Total		1					••		
Decimo (Nort Vinnie	1	1			ł	10	10	115	
naciae (nesc virga Dablic	114)					100			
Paivata	i i	4			4	100	100		
Total	1	4			4				
Gallipolis (Ohio)			• • • • • • • • • • • • • • • • • • •						
Public	Q	14	1	2	12	140	140	210	100
Private	-	1.7	*	č	16	147	142	513	120
Intal	0	14	1	2	12	140	140		100
Gallipolis (West Vi	minia	47	I	2	12	147	143	219	120
Public	3	4			Л	50	50	110	70
Private	Ŷ.	•			T T	50	υc	110	70
Total	3	4			4	50	50	116	70
Greenup (Ohio)			· · · · · · · · · · · · · · · · · · ·						
Public	5	12			12	140	140	415	282
Private					12	110	140	415	<u>, 14</u>
Total	5	12			12	140	1#0	A15	246
Greenup (West Virgi	nia)					1 7U	170	417	640
Public	5	11				320		590	929
Private	-					20		300	232
Total	c	13				240		10	30

.

APPENDIX E

APPROACH USED TO MODEL THE IMPACTS OF DISSOLVED OXYGEN CHANGE ON FISH GROWTH

APPENDIX E

E.1 INTRODUCTION

A 3-tiered approach was taken to analyze the impacts of reduced dissolved oxygen content of river water on fishes (Section 4.1.2.1). The first, most simple tier was comparison of predicted dissolved oxygen concentrations along the river length after hydroelectric installations are completed to applicable state water quality standards for dissolved oxygen. The second analytical tier compared the estimated dissolved oxygen concentrations along the river length to the suite of data presented in the latest EPA water quality criteria document for dissolved oxygen (EPA, 1986). This criteria document includes species-specific data on the life stages generally thought to be most sensitive to low oxygen content of water (juveniles), and it recommends levels of protection somewhat more stringent than current state standards.

This appendix describes the third tier of our analysis of the impacts of dissolved oxygen change on fish, which involves bioenergetics modeling. The oxygen-, temperature-, and sizedependent energetics of juvenile fish growth is modeled to estimate alterations in fish biomass production over a typical growing season due to projected effects of hydroelectric development on dissolved oxygen concentrations of the upper Ohio River system. Bioenergetics models have attained widespread use for estimating impacts of environmental conditions on fishes (Kitchell et al., 1977; Adams and Breck, in press).

Modeling is advisable for several reasons. First, because the impact of dissolved oxygen change on fish metabolism and growth is incremental and continuous rather than discrete over concentrations often found in the natural environment. That is, reduction in dissolved oxygen content of the water, even at generally accepted levels above water quality standards, can have some effect. The effect is small at high oxygen concentrations and it becomes more pronounced at lower concentrations. There is a more discrete "critical oxygen concentration" (usually, but not always, below water quality standards) below which oxygen metabolism decreases rapidly. Second, it is useful to use models to include the important influence of water temperature on fish metabolism throughout the range of oxygen concentrations. Temperature determines rates of metabolism and the levels of oxygen availability that are "critical." A third reason to use models is to incorporate the influence of fish size on metabolism and growth. Models can integrate size-specific effects as fish grow and yield an endpoint of accumulated biomass at the end of a growing season. It is also possible to include effects of various feeding rates, which can be slowed by depressed oxygen levels in the water, and other water quality features (e.g., ammonia) in bioenergetics models.

E.2. CHOICE OF THE BIOENERGETICS-BASED POND CULTURE MODEL

EPA's dissolved oxygen criteria document (EPA, 1986) notes that "A detailed discussion and model for evaluating interactions among temperature, dissolved oxygen, ammonia, fish size, and ration on the resulting growth of individual fish (Cuenco, Stickney, and Grant, 1985a-c) provides an excellent, in-depth evaluation of potential effects of dissolved oxygen on fish growth." Discussions with the author of the EPA document (personal communication, G. Chapman, EPA, Newport, Oregon) amplified the belief that this model is the best currently available for evaluating impacts of changing dissolved oxygen concentrations on long-term fish health in the context of other system dynamics. We, therefore, chose the Cuenco, Stickney, and Grant model for this analysis of the cumulative impacts of installing hydroelectric facilities on the upper Ohio River system.

The Cuenco, Stickney, and Grant model was developed for evaluating growth of channel catfish in a simple pond environment in Texas under varying conditions of aquaculture. Some conditions were controlled (e.g., stocking rate, artificial feeding rate, initial size distribution), whereas others changed according to season, pond productivity, and fish growth and competition (e.g., temperature, light, ammonia concentration, dissolved oxygen, natural food). The model includes an individual fish growth component and a population growth component; a pond environment submodel generates many of the pond's environmental features through a growing season. The size distribution of stocked fish at the end of the growing season and the percentage of fish above a marketable size were the main endpoints of the original application.

Although the Cuenco, Stickney, and Grant model contains elements specific to channel catfish aquaculture in ponds, it is capable of being generalized to other environments and other species. The environmental features with which the fish interact in the model are common

to fish living in any water body. Input parameters such as artificial feeding can be replaced with estimates of natural food availability. Oxygen concentrations that are driven in the original model by biodegradation of uneaten food and phytoplankton can be replaced by inputs from another water quality model of interest, such as from the Ohio River. The attributes are similar wherever applied: the ability to estimate the capacity of a valued fish species to attain weight gain under an array of environmental conditions. For our application, changing dissolved oxygen is of central importance. Growth parameters in the model are specified according to the fish species; parameters in the Cuenco, Stickney, and Grant formulation for population growth were estimated for channel catfish. We substituted parameter estimates for other species as well.

E.3 MODEL DESCRIPTION

E.3.1 Individual Fish Bioenergetics Mode)

The growth of fish is a complex process that represents the net outcome of a series of physiological and behavioral processes beginning with food intake and culminating in deposition of animal tissue (Brett, 1979). For a given time period, the energy consumed must equal the sum of energy losses in the form of wastes, the energy used to perform work (e.g., swimming), and the change in energy of the body (Warren and Davis, 1957):

$$\mathbf{I} = \mathbf{E} + \mathbf{M} + \mathbf{G} \,, \tag{1}$$

where I is the food ingested, E is the total amount of waste excreted (feces, urine, wastes lost through the gills and skin), M is total metabolism, and G is the change in energy of the body or growth. These terms are measured in kilocalories (kcal) of energy (1 kcal = 4.1868 kJ) and refer to a particular time period, T, in days. Intake and excretion are often handled together, and the equation can be rearranged to solve for growth, as:

$$G = AR + I - N, \qquad (2)$$

where AR is the assimilation ratio for the particular food ingested by the fish.

All of these processes are a function of body size:

Ì5Ϊ

where EAT is relative food consumption rate in kcal of food per day per kcal of fish, RES is relative respiration rate in kcal respiration per kcal of fish, GR is relative growth rate in kcal growth per day per kcal of fish, W is mean energy of the body of the fish in kcal, and T is the period of time in days. Substituting Equations 3-5 into Equation 2, we get:

$$GR = AR * EAT - RES .$$
(6)

Thus, from a bioenergetic standpoint, the growth of a fish can be expressed as the difference between the amount of food energy the fish consumes and the distribution of this food into various losses and uses, each of which is proportional to the size of the fish at the moment.

Although Equation 6 is a useful relationship, it describes the fish only under one set of environmental conditions. Growth in freshwater fishes is governed by a variety of environmental factors (Brett, 1979) including water temperature, dissolved oxygen concentration, photoperiod, unionized ammonia concentration, and food availability. Genetic features of the species also determine characteristic growth responses. Water temperature sets the pace of metabolic activity in cold-blooded organisms. Dissolved oxygen is needed for respiration. Unionized ammonia is an index of waste accumulation that depresses growth.

In formulating a growth model for aquaculture, Cuenco, Stickney, and Grant (1985a-b) identified size or weight of the fish (W), water temperature (TP), dissolved oxygen concentration (DD), food (FD), and unionized ammonia concentration (AM) as most important. They likewise will be the most important for evaluating effects of dissolved oxygen changes in natural environments.

Because the effect of any factor on growth is through its influence on food consumption and metabolism, the basic approach used by Cuenco, Stickney, and Grant to structure the model was to define the biological relationships of each factor to food consumption and metabolism and then find an appropriate mathematical expression to approximate the known biological relation. The specific formulations are found in the papers.

Dissolved oxygen is a limiting factor characterized by dependent and independent states. Increasing DO up to a critical value (DOCRIT) at constant levels of all other factors increases the fish's appetite and feeding. There is also a lower level of DO (DOZERO) at which appetite, feeding, and metabolism cease. DOCRIT and DOZERO are both dependent on the respiration rate of the fish (Davis, 1975; Herreid, 1980). High metabolic rates (e.g., from high temperatures) increase DOCRIT and DOZERO, while low metabolic rates depress them.

E.3.2 Population Growth and Pond Culture Models

The pond culture model used for this impact assessment is a combination of the individual fish bioenergetic model just described (Cuenco, Stickney, and Grant 1985a,b), with a population growth model and a pond environment model (Cuenco, Stickney, and Grant, 1985c). The population growth model was developed from the individual fish bioenergetics model by including social interactions, mostly competition for food. It also includes the influences of seasonal and diel temperature fluctuations and of diel oxygen oscillations. The pond environment model calculates various environmental characteristics of a pond ecosystem during a growing season. Many features of the population growth and pond environment submodels are not relevant to the Ohio River system study and were removed or set to constants. The output values, however, were retained.

The following assumptions and limitations define the set of conditions to which the population growth model applies:

- Only the growth stage from fingerling (past yolk sac and well beyond first feeding) to the end of the first growing season is considered.
- No reproductive activity occurs during the growth phase.
- Because diet quality is complex and difficult to quantify, a fixed diet of high quality is assumed.
- Lethal factors are absent.
- 5. The species being evaluated exists in a monoculture.

For the pond culture model, growth of a known number of fish with a specified weight and coefficient of variation is started at a certain date in a known water volume. The water volume is assumed to be well mixed and all water quality features uniform in space but changing in time. The seasonal temperature cycle determines the end of the growing season. Fish growth is terminated at temperatures known to inhibit weight gain of the species.

The minimum diel DO should be used as DO input to the growth model. A few revealing studies have shown that diel fluctuations in DO result in lower growth rates and a higher critical level of DO for fishes than would be expected from the average DO (Fisher, 1963 for coho salmon: Stewart, Shumway, and Dondoroff, 1967 for largemouth bass; Whitworth, 1968 for brook trout). Periods of higher DO in a daily cycle do not compensate for periods of low DO. This require-ment is difficult to apply to the Ohio River assessment, however, because data on diel variations are scanty. Where they do occur, the diel change seems to amount to only a few tenths of a milligram per liter. The dissolved oxygen model used to calculate concentrations in the water quality analyses (Section 4.1.1 and Appendix B) also do not consider diel variations. Thus, the bioenergetics model is run using typical (average) values.

Dutputs of the pond culture model that are used to evaluate effects of DO changes in the upper Ohio River system are (1) average weight of fish at the end of the growing season with coefficient of variation, (2) average absolute growth rate for the growing year, (3) the total weight of the initial group of fish at the end of the growing season kilograms per hectare (kg/ha), (4) the total harvestable weight of the initial group of fish at the end of the growing season (kg/ha) (harvestable being above a certain size limit appropriate for the fishery), (5) average yield (kg/ha/day), and (6) average harvestable yield (kg/ha/day). All of

these results are to be considered relative among the scenarios tested, and not true predictions of yields. Average weight at the end of the growing season appears to summarize the results adequately for assessment of impacts from alternatives.

E.4. PARAMETER ESTIMATION

The bioenergetic growth model requires 17 species-specific parameters. The model as developed by Cuenco, Stickney, and Grant was tested for four species: brown trout, rainbow trout, tilapia, and channel catfish. To estimate those parameters, Cuenco, Stickney, and Grant used data for the particular species when available. In the absence of data for a particular species, available data for the most closely related species or "best judgement" were used. Parameter estimates and data sources for species used by Cuenco, Stickney, and Grant in developing and testing the individual fish bioenergetics model are given in those papers. The channel catfish parameters only were used by Cuenco, Stickney, and Grant for extending the bioenergetics model to a population growth model and the pond culture model.

We reparameterized the model for species other than channel catfish that are important for the upper Ohio River system. Although channel catfish is considered by EPA (1986) to be an oxygen-sensitive species, we assumed that cool-water species such as walleye and sauger will be the most sensitive of those in the upper Ohio River system to low dissolved oxygen and high temperature conditions. Parameter values intermediate between those for channel catfish and two trout species, brown and rainbow, were selected as reasonable for the cool-water sauger or walleye. These values were as follows: WEF = -0.30, A = -80, B = 30, WER = 0.20, C = 0.14, D = 0.14, AR = 0.75, RMC = 0.37, TEOPT = 22, TEHIGH = 32 (these parameters are defined in Cueno, Stickney, and Grant, 1985 a-c). All other parameters remained the same.

E.5 APPLICATION TO THE UPPER OHIO RIVER SYSTEM

Typical monthly dissolved oxygen values for the upper Ohio River system as estimated by water quality modeling (Section 4.1.1 and Appendix B) for various no-hydro and operating scenarios are used to drive the fish growth model. These estimates are made at representative monthly times over the fish growing season from April to November, allowing interpolation of daily DO values. Temperatures used in the water quality modeling are also used for estimating fish growth. Three cases were examined: Case One in which river flows reach summer extremes of the 7010 summer low flow, Case Three in which a more moderate summer low flow occurs from mid-June through mid-September, and Case Three in which monthly average values are used throughout (Section 4.1.2.1.3).

For purposes of this assessment, the upper Ohio River system is considered to be composed of 55 "ponds," each either the upper or lower half of a navigation pool. The upper half is the tailwater of the upstream dam (often having DO values elevated above those in the lower section because of aeration at many dams). Exceptions are the pools bounding the study area: the upper boundary on the Monongahela River is the Tygart tailwater; only the lower pool segment is considered for the uppermost projects evaluated on the Allegheny River (Allegheny L&D No. 7) and Muskingum River (Muskingum L&D No. 3); and only the tailwater segment is considered in the Greenup pool below Gallipolis. There are two upper sections of the Emsworth pool: one in the Allegheny and the other in the Monongahela. Water quality input values for the fish growth model are averaged within the half-pools.

A hatch of juvenile fish prior to the summer-fall period of depressed dissolved oxygen in the system is assumed in each half-pool according to the life history of the fish species. For each species, 3-gram fish (average) are assumed on June 1. The growth of these juvenile fish is followed through the growing season in each half-pool as the half-pool's temperatures and dissolved oxygen concentrations for the modeled scenario change.

Each half-pool ends the growing season with a set of values describing growth of the DOsensitive juvenile fish there, as listed above. Differences in these values, as percent decline from preproject conditions, are plotted along the river system (Figures 4.1.1-3 and 4.1.1-6), in a manner that allows comparison among alternative scenarios (e.g., current nohydro condition, hydro installations as proposed, selected installations only, with and without artificial aeration, etc.). The analysis is thus a comparative one in which it is the relative differences that are important rather than the actual biomass production. Thus, only the relative results are presented. Although it is reasonable to ask about the effect on populations of a reduction in annual accumulation of growth, this question has not been addressed in this analysis. It is clear, however, that failure to grow sufficiently during the growing season can cause overwinter mortality in several species that have been studied, including smallmouth bass (Shuter et al., 1980) and largemouth bass (Adams, McLean, and Huffman, 1982).

E.6 REFERENCES

Adams, S. M., R. B. McLean, and M. M. Huffman, 1982. Structuring a Predator Population Through Temperature-Mediated Effects on Prey Availability, Can. J. Fish. Aquat. Sci. 39:1175-184.

Adams, S. M., and J. E. Breck, in press. Bioenergetics, Chapt. 12 in Methods for Fish Biology, ed. C. B. Schreck and P. B. Moyle, American Fisheries Society, Bethesda, Md.

Brett, J. R., 1979. Environmental factors and growth. pp. 599-675, in Fish Physiology, Vol. VIII, Bioenergetics and Growth, ed. W. S. Hoar, D. J. Randall, and J. R. Brett. Academic Press, New York.

Cuenco, M. L., R. R. Stickney, and W. E. Grant, 1985a. Fish Bioenergetics and Growth in Aquaculture Ponds: I. Individual Fish Model Development, Ecol. Modelling 27, 169-90.

Cuenco, M. L., R. R. Stickney, and W. E. Grant, 1985b. Fish Bioenergetics and Growth in Aquaculture Ponds: II. Effects of Interactions Among Size, Temperature, Dissolved Oxygen, Unionized Ammonia, and Food on Growth of Individual Fish, Ecol. Modelling 27, 191-206.

Cuenco, M. L., R. R. Stickney, and W. E. Grant, 1985c. Fish Bioenergetics and Growth in Aquaculture Ponds: III. Effects of Intraspecific Competition, Stocking Rate, Stocking Size, and Feeding Rate on Fish Productivity, Ecol. Modelling 28, 73-95.

Davis, J. C., 1975. Minimal Dissolved Oxygen Requirements of Aquatic Life with Emphasis on Canadian Species: A Review, J. Fish Res. Board Can. 32, 2295-332.

EPA (U. S. Environmental Protection Agency), 1986. Ambient Water Quality Criteria for Dissolved Oxygen, EPA 440/5-86-003.

Fisher, R. J., 1963. Influence of Oxygen Concentration and of its Diurnal Fluctuations on the Growth of Juvenile Coho Salmon, Masters thesis, Oregon State University, Corvallis.

Herreid, C. G., II, 1980. Hypoxia in Invertebrates: A Review, Comp. Biochem. Physiol. 67, 311-20.

Kitchell, J. G., et al., 1974. Model of Fish Biomass Dynamics, Trans. Am. Fish. Soc. 103, 786-98.

Shuter, B. J., et al., 1980. Stochastic Simulation of Temperature Effects on First-Year Survival of Smallmouth Bass, Trans. Am. Fish. Soc. 109, 1-34.

Stewart, N. E., D. L. Shumway, and P. Doudoroff, 1967. Influence of Oxygen Concentration on the Growth of Juvenile Largemouth Bass, J. Fish. Res. Board Can. 24, 475-94.

Warren, C. E., and G. E. Davis, 1967. Laboratory Studies on the Feeding, Bioenergetics and Growth of Fish. pp. 175-214, in The Biological Basis of Freshwater Fish Production, ed. S. D. Gerking, Blackwell, Dxford.

Whitworth, W. R., 1968. Effects of Diurnal Fluctuations of Dissolved Oxygen on the Growth of Brook Trout, J. Fish. Res. Board Can. 25, 579-84.

APPENDIX F

DIAGRAMS SHOWING APPLICANTS' PLANS FOR ENHANCING RECREATION FACILITIES



Figure F-1. Proposed recreational enhancements at Allegheny River Lock and Dam No. 7 Project. Source: Allegheny Hydropower, Inc., November 1987.



Figure F-2. Proposed recreational enhancements at Allegheny River Lock and Dam No. 4 Project. Source: Allegheny County, November 1984.

.

F-2



Figure F-3. Proposed recreational enhancements at Allegheny River Lock and Dam No. 3 Project. Source: Allegheny Valley orth Council of Governments, November, 1987.

F-3



Figure F-4. Proposed recreational enhancements at Allegheny River Lock and Dam No. 2 Project. Source: City of Pittsburgh, October 1987.



Figure F-5. Proposed recreational enhancements at Tygart Dam Project (FERC No. 7307). Source: City of Grafton, 1987.

.

F.S



Figure F-6. Proposed recreational enhancements at Tygart Dam Project (FERC No. 7399). Source: Noah Corporation, 1987d.

F-6



Figure F-7. Proposed recreational enhancements at Opekiska Lock and Dam Project. Source: Noah Corporation, 1987c.

F-7




Figure F-9. Proposed recreational enhancements at Point Marion Lock and Dam Project. Source: Noah Corporation, 1987a.



Figure F-10. Proposed recreational enhancements at Maxwell Lock and Dam Project. Source: Pennsylvania Renewable Resource, Inc., December 1987b.



.

Figure F-11. Proposed recreational enhancements at Monongahela River Lock and Dam No. 4 Project. Source: Pennsylvania Renewable Resources, Inc., December 1987a.











Figure F-14. Proposed recreational enhancements at Dashields Lock and Dam Project. Source: County of Allegheny, November 1987.







Conceptual Layout Proposed Hontgomery Hydro Project









Figure F-18. Proposed recreational enhancements at New Cumberland Lock and Dam Project (FERC No. 10332). Source: WV Hydro, November, 1987a.



Figure F-19. Existing fishing access facilities at Pike Island Lock and Dam Project. Source: City of Orrville, 1982.



Figure F-20. Proposed recreational enhanceents at Pike Island Lock and Dam Project. Source: City of Orrville, 1982.



Figure F-21. Proposed recreational enhancements at Willow Island Lock and Dam Project (FERC No. 6902). Source: Burgess and Niple, September, 1986.

.



Figure F-22. Proposed recreational enhancements at Willow Island Lock and Dam Project (FERC No. 9999).

-



Figure F-23. Proposed recreational enhancements at Belleville Lock and Dam Project. Source: W. M. Lewis and Associates, November 1984.



Figure F-24. Proposed recreational enhancements at Gallipolis Lock and Dam Project(FERC No. 9042). Source: Gallia Hydro Partners, November, 1987.

.



Figure F-25. Proposed recreational enhancements at Gallipolis Lock and Dam Project (FERC No. 10098). Source: WV Hydro, Inc., 1987.



Figure F-26. Proposed recreational enhancements at Muskingum Lock and Dam No. 3 Project. Source: Mitex, November, 1987.

APPENDIX G

.

ASSESSMENT OF THE ABILITY OF LOCAL LABOR FORCES TO PROVIDE CONSTRUCTION WORKERS

APPENDIX G

ASSESSMENT OF LOCAL LABOR FORCES TO PROVIDE CONSTRUCTION WORKERS

A key aspect of the socioeconomic impacts of the proposed hydropower projects is the ability of local labor forces to provide construction workers for all projects. If the labor requirements of the projects are greater than the supply of workers in the area, a substantial number of workers would have to relocate to the area during the construction period. Such an influx of new, temporary residents would initiate a complex series of impacts, including an increased demand and competition for housing and a requirement for increased public services (e.g., fire and police protection, capacity for additional students in local schools).

The purpose of this analysis is to determine if the combination of projects addressed in the DEIS would be likely to result in the in-migration of a significant number of workers during the construction period. The potential problem of in-migration arises when the proposed projects are considered cumulatively. No single project among those proposed would require more construction workers than reside within commuting distance of the construction site. Simultaneous construction of several projects within a given area, however, could generate a combined demand for workers that exceeded the number of available workers in the local area. In this case, additional workers would have to be brought in from other areas.

Some uncertainty is involved in several of the procedures used in this analysis. Every effort has been made to ensure that the errors associated with all assumptions are conservative (i.e., that they tend to overestimate the demand for workers and underestimate the supply). It is felt that adherence to this principle will help ensure the validity of the results of the analysis.

G.1 ESTIMATING WORK FORCE REQUIREMENTS

Project work force requirements were estimated using a procedure developed by FERC based on an analysis of the agency's experience with hydropower projects (personal communication, Jim Haimes, FERC, Division of Environmental Analysis, March 3, 1988). This analysis found that (1) approximately 20 percent of a project's total direct construction cost was paid as wages and salaries of construction workers and (2) the average wage (including all benefits) of a construction worker was about \$40,000 per year, or \$3,333 per month. Because of the uncertainties inherent in these and other assumptions used in the estimating process, a range of work force requirements was obtained by calculating low and high figures based on 15 percent and 25 percent, respectively, of the construction costs being paid as wages and salaries.

The estimated construction work force required for each project is shown in Table G-1. The total direct construction costs were obtained from the project applications and subsequent materials submitted by the applicants and adjusted to 1988 dollars using an annual adjustment rate of 4 percent. Low and high estimates of construction wages and salaries were calculated by assuming that between 15 percent and 25 percent of the direct construction costs would be paid to workers. The estimates of wages and salaries were divided by the number of months in the construction period (from project applications and subsequent materials provided by the applicants). These estimates of total monthly wages were then divided by the average monthly wage (\$3,333) to obtain low and high estimates of the number of workers required to construct each project.

Proposed project construction schedules submitted by the applicants were used to assess the potential for simultaneous construction of several projects in a given area. This information is summarized in Figure G-1. Although there is considerable variety in the proposed schedules, it is likely that construction activities would be under way on many of the projects during some months. In addition, unexpected developments in obtaining project financing and designing the facilities could alter the schedules. Consequently, a worst-case scenario was adopted, and it was assumed that all projects would be constructed simultaneously. Under this scenario, all workers listed in Table G-1 would be required at the same time.

6.2 ESTIMATING THE SUPPLY OF CONSTRUCTION WORKERS

It was assumed that construction workers would be willing to commute to jobs up to 40 miles from their residences. The number of construction workers expected to be available was estimated for all counties whose main population center was within 40 road miles of any of the proposed projects. The overall 1986 unemployment rate for the county was multiplied by the

	Estimated (in thousand:	project c s of 1988	osts dollars)			
	Total	Esti wages &	mated salaries	Nootha of	Estimated constructi	number of on workers
Project	construction]/	low	high	construction	low	high
Gallipolis (9042)2/	71,699	10.755	17,925	33	98	163
Gallipolis (10098)2/	74,803	11.221	18,701	24	140	234
Belleville (6939)	82,554	12,363	20,639	28	133	221
Willow Island (6902)2/	NÁ 3/					
Willow Island (9999)2/	74,803	11.221	18.701	22	153	255
Muskingum L&D No. 3	19,241	2,886	4,810	22	39	6 6
Pike Island (3218)	46,433	6,965	11,608	25	84	139
New Cumberland (6901)2/	93,904	14,086	23,476	32	132	220
New Cumberland (10332)2/	76,180	11,427	19,045	39	88	147
Montgomery (2971)2/	54,737	8.211	13.684	18	137	228
Montgomery (3490) ^T	33,746	5,062	8,436	26	58	97
Dashields (7568)	44,965	6.745	11.241	25	81	135
Emsworth (7041)	28,169	4.225	7.042	24	53	88
Allegheny L&D No. 2 (40)	7) 25,476	3.821	6.369	30	38	64
Allegheny L&D No. 3 (447	4) 30.359	4.554	7.590	33	41	69
Allegheny L&D No. 4 (790	9) 28.706	4.306	7.176	23	56	94
Allegheny L&D No. 7 (791	4) 28.077	4.213	7.019	22	57	96
Monongahela L&D No. 4 (4	675)12.348	1.852	3,087	24	23	39
Maxwell (8908)	14.641	2,196	3,660	23	29	48
Point Marion (7660)	NA 3/			24	160 4/	130 4/
Hildebrand (8654)	10.304	1.546	2.576	21	22	37
Opekiska (8990)	10,776	1.616	2,694	20	24	40
Tygart (7307)2/	23.842	3.576	5,961	29	37	62
Tygart (7399) <u>2</u> /	58,155	8,723	14,539	30	87	145

Table G-1.	Estimated	construction w	work force	requirements.
------------	-----------	----------------	------------	---------------

1/ Source: Project applications and materials subsequently provided by project applicants. Costs were adjusted to 1988 dollars using an escalation rate of 4 percent per year.

2/ Competing applications.

3/ NA = Not available.

4/ Source: FERC, Office of Hydropower Licensing, Division of Environmental Analysis, 1984. Environmental Assessment, Point Marion Lock and Dam Project, FERC No. 7660--Pennsylvania.

Months of Construction (0	÷	Li	ce	ns	e	aŗ	pr	٥.	/a]).									•																											E
Project	0	1	2	3	4	1 5	5 €	;)	8	39	0	i	2	3	4	5	6	78	39	0	1	2	3	4 5	6	7	8	9	0	1	2 :	3 4	1 5	6	7	8	9	0	1 3	2 3	4	5	6	7	8	9 0	01
Gallipolis (9042) 2/	*	*	+	*			• •		. ,	- +	*	*	*	*	*	*	*	* *	+ +	*	*	*	*	* 1		*	*	*	*	×	* ;	•															
Gallipolis (10098) 2/		÷	*	*	r +	•	k 3			*	*	*	*	*	*	*	*	* *	е и с	*	*	*	* :	* *							. .												+				
Willow Island (6902) 2/	N	A																			-	-	*			-	~	~	~	-	~ '	• •		-	-	Î	-	-			• • •		-				
Willow Island (9999) 2/		• • •				,	• •			* *	*	*	*	*	*	*	*	* 1	• •	*	*	*	*	* *	r 🕈	*																					
Muskingum LAD No. 3													*	*	*	*	* ·	* *	• •	*	*	*	* :	* *	•	*	: * : *	*	*	*	* 1 * 1	k 1 k 1	• • •		,												
New Cumberland (6901) 2/	,			*			e y	- 1		• •	*	*	*	*	*	*	*	* *	• •	*	*	*	*	* *	: #	*	*	*	±	*	* 1	h 4	ł														
New Cumberland (10332) 2	1							•	. 1	t #	*	*	*	*	*	*	*	* 1	t #	*	*	*	* :	* *	t #	*	*	*	*	*	* *	k 1	• •	*	*	*	*	*	• •	* 1	* *	*					
Montgomery (297)) 2/ Montgomery (3490) 2/												*	÷	*	•	*	* *	* 1	• •	*	*	*	* :	* *	. # . #	*	: * : *	*	*	*	* 1	я к 3	t #		. *												
Dashields (7568)										t A	*	*	*	*	*	*	*	* 1	e #	+	*	*	*	* *	*	*	*	*	*	*																	
Emsworth (7041)													*	*	*	*	* ·	* *	* *	*	*	÷	* :	* *	•	*	*	*	*	*	* :	k i	t st L	•													
Allegheny L&D No. 2 (401 Allegheny L&D No. 3 (447	/) 41											F	π	۳	*	*	*			•	*	÷	* :	* *	• *	*	*	*	*	*	* 1			•	*	*	*	*	* 1	* *		*	*	*	×	* i	* *
Allegheny L&D No. 4 (790	ġý						1	- 1		t 1	*	*	*	*	*	*	*	* 1	• •	*	*	*	*	* *	*	*	*	*																			
Allegheny L&D No. 7 (79)	4)	Ľ١					•			* *	*	*	*	*	*	*	* • ·	* 1		· *	*	*	* :	* *	. *		*	*																			
Maxwell (8908)	07	ə)										*	*	*	*	•	*	* 1	• •	*	*	*	* 1	* *	r *	*	*	*	*	ŧ	* i	. ,	ł														
Point Marion (7660)										_	_	_	_		_					_			. '	* *	*	*	*	*	*	*	* i	k 1	* *	*	*	*	*	*	* •	* *	* *	*	*	*	*		
Hildebrand (8654) Opekiska (8990)									1	• *	*	*	*	*	*	*	* *	* 1 * 1	; ≉ ; •	* *	*	*	* 1	* *	· *	*	*	*																			
Tygart (7307) 2/	*	*	*	*	r 🖈	r 1	. ,			* *	*	*	*	*	*	*	*	* +	• +	*	*	*	* 1	* *	*	*	*	*																			
Tygart (7399) <u>2</u> /													*	*	*	*	*	* 1	+ *	*	*	*	* :	* 1	*	*	*	*	*	*	* 1	k 1	* *	*	*	*	*	*	* 1	ł							

.

Figure 1. Estimated project construction time lines. $\underline{1}/$

1/ Source: Project applications and subsequent materials submitted by project applicants.

2/ Competing applications.

.

1980 percentage of the county's labor force which worked in the construction industry to produce an estimate of the number of unemployed construction workers. The resulting estimate is likely to be slightly low because unemployment in construction trades is generally higher than overall unemployment. The estimated number of unemployed construction workers in each county is shown in Table G-2.

G.3 MATCHING THE SUPPLY AND DEMAND

To determine whether the construction work force requirements of the proposed projects could be met without significant relocation or in-migration of workers, the estimated supply of construction workers in each county was apportioned to the projects within 40 road-miles until the supply was exhausted or the estimated demands were filled. In an attempt to compensate for uncertainties inherent in the analysis, "worst-case" conditions were assumed whenever appropriate. The high estimates of work force requirements were used, and, in cases of competing applications for projects at a single site, the project with the highest work force requirement was included.

Because it is reasonable to assume that workers would prefer to work as close as possible to their homes, the apportionment of unemployed workers to projects was performed in several iterations. First, an attempt was made to satisfy each project's demand for workers using the available workers in the county in which the project was located. Any remaining unemployed construction workers in the county were then considered available for employment at other projects. In subsequent iterations, projects with unmet work force requirements drew workers from the next closest counties that had remaining supplies of unemployed workers.

The results of the apportionment are shown in Tables G-3 and G-4 on the basis of this analysis, construction work force requirements of the proposed projects could be met by available workers living within commuting distance of each project, even if all projects were constructed concurrently. Thus, no significant relocation of workers within the region or inmigration of workers from outside the region is expected to occur. It should be noted that no information was available to estimate either the demand for or supply of construction specialists (e.g., welders, pipe fitters, electricians). It is possible that an insufficient number of such specialists will be available in some local areas and that appropriate workers will have to be brought in from outside the areas. However, the number of specialists imported is likely to be small and should not result in significant socioeconomic impacts in the area.

County	Overall unemployment (1986) <u>1</u> /	Construction as percent of labor force (1980) 2	Estimated construction unemployment
Gallia Obio	1.400	8.32	116
Meios, Ohio	1,100	11.33	125
Washington, Ohio	3,600	7.54	271
Mason, W. Va.	1,160	10.07	117
Jackson, W. Va.	1.200	9.58	115
Pleasants, W. Va.	270	11.88	32
Ritchie, W. Va.	520	8.93	46
Tyler, W. Va.	490	8.60	42
Wood, W. Va.	3,160	7.42	234
Belmont, Ohio	4,100	5.78	237
Jefferson, Ohio	3,000	4.74	142
Brooke, Ohio	710	4.68	33
Hancock, Dhio	720	2.95	21
Ohio, Ohio	1,560	5.94	93
Pittsburgh PMSA <u>3</u> /			
(Allegheny, Fayette,			
Washington, &			
Westmoreland, Pa.)	76,200	5.31	4,046
Armstrong, Pa.	3,100	4.99	155
Beaver, Pa.	8,100	5.37	435
Greene, Pa	1,700	7.30	124
Barbour, W. Va.	800	7.71	52
Harrison, W. Va.	2,430	7.00	170
Marion, W. Va.	2,320	5.79	139
Monongalia, W. Va.	1,370	5.25	12
Preston, W. Va.	1,230	7.51	94
iaylor, W. Va.	580	0.93	40

Table G-2. Estimated number of construction workers available by county.

.

- 1/ Sources: Ohio Bureau of Employment Services, 1987, County Labor Force Estimates by Month; Pennsylvania Department of Labor and Industry, 1987, Civilian Labor Force Series by County of Residence, 1977-1986; West Virginia Department of Employment Security, 1987, Monthly Report on the Civilian Labor Force, Employment, and Unemployment.
- 2/ Source: U.S. Department of Commerce, Bureau of the Census, 1983, 1980 Census of Population, Characteristics of the Population, General Social and Economic Characteristics.

3/ PMSA = Primary Metropolitan Statistical Area.

Project	Estimated maximum workers required	Counties supplying workers/ number of workers supplied <u>2</u> /
Gallipolis (10098) <u>1</u> /	234	Gallia, Ohio/116; Mason,
Rolleville (6030)	221	W. Va./11/; Meigs, Ohio/1
Willing Island (9999) 1/	221	NOUL, W. Ya./221 Diessante II Vo./22: Mond
RTTTOR TSTUNCE (5555) 1/	200	19 Va /12 Tulan 4 Va /12
		Difebia W Vo 786.
		Washington Dhio/122
Muskingum L&D No. 3	66	Washington, Dhio/65
Pike Island (3218)	139	Relmont Ohio/139
New Cumberland (6901) 1/	220	Hancock, W Va /21 - Brooke
		W. Va./33; Jefferson, Ohio/142; Pittsburgh PMSA/24
Montgomery (2971) 1/	228	Beaver, Pa./228
Dashields (7568)	135	Pittsburgh PMSA/135
Emsworth (7041)	88	Pittsburgh PMSA/88
Allegheny L&D No. 2 (4017)	64	Pittsburgh PMSA/64
Allegheny L&D No. 3 (4474)	69	Pittsburgh PMSA/69
Allegheny L&D No. 4 (7909)	94	Pittsburch PMSA/94
Allegheny L&D No. 7 (7914)	96	Armstrong, Pa./95
Monongahela L&D No. 4 (467)	39	Pittsburgh PMSA/39
Maxwell (8908)	48	Pittsburgh PMSA/48
Point Marion (7660)	180	Greene, Pa./90; Pittsburgh PMSA/90
Hildebrand (8654)	37	Monongalia, W. Va./36; Marion, W. Va./1
Opekiska (8990)	40	Monongalia, W. Va./36; Marion, W. Va./4
Tygart (7399) <u>1</u> /	145	Taylor, W. Va./40: Barbour, W. Va./62; Harrison, W. Va./43

Table G-3. Ability of local labor forces to meet construction work force requirements.

1/ Project with highest estimated work force requirement of two competing projects at this site.

2/ PMSA = Primary Metropolitan Statistical Area.

County	Estimated construction unemployment	Number of unemployed workers remaining after project assignments
Gallia, Ohio	116	0
Meigs, Ohio	125	124
Washington, Ohio	271	83
Mason, W. Va.	117	0
Jackson, W. Va.	115	115
Pleasants, W. Va.	32	0
Ritchie, W. Va.	46	Ō
Tyler, W. Va.	42	Ō
Wood, W. Va.	234	Ō
Belmont, Ohio	237	98
Jefferson, Ohio	142	0
Brooke, W. Va.	33	0
Hancock, W. Va.	21	0
Ohio, W. Va.	93	93
Pittsburgh PMSA 1/ (Allegheny, Fayette, Washington, &	,	
Westmoreland, Pa.)	4,046	3,509
Armstrong, Pa.	155	59
Beaver, Pa.	435	207
Greene, Pa.	124	62
Barbour, W. Va.	62	 G
Harrison, W. Va.	170	127
Marion, W. Va.	134	129
Monongalia, W. Va.	72	0
Preston, W. Va.	94	94
Taylor, W. Va.	40	0

Table G-4. Amount of local unemployed construction workers employed by projects.

1/ Primary Metropolitan Statistical Area.

APPENDIX H

· · · ·

•

.

LICENSE ARTICLES APPLICABLE TO OHIO RIVER PROJECTS

APPENDIX H

LICENSE ARTICLES APPLICABLE TO OHIO RIVER PROJECTS

H.1 FEDERAL ENERGY REGULATORY COMMISSION TERMS AND CONDITIONS OF LICENSE FOR UNCONSTRUCTED MAJOR PROJECT AFFECTING NAVIGABLE WATERS AND LANDS OF THE UNITED STATES [Form L-6 (Revised October, 1975)]

<u>Article 1</u>. The entire project, as described in this order of the Commission, shall be subject to all of the provisions, terms, and conditions of the license.

<u>Article 2</u>. No substantial change shall be made in the maps, plans, specifications, and statements described and designated as exhibits and approved by the Commission in its order as a part of the license until such change shall have been approved by the Commission: <u>Provided</u>, <u>however</u>, That if the Licensee or the Commission deems it necessary or desirable that said approved exhibits, or any of them, be changed, there shall be submitted to the Commission for approval a revised, or additional exhibit or exhibits covering the proposed changes which, upon approval by the Commission, shall become a part of the license and shall supersede, in whole or in part, such exhibit or exhibits theretofore made a part of the license as may be specified by the Commission.

<u>Article 3</u>. The project works shall be constructed in substantial conformity with the approved exhibits referred to in Article 2 herein or as changed in accordance with the provisions of said article. Except when emergency shall require for the protection of navigation, life, health, or property, there shall not be made without prior approval of the Commission any substantial alteration or addition not in conformity with the approved plans to any dam or other project works under the license or any substantial use of project lands and waters not authorized herein; and any emergency alteration, addition, or use so made shall thereafter be subject to such modification and change as the Commission may direct. Minor changes in project works, or in uses of project lands and waters, or divergence from such approved exhibits may be made if such changes will not result in a decrease in efficiency, in a material increase in cost, in an adverse environmental impact, or in impairment of the general scheme of development; but any of such minor changes made without the prior approval of the Commission, which in its judgment have produced or will produce any of such results, shall be subject to such alteration as the Commission may direct.

Upon the completion of the project, or at such other time as the Commission may direct, the Licensee shall submit to the Commission for approval revised exhibits insofar as necessary to show any divergence from or variations in the project area and project boundary as finally located or in the project works as actually constructed when compared with the area and boundary shown and the works described in the license or in the exhibits approved by the Commission, together with a statement in writing setting forth the reasons which in the opinion of the Licensee necessitated or justified variation in or divergence from the approved exhibits. Such revised exhibits shall, if and when approved by the Commission, be made a part of the license under the provisions of Article 2 hereof.

Article 4. The construction, operation, and maintenance of the project and any work incidental to additions or alterations shall be subject to the inspection and supervision of the Regional Engineer, Federal Power Commission, in the region wherein the project is located, or of such other officer or agent as the Commission may designate, who shall be the authorized representative of the Commission for such purposes. The Licensee shall cooperate fully with said representative and shall furnish him a detailed program of inspection by the Licensee that will provide for an adequate and qualified inspection force for construction of the project and for any subsequent alterations to the project. Construction of the project works or any feature or alteration thereof shall not be initiated until the program of inspection for the project works or any such feature thereof has been approved by said representative. The Licensee shall also furnish to said representative such further information as he may require concerning the construction, operation, and maintenance of the project, and of any alteration thereof, and shall notify him of the date upon which work will begin, as far in advance thereof as said representative may reasonably specify, and shall notify him promptly in writing of any suspension of work for a period of more than one week, and of its resumption and completion. The Licensee shall allow said representative and other officers or employees of the United States, showing proper credentials, free and unrestricted access to, through, and across the

.

project lands and project works in the performance of their official duties. The Licensee shall comply with such rules and regulations of general or special applicability as the Commission may prescribe from time to time for the protection of life, health, or property.

<u>Article 5</u>. The Licensee, within five years from the date of issuance of the license, shall acquire title in fee or the right to use in perpetuity all lands, other than lands of the United States, necessary or appropriate for the construction, maintenance, and operation of the project. The Licensee or its successors and assigns shall, during the period of the license, retain the possession of all project property covered by the license as issued or as later amended, including the project area, the project works, and all franchises, easements, water rights, and rights of occupancy and use; and none of such properties shall be voluntarily sold, leased, transferred, abandoned, or otherwise disposed of without the prior written approval of the Commission, except that the Licensee may lease or otherwise dispose of interests in project lands or property without specific written approval of the Commission pursuant to the then current regulations of the Commission. The provisions of this article are not intended to prevent the abandonment or the retirement from service of structures, equipment, or other project works in connection with replacements thereof when they become obsolete, inadequate, or inefficient for further service due to wear and tear; and mortgage or trust deeds or judicial sales made thereunder, or tax sales, shall not be deemed voluntary transfers within the meaning of this article.

Article 6. In the event the project is taken over by the United States upon the termination of the license as provided in Section 14 of the Federal Power Act, or is transferred to a new licensee or to a non-power licensee under the provisions of Section 15 of said Act, the Licensee, its successors and assigns shall be responsible for, and shall make good any defect of title to, or of right of occupancy and use in, any of such project property that is necessary or appropriate or valuable and serviceable in the maintenance and operation of the project, and shall pay and discharge, or shall assume responsibility for payment and discharge of, all liens or encumbrances upon the project or project property created by the Licensee or created or incurred after the issuance of the license: <u>Provided</u>. That the provisions of this article are not intended to require the Licensee, for the purpose of transferring the project to the United States or to a new licensee, to acquire any different title to, or right of occupancy and use in, any of such project property than was necessary to acquire for its own purposes as the Licensee.

<u>Article 7</u>. The actual legitimate original cost of the project, and of any addition thereto or betterment thereof, shall be determined by the Commission in accordance with the Federal Power Act and the Commission's Rules and Regulations thereunder.

Article 8. The Licensee shall install and thereafter maintain gages and stream-gaging stations for the purpose of determining the stage and flow of the stream or streams on which the project is located, the amount of water held in and withdrawn from storage, and the effective head on the turbines; shall provide for the required reading of such gages and for the adequate rating of such stations; and shall install and maintain standard meters adequate for the determination of the amount of electric energy generated by the project works. The number, character, and location of gages, meters, or other measuring devices, and the method of operation thereof, shall at all times be satisfactory to the Commission or its authorized representative. The Commission reserves the right, after notice and opportunity for hearing, to require such alterations in the number, character, and location of gages, meters, or other measuring devices, and the method of operation thereof, as are necessary to secure adequate determinations. The installation of gages, the rating of said stream or streams, and the determination of the flow thereof, shall be under the supervision of. or in cooperation with. the District Engineer of the United States Geological Survey having charge of stream-gaging operations in the region of the project, and the Licensee shall advance to the United States Geological Survey the amount of funds estimated to be necessary for such supervision, or cooperation for such periods as may be mutually agreed upon. The Licensee shall keep accurate and sufficient records of the foregoing determinations to the satisfaction of the Commission, and shall make return of such records annually at such time and in such form as the Commission may prescribe.

<u>Article 9</u>. The Licensee shall, after notice and opportunity for hearing, install additional capacity or make other changes in the project as directed by the Commission, to the extent that it is economically sound and in the public interest to do so.

<u>Article 10</u>. The Licensee shall, after notice and opportunity for hearing, coordinate the operation of the project, electrically and hydraulically, with such other projects or power systems and in such manner as the Commission may direct in the interest of power and other beneficial public uses of water resources, and on such conditions concerning the equitable sharing of benefits by the Licensee as the Commission may order.

<u>Article 11</u>. Whenever the Licensee is directly benefited by the construction work of another licensee, a permittee, or the United States on a storage reservoir or other headwater improvement, the Licensee shall reimburse the owner of the headwater improvement for such part of the annual charges for interest, maintenance, and depreciation thereof as the Commission shall determine to be equitable, and shall pay to the United States the cost of making such determination as fixed by the Commission. For benefits provided by a storage reservoir or other headwater improvement of the United States, the Licensee shall pay to the Commission the amounts for which it is billed from time to time for such headwater benefits and for the cost of making the determinations pursuant to the then current regulations of the Commission under the Federal Power Act.

<u>Article 12</u>. The United States specifically retains and safeguards the right to use water in such amount, to be determined by the Secretary of the Army, as may be necessary for the purposes of navigation on the navigable waterway affected; and the operations of the Licensee, so far as they affect the use, storage and discharge from storage of waters affected by the license, shall at all times be controlled by such reasonable rules and regulations as the Secretary of the Army may prescribe in the interest of navigation, and as the Commission may prescribe for the protection of life, health, and property, and in the interest of the fullest practicable conservation and utilization of such waters for power purposes and for other beneficial public uses, including recreational purposes, and the Licensee shall release water from the project reservoir at such rate in cubic feet per second, or such volume in acre-feet per specified period of time, as the Secretary of the Army may prescribe in the interest of any pay prescribe in the interest of navigation, or as the Commission may prescribe for the protection of time, as the Secretary of the Army may prescribe in the interest of navigation, or as the Commission may prescribe for the other purposes hereinbefore mentioned.

Article 13. On the application of any person, association, corporation, Federal agency, State or municipality, the Licensee shall permit such reasonable use of its reservoir or other project properties, including works, lands and water rights, or parts thereof, as may be ordered by the Commission, after notice and opportunity for hearing, in the interests of comprehensive development of the waterway or waterways involved and the conservation and utilization of the water resources of the region for water supply or for the purposes of steam-electric, irrigation, industrial, municipal or similar uses. The Licensee shall receive reasonable compensation for use of its reservoir or other project properties or parts thereof for such purposes, to include at least full reimbursement for any damages or expenses which the joint use causes the Licensee to incur. Any such compensation shall be fixed by the Commission either by approval of an agreement between the Licensee and the party or parties benefiting or after notice and opportunity for hearing. Applications shall contain information in sufficient the applicant possesses necessary water rights pursuant to applicable State law, or a showing of cause why such evidence cannot concurrently be submitted, and a statement as to the relationship of the proposed use to any State or municipal plans or orders which may have been adopted with respect to the use of such waters.

<u>Article 14</u>. In the construction or maintenance of the project works, the Licensee shall place and maintain suitable structures and devices to reduce to a reasonable degree the liability of contact between its transmission lines and telegraph, telephone and other signal wires or power trans- mission lines constructed prior to its transmission lines and not owned by the Licensee, and shall also place and maintain suitable structures and devices to reduce to a reasonable degree the liability of any structures or wires falling or obstructing traffic or endangering life. None of the provisions of this article are intended to relieve the Licensee from any responsibility or requirement which may be imposed by any other lawful authority for avoiding or eliminating inductive interference.

<u>Article 15</u>. The Licensee shall, for the conservation and development of fish and wildlife resources, construct, maintain, and operate, or arrange for the construction, maintenance, and operation of such reasonable facilities, and comply with such reasonable modifications of the project structures and operation, as may be ordered by the Commission upon its own motion or upon the recommendation of the Secretary of the Interior or the fish and wildlife agency or agencies of any State in which the project or a part thereof is located, after notice and opportunity for hearing.

<u>Article 16</u>. Whenever the United States shall desire, in connection with the project, to construct fish and wildlife facilities or to improve the existing fish and wildlife facilities at its own expense, the Licensee shall permit the United States or its designated agency to use, free of cost, such of the Licensee's lands and interests in lands, reservoirs, waterways and project works as may be reasonably required to complete such facilities or such improvements thereof. In addition, after notice and opportunity for hearing, the Licensee shall modify the project operation as may be reasonably prescribed by the Commission in order to permit the maintenance and operation of the fish and wildlife facilities constructed or improved by the United States under the provisions of this article. This article shall not be interpreted to place any obligation on the United States to construct or improve fish and wildlife facilities or to relieve the Licensee of any obligation under this license.

Article 17. The Licensee shall construct, maintain, and operate, or shall arrange for the construction, maintenance, and operation of such reasonable recreational facilities, including modifications thereto, such as access roads, wharves, launching ramps, beaches, picnic and camping areas, sanitary facilities, and utilities, giving consideration to the needs of the physically handicapped, and shall comply with such reasonable modifications of the project, as may be prescribed here after by the Commission during the term of this license upon its own motion or upon the recommendation of the Secretary of the Interior or other interested Federal or State agencies, after notice and opportunity for hearing.

<u>Article 18</u>. So far as is consistent with proper operation of the project, the Licensee shall allow the public free access, to a reasonable extent, to project waters and adjacent project lands owned by the Licensee for the purpose of full public utilization of such lands and waters for navigation and for outdoor recreational purposes, including fishing and hunting: <u>Provided</u>, That the Licensee may reserve from public access such portions of the project waters, adjacent lands, and project facilities as may be necessary for the protection of life, health, and property.

Article 19. In the construction, maintenance, or operation of the project, the Licensee shall be responsible for, and shall take reasonable measures to prevent, soil erosion on lands adjacent to streams or other waters, stream sedimentation, and any form of water or air pollution. The Commission, upon request or upon its own motion, may order the Licensee to take such measures as the Commission finds to be necessary for these purposes, after notice and opportunity for hearing.

Article 20. The Licensee shall consult with the appropriate State and Federal agencies and, within one year of the date of issuance of this license, shall submit for Commission approval a plan for clearing the reservoir area. Further, the Licensee shall clear and keep clear to an adequate width lands along open conduits and shall dispose of all temporary structures, unused timber, brush, refuse, or other material unnecessary for the purposes of the project which results from the clearing of lands or from the maintenance or alteration of the project works. In addition, all trees along the periphery of project reservoirs which may die during operations of the project shall be removed. Upon approval of the clearing plan all clearing of the lands and disposal of the unnecessary material shall be done with due diligence and to the satisfaction of the authorized representative of the Commission and in accordance with appropriate Federal, State, and local statutes and regulations.

<u>Article 21</u>. Material may be dredged or excavated from, or placed as fill in, project lands and/or waters only in the prosecution of work specifically authorized under the license; in the maintenance of the project; or after obtaining Commission approval, as appropriate. Any such material shall be removed and/or deposited in such manner as to reasonably preserve the environmental values of the project and so as not to interfere with traffic on land or water. Bredging and filling in a navigable water of the United States shall also be done to the satisfaction of the District Engineer, Department of the Army, in charge of the locality.

<u>Article 22</u>. Whenever the United States shall desire to construct, complete, or improve navigation facilities in connection with the project, the Licensee shall convey to the United States, free of cost, such of its lands and rights-of-way and such rights of passage through its dams or other structures, and shall permit such control of its pools, as may be required to complete and maintain such navigation facilities.

<u>Article 23</u>. The operation of any navigation facilities which may be constructed as a part of, or in connection with, any dam or diversion structure constituting a part of the project works

shall at all times be controlled by such reasonable rules and regulations in the interest of navigation, including control of the level of the pool caused by such dam or diversion structure, as may be made from time to time by the Secretary of the Army.

<u>Article 24</u>. The Licensee shall furnish power free of cost to the United States for the operation and maintenance of navigation facilities in the vicinity of the project at the voltage and frequency required by such facilities and at a point adjacent thereto, whether said facilities are constructed by the Licensee or by the United States.

<u>Article 25</u>. The Licensee shall construct, maintain, and operate at its own expense such lights and other signals for the protection of navigation as may be directed by the Secretary of the Department in which the Coast Guard is operating.

<u>Article 26</u>. Timber on lands of the United States cut, used, or destroyed in the construction and maintenance of the project works, or in the clearing of said lands, shall be paid for, and the resulting slash and debris disposed of, in accordance with the requirements of the agency of the United States having jurisdiction over said lands. Payment for merchantable timber shall be at current stumpage rates, and payment for young growth timber below merchantable size shall be at current damage appraisal values. However, the agency of the United States having jurisdiction may sell or dispose of the merchantable timber to others than the Licensee: <u>Provided</u>. That timber so sold or disposed of shall be cut and removed from the area prior to, or without undue interference with, clearing operations of the Licensee and in coordination with the Licensee's project construction schedules. Such sale or disposal to others shall not relieve the Licensee of responsibility for the clearing and disposal of all slash and debris from project lands.

<u>Article 27</u>. The Licensee shall do everything reasonably within its power, and shall require its employees, contractors, and employees of contractors to do every thing reasonably within their power, both independently and upon the request of officers of the agency concerned, to prevent, to make advance preparations for suppression of, and to suppress fires on the lands to be occupied or used under the license. The Licensee shall be liable for and shall pay the costs incurred by the United States in suppressing fires caused from the construction, operation, or maintenance of the project works or of the works appurtenant or accessory thereto under the license.

Article 28. The Licensee shall interpose no objection to, and shall in no way prevent, the use by the agency of the United States having jurisdiction over the lands of the United States affected, or by persons or corporations occupying lands of the United States under permit, of water for fire suppression from any stream, conduit, or body of water, natural or artificial, used by the Licensee in the operation of the project works covered by the license, or the use by said parties of water for sanitary and domestic purposes from any stream, conduit, or body of water, natural or artificial, used by the Licensee in the operation of the project works covered by the license.

<u>Article 29</u>. The Licensee shall be liable for injury to, or destruction of, any buildings. bridges, roads, trails, lands, or other property of the United States, occasioned by the construction, maintenance, or operation of the project works or of the works appurtenant or accessory thereto under the license. Arrangements to meet such liability, either by compensation for such injury or destruction, or by reconstruction or repair of damaged property, or otherwise, shall be made with the appropriate department or agency of the United States.

Article 30. The Licensee shall allow any agency of the United States, without charge, to construct or permit to be constructed on, through, and across those project lands which are lands of the United States such conduits, chutes, ditches, railroads, roads, trails, telephone and power lines, and other routes or means of transportation and communication as are not inconsistent with the enjoyment of said lands by the Licensee for the purposes of the license. This license shall not be construed as conferring upon the Licensee any right of use, accupancy, or enjoyment of the lands of the United States other than for the construction, operation, and maintenance of the project as stated in the license.

<u>Article 31</u>. In the construction and maintenance of the project, the location and standards of roads and trails on lands of the United States and other uses of lands of the United States. including the location and condition of guarries, borrow pits, and spoil disposal areas, shall

be subject to the approval of the department or agency of the United States having supervision over the lands involved.

<u>Article 32</u>. The Licensee shall make provision, or shall bear the reasonable cost, as determined by the agency of the United States affected, of making provision for avoiding inductive interference between any project transmission line or other project facility constructed, operated, or maintained under the license, and any radio installation, telephone line, or other communication facility installed or constructed before or after construction of such project transmission line or other project facility and owned, operated, or used by such agency of the United States in administering the lands under its jurisdiction.

<u>Article 33</u>. The Licensee shall make use of the Commission's guidelines and other recognized guidelines for treatment of transmission line rights-of-way, and shall clear such portions of transmission line rights-of-way across lands of the United States as are designated by the officer of the United States in charge of the lands; shall keep the areas so designated clear of new growth, all refuse, and inflammable material to the satisfaction of such officer; shall trim all branches of trees in contact with or liable to contact the transmission lines; shall cut and remove all dead or leaning trees which might fall in contact with the transmission lines; and shall take such other precautions against fire as may be required by such officer. No fires for the burning of waste material shall be set except with the prior written consent. of the officer of the United States in charge of the lands as to time and place.

<u>Article 34</u>. The Licensee shall cooperate with the United States in the disposal by the United States, under the Act of July 31, 1947, 61 Stat. 681, as amended (30 U.S.C. sec. 601, <u>et</u> <u>seq</u>.), of mineral and vegetative materials from lands of the United States occupied by the project or any part thereof: <u>Provided</u>, That such disposal has been authorized by the Commission and that it does not unreasonably interfere with the occupancy of such lands by the Licensee for the purposes of the license: <u>Provided further</u>, That in the event of disagreement, any question of unreasonable interference shall be determined by the Commission after notice and opportunity for hearing.

Article 35. If the Licensee shall cause or suffer essential project property to be removed or destroyed or to become unfit for use, without adequate replacement, or shall abandon or discontinue good faith operation of the project or refuse or neglect to comply with the terms of the license and the lawful orders of the Commission mailed to the record address of the Licensee or its agent, the Commission will deem it to be the intent of the Licensee to surrender the license. The Commission, after notice and opportunity for hearing, may require the Licensee to remove any or all structures, equipment and power lines within the project boundary and to take any such other action necessary to restore the project waters, lands, and facilities remaining within the project boundary to a condition satisfactory to the United States agency having jurisdiction over its lands or the Commission's authorized representative, as appropriate, or to provide for the continued operation and maintenance of non-power facilities and fulfill such other obligations under the license as the Commission may reactive, may also agree to the surrender of the license when the Commission, for the reasons recited herein, deems it to be the intent of the License.

<u>Article 36</u>. The right of the Licensee and of its successors and assigns to use or occupy waters over which the United States has jurisdiction, or lands of the United States under the license, for the purpose of maintaining the project works or otherwise, shall absolutely cease at the end of the license period, unless the Licensee has obtained a new license pursuant to the then existing laws and regulations, or an annual license under the terms and conditions of this license.

<u>Article 37</u>. The terms and conditions expressly set forth in the license shall not be construed as impairing any terms and conditions of the Federal Power Act which are not expressly set forth herein.

H.2 SPECIAL LICENSE ARTICLES FOR HYDROPOWER PROJECTS CONSTRUCTED AT U.S. ARMY CORPS OF ENGINEERS DAMS

<u>Article 301</u>. The design and construction of those permanent and temporary facilities, including reservoir impounding cofferdams and deep excavations, that should be an integral part of, or that could affect the structural integrity or operation of the government project shall be done in consultation with and subject to the review and approval of the Corps' Division Engineer. Within 90 days from the issuance date of the license, the licensee shall furnish the Corps and the Commission's Regional Director, for their information, a schedule for submission of design documents and the plans and specifications for the project. If the schedule does not afford sufficient review and approval time, the licensee, upon request of the Corps, shall meet with the Corps' and Commission's staff to revise the schedule accordingly.

<u>Article 302</u>. The licensee shall review and approve the design of contractor-designed cofferdams and deep excavations other than those approved according to Article prior to the start of construction and shall ensure that construction of the cofferdams and deep excavations are consistent with the approved design. At least 30 days prior to start of construction of the cofferdam, the licensee shall file 2 copies with the Commission, and submit 1 copy each to the Commission's Regional Director and the Corps of Engineers, of the approved cofferdam construction drawings and specifications and letter(s) of approval.

Article 303. Within 90 days from the issuance date of the license, the licensee shall enter into an agreement with the Corps of Engineers to coordinate plans for access to and site activities on lands and property administered by the Corps so that the authorized purposes, including operation of the Federal facilities, are protected. In general, the agreement shall not be redundant with the Commission's requirements contained in this license, shall identify the facility, and the study and construction activities, as applicable, and terms and conditions under which studies and construction will be conducted. The agreement shall set forth reasonable arrangements for access to the Corps site to conduct studies and construction activities, such access rights to be conditioned by the Corps as may be necessary to protect the federally authorized project purposes and operations. Should the licensee and the Corps fail to reach an access agreement, the licensee shall refer the matter to the Commission for resolution.

Article 304. The construction, operation and maintenance of the project works that, in the judgement of the Corps of Engineers may affect the structural integrity or operation of the Corps project shall be subject to periodic or continuous inspections by the Corps. Any construction, operation and maintenance deficiencies or difficulties detected by the Corps' inspection shall be immediately reported to the Commission's Regional Director. Upon review, the Regional Director shall refer the matter to the licensee for appropriate action. In cases where construction, operation or maintenance practices or deficiencies may create a situation posing imminent danger to the structural integrity and safety of Corps project, the Corps' inspector has the authority to stop construction, operation, or maintenance while waiting resolution of the problem.

Article 305. At least 60 days prior to start of construction, the licensee shall submit for approval a regulating plan to the Corps of Engineers, describing (a) the designed mode of hydropower operation, and (b) reservoir flow diversion and regulation requirements as established by the Corps for operation of the Corps project during construction. In addition, the licensee, prior to start of power plant operation, shall enter into an operating Memorandum of Agreement (MOA) with the Corps describing the detailed operation of the powerhouse acceptable to the Corps. The MOA shall specify any restrictions needed to protect the primary purposes of the Corps project for navigation, recreation, water quality, and flood control. The Regional Director shall be invited to attend meetings regarding the agreement. The MOA shall be subject to revision by mutual consent of the Corps and licensee as experience is gained by actual project operation. Should the licensee and the Corps fail to reach an agreement, the matter shall be referred to the Commission for resolution. Three copies of the regulating plan and signed MOA between the Corps and the licensee and any revision thereof shall be filed with the Commission and one copy submitted to the Regional Director.

<u>Article 306</u>. The licensee shall have no claim under this license against the United States arising from the effect of any changes made in the operation or reservoir levels of the Corps of Engineers' project.

<u>Article 307</u>. The licensee shall provide the Regional Director two copies of all correspondence between the licensee and the Corps of Engineers. The Regional Director shall not authorize construction of any project work until the Corps of Engineers' written approval of construction plans and specification has been received by the Regional Director.

APPENDIX I

ADDITIONAL INFORMATION AND EVALUATION OF POTENTIAL IMPACTS ON THE PINK MUCKET PEARLY MUSSELL (<u>LAMPSILIS ABRUPTA</u>) SUBMITTED TO THE U.S. FISH AND WILDLIFE SERVICE
APPENDIX 1

The Notice of Availability of the draft environmental impact statement (DEIS) was published in the Federal Register on May 20, 1988. The DEIS was mailed to Federal, state, and local agencies and individuals for comments on May 12, 1988. Chapter 8 of the final environmental impact statement (FEIS) contains a listing of those agencies and individuals that were sent copies of both the DEIS and FEIS. A public meeting on the DEIS was conducted on July 15, 1988 in Pittsburgh, Pennsylvania, to allow participants to express their views on the DEIS.

All timely letters of comment that address specific analyses in the DEIS were reviewed by FERC staff. Suggestions for correcting text or data and requests for further discussion of a subject have been considered. Those editorial changes and suggestions which were practicable, reasonable, and which improved the quality of the EIS are incorporated herein.

Constructive criticism presenting a major environmental point of view or one in opposition to staff, when persuasively supported, is treated by making revisions in the appropriate part of the FEIS. When the major point of view is not persuasive, reasons are given why the staff did not change its point of view.

The sections or pages of the FEIS that have been modified as a result of comments received are identified in the staff responses to the right of the letters of comments. Other FERC staff responses are self-explanatory.

A "no response required" response is given to comments that are statements that raised no questions concerning treatment of subject matter in the DEIS. A "your opinion has been noted" or "comment noted" response is given to comments that are considered to be statements or opinions.

Letters were received form the U.S. Soil Conservation Service, U.S. Public Health Service, and the Ohio State Clearing House which acknowledged receipt of the DEIS but offered no comments.

The respondents and the page on which their letters occur are as follows:

Name P	age
Department of the Army; Corps of Engineers.	
Department of the Interior; U.S. Fish and	
Wildlife Service.	
Department of Interior; Office of Environmental	
Project Review ,	
Mr. Chris Clower, U. S. Fish and Wildlife Service.	
Testimony at Public Meeting	
U.S. Environmental Protection Agency.	
Chio Department of Natural Resources.	
Ohio Environmental Protection Agency.	
Pennsylvania Department of Environmental Resources.	
Pennsylvania Fish Commission.	
West Virginia Department of Culture and History	
West Virginia Department of Natural Resources	
American Electric Power Service Corporation	
American Rivers, Inc., and Friends of the Farth	
Darwin F. Johnson	
Ohio River Valley Water Sanitation Commission	
Allegheny Electric Cooperative. Inc.	
Allegheny Hydropower, Inc.	
Morrison-Knudsen Engineers, Inc.	
July 14th	
Morrison-Knudsen Engineers, Inc.	
August 3.	
LeBoeuf, Lamb, Leiby & MarRae	
City of New Martinsville, West Virginia	
Testimony at Public Meeting	
County of Alleoheny	
Mitex. Inc., re: Gallionlis Hydro Project	
Noah Corp. and W. V. Hydro. Inc	

Sithe Energies U.S.A., Inc., re: Alleghe	eny	8	\$.						
and 9 Hydro Partners			-						
Mitex, Inc., re: Muskingum L&D No. 3.		•	•					•	•
National Renewable Resources, Inc.,	•						-	,	
re: Maxwell L&D									
National Renewable Resources, Inc.,	٠			٠		·		•	
re: Monongahela L&D					-		-		
National Renewable Resources, Inc.,							-		
re: Tygart Dam (7307)									
Mitex, Inc., re: Allegheny L&D Nos. 5 &	б								
Green International Affiliates, Inc.,.				-					
re: Montgomery L&D. (3490)			•		-	-	•		

.

,

FEDERAL ENERGY REGULATORY COMMISSION

WASHINGTON D C.20426

SEP 0 8 1988

Mr. Christopher Clower Elkins Suboffice U.S. Fish & Wildlife Service P.O. Box 1278 Elkins, WV 26214

Dear Mr. Clower:

This is in partial response to the letter dated August 1, 1988, from the United States Department of the Interior (Interior) that provides comments on the draft environmental impact statement (DEIS) for proposed hydropower projects in the upper Ohio River Basin, Docket No. EL85-19-114 (Enclosure 1).

Comments on pages 3, 4, and 5, in the August 1, 1988, letter reference the need for completion of a biological assessment and consultation on the pink mucket pearly mussel, a federally-listed endangered species that occurs in the Ohio River Basin, pursuant to Section 7 of the Endangered Species Act, as amended.

Consistent with Section 7(c)(1) of the Endangered Species Act, as amended, and 50 C.F.R. § 402.06 (1987), staff's biological assessment was included in the DEIS transmitted to your office by letter dated June 27, 1988 (Enclosure 2). That letter specifically addressed the DEIS assessment of impacts on the endangered mussel and requested your comments. Pursuant to 50 C.F.R. § 402.12(j) (1987), which permits an agency to initiate formal consultation concurrently with its submission of the biological assessment, it was our intent that the request constitute our request for formal consultation. In response to Interior's request for additional data included in its DEIS comments, staff has compiled additional information in the attached document concerning the species (Enclosure 3). This information will be included in the final environmental impact statement.

As reflected in the DEIS and the enclosed supplemental information, staff has determined that the proposed development of hydroelectric projects in the upper Ohio River Basin could have some effect on the species but that the effect is not likely to jeopardize the continued existence of the species or result in the destruction or adverse modification of its critical habitat. The effect varies with the alternative considered, and the impact of the preferred alternative is so small as to be considered to have no demonstrable effect. Therefore, we request that consultation be concluded expeditiously and that you proceed to prepare your biological opinion regarding the endangered pink mucket pearly mussel.

If you have any questions on this matter, please telephone George Taylor at 202/376-1900.

Sincerely,

Dean L. Shumway Director, Division of Project Review

Enclosures:

Draft Environmental Impact Statement for Hydroelectric Development in the Upper Ohio River Basin, Docket No. EL85-19-114. Letter from Dean Shumway, FERC, to Christopher Clower, USFWS, dated June 27, 1988. Additional Information on the Effects of Hydroelectric Development in the Upper Ohio River Basin on the Pink Mucket Pearly Mussel. Lampsilus abrupta (Say 1831)=L. orbiculata.

Enclosure 2

Mr. Christopher Clower Elkins Suboffice U.S. Fish and Wildlife Service P.O. Box 1278 JUN 2 7 1388 Elkins, WV 26214

Dear Mr. Clower:

Enclosed is the Draft Environmental Impact Statement (EIS) for Hydroelectric Development in the Upper Ohio River Basin, Docket No. EL85-19-114.

The EIS assesses the impacts associated with 24 proposed hydroelectric projects located at 19 sites in the upper basin, including the potential impacts of the proposed projects on the endangered pink mucket pearly mussel.

Page 3-19 of the EIS provides information on the distribution of the endangered mussel in the upper basin. This section of the DEIS also identifies suitable habitat for this species in the downstream areas of the proposed hydropower projects located at Muskingum River Lock and Dam (L&D) No. 3 (FERC No. 6998), Belleville L&D (FERC No. 6939), and the competing proposals at Gallipolis LAD (FERC Nos. 9042 and 10098).

Page 4-62 discusses the effect of the hydropower proposals on the endangered mussel. Page 5-21 indicates staff's recommended alternative, alternative 4. Under this alternative, there would be no hydropower development at Muskingum River L&D No. 3 and, therefore, no effect on this endangered species. Page 5-29 provides staff's recommendations for the protection of this species associated with hydropower development at Belleville L&D and Gallipolis L&D.

We would appreciate receiving your review comments within 30 days, regarding the endangered pink mucket pearly mussel.

Sincerely, Dean L. Shumway, Director, Division of Project Review

Enclosure: Draft Environmental Impact Statement for Hydroelectric Development in the Upper Ohio River Basin, Docket No. EL85-19-114.

.

ADDITIONAL INFORMATION ON THE EFFECTS OF HYDROELECTRIC DEVELOPMENT IN THE UPPER OHIO RIVER BASIN ON THE PINK MUCKET PEARLY MUSSEL, <u>Lampsilus abrupta</u> (Say 1831) = <u>L. orbiculata</u>

September 7, 1988

Federal Energy Regulatory Commission Washington, D. C. ADDITIONAL INFORMATION ON THE EFFECTS OF HYDROELECTRIC DEVELOPMENT IN THE UPPER OHIO RIVER BASIN ON THE PINK MUCKET PEARLY MUSSEL, Lampsilus abrupta (Say 1831) = L. orbiculata

September 7, 1988

Federal Energy Regulatory Commission. Washington, D. C.

.

-1-CONTENTS

ABSTRACT	2
1. INTRODUCTION	3
2. DESCRIPTION OF PROPOSED PROJECTS	3
3. ECOLOGY OF THE PINK MUCKET PEARLY MUSSEL	6
3.1 Taxonomic Status	6
3.2 Distribution	6
3.2.1 Muskingum River	14
3.2.2 Ohio River Below Gallipolis L&D	15
3.2.3 Elsewhere in the Upper Ohio River	15
3.3 Discolved Orvien Reminments	15
3 A Fich Hote	
	17
4. ASSESSMENT OF IMPACTS	13
4.1 Impacts of Projects as Proposed	18
4 1] Dissolved Overen	<u>ב</u> קו
A 1 2 Dimerical Habitat	10
4.1.2 Physical habitat	20
$4.1.3 \text{Host Fish} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	22
4.2 Impacts of Project Operation to Meet Dissolved Oxygen	
Standards	22
4.2.1 Dissolved Oxygen	22
4.2.2 Physical Habitat and Host Fish	23
4.3 Impacts of Project Operation to Meet Antidegradation	
with in	
	25
4.3.1 Dissolved Oxygen	23
4.3.2 Physical Habitat and Host Fish	27
4.4 Projects Selected to Minimize Impacts To All Target	
Resources	27
4.4.1 Dissolved Oxygen	27
4.4.2 Physical Habitat	27
A A 2 Hoot Pick	27
	27
4.5 Significance of impacts	27
5. ADEQUACY OF DATA	30
6. SUMMARY	31
7. REFERENCES	33
3. APPENDICE:	
ADDENDIY I Letter from K E Vacance USENC to	
AFFINDIA I. LEVUCI IIV. A. E. ALUDRERYEF, UDIND, LO	.
w. rowler, Mitex, Inc., dated April 6, 198/.	36
APPENDIA 11. Letter Ind Charles. J. Kulp, USFWS, to	
Jazon Scory, FEM. dated September 28, 1987.	40
APPENDIX III. Letter from Dean Sounway, FERC, to C. Clower,	
USFWS, dates Sume 27, 1988	
APPENDIX IV. USING Contents on the FFRC DEIS	61
APPENDIX V. Summary of Expert Contacts	/0
	~ 2

ADDITIONAL INFORMATION ON THE EFFECTS OF HYDROELECIRIC DEVELOPMENT IN THE UPPER OHIO RIVER BASIN ON THE PINK MUCKET PEARLY MUSSEL, <u>Lampsilus abrupta</u> (Say 1831) = <u>L. orbiculata</u>

ABSTRACT

Proposed hydropower development of the upper Ohio River basin has been determined to have the potential for some impacts on the federally listed erdangered freshwater mussel, Lampsilus abrupta (Say, 1931). The Federal Energy Regulatory Commission (FERC) is considering licensing 24 hydroelectric projects at 19 sites in the upper basin. Impacts at one confirmed locality, one recently occupied locality, and two localities anticipated by the U.S. Fish and Wildlife Service to contain the species have been analyzed. The evaluation is based on the best estimates of the environmental requirements of the species (using analogy with other freshwater mussel species where necessary), the locations of confirmed and suspected beds of the species, and the estimated changes in water quality (DO), habitat, and fish hosts that might be caused by hydropower development. There are important uncertainties in this information, especially the dissolved oxygen criterion for continued long-term mussel growth and production, that prevent precise definition of impacts. The impacts were determined to differ with alternative development scenarios. FERC staff believes that none of the four development alternatives examined in the DEIS would be likely to jeopardize the continued existence of L. abrupta or result in the destruction or adverse modification of L. abrupta critical habitat, while the preferred alternative (Projects Selected To Minimize Impacts To All Target Resources) will have no demonstrable effect on populations of this freshwater mussel. The proposed licensing action, therefore, would not jeopardize the continued existence of L. abrupta, in accord with requirements of the Endangered Species Act of 1973, as amended.

1. INTRODUCTION

The pink mucket pearly mussel, Lampsilus abrupta (Say 1831), which is synonymous with L. orbiculata (Stansbery 1985), is listed by the U. S. Fish and Wildlife Service (USFWS) as an endangered species (USFWS 1985, 41 FR 24062-24067). In compliance with Section 7(a) of the Endangered Species Act of 1973, as amended, the Upper Mississippi Water Company, which seeks a license from the Federal Energy Regulatory Commission (FERC) to construct a small hydropower project on the Muskingum River (FERC No. 6996-001), supplied the USFWS with biological and physical information on its project and its impacts on L. abrupta. (Mitex, Inc. 1987). The USFWS responded to Mitex, Inc. with a letter approving mitigation plans (Appendix I). This species has occupied habitats immediately downstream of Muskingum Lock and Dam 3, the site of the proposed hydropower project. Correspondence between FERC and the USFWS has revealed that L. abrupta was identified in 1987 from another site in the FERC study area, about 14 miles below the Gallipolis Locks and Dam on the Ohio River (Tolin et al. 1987, Appendix II). The USFWS believes habitats in other locations in the Ohio River, e.g., below Willow Island Locks and Dam and Belleville Locks and Dam, are also suitable and the species may be found there (Appendix II). FERC staff prepared a biological assessment in compliance with section 7(c) of the Endangered Species Act of 1973, as amended, on L. abrupta and incorporated it into the Draft Environmental Impact Statement (DEIS) on the cumulative impacts of hydroelectric development in the upper Ohio River basin of Ohio, Pennsylvania, and West Virginia in accordance with the National Environmental Policy Act of 1969 (FERC 1988; FERC Docket No. EL85-19-114). By letter of June 27, 1988, the FERC indicated to USFWS that it had conducted an evaluation of L. abrupta as part of the DEIS and requested comments (Appendix III). USFWS did not respond specifically to this letter. Comments by the USFWS on the DEIS indicated that additional assessment of upper basin hydroelectric development on L. abrupta should be undertaken. (Appendix IV).

This document presents additional information regarding the previously prepared biological assessment of potential impacts of hydropower development in the upper Ohio River basin on <u>L. abrupta</u> (= <u>L. orbiculata</u>). It describes the methods, results, and implications of studies conducted by license applicants, university researchers, the USFWS, state agencies, and others. This report was prepared by the Oak Ridge National Laboratory (ORNL) for the FERC staff.

2. DESCRIPTION OF THE PROPOSED PROJECTS

The proposed action is the licensing of 24 hydroelectric projects at 19 sites (5 sites have competing applications) located in the upper Ohio River basin in the states of Pennsylvania, West Virginia, and Ohio (Figure 1). The projects (Table 1) can be licensed if they can provide energy in an environmentally acceptable manner that is more economically feasible than the least-cost thermal alternative. A more complete description of the projects can be found in the DEIS (FERC 1968). The projects are located for the most part (17 sites) at locks and dams (L&D) constructed and operated by the U. S. Army Corps of Engineers (Corps). One site is on the Muskingum River where navigation dams are operated by the State of Ohio, and another site is on a high-head storage dam on the Tygart River. Power generation would be controlled by river flows and water use for navigation locks (Tygart Dam is a



Figure 1. Schematic representation of projects within the FERC DEIS study area. (Table 1 gives project names and FERC project numbers for each of the abbreviations used in this figure). Source: FERC (1988).

Project name, abbreviation	FERC project no.
Allegheny River L&D No. 7, 47	7914-003
Allegheny River L&D No. 4, A4	7909-002
Allegheny Eiver L&D No. 3, A3	4474-00
Allegheny River LLD No. 2, A2	4017-002
Tygart Dam, TD	7307-000 7399-000
Opekiska L&D, O°E	8990-000
Hildebrand L&D, HIL	8554-001
Point Marion L&D, PM	7660-000
Maxwell L&D, MAX	8908-000
Monongahela L&D Ko. 4, H4	4675-002
Emsworth L4D, EMS	7041-001
Dashields L&D, DAS	7558-001
Hontgomery, MONT	2971-002 3490-003
New Cumberland L&D, NC	6901-001 10332-000
Pike Island, PI	3218-001
Willow Island LAD, WI	6902-003 9995-000
Belleville, BEL	6939-001
Sallipelis L&D, GAL	9042-000 10098-000
Muskingum River L&D No. 3, MUSK3	6998-001

Table I. Hydroelectric projects with pending FERC license applications evaluated in the DEIS.

special case not relevant to <u>L. abrupta</u>). All of the projects would be operated as run-of-the-river plants, producing base-load power. Operating agreements would be reached between the licensee and the Corps before operation begins.

Features common to most of the projects include: (1) the proposed powerhouses would be located at the end of the navigation dam opposite the locks, (2) projects propose to install bulb or propeller turbines, (3) projects would operate automatically and/or remotely with wicket gates and turbine blade angles determined by computer to maximize power, (4) and projects would shut down when river flows are either too low or too high. A typical project section and layout are shown in Figures 2 and 3. Summaries of the individual projects are included in the DEIS (pp. 2-5 to 2-24); details are found in individual project applications which have been reviewed by the USFWS.

FERC conducted an environmental assessment (EA) (FERC 1987) that concluded that the proposed projects would interact with one another in a manner that would contribute to significant adverse impacts to target resources, which were identified as dissolved oxygen, recreational fishing, and river navigation. From the analysis in the EA, the staff determined that the proposed construction and operation of multiple hydropower projects in the basin warranted the preparation of an EIS to address the specific and cumulative environmental impacts of licensing the proposed projects.

Although many of the potential impacts from hydroelectric development in the Upper Ohio River Basin are site specific, interactions and cumulative impacts were identified as a major concern in the EIS scoping process, including comments from the USFWS (FERC 1988). The primary mechanism for project interactions and cumulative effects is through flow regulation at navigation dams and subsequent impacts on water quality, river hydraulics, fish populations, and recreational fisheries. The pink mucket pearly mussel is a federally-listed endangered species in the basin (DEIS pp. 3-4, 3-16, 3-19, 3-20) that may be affected by one or more of these mechanisms and by other cumulative impacts. Cumulative impacts are discussed in general in the DEIS Section 1.4 and in detail in the resource impact analyses (DEIS Section 4).

3. ECOLOGY OF THE PINK MUCKET PEARLY MUSSEL

3.1 TAXONOMIC STATUS

Stansbery (1985) conducted a review of the taxonomy of the species in order to resolve matters of synonomy that had been raised in the USFWS Recovery Plan (USFWS 1985). In order to locate and retrieve information from both the literature and from museum collections, it was necessary to become familiar with the various names under which the information and museum specimens were listed. He resolved that the valid name is <u>Lampsilus abrupta</u> (Say, 1831) having major synonomy with <u>L. orbiculata</u>. Some specimens were judged to have been misidentified as the subject species rather than the appropriate sibling species, <u>L. higginsi</u> (Lea, 1856).

3.2 DISTRIBUTION

Stansbery (1985) summarized the distribution of the species as entirely limited to the Mississippi River drainage basin. A few records outside the



10 TRASH BACK.

.

Figure 2. Longitudinal section of a hydropower station proposed for the upper Ohio River, showing typical turbing teatures. Source: FERC (1988).

69 DOWNSEREAM GATE OPERATOR



DANS-OWD BERLAND LOCK AND DAM PROJECT FERC NO. 6901-001

Figure 3. Layout diagram for the New Cumberland Look and Dam hydropower project, showing typical features for upper Chio hydropower proposals that consist of a powerhouse along the shoreline opposite the locks. Source: FERC (1988).

basin were attributed by Stansbery to <u>L. hioginsi</u> or to improper changes in museum labels. The species was originally widespread, although generally uncommon or rare, historically occurring in 25 river systems (USFWS 1985). Collection records listed by Stansbery show that it has been distributed mainly in the Ohio River system, including the Tennessee and Cumberland rivers (Figure 4). A population has existed, however, in the western part of the Mississippi basin, where thriving populations have been found in the Black River drainage of Arkansas-Missouri, and three river basins in Missouri: the Meramec-Big, Gasconade, and Osage river basins (USFWS 1985).

L. abrupta has been known recently from 16 rivers representing three major geographic regions, with the greatest concentration reported from the Tennessee River system (USFWS 1985). Age structure suggests continued reproduction at most sites (USFWS 1985). J. Jenkinson (pers. comm.; Appendix V) notes that it is fairly widespread in the Tennessee River system, comprising 0.1 to 0.2 of the mussel community; it seems to occur in small numbers whenever the mussel species list amounts to 12 to 14 species. Nearly all sites are in the largest rivers (4th or 5th order) rather than in tributaries; the Spring River, Missouri, is uncharacteristically small for the species. The species seems to have adapted to dam tailwaters in the Tennessee River system (USFWS 1985).

Stansbery (1985) hypothesizes that the original range in the Ohio River drainage once extended from its mouth at Cairo, Illinois, through the river's entire length to Pittsburgh, Pennsylvania, and in its major tributaries above Pittsburgh. Ortmann (1921) notes that this species (which he calls <u>L.</u> <u>orbiculata</u>) was a major constituent of the shell beds of the Ohio River between Pittsburgh and Cincinnati. Museum lots exist from the mainstem Ohio River, and the Allegheny, Monongahela, and Muskingum rivers of the FERC study area (Table 2: Stansbery 1965). The Kanawha River, which enters the Ohio in the Gallipolis pool, has a good population in a small stretch below Kanawha Falls (Clarke 1982, Taylor 1983). There are live populations in the lower Ohio River 50-100 miles above the mouth (USFWS 1985, S. Ahlstedt and W. Tolin pers. comm., Appendix V).

The species had not been found alive in the upper Ohio River and its tributaries for many years until recent discoveries (see DEIS pp. 3-20) below Gallipolis L&D (Tolin et al. 1987; discussed below). A relatively recent study of the unionid freshwater mussels of the Ohio River in Pennsylvania did not find any evidence of the continued existence of <u>L. abrupta</u> (Dennis 1970). Neither did surveys of mussels of the Ohio from Pittsburgh to Greenup Locks and Dam in 1979, which turned up only subfossil <u>L. abrupta</u> shells (Taylor 1980), or a 1983-1985 survey of the same reach (Zeto et al. 1987).

Because of the number of sites confirmed to contain <u>L. abrupta</u>, some attention has been given to delisting the species or downgrading it from endangered to threatened (Clarke 1982, USFWS 1985, Ahlstedt, pers. comm., Appendix V). Delisting criteria are described in the USFWS recovery plan (USFWS 1985). There apprars to be no effort currently underway to delist.

The habitat of the species is judged by Stansbery (1985), USFWS (1985) and all experts contacted, to be restricted to the riffles and runs of the larger rivers within its range. This habitat consists of relatively shallow (0.5 to 8.0 m), fast-flowing water over a firm, stable, gravel-cobble substrate. The



Figure 4. The distribution of Lampsilis abrupta (Say, 1831). Source: Stansbery (1985).

-10-

Table 2. Muskingum River specimens of Lampsilis abrupta (Say, 1831) in the collections of the Ohio State University Museum of Zoology, November 1985. Source: Stansbery 1985. o OSUM 47283 Locale: "B.Musk.", (MRM ?) Date: ca. 1830 Collector: Samuel P. Hildreth Specimens: 2 dry shells, males o OSUM 47284 Locale: "Muskn.", (MRM?) Date: ca. 1830 Collector: Samuel P. Hildreth Specimens: 3 dry shells, females o OSUM 19261 Locale: "Ohio River at Marietta," (MRM?) Date: 1879 Collector: William Holden Specimens: 5 dry shells: 2 males and 3 females o DSUM 43980 Locale: Muskingum River below Luke Chute Dam [3.9 mi, S. of Stockport, 4.5 mi. WSW of Beverly, Windsor/Waterford Twp.], Morgan/Washington Co., Ohio, (MRM 34) Date: 4 Aug. 1929 Collector: H. Ray Eggleston Specimens: 1 dry shell, male c OSUM 46951 Locale: Muskingum River at West Side Beach, just above its mouth, at Marietta, Washington Co., Chio, (MRM D --> 1) Date: 5 April 1930 Collector: H. Ray Eggleston Specimens: 1 left valve only, weathered dry, male o OSUM 44243 Locale: Muskingum River at Swift, [4.5 mi. W of Beverly, Windsor/Waterford Twp.], Morgan/Washington Co., Ohio (MRM 33.0) Date: 27 July 1930 Collector: H. Ray Eggleston Specimens: 1 dry juvenile shell, female? o OSUH 46955 Locale: Muskingum River below Luke Chute, [3.9 mi. ESE of Stockport, 4.5 m. WSW of Beverly, Windsor/Waterford Twp.], Morgan/Washington Co., Ohio, (MRM 34.0) Date: 3 Aug. 1930 Collector: H. Ray Eggleston Specimens: 5 dry and 1 weathered dry shells: 2 males, 4 females

Table 2. (continued) * OSUM 7195 Locale: Muskingum River at Lowell below dam, Adams Twp., Washington Co., Ohio (MRM 14) Date: 19 Jan. 1963) Collector: David H. Stansbery, Carol B. Stein, Mary Lightner and William Davies Specimens: 1 dry left valve, male o OSUM 8817 Locale: Muskingum River below Luke Chute Dam, Windsor Twp., Morgan Co., Ohio, (MRM 34) Date: 25 Sept. 1963 Collector: David H. Stansbery, Carol B. Stein Specimens: 1 badly weathered dry right valve, male * OSUM 16424 Locale: Muskingum River at Luke's Chute, Windsor Twp., Morgan Co., Ohio, (MRM 34) Date: 21-22 June 1966 Collector: Milton B. Trautman Specimens: 1 weathered dry shell, female; 1 weathered dry left valve, male * OSUM 17074 Locale: Muskingum River below Lowell Dam, Adams Twp., Washington Co., Ohio, (MRM 14) Date: 22 Oct. 1966 Collector: David H. Stansbery Specimens: I weathered dry shell, male; I weathered dry right valve, male o OSUM 45612 Locale: Muskingum River 1.9 mi. SW of Coal Run, 5.5 mi. WWW of Lowell, Waterford Two., Washington Co., Ohio, (MRM 22) Date: 25 May 1977 Collector: Robert Rothwell Specimens: 3 subfossil valves, sex uncertain o OSUM 50021 Locale: Huskingum River 2.6 m. N. of Luke Chute Dame, 4.5 mi. WNW of Beverly, Windsor/Waterford Twp., Morgan/Washington Co., Chio, (MRM 30.9 - 31.8) Date: 3 Oct. 1980 Collector: William N. Kasson, A. E. Spreitzer Specimens: 1 badly weathered dry right valve, female

Table 2. (continued) o CSUM 50889 Locale: Muskingum River 1.6 mi. ENE of Beverly, 5.9 mi. WNW of Lowell, Waterford Twp., Washington C., Ohio, (MRM 22.0 - 22.6) Date: 15 Oct. 1980 Collector: William N. Kasson, A. E. Spreitzer Specimens: 1 subfossil shell, male; 2 unmatched weathered dry valves, male o OSUM 51307 Locale: Muskingum River D.3 m. S of Lowell, B.6 mi. KWN of Marietta, Adams/Muskingum Twp., Washington Co., Ohio, (MRM 13.2 - 13.7) Date: 6 Nov. 1980 Collector: William N. Kasson, A. E. Spreitzer Specimens: 1 badly weathered dry left valve, male * OSUM 50938 Locale: Muskingum River 2.1 mi. SE of Lowell, 7.2 mi. N of Harietta, Adams/Muskingum Twp., Washington Co., Ohio, (MRM 10.7 - 12.8) Date: 16 Aug. 1981 Collectors: William N. Kasson, A. E. Spreitzer Specimens: I weathered dry right valve, male

o = Records based upon pre-1950 collections, subfossil specimens or cadly
weathered (old) specimens
* = Percords based upon record publicance (1965 --> 1985)

^{* =} Records based upon recent evidence (1960 --> 1985) MRM = Muskingum River Mile

hypothesis that this characterization has been skewed because of a preponderance of collections in the swift tailwaters compared to other sites has been dismissed by Stansbery (1985) on the basis of his own collections that ranged across habitat types and which used a more comprehensive technique. A. Miller (pers. comm., Appendix V) notes that the species has usually been found in gravel burs intermixed with dense mixed-species mussel communities, although it has also been found occasionally in sandy stretches with few other mussels.

Within the FERC study area there are two locations where the species has been confirmed or is strongly suspected based on fresh shell material. These locations are the Muskingum River below L&D 3 at Lowell, Ohio and a zone 12-15 miles downstream of Gallipolis L&D on the mainstem Ohio River (DEIS pp. 3-20) The USFWS has indicated a strong likelihood of occurrences at two other sites. These lie below Willow Island L&D between the confluences of the Muskingum and Little Kanawha rivers and below Belleville L&D from the dam to the downstream end of Buffington Island (Appendix II) although these localities have not been confirmed (DEIS pp. 3-20).

3.2.1. Muskimum River

Stansbery (1985) summarizes the Muskingum River locality as follows:

"This species would most likely be found in the Muskingum River nainstem only; and only from Luke Chute Dam (No. 5) downstream. ... Within the lower Muskingum River range described above, it is most probable that this species occurs in the first 1-3 miles downstream from dams 2, 3, 4, and 5. ... A greater number of fresh or relatively fresh shells of <u>L. abrupts</u> have been found at Lock and Dam Number 3 at Lowell than at any other site on the Muskingum River in recent years. We have not seen any living specimens from the Muskingum River in recent years even though the occasional collection of fresh shells indicates their continuing presence."

The strength of evidence for living specimens in the tailwaters of L&D 3 (DEIS pp. 3-79) is controversial. Historic occupation is not in question because of the numerous museum specimens available for study (listed in Stansbery 1985). The hydropower license applicant, the Upper Mississippi Water Company (Mitex, Inc. 1987), emphasizes the doubt that the species occurs at L&D 3 because the only specimens found lately have been "a few fossil and sub-fossil shells" rather than live specimens, and expresses the opinion that "It is quite possible that the species has already been extirpated from this reach of the Miskingum." The applicant acknowledges, however, that there is "no conclusive evidence that L. abrupta (sic) does not currently exist in the project impact area." The USFWS, however, quotes Stansbery's assertion "that the occasional collection of fresh dead shells in recent years indicates the continued presence of the species." (letter of August 7, 1986 from K. E. Krooneneyer to B. Fowler of Mitex Inc.).

Lack of evidence for living specimens in the Muskingum River is apparently not due to lack of effort expended to find them. Stansbery himself acknowledges that, "The evidence of [L. abrupta's] continued existence there has become noticeably weaker over the years in spite of increasing collecting effort." (Stansbery 1985). The chronology of these mostly fruitless collecting efforts

3.2.2 Ohio River Below Gallipolis L&D

By letter of September 28, 1987, the USFWS notified FERC that <u>L. abrupta</u> had been collected in the mainstem Ohio River on August 13, 1987 (Appendix II). The find was reported promptly in the scientific literature (Tolin et al. 1987). Two live specimens, a male and a female, were collected; identifications were confirmed by D. Stansbery (the male specimen resides in the collection of the Ohio State University Museum of Zoology) and Dr. Andrew Miller of the U. S. Army Corps of Engineers Waterways Experiment Station. The location was the upper Greenup pool between river mile 292.0 and 292.4 near the Greenbottom light and daymark along the right bank just downstream of Shoal Run. The location is approximately 12 miles downstream of the Gallipolis Locks and Dam, the most downstream hydroelectric project included in the FERC impact analysis(DELS pp. 3-16). <u>L. abrupta</u> had not been discovered in a previous sampling of that general area in 1985 (Zeto et al. 1987). The area but not the exact bed had been listed as a source of muscal material by Taylor (1980).

3.2.3 Elsewhere in the Upper Ohio River Basin

In its September 28, 1987, letter to FERC, the USFWS identified three reaches in the upper Ohio River basin where it believed the endangeted <u>L. abrupta</u> might be found (Appendix II). These areas are: (1) river miles 280 to 305, Gallipolis Locks and Dam to the confluence of the Guyandotte River, which includes the confirmed locality; (2) river miles 204.0 to 218.0, Belleville Locks and Dam to the toe of Buffington Island; and (3) river miles 172.0 to 184.0, between the confluences of the Muskingum River and the Little Kanawha River in the Belleville pool ten river miles below the Willow Island LAD (DEIS pp. 3-20). Although <u>L. abrupta</u> has not been located in the latter two reaches, the USFWS believes that conditions there are suitable for their existence (Appendix II and W. Tolin, pers. comm., Appendix V). Additional fir d sampling by the USFWS and the West Virginia Department of Natural Resources in these areas is planned for late August and September 1988.

3.3 DISSOLVED OXYGEN REQUIREMENTS

Based on its occurrence in riffle-run habitats of big rivers, Stansbery (1985) presumed that <u>L. abrupta</u> requires well-aerated water having high dissolved oxygen and low carbon dioxide concentrations. The habitat also suggested to him that the species might be intolerant of large amounts of organic debris and silt in suspension over long periods of time. No experimental studies have been conducted to verify such requirements for the species, however, as determined by a search of the scientific literature and discussions with mussel experts (personal communications listed in Appendix V). The best that can be done is to infer <u>L. abrupta</u>'s dissolved oxygen requirements by analogy with species that have been studied. All experts contacted agreed that the species requires a well-washed, stable substrate, usually of small gravel, that will prevent sufficient.

There is a general belief that freshwater mussels as a group are tolerant of low dissolved oxygen concentrations. Cole (1926) established that Anadontoides ferrussacianus Lea, a species found in organic mud and silt, could survive at nearly zero dissolved oxygen concentrations for several days in an early experimental study. Anodonta implicata (Say 1829) could survive when the dissolved oxygen concentration was exhausted (Eddy and Cunningham 1934), and Hiestand (1938) demonstrated that A. imbecilis could respire normally at about 0.73 mg/L of oxygen. The largest mussels had the lowest metabolic rate and thus were the least sensitive to low oxygen concentrations (Hiestand (1938). Imlay (1971) found a pool species, Amblema plicata, survived for 10 weeks at 0 mg/L.

Two traits seem to assist in tolerance of hypoxia--(1) behavioral, structural, and metabolic adaptations that allow mussels to clamp their shells together very tightly to seal themselves off from adverse conditions and maintain a lowered metabolic rate of dormancy and (2) a physiological amplitude for surviving at low oxygen tensions, seen mostly in the Anodontae (Fuller 1974). Freshwater mussels exhibit "rest periods" during which their oxygen consumption is much lower than during periods of activity, although it does not drop to zero (Salanki and Lukacsovics 1967).

Badman and Chin (1973; Pleurobera coccineum) and Dietz and his colleagues (e.g., Silverman et al. 1983; Ligumia subrostrata) have studied the physiological adaptations that allow mussels to live with minimal amounts of oxygen for long periods. These experiments have shown that the animals are capable of stating energy reserves in the form of simple sugars which can be metabolized with or without oxygen. There is a well-developed oxygen debt capacity for animals exposed to periods of shell closure, which is correlated with the disappearance of glycogen under anaerobic conditions (Badman and Chin 1973). Silverman et al. (1983) studied a species that has calcium concretions in the gills and other tissues that serve as a reservoir of calcium that is lost from the shell when blood pH drops during extended periods of hypoxia. The test organisms remained alive in hypoxic water for 2-4 weeks. Steffens et al. (1985) showed that the concretions, and presimably the adaptations to hypoxia, were evident in other species as well, including Anodonta grandis, Margaritifera herbeli, and Elliptio crassedens. "Adequate" oxygen concentrations that did not stimulate compensatory metabolic shifts were as low as 3% of saturation for P. coccineum (Badman and Chin 1973).

Riffle species may not fit the pattern observed for the more common slackwater species, however. Imlay (1971) examined the low oxygen tolerances of several unspecified "riffle species" of mussels in the laboratory and found that they required 2.5 mg/L of dissolved oxygen for survival at temperatures corresponding to summer. Imlay (1971) expressed the opinion that all species (both riffle and pool) require 6 mg/L for normal growth, based on as-yetunpublished experiments. Ellis (1931) reported that mussels became inactive when the saturation level of dissolved oxygen was less than one-fifth of atmospheric. Grantham (1969) found no live mussels in the Mississippi River where oxygen concentrations dropped as low as 3 mg/L even for short periods.

Low dissolved oxygen concentrations below some dams is providing an in situe experiment that indicates mussel sensitivities. J. Jenkinson (pers. comm., Appendix V) indicated that <u>L. abrupta</u> has been found in mixed mussel

communities below TVA dams where there has been periodic low dissolved oxygen. He described a survey in 1986 that showed no mussel mortalities at several sites when there was a minimum of 1 mg/L recorded at monitoring stations for more than one week. However, in 1988 there has been 0 to 0.5 mg/L dissolved oxygen below Watts Bar Dam (Tennessee River) for two weeks and adult mussels are being killed (the stressful period is underway as this is written in mid-August). The kill is not species-specific, and rarer species such as L.

Coon et al. (1977) found the sibling species L. higginsii (Lea) to have disappeared from Mississippi River Pools 8, 9, and 10 where it had been found in low numbers in a survey 45 years earlier prior to damming. The combined impacts of navigation pools (e.g., sedimentation, lower velocities, and barge traffic) is detrimental to these closely related species, but the critical environmental factor in their demise is not known.

Reproduction causes strains on mussel respiration which might affect survival. Portions of the gills are used in producing the dispersal phase, the glochidia, which renders these gill portions unsuitable for gas exchange (Matteson 1955, Fuller 1974).

S. Ahlstedt (pers. comm.) expressed the opinion that the juvenile mussel, immediately after release from the gills of the host fish, is the life stage most sensitive to low dissolved oxygen. He bases this opinion on unpublished observations of laboratory cultures in which mortality of early juveniles was high. Isely (1911) included abundant dissolved oxygen as a requirement for successful colonization of riffle substrates by juvenile mussels released from host fish. He reported that mussels radiate to other, more sandy or silty habitats as they grow larger.

In summary, freshwater mussels like <u>L</u>, <u>abrupta</u> that inhabit riffle habitats probably need fairly high dissolved exygen concentrations, perhaps near 6.0 mg/L, for normal growth and production. Like other mussels, however, the adults may be capable of tolerating quite low concentrations for periods of time that could extend to a few days. Low dissolved exygen concentrations in the Ohio River have probably exceeded these tolerance durations in historical times.

3.4 FISH HOSTS

Unionid molluses require a discrete fish species as an intermediate host for the glochidia stage. There seems to be confusion over the host for <u>L. abrupta</u> = <u>orbiculata</u>, and assignment of a host is currently considered speculative (S. Ahlstedt, pers. comm.). The host has not been identified for <u>L. abrupta</u> according to Stansbery (1985). Taylor (1980) lists "sauger?" as the host fish for <u>L. orbiculata</u>, which he presumed to be extinct in the Ohio River. However, Fuller (1974) tabulates the host fish for <u>L. orbiculata</u> as the sauger Stizostedion canadense, citing as authorities Coker et al. (1921), Surber (1913), and Wilson (1916). W. Tolin of the USFWS (pers. comm.), who collected the species alive below Gallipolis, feels that there could be multiple host fish species for <u>L. abrupta</u>.

Historical reproductive failure of L abrupta may have been caused by decline in populations of the required fish host, because it is well documented that

the fish fauna, including the sauger, became depauperate in the Ohio River during years of severe pollution (Pearson and Krumholz 1984). There are now abundant sauger in much of the upper Ohio River basin (DEIS pp. 3-35), including the confirmed and probable sites of <u>L. abrupta</u>.

4. ASSESSMENT OF IMPACTS

Installation of hydroelectric turbines on 18 navigation dams in the upper Ohio River basin has the potential for (1) reducing dissolved oxygen concentrations in the river below Gallipolis Locks and Dam (where <u>L. abrupta</u> is confirmed) and at the two possible sites identified by the USFWS (Appendix II), (2) altering the physical habitat in the tailwaters of Muskingum Lock and Dam No. 3 (where occurrence of the species is suspected), and (3) lowering population numbers of the fish host species through entrainment or low dissolved oxygen (FERC 1988). Whereas the possible habitat change at Muskingum No. 3 is a site-specific impact (DEIS pp. 4-23 and 4-24), the possible change in dissolved oxygen concentration of the mainstem Ohio River is a cumulative impact of all upstream (DEIS pp. 4-8), facilities, and loss of fish may be a cumulative impact of losses at all nearby facilities, both upstream and downstream (although the extent of fish movements is not documented).

The DEIS (FERC 1988) analyzed four alternatives for licensing action: (1) Projects as proposed, (2) Project operation to meet dissolved oxygen standards, (3) Project operation to meet antidegradation criteria, and (4) Projects selected to minimize impacts to all target resources. Impacts to <u>L</u>. <u>abrupta</u> are further analyzed here for these four alternatives. Each alternative is briefly summarized here; additional details are in the DEIS.

4.1 IMPACIS OF PROJECTS AS PROPOSED

4.1.1 Dissolved Oxygen

The DEIS (FERC 1988) estimates, on the basis of water quality modeling (DEIS pp. 4-4 to 4-9), that average daily dissolved oxygen (DO) concentrations in the Greenup pool below Gallipolis Locks and Dam could be depressed by a maximum of approximately 0.5 mg/L when all upstream projects operate as proposed under summer moderate flow conditions (Figure 5a). This decrease in DO is a conservative estimate of the greatest impacts of hydropower, made by assuming no aeration would occur at hydroturbines. In reality, slight amounts of aeration may occur at turbines, and the decrease in DO may be less than 0.5 mg/L. The impacts of hydropower on DO in this reach are minor because the 5 upstream dams provide little aeration. Little or no aeration would thus be lost (even though further reduction could be more significant for the ecosystem than if the water was well aerated). The estimated 0.5 mg/L maximum depression could cause current DO concentrations that are slightly above 6.0 mg/L to be depressed to slightly below 6.0 mg/L. At the summer low flow conditions (7010; Figure 5b) a similar DO decrease is estimated, but occurring between 5.5 and 6.0 mg/L (again, the estimated difference is the maximum expected impact of hydropower). At these lower flows, many projects cease operation.

Duration of low DD concentrations can be estimated from the ORSANCO monitor at Gallipolis LAD. DD concentrations at Gallipolis have fallen below 6 mg/L about 25% of the time in the critical high temperature-low DD summer months of



Figure 5. On't River dissolved oxygen model results for (a) summer moderate flow conditions and (b) summer low flow conditions (7010) for projects as proposed, in relation to confirmed and suspected beds of *Lampsills abrupta*. Modilied from FERC (1936).

July to September over the period 1980 to 1986 (Figure 6). A deficit of about 0.5 mg/L due to hydropower could extend the duration to 30 to 40 % of the time.

The estimates of current and project-impacted DO concentrations over the Greenup Pool mussel beds during typical summer conditions straddle the 6.0 mg/L DO concentration believed necessary for long-term growth of freshwater mussels, although this value is poorly substantiated as discussed in Sect. 3.3 (Imlay 1971). Thus, some small reduction in growth of <u>L. abrupta</u> may occur, based on analogy with other mussel species. Daily fluctuations, although reported to be small in this reach historically (FERC 1988), could further lower instantaneous mussel growth. The DO concentrations are not projected to be reduced to anywhere near what could be considered an acutely lethal level, based on analogy with other mussel species.

In the Ohio River reaches in which the USFWS suspects that <u>L. abrupta</u> may occur (Appendix II), the incremental change under summer moderate low flow conditions is also estimated to be about 0.5 mg/L (Figure 5a). The tailwaters of Belleville Locks and Dam present a near identical picture to that below Gallipolis—a shift in DO concentrations from just above 6.0 mg/L to just below that value. Impacts on <u>L. abrupta</u> should also be similar. Between the confluences of the Muskingum and Little Kanawha rivers with the Ohio, the incremental change occurs between about 6.8 and 6.3 mg/L (DEIS pp. 4-7, Figure 4.1.1-7), each above a level that might cause loss of growth of mussels.

Summer low flow conditions (7010) present a more severe pattern at the suspected <u>L</u>, <u>abrupta</u> site below Belleville LAD, however (Figure 5b). D0 concentrations there are estimated to fall to near 4.5 mg/L both with and without hydro (DEIS pp. 4-7, Figure 4.1.1-7), with a differential due to hydro of only a few tenths of a mg/L (within model error). This level would be inimical to long-term productivity of mussels such as <u>L</u>, <u>abrupta</u> but could probably be tolerated for short periods. Such levels may be a limiting factor currently during periods of low flows and high temperatures. Between the Muskingum and Little Kanawha Rivers there is expected to be no change in D0 concentrations near 6.0 mg/L at 7010 flows due to cessation of most hydropower operations at such low flows.

4.1.2 Physical Habitat

Although no flow or bottom habitat modifications at the Greenup pool mussel beds are anticipated by hydropower development at the Gallipolis L&D and above, the habitat will be altered by the project proposed at Mushingum No. 3. The DELS indicates that the tailwater of this proposed facility will be significantly altered with reduction in habitat for all mussels, including <u>L</u>. <u>abrupta</u> if it occurs there (DELS pp. 4-24). Habitat change could be in the form of a shift of current flow to one side of the river with significant reduction in flow of the existing tailwater for a distance of about 1 mile downstream. This change could occur in spite of the minimum spillage over the dam of 1520 cfs agreed to by the USFWS and the applicant (Mitex Inc. 1907; Appendix I). Channel erosion by the diverted flow is expected to cause substrate instability and sedimentation until the channel reequilibrates. These impacts may be severe enough to significantly reduce the available habitat for a period of time long enough to extirpate any extant <u>L</u>. <u>abrupta</u>.



Figure 5. Frequencies of summer dissolved oxygen concentrations at the ORSANCO monitor at RM 279, Gallipolis L&D, 1980-1986. Source: FERC (1998).

4.1.3 Host Fish

Some losses of the presumed host fish, sauger, might occur if the projects are built and operated as proposed (FERC 1988). Reductions in DO of magnitudes discussed above could affect the growth and production of coolwater fish according to USEPA (1986; Table 3) and a bioenergetics model applied by FERC staff (DEIS pp. 4-15 t 4-18 and Appendix E), although levels are not in the acutely lethal range. In the vicinity of the known or presumed L. abrupta beds, the zone with the greatest impact of DO on fish would be in the reach below Belleville LAD. There, DO levels for the maximum expected impact under summer moderate flow conditions are depressed to the zone of slight production impairment for all life stages (Figure 5a). Under summer low flow (7010; Figure 5b), conditions both with and without hydropower are well into the zones of moderate to severe growth and production impairment below Belleville, with hydropower causing little further DO decrease. Below Gallipolis and Willow Island, there would be slight production impairment. Slight to moderate reduction in growth and production might result in fewer numbers of host fish in the river, although the relationship is speculative.

Entrairment of larger fish through the hydropower turbines is likely to kill from 0 to 10% of those entrained, although experimental evidence for that range is poorly surported (FERC 1988). Small fish have a much lower mortality rate. Vulnerability of sauger to entrainment may be low, for Holland et al. (1984) found movements of sauger in the Mississippi River usually did not take them through the dams, and most interpool movements occurred at high water when turbines would not operate (DEIS pp. 4-26 and 4-28). Elsewhere, e.g., in the Tennessee River, combined navigation-hydropower dams are not detrimental to sauger populations, for the most productive fisheries for the species are below them. Fish protection devices with proven effectiveness for excluding sauger from turbines under conditions such as the Ohio River are not available (FERC 1988, Sect. 4.1.2.3). Therefore, there may be residual losses of fish hosts of L. abrunta that cannot be mitigated with present technology. Whether these losses will be compensated by fish populations is uncertain. Multiple host species, as suggested by Tolin (pers. comm.), would reduce the risk of loss of host fish.

4.2 IMPACIS OF PROJECT OPERATION TO MEET DISSOLVED OXYGEN STANDARDS

4.2.1 <u>Dissolved Oxvaen</u>

This alternative was selected to allow hydroelectric generation so long as spill flows at dams can assure maintenance of the standard of 5.0 mg/L that has been adopted by each of the relevant states (DEIS pp. 4-71, 4-74, 4-75). Because violations of the standard now occurs without hydropower, the criterion held violations to be no more frequent nor over more river miles than at present (FERC 1988; Sect. 4.2).

The moderate summer flows of Figure 5a showed no violation of state standards, so no additional spill would be required and the estimated dissolved oxygen concentrations and impacts on <u>L. abrupta</u> would be the same. Additional spill flows would be needed to meet the standard at flows below about 9,000 cfs at Pittsburgh (DEIS pp. 4-75). When summer spill flows are regulated to maintain the state standard at these flows, there are only small improvements in

dissolved oxygen between operating the projects as proposed and those estimated to maintain state standards at the lowest point in the sag curve. That point is near the suspected <u>L</u> <u>abrupta</u> beds below Belleville L&D (Figure 7). The greatest estimated improvement would be in the suspected locality below Willow Island L&D, where the difference (which is minor compared to natural variation) would approximate 0.5 mg/L. This largest charge would be in a range just above 6.0 mg/L whereas the charges in estimated concentrations at the other two sites are in the range of 5 to 6 mg/L. Spilling to maintain 5.0 mg/L at flows below 9,000 cfs will not prevent dissolved oxygen concentrations in the verified <u>L</u> <u>abrupta</u> bed and one suspected bed from dropping below 6.0 mg/L where growth might be inhibited.

For summer low flows (7010), the DO profile essentially matches the profile with no hydropower (Figure 5b). This is because only the projects at dams that are poor aerators would be allowed to operate (FERC 1988).

4.2.2 Physical Habitat and Host Fish

There would be little change in the impacts on physical habitat at Muskingum L&D No. 3 with this alternative (DEIS pp. 4-78). Increased spillage requirements would slightly reduce the time when large fish would be entrained. The difference is small, however, and some entrainment damages would continue.

4.3 IMPACIS OF PROJECT OPERATION TO MEET ANTI-DEGRADATION CRITERION

4.3.1 Dissolved Oxygen

This alternative allows hydroelectric generation with sufficient spill flows to maintain a DD concentration of 6.5 mg/L or better in all locations where this level occurs without hydropower (DEIS pp. 4-78, and 4-79). The value of 6.5 mg/L was selected from biological data on fishes as a minimum concentration that avoids deleterious effects (Table 3), mostly to early life stages (EPA 1986). DD concentrations below 6.5 mg/L occur frequently in the Ohio river basin even without hydroelectric development. However, spill flows were selected by the FERC (1988) optimization model that would allow hydropower generation yet maintain this oxygen level wherever it occurs now.

Under summer moderate flow conditions (Figure 8), efforts to maintain 6.5 mg/L where it occurs presently would still cause a few tenths of a mg/L drop in dissolved oxygen concentration in the Gallipolis pool at the <u>L</u>. abrupta bed (DEIS pp. 4-80, Figure 4.3.1-3). This estimate is minor compared to natural variability, however. Concentrations could be depressed from slightly above 6.0 mg/L to just below that value. Below Belleville L&D at the presumed <u>L</u>. abrupta bed, DD could be reduced slightly to near 6.0 mg/L. There would be little change in concentrations at presumed beds below Willow Island L&D, where concentrations are likely to be in the 6.5 to 7.0 range under these conditions. The biological effect on <u>L</u>. abrupta from these changes could amount to a small decrease in longterm growth and production by lowering the dissolved oxygen concentrations at summer moderate flow conditions at the known site below Gallipolis L&D. It should be emphasized, however, that the predicted decreases in dissolved oxygen are the maximum expected inpact of hydropower and are minor compared to natural variability in DO.



Figure 7. Ohio River DO model results for summer flow conditions, 9,000cfs at Pittsburgh, with spill flows to maintain state standards of 5 mg/... Source: FERC (1958).

Table 3. Levels of impairment to be expected for fishes at two age classes in nonsalmonid waters at different DD concentrations (USEPA, 1986):

a. Early Life Stages

b.

,

0 0 0	No production impairment = 6.5 Slight production impairment = Moderate production impairment Severe production impairment = Limit to avoid acute mortality	and above 5.5 = 5.0 4.5 = 4.0
Other	Life Stages	
0 0	No production impairment = 6.0 Slight production impairment =	and above 5.0
0	Moderate production impairment Severe production impairment = Limit to avoid acute portality	= 4.0 3.5 = 3.0



Figure 8. Ohio River DO model results for summer moderate flow conditions, with spill flows to maintain 6.5 mg/L wherever they occur prior to hydropower development. Source: FERO (1988).

At summer low flows (7010), few projects would operate. Thus, the DO concentrations for this case are essentially the same as with no hydropower. There would be little additional impact on <u>L. abrupta</u> at any site beyond the naturally stressing conditions (Figure 5b).

4.3.2 Physical Habitat and Host Fish

As with Alternative 2, there would be little change in the physical habitat difficulties at Muskingum L&D No. 3 with this alternative (DEIS pp. 4-81). Increased spillage requirements would further slightly reduce the time when large fish would be entrained. The difference is small, however, and some entrainment damages would continue.

4.4 PROJECTS SELECTED TO MINIMIZE IMPACTS TO ALL TARGET RESOURCES

4.4.1 Dissolved Oxygen

The dissolved oxygen criterion under this alternative was the same as for the antidegradation alternative; other constraints were applied, however, that altered the projects that would operate and the spill flow regime (DEIS pp. 4-81 to 4-83). Dissolved oxygen concentrations and their expected biological effects at the known or presumed <u>L. abrupta</u> sites would be essentially identical under this

alternative to those seen under the nondegradation alternative (Figure 9).

4.4.2 Physical Habitat

The principal effect of this alternative would be elimination of potential damages to any surviving population of <u>L. abrupta</u> in the lower Muskingum River below the proposed Muskingum No. 3 project. This project would not be recommended for development under this alternative in order to protect the tailwater habitat from physical alteration (DEIS pp. 4-84).

4.4.3 Host Fish

Projects near the confirmed or suspected <u>L. abrupta</u> beds would be licensed to operate under this alternative, and spill flows would be the same as in Alternative 3. Therefore, fish mortalities due to entrainment should be the same as in Alternative 3.

4.5 SIGNIFICANCE OF IMPACTS

The Endangered Species Act of 1973, as amended, requires the federal agency to determine whether its action will "affect listed species or their habitat". FERC staff has determined in its biological assessment incorporated into its DEIS that the proposed development of hydroelectric projects in the upper Ohio River basin could have some effect on <u>L. abrupta</u>, the pink mucket pearly mussel. The additional information provided in this report confirms staff's previous conclusions concerning this species presented in the DEIS. Staff's conclusions on these effects, contained in the previously prepared biological assessment and those included herein, are based on the best estimates of the environmental requirements of the species (using analogy with other freshwater mussel species, and the estimated changes in water quality (DO), habitat, and



Figure 9. Ohio River DO model results for summer moderate flow conditions, with spill flows to maintain 6.5 mg/L wherever they occur prior to hydropower development under Alternative 4 that protects all target resources. Source: FERO (1958).

fish hosts that might be caused by hydropower development. There are important uncertainties in this information, however, that prevent precise definition of impacts. These uncertainties are discussed in Section 5.

The impacts were determined to differ with alternative development scenarios. FERC staff believes that the preferred alternative 4, "Projects Selected To Minimize Impacts To All Target Resources, will have no demonstrable effect on populations of <u>L. abrupta</u>, for the following reasons:

a. Dissolved oxygen concentrations will remain close to levels now occurring in the vicinity of a confirmed population of <u>L. abrupta</u> downstream of the Gallipolis L&D and two suspected population sites below Belleville L&D and Willow Island L&D. Small estimated reductions of 0.5 mg/L or less in average concentrations are insignificant compared to existing variability in DO, and they would be largely offset by aeration of riverflow by turbines, which may amount to a few tenths of a mg/L.

The estimated maximum oxygen reductions below Belleville and Gallipolis due to hydropower development are to levels only slightly below the concentration of 6.0 mg/L suggested in the literature to be the level below which normal long-term growth of freshwater mussels is impaired. This limit is poorly documented in the scientific literature, however, and may have an error associated with it that is as great as the estimated decline in DO levels. Freshwater mussels as a group are known to tolerate much lower DO concentrations (to nearly zero) for periods of days, suggesting that they could survive short-term transient low DO concentrations so long as the estimated average is maintained. DO concentrations significantly lower than 6.0 mg/L would occur only as a result of conditions occurring without hydropower; the proposed hydropower projects are not capable of lowering DO to lethal levels.

- b. The tailwater of Muskingum L&D No. 3, which has recently held a population of <u>L. abrupta</u> and may still do so (although no recent collectors have found live specimens), would not be recommended for development in order to protect riffle and run species. This is the only locality where significant habitat change detrimental to a known or likely population of <u>L. abrupta</u> was predicted.
- c. Although entrainment in hydropower turbines will kill some fishes, it is uncertain but viewed as unlikely that this source of additional mortality would significantly reduce populations of the presumed fish host, the sauger Stizostedion canadense, or of any other host. There is no information available, however, relating fish numbers to the strength of mussel populations.
- d. Maintenance of dissolved oxygen of 6.5 mg/L in areas of the river where these levels now occur might allow oxygen to be only slightly degraded (by 0-0.5 mg/L) to levels which cause only moderate to slight impairment of fish production in the vicinity of <u>L</u>. <u>abrupta</u> beds. It is speculative whether this decline in production will be enough to cause a loss of numbers of host fish that is significant for mussel populations.

If other alternatives considered by FERC staff (1988) are selected for licensing, the results of this assessment would be somewhat different.
Differences among alternatives are not great for dissolved oxygen in the vicinity of the confirmed or suspected <u>L. abrupta</u> beds between Willow Island L&D and mid-Greenup pool (Figures 5, 7-9). All of the differences are minor because the proposed projects have little effect on DO in these reaches. Reduced spillage requirements could cause a somewhat larger entrainment loss of potential fish host(s). A decision to license a project at Muskingum L&D No. 3 might eliminate or reduce the value of this tailwater as a site for the present population (if it exists), which is likely to be disturbed by realignment of the flows in the river channel. Surviving individuals or a substrate equilibrium, however.

5. ADEQUACY OF THE DATA

FERC staff believes the information provided in the DEIS was adequate to determine the effects of the proposed hydropower development on <u>L. abrupta</u>. The additional information provided in this document references many of the conclusions reached in staff's biological assessment incorporated into its DEIS. Based on that assessment, and the additional information provided herein, staff therefore concludes that sufficient information exists to determine that the proposed hydropower development of the upper Ohio River Basin, under each of the four alternatives, is not likely to jeopardize the continued existence of <u>L. abrupta</u> or result in the destruction or adverse nodification of its critical habitat.

Although the FERC considers that overall the information is adequate for making a reasonable judgement, certain types of information were lacking in the scientific literature and were not adequate to predict impacts with a great deal of certainty. This information concerns the following subjects:

- a. There are no thorough locality records for <u>L. abrupta</u> in the study area. If additional confirmed localities are determined, however, then the endangered status of the species, already questioned in the USFWS Recovery Plan, would be in doubt. Further surveys of the Ohio River basin are desirable, especially in the zones in which the USFWS has suggested that populations should exist (Appendix II).
- b. The species' physiological and ecological requirements for physical habitat, dissolved oxygen, and other features relevant to hydropower development are not known with certainty. Much of the information used herein has been by analogy with other species of freshwater mussels or with the general mussel community with which <u>L. abrupta</u> associates. Although this information is useful, a more certain judgement could be made with more species-specific data.
- c. Data on long-term and short-term requirements for dissolved oxygen by <u>L</u>. <u>abrupta</u> (and all mussels) are especially needed. The experimental data which forms the basis of an opinion that 6 mg/L DO is needed for normal growth of freshwater mussels (Imlay 1971) should be published in the refereed scientific literature and/or confirmed by independent research. It is difficult to obtain such experimental data on a species whose occurrence is so limited, however.

- d. The definitive fish host(s) remain unclear in the recent literature, even though there seems to be some concensus in the older literature that it is sauger. This uncertainty makes it difficult to assess the possibility that loss of host(s) could cause loss of the species. Detailed life history information would assist preservation of the species.
- e. Entrainment damages to fish populations are not well quantified, so damages to the mussel's fish host(s) remain uncertain. Furthermore, there is no information available relating fish numbers to the strength of mussel populations. As discussed in detail in the DEIS (FERC 1988), neither accurate estimates of the impact of entrainment on Ohio River fish nor effective fish protection devices for the Ohio are available. Resolution of the entrainment issue would aid in evaluating impacts on freshwater mussels.

6. SUMMARY

Installation of hydroelectric facilities at navigation dams in the upper Ohio River Basin is estimated to have some impact on the endangered freshwater mussel, <u>Lampsilus abrupta</u>, that varies with the development scenario selected. FERC staff judges the impact of the preferred alternative to be so small as to be considered no demonstrable effect. FERC staff believes that none of the four hydropower development alternatives examined in the DELS would be likely to jeopardize the continued existence of <u>L. abrupta</u> or result in the destruction or adverse modification of <u>L. abrupta</u> critical habitat.

Under Alternative 1 (Projects as Proposed), reduced dissolved oxygen concentrations (that are the cumulative result of reductions in aeration capacity of existing structures) by an estimated maximum of 0.5 mg/L at a confirmed <u>L. abrupta</u> locality about 14 miles below Gallipolis L&D could lower mussel growth there by an unquantifiably small amount during 30-40% of the time during summer months. The DD reduction would occur from just above the 6.0 mg/L concentration that is the reported (but unsubstantiated) lower limit for normal growth to just below it. The DD reduction that could impact growth would, therefore, be <0.5 mg/L. Impacts to <u>L. abrupta</u> from lowered DD at two localities that the USFWS considers likely to contain the species are estimated to be similar or less than at the confirmed site, although existing DD conditions are less satisfactory at the site below Belleville L&D.

Detrimental changes in the physical habitat in riffles and runs downstream of Muskingum L&D No. 3, where <u>L. abrupta</u> has occurred recently, are predicted with the project as proposed, and the DEIS does not recommend the project for licensing at this time. Physical habitat changes are not anticipated at other <u>L. abrupta</u> locations.

Reductions in numbers of fish hosts (presumably sauger) for glochidia dispersal due to entrairment nortality at nearby hydroprojects might occur. There is little information available for quantitatively linking host fish abundance to the strength of mussel populations. The impact is judged to be small, however, because sauger are not especially prone to movement between pools (based on migration studies on the Mississippi River) and some of the most productive sauger fisheries in other river systems are below multiple-use dams (e.g., on the Tennessee River). For Alternative 2 (Project Operation to Meet Dissolved Oxygen Standards), when summer spill flows are regulated to maintain the state standards, there are only minor improvements in DO concentrations (compared to Aternative 1) at the suspected <u>L. abrupta</u> beds below Belleville L&D and Willow Island L&D. Spilling to maintain DO standards, however, will not prvent DO concentrations in the verified <u>L. abrupta</u> bed and one suspected bed from dropping below 6.0 mg/L where growth might be inhibited. There would be little change in the impacts on physical habitat (compared to Alternative 1) at Muskingum L&D No. 3 with this alternative. Increased spillage requirements would slightly reduce the time when large fish would be entrained; however, the difference is small.

Alternative 3 (Project Operation to Meet Anti-Degradation Criterion) would cause a few tenths of a mg/L drop in DO concentrations, from slightly above 6.0 mg/L to just below that value, at the <u>L. abrupta</u> bed in the Gallipolis pool. At the presumed <u>L. abrupta</u> bed below Belleville L&D, DO could be reduced slightly to near 6.0 mg/L. There would be little change in DO concentrations at presumed beds below Willow Island L&D. The predicted DO decreases are the maximum expected and are minor compared to natural variability in DO. As with Alternative 2, there would be little change in the physical habitat difficulties at Muskingum L&D No. 3 with this alternative. Increased spillage requirements would further slightly reduce the time that large fish would be entrained.

Under Alternative 4 (Projects Selected to Minimize Impacts to All Target Resources), D0 concentrations and their expected biological effects at the known or presumed <u>L. abrupta</u> sites would be essentially identical to those seen under Alternative 3. Spill flows would be the same as in Alternative 3, therefore fish mortalities due to entrainment should be the same as in Alternative 3. The principle effect of this alternative would be the elimination of potential damages to any surviving population of <u>L. abrupta</u> in the lower Muskingum River below the proposed Muskingum No. 3 project.

7. REFERENCES

Badman, D. G., and S. L. Chin. 1973. Metabolic responses of the fresh-water bivalve, Pleurobema coccineum (Conrad), to anaerobic conditions. Comp. Biochem. Physiol. 44B:27-32.

Bates, J. 1970. Ohio mussel fisheries investigation, May 15, 1967 - September 1, 1970. Final Report, Part 1. Mussel Studies. Report to Division of Wildlife, Ohio Department of Natural Resources, Columbus, Ohio.

Clarke, A. H. 1982. Survey of the freshwater mussels of the upper Kanawha River (RM 91-95), Fayette County, West Virginia, with special reference to Epioblasma torulosa (Rafinesque) and <u>Lampsilus abrupta</u> (Say) (= <u>Lampsilis</u> <u>orbiculata</u> (Hildreth), of authors). Final Report, USFWS, Newton, Massachusetts.

Coker, R. E., A. F. Shira, H. W. Clark, and A. D. Howard. 1921. Natural history and propagation of fresh-water missels. Bull. U. S. Bur. Fish. 37:77-181.

Cole, A. E. 1926. Physiological studies on fresh-water clams. Carbondioxide production in low oxygen tensions. J. Exp. Zool. 45:349-359.

Coon, T. G., J. W. Eckblad, and P. M. Trygstad. 1977. Relative abundance and growth of mussels (Mollusca:Eulamellibranchia) in pools 8, 9, and 10 of the Mississippi River. Freshwater Biology 7:279-285.

Davies, W. D. 1963. Some effects of condenser discharge water on teh growth and distribution of the mussel population of a larege river. Masters Thesis, The Ohio State University, Columbus, Ohio.

Dennis, S. D. 1970. Pennsylvania mussel studies, June 1, 1968 - October 1, 1969. Final Report. Center for Aquatic biology, Eastern Michigan University, Ypsilanti, Michigan.

Eddy, N. W., and R. B. Cunningham. 1934. The oxygen concumption of the fresh-water mussel, Anodonta implicata. Proc. Penn. Acad. Sci. 8:140-143.

Ellis, M. M. 1931. Some factors affecting the replacement of the commercial fresh-water mussels. U. S. Bur. Fish, Fish Circ. No. 7, 1-10.

EPA (U. S. Environmental Protection Agency). 1986. Ambient water quality criteria for dissolved oxygen. EPA 440/5-86-003, Washington, D. C.

FERC (Federal Energy Regulatory Commission). 1987. Environmental Assessment for the Ohio River Basin. Office of Hydropower Licensing, Washington, D. C.

FERC (Federal Energy Regulatory Commission). 1988. Draft Environmental Impact Statement: Hydroelectric Development in the Upper Ohio River Basin. PERC Docket No. EL85-19-114. Ohio, Pennsylvania, West Virginia. Office of Hydropower Licensing, Washington, D. C.

Fuller, S. L. H. 1974. Clams and mussels (Mollusca: Bivalvia). pp. 215-273

in C. W. Hart and S. L. H. Fuller (eds.), Pollution Ecology of Freshwater Invertebrates. Academic Press, New York.

Grantham, B. J. 1969. The fresh-water pelecypod fauna of Mississipi. Doctroral Dissertion, Univ. of Southern Mississippi, Hattiesburg.

Hiestand, W. A. 1938. Respiration studies with fresh-water molluscs: I. Oxygen consumption in relation to oxygen tension. Proc. Indiana Acad. Sci. 47:287-292.

Holland, L., D. Huff, S. Littlejohn, and R. Jacobson. 1984. Analysis of existing information on adult fish movements through dams on the upper Mississippi River. U. S. Fish and Wildlife Service, National Fishery Research Laboratory, LaCrosse, Wisconsin.

Imlay, M. J. 1971. Bioassay tests with naiads. pp 38-41 in S. E. Jorgenson and R. W. Sharp (eds.). Proceedings of a symposium on rare and endangered mollusks (naiads) of the U. S. U. S. Department of the Interior, Fish and Wildlife Service, Bureau of Sport Fisheries and Wildlife.

Isely, F. B. 1911. Preliminary note on the ecology of the early juvenile life of the Unionidae. Biol. Bull. 20:77-80.

Matteson, M. R. 1955. Studies on the natural history of the Unionidae. Amer. Midl. Natur. 53:126-145.

Mitex, Inc. 1987. Biological assessment of the effects of the Muskingum Lock and Dam No. 3 Hydroelectric Project on the pink mucket pearly mussel. Revised edition. Boston, Massachusetts.

Ortmann, A. E. 1921. A monograph on the Naiads of Pennsylvania. Annals of the Carnegie Museum. 8:1-384.

Pearson, W. D., and L. A. Krumholz. 1984. Distribution and status of Ohio River fishes. ORNL/SUB/79-7831/1. Oak Ridge National Laboratory, Oak Ridge, Tennessee.

Rothwell, R. J. III. 1979. The effect of elevated temperature on the growth of freshwater clams (Mollusca: Pelecypoda). Masters Thesis, Edinboro State College, Edinboro, Pennsylvania.

Salanki, J., and F. Lukacsovics. 1967. Filtration and oxygen consumption related to the periodic activity of the freshwater mussel (<u>Anodonta cygnea</u> <u>L.</u>). Annal. Biol. Tihany. 34:85-98.

Silverman, H., W. L. Steffens, and T. H. Dietz. 1983. Calcium concretions in the gills of a freshwater mussel serve as a calcium reservoir during periods of hypoxia. J. Exp. Zool. 227:177-189.

Stansbery, D. H. 1985. The status of <u>Lampsilus abrupta</u> (Say, 1831) with special attention given to the Muskingum River population (Mollusca: Bivalvia: Unionoida). Report for WAPORA, Inc. The Ohio State University Museum of Zoology, Columbus, Ohio.

Stansbery, D. H. and C. C. King. 1983. Management of Muskingum River mussel (unionid mollusk) populations. Ohio State Univ. Museum of Zool. Reports 1983(5):1-79.

Steffens, W. L., H. Silverman, and T. H. Dietz. 1985. Localization and distribution of antigens related to calcium-rich deposits in the gills of several freshwater bivalves. Can. J. 2001. 63:348-354.

Stillwell, , J. M., M. C. Mathews, and D. E. Bell. 1974. Final report on a study of the effects of power plant cooling water systems upon the Muskingum River biota. Phase III. Final Report to Columbus and Southern Ohio Electric Co. by Battelle, Columbus Laboratories, Columbus, Ohio.

Surber, T. 1913. Notes on the natural hosts of fresh-water mussels. Bull U. S. Bur. Fish. 32:110-116.

Taylor, R. W. 1980. A survey of the freshwater mussels of the Ohio River from Greenup Locks and Dam to Pittsburgh, Pa. Report for the U. S. Army Corps of Engineers, Huntington District. Marshall University, Huntington, West Virginia.

Taylor, R. W. 1983. A survey of the freshwater mussels of the Kanawha River. Prepared for Huntington District, Corps of Engineers, Huntington, West Virginia.

Tolin, W., J. Schmidt, and M. Zeto. 1987. A new location for the federally-listed endangered unionid <u>Lampsilus abrupta</u> (Say 1831) (=<u>L.</u> <u>orbiculata</u> (Hildreth, 1828)) in the upper Ohio River bordering West Virginia.

Malacological Data Net 2:18.

USFWS (United States Fish and Wildlife Service) 1985. Recovery Plan, Pink mucket pearly mussel <u>Lampsilis orbigulata</u>. Atlanta, Georgia.

Wilson, C. B. 1916. Copepod parasites of fresh-water fishes and their economic relations to mussel glochidia. Bull. U. S. Bur. Fish. 34:331-374.

Zeto, M. A., W. Tolin, and J. Schmidt. 1987. The freshwater musels of the upper Ohio River, Greenup and Belleville Pools, West Virginia. The Nautilus 4:182-185.

APPENDIX I. Letter from K. E. Kroonemeyer, USFWS, to W. Fowler, Mitex, Inc., dated April

6,1987.



United States Department of the Interior

FISH AND WILDLIFE SERVICE

IN REPLY AFTER TO:

Columbus Field Office 6950-H Americana Parkway Reynoldsburg, Obio 43563

April 6, 1987

Mr. William Fowder -> Mitex, Inc. 91 Newbury Street Boston, Hassachusetts - 02116

Dear Mr. Fowler:

We have reviewed the February 26, 1987 revised Biological Assessment of the effects of the Muskingun Lock and Dan No. 3 hydroelectric project on the pink mucket pearly mussel. Lamosilis orbiculate or Lamosilis abrupta.

This letter is prepared in accordance with provisions of the Endangered Species Act of 1973, as amended and provides informal consultation.

Muskingum Lock and Dat No. 3 consists of an 840-foot long uncontrolled, rock fill crib dam with a concrete cap. The lock is located in a 5,400-foot long, 75-foor wide canal along the left bank of the river. The proposed hydroelectric project would redirect flows from the dam spillway through two hydraulic turbines. As proposed, an intake channel would be constructed upstream of and around the right abutment of the dam. Flows would pass through the proposed powernouse and be released to the river via a 200-foot long tailrace. The powernouse would contain two 3,500 cfs turbines requiring a total capacity of 7,000 cfs.

Dr. David H. Stonsbery, of the Chio State University Huseum of Zoology, provided Mitem with a report on the status of <u>Lapsilis abrupts</u> with special attention given to the Muskingum River population. Stansbery's report reached the following conclusions:

- The Shio Nucket, Labosilis abtuntu, has a distribution limited to the Mississiphi River drainage basin. A single record for western Lake Erie and another for the Mispari River are very doubtful and are suspected of being transpositions of labels of specimens.
- The habitat of this species is the fitfles and runs of the largest rivers within its range.

- 3. The above habitat (2) consists of relatively shallow, fast flowing water over a firm, stable, gravel-cobble substrate.
- 4. This species would most likely be found in the Muskingun river main stem only; and only from Luke Chute Dam (No. 5) downstream.
- 5. Within the lower Muskingum River range described above, it is most probable that this species occurs in the first 1-3 miles downstream from dams 2, 3, 4, and 5. (Dam No. 1 has been removed for navigation purposes.)
- 5. A greater number of fresh of relatively fresh shells of <u>Lampsilis</u> <u>abruots</u> have been found at Lock and Dan Number 3 at Lowell than at any other site on the Nushingum River in recent years. We have not seen any living specipens from the Muskingum River in recent years even though the occasional collection of fresh shells indicates their continuing presence.

Although no live specimens have been seen or collected at Lock and Dam No. 3 in recent years, there is no conclusive evidence that <u>Lampsilis abrupta</u> does not exist in the proposed project area. With this in mind, Dr. Stansbery recommended the following mitigation measures to preserve, improve, or expand the habitat for Lampsilis abrupta at the site.

- Maintenable of the natural regimen of water flow and water temperature.
- Assurance of adequate suspended organics of the type used as food by the unionids in question.
- Precautionary measures to avoid adding any toxic materials to the rivers or the removal of any materials necessary to the continued life of these species.
- 4. Avoidance of any changes which would significantly decrease the species diversity of the fishes at this site, reduce population sites or alter their movements (feeding forays, breeding runs, etc.)
- 5. Taking measures to insure that the water cifluent from the generating unit would rementer the river at the present dam site with the same or improved quality of the water currently passing over the dam. As I understand it, the only resource sought in this retrofit hydroelectric project is the energy held by the water above the dam by virtue of its higher elevation. Theoretically, at least, this could be accomplished without the alteration of any factors critical to the block of the river.

We have recommended a minimum flow of 2,280 cfs (30 percent of mean annual flow) for the April 1 - June 30 time period and 1.520 cfs (20 percent of mean annual flow) for the remainder of the year, July 1 - March 31. Rowever, we have agreed, that if the proposed modeling soudies include that a flow of 1,000+ cfs during the July-March time period provides adequate velocity and depth over the substrate, we will accept 1,000 cfs as the minimum flow. The recommendation of 2,280 cfs for the April - June time period would still remain. With this understanding, Mitex, the consultant for Upper Mississippi Water Company, has developed the following mitigation proposals:

- Project operation will be subject to a continuous minimum flow as confirmed in physical model tests. The range will be 1,000 to 1,520 cfs from July to March. These are 165 percent and 235 percent respectively, of the 7010 flow. A minimum flow of 2,230 cfs will be maintained from Arril 1 to June 30.
- A detailed erosion and sediment control plan will be prepared prior to commencement of construction. Upper Mississippi will consult with the ODNR, OEPA, USFWS, and USEPA in preparetion of this plan. A primary gcal will be to minimize introduction and resuspension of river sediments.
- Upper Mississippi will operate the facility in a strict "run-of-river" mode.
- 4. Upper Mississippi will operate D.O. monitors at the project. In the event levels drop below the state standard as a result of hydropower operations, aerotors will be run, or minimum flows increased to whatever level is necessary.
- 5. Upper Mississippi will field survey all areas to be dredged prior to actual dredging. In the event any specimens of <u>Lampsilis abrupta</u> are found, they will be transplanted, using methods approved by the FWS, to locations approved by the FWS.
- Upper Mississippi will schedule dredging and other disruptive in-river operations to avoid times of maximum mussel sensitivity to turbidity. The best time would be from late fall to early winter.

Should the above conditions be incorporated into the project, this precludes the need for further action on this project as required by the 1973 Endangered Species Act, as anended. Should the project be modified of the above conditions not be incorporated or new information become available that indicates listed or proposed species may be affected, consultation should be initiated.

In item Number 5 above, it states that if any <u>Lampsilis Abrupta</u> are found, they will be transplanted. This action would require that a federal endangered species permit be obtained. The permit can be obtained from our Regional Office in Twin Gitles, Minnesota. We will assist the applicant in obtaining this permit prior to any relocation efforts. We appreciate this opportunity to comment on your proposed project. If you have any questions, please call Ken Hulterer at 641/469-6923.

Sincerely yours,

Nent C. Moorimujer Rene E. Kroonemeyer Supervisor

4.

cc: Chief, Ohio Division of Wildlife, Columbus, OH Ohio DNE, Outdoor Recreation Serv, M. Colvin, Columbus, OH APPENDIX II. Letter from Charles. J. Kulp, USFWS, to James Keany, FERC, dated September



UNITED STATES DEPARTMENT OF THE INTERIOR 987 OCT 6 Fr 4 33 OFFITE OF FILEC new start Carlor State College, Pennsylvania 16801

September 28, 1987

Mr. James Keany Ecologist (308RB) Pederal Energy Regulatory Commission 825 North Capitol Street, N.E. Washington, D. C. 20426

Dear Mr. Keany:

This alerts you to significant new biological data from the Chio River pertaining to a federally-listed endangered mussel. On August 13, 1987, biologists from the Fish and Wildlife Service and the West Virginia Department of Natural Resources, Water Resources Division, collected the pink mucket pearly sussel, Lappeilis abrupta from the upper Greemup pool between rivernile 292.0 and 292.4 near the Greenbottom light and daymark along the right bank just downstream of Shoal Run. Lemosilis abrupta is listed as an endangered species by the Service under the Endangered Species Act of 1973. A mussel population at the site was discovered by Service and WUDNE personnel in 1985, but L. abrupta was not encountered at that time. These data are expected to be published in The Nautilis in October 1987. This reach of the Ohio River was also identified in the Service's report entitled A Physical and Biological Survey of the Chio River Islands as having significant mussel resources.

Dr. Ralph Taylor identified ten sites in the reach between Gallipolis Locks and Dam and the confluence of the Guyandotte River (rivermiles 280 - 305) which contained shell material on the shore. Two sites were brailled by Dr. Taylor but no mussels were apparently discovered. Dr. Taylor did not find evidence of missels in the vicinity of the Greenbotton site. Wenty-one species are known to exist in the reach surveyed by Dr. Taylor and may exist in the Greenbotton missel bed as well. Sixteen species have been collected from riveraile 292-292.4 including L. abrusts. We suspect that L. abrusts exists at other sites which contain suitable habitat in the Onio River, especially in this reach.

We collected mussels, including L. <u>abrupta</u>, at the Greanbottom site by brailling (dove tail). The mussel bed, of unknown size and shape, is located in 14 to 23 feet of water over sand, gravel, comble, and boulder substrate. L. abruna was located in water 16 to 18 feet deep approximately 200 feet off the shoreline. The dominant species in the bed are <u>Elliptic</u> crassidens crassidens, (elephant ear); <u>Actinomaizs ligamentina</u> carinata, (mucket); Quadrula pustulosa pustulosa, (pirpleback), and Pleurobena condatum, (Onio River pictoe).

The elephant ear and market were presumed extirpated from the Chio River. <u>Pleurabema. cordatum, Quadrula metanevra</u>, (monkeyface), and <u>Plethobasus</u> <u>cyphyus</u>, (bullhead), were relatively abundant in the population but are considered rare by the State of Chio. Two adult specimens of <u>Fusconaia</u> ebena, (ebony shell), represent a significant upriver range extension for this species. The collection of <u>Truncilla</u> truncata, (deertoe), is also unique since it was once very abundant in the mainstem Chio River but has not been reported for approximately 75 years.

Two adult specimens of <u>L</u>. <u>abrupta</u> were collected, one male and one female. The male was sacrificed at the site for identification. The female was kept alive and later shipped to Dr. Andrew Miller at the Waterways Experiment Station for research purposes. The male <u>L</u>. <u>abrupta</u> and representatives of our collections from the site are accessioned at the OSU Museum of Zoology.

Section 7 of the Endangered Species Act of 1973, as amended (16 U.S.C. 1531 et seq.), requires that federal agencies, in consultation with the Service, insure that their actions are not likely to jeopardize the continued existence of any endangered or threatened species or result in the destruction or adverse modification of critical habitat. Therefore, Service recommends that special attention be given to potential impacts on endangered mussels from orgoing and future projects within the following three reaches of the Ohio River: rivermiles 280.0 - 305.0, Gallipolis Lock and Dam to the confluence of the Guyardotte River; rivermiles 204.0 to 218.0, Belleville Lock and Dam to the toe of Buffington Island; and rivermiles 172.0 to 184.0, between confluences of the Muskingum River and the Little Karawha River. Although L. ahrupta has not been located in the latter two reaches, habitat conditions are suitable for their existence.

In accordance with Section 7 of the Endangered Species Act, as amended, informal consultation with the Service will be required on any federally funded, licensed, or permitted project within these three Ohio River reaches. This informal consultation will be required on all projects which may affect freshwater missels, such as connercial sand and gravel dredging, navigation maintenance dredging, fleeting, and/or mooring facilities which are located in the three previously mentioned reaches of the Ohio River. The extent of the informal consultation will be determined on a case-by-case basis but should include missel surveys by gualified consultants approved by the Service.

If you require additional information on this matter, please contact us.

Sincerely,

Charles J. R.D.

Pield Spervisor

Enclosure

Table 1. Check list of freshwater mussels collecte Navigation Pool, with specific reference containing <u>L. abrupta</u> , Mason and Cabell O West Virginia. Data from 1985 and 1987 o	Theck list of freshwater mussels collected in the upper Greenup Wavigation Pool, with specific reference to the mussel bed xontaining L. <u>abrupta</u> , Mason and Cabell Counties, West Virginia. Data from 1985 and 1987 combined.				
SCIENTIFIC NAME	COMMON NUME				
Strophitus undulatus undulatus (Say, 1987)	Squaw foot				
* <u>Magnonaias nervosa</u> (Raf, 1820)	Washboard				
Lasmigona complanata (Barnes, 1923)	White heelsplitter				
*Quadrula quadrula (Raf, 1820)	Mapleleaf				
"Quadrula metanevra (Raf, 1820)	Monkeyface				
*Quadrula pustulosa pustulosa (Lea, 1831)	Pinpleback				
*Amblema plicata plicata (Say, 1817	Three ridge				
*Pusconaia ebena (Iea, 1831)	Ebony shell				
*Fusconaia flava (Raf, 1820)	Pigtoe				
*Plethobasus cyphyus (Raf, 1820)	Bullhead				
*Pleirobena cordatum (Raf, 1820)	Chio River pigtoe				
*Elliptio crassidens crassidens (Lamarck, 1619)	Elephant ear				
*Obliquaria reflexa (Raf, 1820)	Threehorn				
*Actinonaias ligzmentina carinata (Barnes, 1823)	Hucket				
Tranilla truncata (Raf, 1820)	Deertoe				
Leptodea fragilis (Raf, 1620)	Presile pepershell				
*Potamilus alatus (Say, 1817)	Pink heelsplitter				
*Ierreilis radiata luteola (Ierarck, 1619)	Fat nucket				
*Liguria recta (Lanarck, 1819)	Black sandshell				
*Lerosilis abrupta (Say, 1831)	Pink mucket				
*Izmosilis ventricosa (Barnes, 1623)	Pocketbook				

*Represented in the mussel bed occupied by L. abrupts at the Greenbottom site.

APPENDIX III. Letter from Dean Shumway, FERC, to C. Clower, USFWS, dated June 27, 1988.

Mr. Christopher Clower Elkins Suboffice U.S. Fish and Wildlife Service P.O. Box 1278 Elkins, WV 26214 JUN 2 7 1388

Dear Mr. Clower:

Enclosed is the Draft Environmental Impact Statement (EIS) for Hydroelectric Development in the Upper Ohio River Basin. Docket No. EL85-19-114.

The EIS assesses the impacts associated with 24 proposed hydroelectric projects located at 19 sites in the upper basin, including the potential impacts of the proposed projects on the endangered pink mucket pearly mussel.

Page 3-19 of the EIS provides information on the distribution of the endangered mussel in the upper basin. This section of the DEIS also identifies suitable habitat for this species in the downstream areas of the proposed hydropower projects located at Muskingum River Lock and Dam (L&D) No. 3 (FERC No. 6998), Belleville LAD (FERC No. 6939), and the competing proposals at Gallipolis L&D (FERC Nos. 9042 and 10098).

Page 4-62 discusses the effect of the hydropower proposals on the endangered mussel. Page 5-21 indicates staff's recommended alternative, alternative 4. Under this alternative, there would be no hydropower development at Muskingum River L&D No. 3 and, therefore, no effect on this endangered species. Page 5-29 provides staff's recommendations for the protection of this species associated with hydropower development at Belleville L&D and Gallipolis L&D.

We would appreciate receiving your review comments within 30 days, regarding the endangered pink mucket pearly mussel.

Sincerely, Sincerely, Dean J./Shumway, Director Division

Director, Division of Project Review

Enclosure: Draft Environmental Impact Statement for Hydroelectric Development in the Upper Chio River Basin, Docket No. ELS5-19-114.

-43-

-44-

APPENDIX IV. USFWS Comments on the FERC DEIS.



United States Department of the Interior

OFFICE OF ENVIRONMENTAL PROJECT REVIEW WASHINGTON, D.C. 20240

ER 88/419

AUG 1 1998

TAKE .

PRIX IN LINERICL

:

Ms. Lois D. Cashell Acting Secretary Federal Energy Regulatory Commission 825 North Capitol Street/N.E. Washington, D.C. 20426

Dear Ms. Cashell:

- -

The Department of the Interior has completed its review of your draft environmental impact statement for Hydroelectric Development in the Upper Ohio River Besin, Ohio, Fennsylvania, and West Virginia. We have the following comments and recommendations on the document. Specific comments are enclosed as a separate document.

FISH AND WILDLIFE COORDINATION ACT AND SECTION 10()) ISSUES

It is our understanding that FERC intends to have the final environmental statement serve as a licensing document. In this context, FERC has stated (on page 4-87) that Departmental comments filed pursuant to the National Environmental Policy Act (NEPA) will be considered by FERC to be the most current recommendations filed pursuant to the Fish and Wildlife Coordination Act (FWCA). Departmental comments on a particular impact statement are provided pursuant to NEFA and do not represent comments pursuant to FWCA or Section 7 of the Endangered Species Act (ESA). Comments under Section 2(b) of FWCA are provided when draft applications are made available for each project. Comments under Section 7 of ESA are provided when FERC initiates the consultation process.

FERC also states on page 4-87 that its response to our comments on this environmental statement will be used as a mechanism to comply with Section 10(3) of the Federal Bower Act (FPA). In our opinion, a response to comments on a general environmental statement covering numerous projects does not constitute issue resolution under Section 10(1) since there is insufficient information to determine whether the staff's recommendetions are consistent with those of the FWS. Furthermore, this does not reflect the spirit of the FPA since FERC's action may limit future opportunities to resolve issues on individual Ucensing actions.

We believe that all comments made during the licensing process of any project should be part of the administrative record for that project and there should be dialogue between parties at any time if conflicts arise. If FERC intends to use these environmental documents as part of the Section 10(j) process, then 10(j) discussions should be presented in a senarate section that contains specific recommendations for each project, a detailed discussion of any disagreements with the FWS, and adequate documentation of the FERC staff's position.

Ms. Lois D. Cashell

CUMULATIVE IMPACT ANALYSIS

On June 29, 1987, the FWS commented on FERC's notice of intent to prepare an environmental impact statement on the potential cumulative impacts of hydroelectric projects in the Ohio River Basin. At that time, FWS concurred with FERC's decision to prepare such an impact statement; however, concern was expressed about which projects were to be included. Since that time FWS has maintained the position that FERC's proposal to examine the potential cumulative impacts of 24 projects (with pending prelicense action at 19 sites) does not and cannot adequately address the potential cumulative adverse impacts of hydropower development throughout the basin. All proposed, constructed, and reasonably potential (regardless of licensing status) hydroelectric projects should be included in the cumulative impact analysis. This is consistent with the definition of cumulative impacts contained in the FERC regulations implementing NEPA.

Under this definition "cumulative impact" is the effect on the environment that results from the incremental impact of the action when added to past, present, and future actions regardless of the agency or person undertaking such actions. In our opinion, the Ohio River cumulative impact study is not being conducted in accordance with this definition. FERC argues that "...increasing the size and scope of the study area (additional licensed or exempted projects) ...would introduce sufficient additional uncertainty to make the results of the impact statement of limited value..." (August 10, 1957, Scoping Document II, page 15). Contrary to that argument, we believe that the draft, as written, precludes meaningful analysis of potential or real adverse cumulative impacts on fish and wildlife resources in the Ohio River Basin from existing and proposed hydroelectric projects because it does not include such projects. Therefore, we believe that PERC should prepare and circulate for comment a revised draft statement including all licensed, exempted, and proposed projects in the Ohio River Basin. We have enclosed a list of additional projects which we believe must be included in any revised environmental document.

The draft statement does not indicate which competing application would be licensed at Tygert Dam, Montgomery Locks and Dam, New Cumperland Locks and Dam, Willow Island Locks and Dam, or Gallipolis Locks and Dam, although the proposed action is to license hydroelectric projects at these sites. The draft states that there are no major differences among these projects; therefore, no preference is indicated. In our opinion, major differences do exist. For example, one of the applicants for the Gallipolis project plans to install our turbines, two on each side of the river. The competing application only plans to install two turbines at one abutment of the dam. The effects of these projects on water currents and patterns would differ significantly, as would proposed fishing access and aquatic habitat requirements. Furthermore, recreational areas proposed at several sites on government-owned lands are of insufficient size to allow the development. FWS previously described major concerns with several of these projects during review of individual license applications. Unfortunately, FERC has not responded to those concerns. Therefore, the draft should be revised to show which competing application is likely to be licensed and what mitigation measures would be required.

ALTERNATIVES ANALYSIS

Conservation and load management is briefly discused as a principal non-generating alternative to the proposed projects. However, the concept was dismissed by the conclusion that implementation of such measures has been adopted in many cases. That assumption is contrary to the conclusions drawn and recommendations made in the Department of Energy's (DOE) study of alternatives to the Davis Power Project released in 1980. That study concluded that implementation of any of the three conservation and load management scenarios would result in significantly lower costs than building a proposed 1.000 MW pumped storage facility. The DOE report also showed that implementation of conservation and load management could, in end of itself, preclude the need for 1.000 MW of peaking power. As such, a detailed analysis of conservation and load management is appropriate for the present study and highly recommended by FWS. Further, the conservation and load management alternatives presently being used should be discussed in concert with all economically available alternatives.

-46--

The draft states that hydropower alternatives were developed to "...give equal consideration to power generation and environmental quality values." Contrary to this statement, the data projecting dissolved oxygen (DO) content (Section 4 and Appendix B) clearly show degradation of water quality values with each of the four proposed hydropower alternatives. Specifically, the DO model for Alternative 3 (Project Operation to Meet Antidegradation Criterion) and Alternative 4 (Project Selected to Minimize Impacts to All Target Resources) predicts DO reductions below Willow Island Locks and Dam for approximately 160 river miles (see Figures 4.4.1-1, 4.4.1-2, 4.3.1-3, etc.). FERC's recent decision (Commission Order 464) to waive Section 401 water quality certification for 11 projects does not reflect equal consideration of power generation and environmental quality.

ENDANGERED SPECIES

The data contained in the draft are insufficient to demonstrate that adverse impacts will not occur to the pink mucket pearly mussel (<u>Lampsilis orbiculate</u>), a listed endangered species. In fact, FWS believes adverse impacts to that species would occur. Our concerns are based on the potential depressed DO levels resulting from the proposed hydropower projects. Additionally, FWS believes a significant reduction in numbers of finfish, as indicated, may also adversely affect the pink mucket pearly mussel since fish are an integral part of their life cycle requirements.

The endangered pink mucket pearly mussel occurs in the area below Gellipplis Look and Dam and may also occur below Willow Island and Belleville Looks and Dams. Section 7(c) of ESA requires the Federal agency proposing a major construction activity significantly affecting the quality of the human environment to conclust and submit an assessment to determine the effects of the proposal on listed and proposed species. The biological assessment must be completed before any construction contracts are let. We do not believe that we can adequately assess the effects of proposed actions on the pink mucket pearly mussel without a complete assessment. When conducting a biological assessment, the following may be considered for inclusion:

1. The results of an on-site inspection of the area to be affected by the action to determine if listed or proposed species are present or boour seasonally.

Ms. Lois D. Cashell

- 2. The views of recognized experts on the species at issue.
- 3. A review of the literature and other information.

4. An analysis of the effects of the action on the species habitat, including consideration of the cumulative effects and the results of any related studies.

-5. An analysis of alternate actions considered by the Federal agency for the proposed action.

After FERC has completed and reviewed the assessment, it has the responsibility to determine if the proposed action "may affect" the pink mucket pearly mussel. If it determines that the project "may affect" a listed species, FERC must request formal consultation from the appropriate field office of the FWS. When FERC provides the biological assessment to the FWS, it should include any other relevant information used in reaching its conclusion.

Section 7(d) of ESA underscores the requirement that the Federal agency or the applicant shall not make any irreversible or irretrievable commitment of resources during the consultation period which, in effect, would deny the formulation or implementation of reasonable and prudent alternatives regarding their actions on any endangered or threatened species.

MINERAL RESOURCES

On page 3-27 we believe the document is deficient in its listing of mineral resources and resource-based industries. The list should include: coal, oil and gas, steel, ferroalloys, zinc, lime, dement, day, and sand and gravel. The document also states that impacts on mineral resources from this project consist of land impairment and utilization of mine guarry and pit sites for waste (spoil) disposal. Mineral deposits are apparently viewed only as convenient dump sites that have no other interest or value. Also, the document does not mention the network of natural gas and petroleum product pipelines that may be affected by the various proposed construction and distribution lines.

We recommend that the draft include a section on mineral resources. The mineral deposits and/or mineral-related facilities, including pipelines, pertinent to each site or facility indits ancillary components should be described, the impacts of the projects on them should be discussed, and mitigating measures should be described.

<u>SUAIMARY</u>

The draft does not provide adequate coverage of the following issues which were identified in the scoping process as issues of primary concern: water quality, endangered species, hydraulies, altered flow patterns, flow modifications, turbina-induced fish mortality, representional fishing, and dredged spoil disposal. The draft does not provide adequate recommendations or articles to protect fish and wildlife resources or the public use thereof. No indication is given as to which competing applications will be licensed. Conservation and load management was not adequately addressed as a project alternative.

4

Ms. Lois D. Cashell

FERC claims to have complied with the Section 7 consultation process of ESA by providing FWS with a review copy of the draft statement. This does not constitute consultation which has yet to be initiated by FERC. Further, the FWS does not concur with staff recommendations to postpone resolution of the previously identified issues regarding fish turbine mortality and fish passage until the post-licensing phase. Recause of these concerns, and the exclusion of numerous basin projects from the analysis, we recommend that the draft be revised and recirculated. A final statement should not be released until these issues are resolved.

Unless these issues are adequately resolved, the FWS may recommend that this project be referred to the Council on Environmental Quality (CEQ) under Section 1504 of its regulations. The referral would be based on the precedent-setting nature of this type of cumulative impact study and the severity of potential water quality impacts. However, we wish to coordinate fully at the earliest possible time because a solution to our concerns can be implemented with a minimum of delay and could preclude the necessity for referral. Coordination can be initiated by contacting the Field Supervisor, U. S. Fish and Wildlife Service, Suite 322, 315 South Allen Street, State College, Pennsylvania 16801, (814) 234-4090.

Sincerely,

Fruce Blanchard, Director

cc: Mr. Dean Shumway
(Rm. 204RB)

5

-49-

APPENDIX V. Summary of Expert Contacts

The following contacts have been made with USFWS staff and individuals in other appropriate governmental agencies and private organizations to obtain information used in this biological assessment. A statement of relevant expertise is included. Specific information communicated is cited in the text by personal communication.

Mr. Stephen Ahlstedt Biologist Office of Natural Resources Tennessee Valley Authority, Norris, TN (615) 632-1790 Author of USFWS Recovery Plan for *L. orbiculata* (= *L. abrupta*); freshwater musset surveys in the Tennessee Valley and Arkansas.

Mr. Ronald R. Cicerello Aquatic Biologist Kentucky Nature Preserve Commission 407 Broadway Frankfort, KY 40601 (502) 564-2886 Surveys of endangered mussels in the lower Ohio River.

Mr. John Jenkinson Office of Natural Resources Evans Building Tennessee Valley Authority Knoxville, TN 37902 (615) 632-3516 Mussel surveys at TVA facilities and in the Tennessee Valley region.

Mr. John H. Marshall Environmental Affairs Specialist Division of Wildlife, Onio Department of Natural Resources Fountain Square Columbus, OH 43224 (614) 265-6306 Ten years of experience working on the Ohio River, including mollusks; general environmental quality and impact assessment.

Dr. Andrew C. Miller Research Limnologist WESER-A Waterways Experiment Station U. S. Army Corps of Engineeers Vicksburg, MS 39180-0631 Freshwater mussel specialist; conducted surveys of mussel populations in the lower Ohio River for 10 years. Mr. Kenneth Multerer Endangered Species Coordinator Columbus Field Office U. S. Fish and Wildlife Service P. O. Box 3990 Columbus, OH 43216-5000 (614) 231-3416 USFWS coordinator of endangered species for Ohio.

Dr. David H. Stansbery Director, Museum of Zoology Department of Zoology Ohio State University 1813 N. High Street Columbus, OH 43210 (614) 292-8560 Leading freshwater mussel taxonomist; curator of mussel collections; author of 1985 review of status of *L. abrupta* with special reference to the Muskingum River.

Dr. Ralph W. Taylor Professor, Department of Biological Sciences Marshall University Huntington, WV 25701 (304) 696-2338 10 years experience conducting mussel surveys of the Upper Ohio River, especially between Greenup Locks and Dam and Pittsburgh.

Mr. William A. Tolin Biologíst, U. S. Fish and Wildlife Service P. O. Box 1278 Elkins, WV 26241-1278 (304) 638-6588 Twelve years experience conducting mussel and other aquatic habitat surveys of the Ohio River; discoverer of *L. abrupia* in the reach below Gallipolis Looks and Dam in 1987.

Dr. John C. Williams Professor of Biblogical Sciences (retired) Eastern Kentucky University home phone (606) 299-4072 Mussel surveys of Chio River

APPENDIX J

LETTERS OF COMMENT ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT AND STAFF RESPONSES

The Notice of Availability of the draft environmental impact statement (DEIS) was publishe in the Federal Register on May 20, 1988. The DEIS was mailed to Federal, state, and local agencies and individuals for comments on May 12, 1988. Chapter 8 of the final environmental impact statement (FEIS) contains a listing of those agencies and individuals that were sent copies of both the DEIS and FEIS. A public meeting on the DEIS was conducted on July 15, 1988 i Pittsburgh, Pennsylvania, to allow participants to express their views on the DEIS.

All timely letters of comment that address specific analyses in the DEIS were reviewed by FERC staff. Suggestions for correcting text or data and requests for further discussion of a subject have been considered. Those editorial changes and suggestions which were practicable, reasonable, and which improved the quality of the EIS are incorporated herein.

Constructive criticism presenting a major environmental point of view or one in opposition to staff, when persuasively supported, is treated by making revisions in the appropriate part of the FEIS. When the major point of view is not persuasive, reasons are given why the staff did not change its point of view.

The sections or pages of the FEIS that have been modified as a result of comments received are identified in the staff responses to the right of the letters of comments. Other FERC staff responses are self-explanatory.

A "no response required" response is given to comments that are statements that raised no questions concerning treatment of subject matter in the DEIS. A "your opinion has been noted" o "comment noted" response is given to comments that are considered to be statements or opinions.

Letters were received form the U.S. Soil Conservation Service, U.S. Public Health Service, and the Ohio State Clearing House which acknowledged receipt of the DEIS but offered no comments

The respondents and the page on which their letters occur are as follows:

3
-
0
-
5
1
â
í
È.
õ
7 8
บ า
1
102
10
11
13
29
44
54
57
73
75
78
34

.

.

Sithe Energies U.S.A., Inc., re: Allegheny 8	
and 9 Hydro Partners.	J-190
Mitex, Inc., re: Muskingum L&D No. 3	J-192
National Renewable Resources, Inc.,	
re: Maxwell L&D	J-200
National Renewable Resources, Inc.,	
re: Monongahela L&D	J-202
National Renewable Resources, Inc.,	
re: Tygart Dam (7307)	J-204
Mitex, Inc., re: Allegheny L&D Nos. 5 & 6	J-206
Green International Affiliates, Inc.,	
re: Montgomery L&D. (3490)	J-209

	· · · · · · · · · · · · · · · · · · ·	lu rio ryski chto dator Bogi - Liki srajos fullona Midel No sz intisci s Hideba cerioszt ne Hideba cerioszt ne Hideba cerioszt sora Liz A, And Hideb rajozi sora trially.	aana kuntut) araan ututlangeest71. 11.	L altreit A. S. Aray
на на селото на селот на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото н на селото на селото на селото	74, 169 B. (Merall, 2010) Beenergte Seteral Tomory Bernskart, 2010 125 Berth Depter Prior, 2011 2000, 61 - 2015	D. B. Ma, Castall. D. B. Car Ch. Sydrawherter have loganet Unrian L. Ren Frankewell KK. Lechnicol accurate An entry Princy representing the correct principal Cart's C. J. 1968. The environment of the Carter C. P. 1968. The environment of the Carter C. P. 1968. The environment of the Carter C. Carter P. Solar endlaced to the Cartanet P. Cart. Endlaced to the control of a licentage of share. Sater Endlaced to the A. Starteg of share. Sater Endlaced to the control of licentage of share.	li se y he ef fyther assirtant. Ps. dura Seyler of y staft, as may te y sumstal	

.

End Repaired

89 JUL 13 AH **9:44**

.

÷

·

1

2

- 4

5

d the definition of a subsection of a

A CARL CONTRACTOR OF A CARL CO

2. Structure interpretential of rights in the end of a structure in the property interpretent of the structure interpretent of the property interpretent of the structure interpretent

for the parameters 2.2.2.3. # 1.4. and others, it is to involve that the zening forten in lanke ukerse tim marstriftigen uf utterantrigen immediately an and don treasfrom as reased but for with the sea be detunctioned doins a the physical askilling program in log comm empiremental country how were astern within the APPL were not welchicles, etc., while residence - and newighting intervals - fr is this described that the floor transdown only restaund to Alterial we Flore be conditioned to that is although we measured and dependenced to be possible, then additional musiderstion with he grown as them. inclusion in the system. At AR Dat #7, for miled of river will be physically values to the condition approxematic sectors of the sector of propared by the Correr of Side propagate. The sense relates the dama of the he seamentd for the motor comp History to mue and the twotaker thlands in their entrystrate south and seem would extend 0.4 with the to believed that many mittentive meaning, would be examined In a model this Carnel.

Recognizing that fills is primarily an environmental report, there 3 may be derify in our instances, to consider the economic and technical function making.

4. The Division is familiar with publicab factor in the acopying of this study. It is call that at thest one table endors the browing all main size and trustancy, existing and prend of, bydra projects in the upper this there there exists a the interval of the entory of the entory of the entory of the entory of the presented of the expect to determine the table entory.

5. In view of the lock of experience with concepts as and behavioral barriers to fick encodiment at lead-hand hydro facilities of shift viewes, we contar in the recommendation of FAPO stational page Substitute device the recommendation of FAPO stational page association with one of more of the hydro probably the light page River System. This needspace funded facility monact super the state providing two protections to be lead to find the space the state structure of the needspace funded facility monact super the state providing two protections to be lead to find the state to state any two protections to be lead to the state of a providing the facility should also take the state to replace receive all of the distribute opport that so has the set of the replace flow. We be there that the cardinate of the hydro new legit to the the so a impact alternation of AFT employed and the replace of the state of the state that the cardinate of the hydro new legit to the the so a

Carlenger 1

1. Text has been modified.

- 2. Staff is aware of the Corps' requirement for licensees to conduct physical modeling studies at Corps dam structures prior to construction of the hydropower project. The primary purpose of the modeling is to determine the effects of hydropower operation on river flows and to design the hydropower project to eliminate any effect on river mavigation. The modeling study requires final detailed design plans for the hydropower facility that are not available until after a license is issued and does not provide the information to staff needed to assess the environmental impacts prior to licensing. Staff has recommended in the fEIS (Section 5.4.1) that the three projects for which inadequate mitigation has been proposed to date be reconsidered should adequate proposals for mitigation be proposed in the future.
- 3. Staff has analyzed the economic benefits of each of the proposed projects and has summarized this information in Section 5.2. The Commission has long recognized the need to study the economics of proposed projects. A project that is not economically beneficial is not in the public interest, unless the applicant or the record demonstrates overriding public interest or benefits of licensing the project. Technical feasibility of proposed projects and mitigation measures have been considered to the extent possible without conducting detailed engineering studies.
- 4. Two new maps have been provided in the FEIS (Figures 1.1-1 and 1.3-1).

5. Staff appreciates the support of the Corps in its recommendation of a bioengineering test facility for evaluating fish protection and guidance systems in the Dhio river. Staff concurs that the commitment must remain open-ended to be successful at meeting unforeseen opportunities for mitlegation. ې. 4 5. It would as preter the to reveally the operating transmost point to instantiation is continuing a large and the problem bids, tage 2463; to down, Astrone 7 is GBCD, page 6 to enater than the large, prove to the trans of these place <u>limitize</u>, such entry that is operating.

(1) It is some over elect the first statement of there for para, posed of the follower ("expendix to substate recent entropy for the statement of the statem

C. It is exampled that the first websees if the fideth pate, pape Arbs, and 4.1.7, this is replated as follows: "The reduction be test exception of involvent data absolute the river's velocity and extreme transport detracks (Section 4.1.5)."

2. It be congressed that an eventsh part scrapped to see to show $[\,\,g\,$ on the title paper in According C.

10. In interface bound and in Compage from the FFTS, it is not subject for despite participation on the basin whose value quality management of the one of the fed.

- 11. In control to be numberous statements in the DETS connecting the period (a) for roll drawiows taking fixed-reset does due to hydro period for (and should charly that the Corre will require that a bindow of which drawing that the corre will require that dation all periods is hydro contained over these does
- 12. If is concerned that a Table Le Added to the DFTS to menter [12 illustrate the flows that are discussed in Section 5.4.4.

1. FFR staff would add a statement early in the tent of the DEFF 13 which income the public fand othern) that the authorized purprise at the lock and dur throutines in revigation, and that hydro plast opposition income this prepare institute.

14. The dypets these systems described under Recommutation 4 an paper 14 5:73 PTLL, would be, be user-specty feet use file would pape user the day set faxed or st structured and unter the dates fat ested day of Machaner 1 words the bythe pilots are not usersting.

- 6. As required by article 305 of Appendix H.2 of the FEIS, each licensee must enter into an operating agreement with the Corps for development of hydroelectric facilities at a federal dam. In accordance with the MOU (1981) between FERC and the Corps, the operating agreement must be made prior to start of power plant operation.
- Text has been modified to indicate that aerial photographs of all L&Ds are provided in Appendix C.
- 8. Text has been modified.
- The footnote indicating no aerial photograph is available for Tygart. Dam has been deleted.
- 10. The purpose of recommendation 7 is to allow increased power generation and to improve water quality management, by implementing system-wide water quality and flow monitoring and coordinated operation of the hydropower projects. The project owners would benefit financially because spill flows could be reduced below those recommended by staff at some times. The Corps and other agencies involved in water quality management would benefit from the improved monitoring and control of the system. Funding for this system should be shared by each of the beneficiaries. However, since hydropower project owners could obtain substantial economic benefits from the relatively low additional cost of maintaining a system-wide monitoring program, project developers should be willing to contribute to such a program. For example, a program could be designed that would use system-wide monitoring and modeling to determine instantaneous spill flow requirements for any projects that have participated in financing the program; projects that have not participated in the program would still be subject to the spill flow requirements recommended by staff.

ے

ΰn.

- The spill flows recommended by the staff for projects at the fixedcrest dams will guarantee that a minimum depth will be continuously passed over these structures.
- 12. The spill flows recommended in Section 5.4.3 are summarized in Table 2.3.1.1; text has been modified to reference this table.
- 13. A sentence mentioning the authorized navigational purpose of the locks and dams has been added to Section 1.1. FERC has a responsibility to balance the addition of hydropower against other project purposes.
- 14. Flows passing over the dam or under the gates may not be sufficiently close to the shoreline access points where there are developed fishing access facilities and, therefore, may not be comparable to the tailrace currents which normally would attract fish to these areas. Recommendation 4 of Section 5.4.2.3 has been revised to address the site-specific issues which would need to be resolved during physical hydraulic modeling. Developers would need to file a plan with the Commission for providing flows in the tailrace fishing areas when the turbines are inoperative (e.g., via selective gate openings and/or bypass flow systems) based on the results of hydraulic modeling and consultation with appropriate state and federal agencies.

THE CHEVER SYSTEME THESE PROTOBLESS IN STREET

-

PAGEPARACHIE	REMARKS	-		
1.12.11.11.11.11 1.12.11	Twent Lake is a multiplicities provid autorized Cardona casher and two fow anyon Parison The Information should be income a			Text has been modified.
1-6-1.3	Raisling water levels, in os rend na ditter thoogs an Noort, especials, intendid te included in the lasses.	15	15.	Text has been modified, referencing flood elevations in No. 1 in the list of issues addressed in the FELS.
2.5-2.5.1.1.0	Change over to all			
2.523122	Charsen (mar 10 %)			
2 - 3	Span is we should be further former in that they are subard to charge to Coros and possibly by other conversional agencies.	16	16.	Text has been modified in Section 2.1.3 and 5.4.3 to reference the recommendation that spill flows be subject to temporary modification for water quality management.
2-5/2 1.1.3.39	Belownoe to contribute and 130 met coll section to Fonder copples	17	17.	A July 26, 1988, letter from the Corps' Pittsburgh District concerning Allegheny River L&D No. 3(FERC No. 4474-003) indicated that the Corps would consider the installation of facilities to control pool levels
1 birgal 15	Gublet dibreud ett appli mette veneret för Garps för provere readet specificaneet			during periods of low flow, such as crest gates that are fully controllable. Staff has analyzed the proposed project at this site to include this consideration in the FEIS.
2-1-2-1-1-5/2	Change 704 151 4 to 1514 704 - The tolater pressared an official initial point, which it isn't since taken Filler ands at RM 129.72			
2 18-2,1,t,E 4	Remove most and add atter isobreards d, the phrane, lioxcupt during very low flows "			
2(10/2)1.1 1912	R(modelft).			
R - 1 6	Some of Bay larger flows shown in the carboan column are ret now always achieved	18	18.	Comment noted. When the river flow minus the minimum turbine flow required for generation is less than the required spill flow, the projects will be unable to operate.
1. C 772 († 14	Opliane will have to be increased it minister distills - over fixed most spillways tally below 0.5 th after powentouse received.com	19	19.	See response to comment #11.
された(2014年4月) 1989年	If would be hotter to substitute "deeber" for "mhasasized". Sonur areas occur bidory reporting the shanballited areas, the latter of which using show up a winder-an at the eads of the factor.			Correction made.
5 34-49 3 1 3 3	Contions of the and tig. 3 1 9 2° should the inter- drawed.			Corrections made.
	4			

РАСЕ РАНА НИЕ	HEMARKS			
9-97 S 2 H 88 H	Particular and a particular and a sample ray of the particular form	20	20.	Text has been modified.
9 90 M	and the second of the second	ţ		
3×30×11,3 - 1	De Reid Hernere (Ling Hottlich), addressing i de Schwerker (Leeke Priedlich), der Schwerker (Leepdelinders), de pdelinders (Litt) ade zichtig (D. A. 200	21	21.	See response to comment #)1.
②→4代 4 3.2	List sent ware cell totally true. Payon model: stady of starup potent in ter tas dans on poly them energy and terbulent lows planking longet most forw ranges except at very high stopes when a nergy minimulated. Turbulence was so service that rail rack task south not be should. As a result, success promise the stope and screateds subservice to bornes tilled with stope and screateds subservice.	22	22.	Text has been medified to state that aeration at Allegheny L&D 7 is probably low because the apron, which is longer than at other Allegheny River dams, allows less plunge than at other dams.
3+60-3, * 6;;	Provise the decond semi-menus in the Unit Crace Consignion processes real visities exemptions when conclused and a place possibles made price by construction to componiate for project-induced flood effects and, for exclude hydropowers where recessory.	73	23.	Text has been modified.
a-s't	Califier figure, show or dilet of in the legend box, a flaw and dot configuration tabeled, 1900-year figed with Hydro			Correction made.
2-56,2513-4-4-3	Tables 3.4.4.1 Incugh 3.4.4.4. It is assumed they refer to Michophola L(D, 2.3.4 and 7 rother that like numbers on the Alleutieny - Also, the projects in these holdes are not orranged in any Club, 15th the homas of lable 3.4.4.6 for an all of the descention and forms of these projects.			Correction made.
3-521 e 4 871	Applied third sentence wy mit pated in prote 3(36 above			Correction made.
3 63	For 3.4.67 fills should be "Flood Product Havabors of mid- at of "Expected 1 port Elevation"			
3+67/1-2	Aller island should build ford from the version must 4 dischargers a menonum et 60, betav stor, ich delweidtmene proj	24	24.	lext has been modified to delete reference to the dam designs. However, Pike Island is retained in the list of dams that are effective aerators. A statistical analysis of dam aeration data
3 67 7 5 9 9 1	Common on " should be planat.			measured by the Fittsburgh Uistrict indicates that flow has a relatively low influence on acration at this dam. Some observations taken at low flows show little acration, but significant acration was measured during other low flow periods.

19 2010 B

.

220+954 0 - 88 - 18

.

J-7

THE REPORT OF THE SECTION OF THE SEC

FAGETARA L'H	BEMAPUS			
3 78-0 5.5 A+8	Prindmant will and provide of Alfree writer simply writer of the Child Facult where write private the types using forsignifies of detropation Programs where the Alfree to the United Factors Provide Child Action of the to every manufacture of Alfree to the Child Action of the every manufacture of Alfree to the Child Action of the every manufacture of Alfree to the Child Action of the every manufacture of Alfree to the Child Action of the every the Provide Wild Alfree to the Sector of the Neurophic Provide Wild Alfree to the Alfree to the Child Capping and Alfree to Child Action of Alfree Child Action of the Sector of Child Action of the Provide The Alfree to the Alfree to the Child Capping and Alfree to Child Action of the Alfree to the Child Capping and Alfree to Child Action of the Alfree to the Alfree Child Action of the Alfree to the Alfree to the Alfree to the Alfree Child Action of the Alfree to the Alfree to the Alfree to the Alfree Child Action of the Alfree to the Alfree to the Alfree to the Alfree Child Action of the Alfree to the Alfree to the Alfree to the Alfree to the Child Action of the Alfree to the Alfree to the Alfree to the Alfree to the Alfree Child Action of the Alfree to the Al	25	25.	Comments noted; text has been modified; see Section 3.5.5.
(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Frazia, periotecien a net of Rice Periodappi primi. Junet can Partific end developed and angliange a wide neuro of this become planent provide of book Galiber from pred wide to	26	Z6.	Comments noted.
9.19 N 0.8 M	Second to only or double of the double page 3.50 and a			Correction made.
2.20	Fey (3.5 th 4). Which we fille yes give protect entry on physical 5 th			Correction made.
4 2	Capite 4.3 Met — Louis e el Call Mérica andrés de graves Mente la el antra bet			Correction made.
4-00 * f	Andd - 11 or 75 rps, end and in that religion at 8, ma den in orders beits occaveled and 1, multiwear to do safy presented in tadwater from segment.			Correction made.
4.22333233	At bottom of prigst, thesis after "Specific sinis", the private "Hinos abried to comparative with previous monter study :"			Correction made,
al - €.2; - t	White Men Hiver Leo 4 while projinally density ded as a fixed-creat rism, it was reported or the track to the 1960 p to provide a path dicrest	27	27.	Camment noted.
2 (44) X (4) (4)	"Ro parage-polynemic ter relevaniese, it's Basis- tocods are toolo presently and/of-of-model evenis	28	28.	See résponse to comment #17.
4.573353	Refer paragraph, and go 'Meda' to 'Srine' - In 5th paragraph, the Dom nembers should be reversed: Lin, 20 Je Dam 45 and 18 Je Dain #0			Correction made.
4 (3.92	Consistent: The plan maryleased to see recommendations to interm- planing any insurgament operating of water filled bag was easing platitia a lease begits notilis?" (bid owners)			Comment noted.

с. as

- PARA AND A SAME OF A SEA OF A CARD AND

.

FASE PARA ERIE	REMARKS			
● 15 位はい!	Sabpara mariel El This innt lati fyglan i struaist be Processed for allts processional that the Corps wai princtname forwills around priors to possategy or a Pretery III - Simikhows as (Schoolderd Richerpo) Prathonopriori 4. Bretarens in made at the Corps of Postpropriori 4. Bretarens in made at the Corps of Postpropriori 4. Bretarens in made at the Corps of	29	29.	The third basin-wide water quality recommendation in Section 5.4.2.1 has been modified to refer to Section 5.4.3 for the issue of flows at lygart Dam. Section 5.4.3 now includes recommendations for walleye flushing flows and for flow regulation in accordance with Corps prohibitions on peaking flows.
N 1 ← A II 1 ← F	Praidic to the statilities may one. To a survey interpret process that the respect of this CEIS is to the full of more to beneficial to serve that the rations root fitter to serve that the rations root fitter to the server knows of AG fitter to the survey survey of expectation were discover at full to the server to serve the server discover at full to the server to the server of server at full to the server to the server of server at full to the server of server to the server of server at full to the server of server of server at full to the server of server of server at full to the server of server of server of server at full to the server of se			Esaments noted.

DHIO HIMEN SLACEWATER ENDAMMENTS CHEATER IN THE FITTERINGH INCLOSE SINCE 1969

Employments over 500 feet is length that ere leaded is reaches that have been affected by the Ohio River mederization program since 1950 med interd in Appendix A. There is a total of 43 to hier expects within the 50 Trile long reach of the Ohio River in question. If of which are in Hermitel Peel The total length of the shoreward employments in Quenthel Peel is note than 15 miles. Must of the Ohio River entry marks within the curic distribution of the Pitteburgh District are relatively marks. There are, however, research exceptions where the embloyments within the mentils of his tributary streams. The East notable embloyments in this respect on listed below.

Approximetr Meximum <u>Width in Feet</u>							
1,500							
3,400	(35 hores)						
702							
64 C							
F:02							
5.30							
4.00							
	Approximate Me <u>Midth in</u> Fo 1,500 1,400 700 601 601 601 601 601 601 601 601 601 6						

30

30. Comments noted; text has been modified; see Section 3.5.5.

.

Two pages of typographical error corrections also provided by the Corps are not reproduced here.

APPENDIX A

Slack-afer Fulkaments (ner 500 feel Loup an the Olio River Frim River Mile 7.17 to 1951

Loration Ohio Rive Mile	Note of Tributery Streng.jggled	Emiff bind Stackwater Fiolayment (McTes)
	New Cumberland Peol at Normal Lool Elevation Eld.5 m	el.
29.5	Little Benves River	0.8
40.1	Mill Cruck	e. 2
48.6	Course anglueme Roma	t: 1
47.)	Little Tellow Creek	U. 4
47.6	Conge Bun	0.2
50.5	Yellow Creek	$\mathbb{C} \times \mathbb{1}$
52.8	Tontinson fim	0.,8
	Fike Island Fool at Normal Fool Elevation 644.0 ms	!
60.1	King: Creak	0.3
62.3	To Land Creek	Q.2
ES.7	Harmon Creck	0.3
72.4	Indino Cross Creth	1.0
72.4	Virginia Cross Creek	D.7
75 3	Buffalo Creek	1.2
77.1	Sult Ban	0.2
78.0	Beech Bottop Hup	10. I
75.0	Rush Bau	0.2
K2.6	Virginia Short Creek	C.4
82.7	Indian Buort Creek). c
	Hannibal Fool at Narral Pool Elevation 623.0 ±sl	
90.8	Wheeling Creek (MV)	1.1
94.7	Mr.Mahon Cteek	0.9
981.7	Wegen Ctnex	0.2
101.4	Little Grave Creek	0.5
102.4	Grave Creek	1.0
104.1	Big Run	0.2
105.0	Pype Creek	£ 3
107.7	Fisch Rath	(, ł
108.1	Little Coption Creek	\mathbf{c} , \mathbf{c}
109.6	Ceptine Crenk	2.4
110.8	Fish Creek	3.3
	(long Hur Sul-subsymmett)	0.5

.

-

.



United States Department of the Interior

2013 LISH AND WITDLIFE SERVICE.

Suite 322 310 South Allen Street State College, Brunylvania 16661

July 27, 1988

M. R.D.S. D. Controll, Actaux Decretary Neural Decrysterial activity (Longinger B.S. Nerth Capital Street, MR Neural Decryster, 1997) Automatics, U.S. (2014)

Deer PS. Cashell:

. 58 AUG -

The Fish and Wildlife Service Derived has reviewed the Traft Invirce-State Internet (15:15) for Hydrelectric Levelopment in the Open Chicgiver Solid. Chic, Foresylvania and Next Vitylifa as represented. These comments are submitted in autordance with the Fish and Wildlife Condination Art (16:0.5.C. 662 et style) to supplement concerns abrevely submitted to the Prioral Pringly Resultatory Commission (Commission) for each license application evaluated in the DEES (Table 1.1-1.). These comments do not preclude separate evaluations and Commission for pendits under Sections 10, 402, or 404 of the Chan Nater Art.

Cur review of the DEIS leads on the conclude that the Service is not able to SUPART many of the analyses and conclusions contained in the document. The Service requests that the DEIS be remased and revicedlated and suggests changes to protect and develop fish and wildlife resources in the basin.

General Comments

The Commission intonly to have the final Environmental Lepact Statement (FIIS) 31 serve at a Licensing document that will be part of the record from which the Commission will make forure licensing decisions in the chic Kivet Lasin. In this context, the Commission has stated that compents received on the DEES will be usualdered by the Commission to be the most current set of recommendations filled pursuant to the Fish and Wildlife Coordination Act during the onceing proceedings of 24 license applications, and that the Commission's response in the FEIS to comments will constitute issue resolution pursuant to Section 16(j) of the Federal Power Act. In our opinion, this does not reflect the spinit of the Federal Power Act, as mended by the Electric Consumers Protection Act, since the action would limit futures opportunities to resolve inspect inducing these made during attained at all comments in the during the Licensing process of any propert should be cort of the administrative record for that propert and that there should be dialogue backers parties if conflicts arise,

31. Staff considered all previous comments in the preparation of the DEIS on these project proposals, including comments contained in the individual project applications, filed during the comment period noticed for each application, and submitted during the EIS scoping process. State and federal Fish and wildlife agencies' comments provided for the DEIS may be considered the most current set of comments and recommendations since these most recent comments may reflect new recommendations for the projects included in the FEIS have been developed after considering all comments (agency, applicant, and individual) filed to date.

Comments and recommendations provided in the DEIS have been used by staff in developing and revising its recommendations as needed for the FEIS. The FEIS, including the responses to comments, indicates documentation of staff's position and evidence of disagreements hetween staff's position and appropriate fish and wildlife agencies' positions on resource protection. On June 29, 1987, the Service connected on the Commission's notice of interato propose an unvironmental impact statement (FUS) on the protectual complative impacts of hydroelectric protects in the Chick Parer Sasin. At that the, the Service concurred with the Commission's decision to prepare such an ELS, but expressed concern about which projects were to be included. The Service's position remains that the Commission's proposal to commine the potential complative impacts of 24 projects (with persing pro-licensing action at 19 sites) connot adequately address the potential complative adverse impacts of hydropower development throughout the basin. Moreover, the Service maintains that all licensed and operating, licensed and unconstructed, exempted, proposed, and reasonably potential hydroelectric projects should be included in the cumulative impact onalyris.

- The Condition argues that "increasing the size and score of the study area" (additional licensed and/or exempted projects) "would introduce sufficient additional uncertainty to make the results of the EIS of limited value," (August 10, 1997, Scoping Excurrent II, page 15). Contrary to that argument, the Service relieves that the USIS, as written, precludes meaningful analysis of potential or real adverse complative inpacts on fish and wildlife resources in the Orio River Basin from existing and proposed hydroelectric projects (40 CPR 1502.9). Therefore, the formion requests that the Condission prepare and circulate a revised DEIS that includes all bicensed, exempted and proposed hydroelectric projects in the Chio Piver Fusin. A list of other projects that should be included in the revised DEIS analysis is enclosed.
- The DEIS does not indicate which competing application would be licensed at Tygart Dam, Montgomery Locks and Dam, New Conterlard Locks and Dam, Willow Island Locks and Dam, or Gallipolis Locks and Hum, although the proposed action is to license hydroelectric projects at these sites. The EMIS states there are no major differences between these projects and no preference is indicated. It our opinion, major differences exist. For example, one of the applicants for the Gallipolis project plans to install four turbines, two on each side of the river, while the competing application proposes only two turbines at the abutment and of the data. The offects of these projects on water currents and patterns would differ significantly, as would proposed fishermen access and aquatic halitat requirements. Also, recreational areas proposed at several other situs on government-owned lands are of insufficient rize to allow development. The feartheat of the Interior (Repartment) previously described puter concerns with neveral of these projects during review of individual license applications. The commission has not responded to those concerns. Therefore, the DDIF should be revised to show which competing application will be licensed and what mitigating measures would be required.

Conservation and load management is triefly discussed as a principal nongenerating alternative to the proposed projects. However, the concept was summarily dismissed by the conclusion that implementation of such measures has here adopted in many cases. That accumption is contrary to the conclusions drawn and recommunications made in the Department of Energy's study of alternatives to the Davis Fower Project released in 1980. That study concluded that inglementation only of the three conservation and load rangement scenarios would remark in significantly lower costs than building a proposed 1,000 MW purped storage facility. The DEF seport also showed that implementation is construction and load rangement could, in and of itself, preclude the need for 1,000 MW science of peaking rever. A detailed analysis of

- 32. Numerous comments on the geographical scope of the staff's study have been received during preparation of the EIS recommending the inclusion and exclusion of hydropower projects different from the 24 hydropower projects analyzed by staff. Staff maintains that the study area contained in the FEIS encompasses the most concentrated stretch of pending and licensed hydropower projects in the Ohio River Basin. In addition to the 19 dams where hydropower projects are pending in the upper basin, 5 dams with no pending license applications and 6 dams where hydropower licenses have been issued by the Commission have also been evaluated in the water quality modeling portion of the fEiS. Staff believes that the licensed projects within the study area represent the existing conditions in the basin, which then can be compared to the impacts contributed by the proposed projects. Preliminary hydropower permits are not viewed collectively as reasonably foreseeable actions. Preliminary permit applications do not contain sufficient information to allow study of the environmental impacts of the development, do not contain sufficient agency consultations, and do not address the feasibility of the proposed development. Because of their incomplete status, preliminary permits cannot collectively be compared to pending license applications. The staff contends that the scope of the FEIS is therefore to address the cumulative and site-specific environmental impacts of licensing up to 19 proposed hydropower projects and to provide recommendations on those 19 proposed hydropower sites.
- 33. See response to comment #32. Although there are other existing and proposed hydropower projects in the basin, staff does not believe that any other projects with accepted applications would have cumulative impacts in the study area. The DD model indicates that, because the dams between Ohio River miles 100 and 200 du not provide much aeration without hydropower projects, significant changes in DD due to the proposed projects do not extend down the Ohio beyond the study reach. Impacts of licensed projects were included in the analysis and, in fact, the need to reconsider spill requirements at some licensed projects in the basin, the impacts of such projects will be assessed in relation to any other projects, proposed or existing, with which there would be cumulative or interactive impacts.
- 34. Staff has compared the competing hydropower applications at Tygart Dam, and at New Cumberland, Willow Island, and Gallipolis tocks and Dams. Staff has concluded no significant differences occur between the competing applications, either in environmental acceptability or in power generating capabilities. Staff believes the implementation of its recommended mitigative measures at the competing sites renders projects at these sites to have similar environmental effects. However, at the competing Montgomery L&D site neither of the competing proposals provide adequate mitigation to offset significant adverse environmental impacts due to hydropower development at this site.

In the event that sufficient lands are not available for construction of the recreational developments, a recreational compensation plan would need to be filed with the Commission. The plan would be developed in consultation with appropriate state and federal agencies.

If it is determined that the plans of competing applicants are equally well adapted to develop, conserve, and utilize the water resources of the region in the public interest, the Commission may issue a license for the applicant with the earliest applicant acceptance filing date [Commission Regulation 4.37(b)[2]]. However, the Commission may issue a license for the applicant with the more recent filing date if it is a municipality or state [Commission Regulation 4.37(b)[3]]. The Commission will make the final licensing decisions on the projects. Including the competing projects, in the public's interest.

.

34

-35
emperation and had meangement is appropriate for the present start and highly recommended by the Strive. Further, the conservation and load meanement electrotives presently being used should be discussed in opport with all economically available alternatives.

The DEDS states that bydropower siternatives were developed to "give equal consideration to power generation and environmental quality values". Contrary to this statement, the data projecting dissolved mayner (b) content (Section 4 and Appendix B) clearly slow demadation of white quality values with each of the four proposed bydropower alternatives. Specifically, the to model for Alternative 3 (Broject Quantum to Next Antoniegnalation Chiterien) and Alternative 4 (Broject Quantum to Next Antoniegnalation Chiterien) and Alternative 4 (Broject Selected to Nonimize Imparts to Al Donet Messagere) predicts like Figures 4.4.1-1.4.4.1-2.4.3.1-3, etc.7. The Compassion's recent decision (Complex) (bed were serviced at the Section 40 Kater Quality Certification for eleven projects appears to indicate less consideration for environmental quality than hydropower (see lopment.

The DEIS does not support the Commission staff's decision to exclude certain 37 projects from the completive inner assessment. Yet the Compission staff states in Section 3 of the DELF that including certain projects would increase the complexity of the analysis, without contributing to an understanding of the standative impacts. However, the document also states that hydroxewer projects in the lasin warrant an analysis of specific and cumulative environmental imputs and that " ... the value of the objective function at any portionlar site is a function of the values of the derision variables at all sites upstream (emphasis added)." And, "A cumulative system-wide temphasis added) modeling analysis ... is required.". The DELY also indicates that energy production and spillage required at a site are dependent on energy produced at all upstream siles (Amendix B, page 27) and that the studies indicate it may be beneficial to re-evaluate spillage at projects previously licensed on the Allenheny River (page 5 26). In view of the above, the Service cannot agree with the decision to exclude manarous hasin-wide projects, and recommends the analysis be revised to include all projects which will have untential adverse curulative effects.

The Service believes that the data contained in the DEUS are insufficient to demonstrate that adverse impacts will not court to the puck muchet (<u>Langalis</u> <u>abrupta</u>), a listed endangered species. In fact, the Service helieves adverse impacts to that species could occur from pretential depressed DD levels resulting from the proposed hydropower projects. Advitionally, the Service believes a significant reduction in numbers of finfish, an indicated, may also adversely affect the pick mucket since fish are an integral part of their life cycle requirements.

The endangered pink mucket occurs in the area below Gallipulis Lock and Dam and may also occur below Willow Island and Belleville Locks and Dams. Section 7(c) of the Endangered Species Act of 1973, as anonded, requires the Federal agency proposing a major construction activity significantly affecting the quality of the herear environment to conduct and submit an assessment to determine the effects of the proposal on listed and proposed species. The biological assessment should be completed within 160 days after the date of 35. The potential benefits of conservation and load management are well understood by the electric utilities distributing power to end-use customers and the utilities have been making a serious effort to educate their end-use customers concerning available cost-effective conservation practices. The historic and projected effects of conservation and load management on peak capacity demands and annual net energy requirements are quantified and submitted to the concerned Reliability Council by each utility in a Reliability Council area. Each Reliability Council totals these projections for its Council area and reports these totals annually in the Department of Energy Code IE-411 Report (Report). The Report covers a 10-year planning period and presents historic data for the first year and projected for the nine future years. The Report does not give these data for the individual reporting utilities but gives the totals for the entire Council area.

Load management affects peak capacity demand, but has little effect on annual net energy requirements. Conservation affects both peak demand and annual net energy requirements. As a result, the if-All Report gives the effect of load management on peak demand as a separate item. Since conservation affects the peak demand, this effect is accounted for in the IE-All Report in the reported values for "internal demand" -which are reduced by conservation. Conservation also reduces annual net energy requirements and this reduction is subtracted from gross requirements to obtain the net requirements shown in the Report.

36. The dissolved oxygen model projections show some decreases in DO under each of the four hydropower alternatives; it is impossible to have any generation without some decrease in DO at dams that agrate because of the lost dam aeration. As stated in Sections 2.1.3 and 2.1.4, the DO objectives of Alternative 3 and Alternative 4 [the recommended alternative) are to allow hydropower generation without allowing decreases in DO to levels (less than 6.5 mg/L) that would affect aquatic life. Staff believes this objective gives full consideration to environmental quality, as it is designed to prevent significant decreases in water quality. In Figs. 4.3.1-3 and 4.4.1-2, showing predicted D0 concentrations in the Ohio Alver for Alternatives 3 and 4. DO concentrations for these alternatives are shown to be slightly less than without hydropower development below about RM 200. Under the conditions used for the model analyses. DO concentrations were less than 6.5 mg/L without hydropower in these reaches, and additional spill at any of the dams would not be effective in increasing DO. The difference between the prodicted DB without hydropower and under alternatives 3 and 4 below RM 200 is caused by the conservative assumption that the hydropower turbines would provide no aeration. In reality, some small increases in DO concentration will probably occur at the hydropower plants, similar to those occurring without hydropower at the submerged-outflow gated dams in these reaches, further reducing the differences in DO concentration caused by the proposed hydropower projects and those occurring without hydropower below Ohio RM 200. None of the decreases in DO concentration predicted for Alternatives 3 or 4 would significantly reduce the quality of the aquatic environment.

this letter of a time motually arrest upon between the asympty and the Service. If The asymptotic much in completed before any construction contracts are let un begun. We do not helicky that we can adequately assess the effects of proposed actives on the park sucher without a complete assessment. When conducting a bickwice's assessment, you should, a a minumer:

- Conjust a submittically sound on-site inspection of the area to be affected by the action. This inspection must, unlers otherwise directed by the ferture, include a detailed survey of the error to determine if listed to rithered optices are present and whether suitable haltest exists within the area for either expecting the existing perilation or presential controlation of perilations.
- Intriview recognized experts of the species at issue, including these within the Dervice, state concervation agencies, universities, and enhans who may have data and yet found in scientific diterature.
- Review literature and other scientific data to determine the species' distribution, hubitat needs, and other biological requirements.
- Review and analyze the offects of the action on the species, in terms of individuals and populations, including consideration of the cumulative offects of the action on the species and habitat.
- 5. Analyze alternative actions that may provide conservation manuros.
- Ormioni may studies measure to fulfill the requirements of (1) through (5) above.
- 7. Review any other relevant information.

After your equary has completed and reviewed the assessment, it is your responsibility to determine if the properdiaction "may affect" the pink market. If you determine that the project "may affect" a listed species, you must request formal consultation from the appropriate field office of the Service. More you provide the biological assessment, you should include any other relevant information that assisted you in reaching your conclusion.

Tour attention is also directed to Section 7(b) of the Enlangered Species Act, as anended, which underscores the requirement that the Federal agency or the applicant shall not take any irreversible or irretrievable commitment of resources during the consultation period which, in effect, would deny the formulation or implementation of reasonable and prodent alternatives regarding their actions on any enhangement or threatened species.

- 37. The quotes cited in this paragraph reflect staff's opinion that water quality and power production for the proposed new projects should be analyzed and managed as an interacting system. However, staff disagrees that the system considered should include additional projects on tributaries of the Allegheny, Konongahela, and Ohio rivers. A system-wide analysis is not the same as a watershed-wide analysis; the system considered in the EIS includes all projects (existing and those with license applications accepted) whose operation can reasonably be exected to have cumulative and interactive effects on water quality.
- 38. Text has been modified to include a discussion of the impacts of depressed DD levels on freshwater mussels, particularly the endangered species jampsiljs abrupta (~ L, <u>orbiculata</u>). Additional information and evaluation of L, <u>abrupta</u> have been submitted to the USFMS to supplement the biological assessment contained in the UEIS in order to comply with Section 7(c) of the Endangered Species Act. Physical habitat issues for mussels were addressed in Section 4.1.2.2.3, and one project (Muskingum t&D No. 3) was determined by staff to cause significant adverse impacts to aquatic habitat, including habitat for mussels. Appendix i of the FEIS includes the supplemental information and analysis impacts for the four alternatives on L, <u>abrupta</u>. Staff concludes that whereas any alteration of river flow or water quality will have some affect on the mussel fauna of the OH River and Its tributaries, selection of the preferred alternative (Alternative 4) will cause no demonstrable effect on L, <u>abrupta</u>.

Specific Consenses

Section 2.1 pages 2 t that 2.27 Projective Generation Alternatives. Funfi has second there is criterian (1) there judgments the applicants, (2) State Standards (500 (1), epi (2) an an appendix of criterian (4.5 mg/1). This the forwards that second that criterian (4.5 mg/1). This the forwards that second that criterian (4.5 mg/1). This the forwards that second that criterian (4.5 mg/1). This the alternatives, we would not expect to that contain the of project by COMERSHIP (100 mg/1) and the comparison of the second that contain the of project by COMERSHIP (100 mg/1) and the comparison of the transition of the project by COMERSHIP (100 mg/1) and the comparison of the transition of the second terms of a second terms of the transition of the second terms of terms of the second terms of terms

Lage 2-25. Given that there is insufficient information to allow definition of existing 10 continue, the approach rescarce admetes recommended during the scoping process should have been ucilized. That approach: i.e., ifiling the DO date cope prior to hieraring, would elimitate namy unknews, recarding barin while hydro development. If it is not product to rely on unproven technology to monstain water guality, then it certainly is not product to rely on insidentiate data to develop models to determine "existing conditions" and to project future contributes.

40

42

Page 2-30. The Service does not across that potential fish parsage mortality [4] Can be considered a moderate or minum impact. Previously, staff stated (see Section 4.1.2.3) that (1) the results of existing turbine mortality studies are highly varied; (2) all entrainment studies completed in the basin are incomplete and inconclusive; (3) a firm basis for taking mortality estimates awits better results; and (4) much more work runt he done before a fith protective device can be achered. An analyzis of existing data does not lead off to the conclusion of many results inconclusion of many results.

<u>Iage 3-20, purgonalty</u> 2. In addition to federally listed species, one fish species to being studied for possible future listing. The blue sucker (<u>Orderbug Florgerue</u>), known from the mainstem of the Char River in the immediate study atem is classified by the Scruick as Category 2. Category 2 comprises taxa for which information new in possession of the Service indicates that projecting to list an unbangered or threatened is pussibly appropriate, but for which conclusive data on biological vulnerability and threat are not currently atailable to support proprised rules. (Reference <u>Federal Register</u>, Vol. 51, No. 106, Pulse and Regulations, gage 1984), third column, 4th paragraph.)

Face 3-20, 3.1.6.2. State Listing. Species list, payagraph 3. The smellmouth buffalo (intictus humalue) and the black reduces (Mixosuma dagassimi) have lists removed from the special concern category.

fage 3-20, 3.1.6.2. State Listing, Mullusks. Chie Species List, Foragraph 6. An the instartly it winds lineolate).

5

39. The Department of Interior (DOI) and several other agencies have requested that the FEIS include an alternative that would require maintenance of "preproject DO conditions", in accordance with antidegradation policies for water quality. This request was made by DOI in scoping meetings and was considered in preparing the EIS (see Section 2.1.3). Staff believes that (i) DOI and other agencies have not recognized significant technical problems that arise in determining what "preproject conditions" are; and (2) requirements for strict maintenance of preproject DO concentrations would prohibit power generation, without resulting in significant water quality benefits during times when DO concentrations are not low enough to affect aquatic organisms.

At dams that aerate well, any amount of hydropower generation (with its necessary decrease in spill flows) would reduce 00 concentrations downstream when 00 deficits occur. Strict maintenance of "preproject conditions" would prohibit hydropower generation at such dams. DO] and the states apparently are requesting that some level of spill flows or mechanical acration be implemented that would allow power generation without decreasing 00 concentrations. Staff's water quality and economic analyses indicate that requiring rates of either spill flow or mechanical aeration [1f mechanical aeration is shown effective) high enough to reproduce existing aeration rates year-round would make projects at dams that aerate well infeasible.

60 concentrations at any point vary continuously over a wide range with concentrations being controlled by temperature, waste loadings, river flows, and biological processes. There is no single set of preproject D0 concentrations for the system. Staff did evaluate preproject D0 conditions for the DEIS in the following ways: [1] historic D0 and temperature frequency distributions are presented for the ORSANCO electronic monitors, the only locations where D0 measurements are extensive enough to do so; [2] D0 concentrations measured by the Corps during typically poor water quality conditions are presented; [3] parameters that mathematically describe the dam aeration available under existing conditions are presented; and [4] in all the modeling analyses, D0 concentrations without the proposed hydropower projects are presented for comparison to concentrations with the projects (including model analyses based on monthly mean conditions presented in Appendix B).

DOI and other agencies request that DO concentrations be monitored for one or two years at each proposed hydropower site to define "existing conditions." One or two years of DO measurements are not adequate to define baseline conditions. For example, DO measurements made in the summers of 1987 and 1968, as recommended by several agencies, represent unusually low DO conditions and if used as a baseline could result in significant degradation. In addition, the experience of project applicants that attempted to monitor DO at their sites indicated that frequent gaps in monitoring data must be expected and that the monitors tend to underestimate DO concentrations. Staff believes that having applicants monitor DO for one or two years would not provide information adequate to base spill flow requirements on, and would be an unjustified burden on license applicants. DOI and other agencies have not defined what is meant by "preproject or "existing conditions", or how these conditions would be determined from measured DO data for application to anti-degradation criteria. Staff does not believe it is justifiable to require collection of data that (1) will contain gaps and errors due to monitoring problems, {2} will not define the historic range of conditions adequately, and (3) will be used to determine DO criteria when the methods used for such a determination have not been specified and may not take into account effects on either aquatic organisms or generating Capacity.

Staff believes that it is more prudent as well as more practical to maintain a specific D0 criteria, based on the best available scientific research and selected to avoid impacts to fish, then to try to maintain undefined and poorly measured "existing conditions". This approach is especially appropriate in rivers whose water quality is as highly affected by development as are the Allegheny, Monongahela, Muskingum, and Ohio rivers. Staff believes that much more success will be had in implementing a relatively simple D0 management objective (maintainting concentrations of at least 6.5 mg/L) than in attempting to implement D01's proposed anti-degradation policy on rivers whose water quality is already heavily affected by man. The cumulative nature of hydropower impacts on D0 would make implementation of anti-degradation D0 criteria that vary between sites and seasons extremely complex and difficult to enforce.

Requiring no degradation of DO could essentially preclude development at dams that currently are important serators. At such dams, the only ways to strictly maintain existing aeration: rates are to either spill all the water or to aerate mechanically. Mechanical aeration is expensive and could make projects economically infeasible if required continuously. During much of the year hydropower can be generated without causing DO concentrations low enough to affect aquatic life, even though some decreases in aeration would occur. (The model analyses based on monthly mean conditions presented in Appendix B. Figs. 8-9 through B-17, show how the projects as proposed would decrease DO concentrations in spring and fall without causing concentrations low enough to be of concern for aquatic life.) This hydropower would be used to displace power currently generated by other sources, mainly coal-fired and nuclear plants, that cause significant environmental impacts.

West Virginia, the only state whose comment letter implied they had antidegradation regulations that apply to the waters considered in the EIS, has designated the Monongahela and Ohio rivers as "high quality waters". West Virginia's water quality regulations (46 CSR Section 46.1-4.1.e) states that such designated high quality waters "must be maintained at their existing high quality unless it is determined after opportunity for public comment and hearing that allowing lower water quality is necessary to accommodate important economic or social development in the area in which the waters are located. If limited degradation is allowed, it shall not result in injury or interference with existing stream water uses or in violation of State or Federal water quality criteria that describe the base levels necessary to sustain the national water quality goal uses of protection and propagation of fish, shellfish and wildlife and recreating in and on the water." Staff believes that the environmental and economic benefits of hydropower development (as recommended by staff) that would not cause violations of the State criteria would justify the limited degradation allowed in this regulation.

To assure that hydropower projects do not cause significant impacts due to low BO concentrations, staff has recommended that DO concentrations upstream and downstream of each project (recommendation 4, Section 5.4.2.1) be monitored and the data provided electronically to ORSANCO, and that water quality management agencies and fERC be authorized to require increased spill flows during low BD events (recommendation 2, Section 5.4.2.3). In addition, formation of a basin-wide group to monitor and manage DO at the projects is also recommended (recommendation 7, Section 5.4.2.1). Staff feels these recommendations [1] will fully protect the aquatic environment, (2) are practical to implement, and (3) will allow generation of significant amounts of power.

The staff's recommended alternative is not to maintain 6.5 mg/L of D0 immediately downstream of hydropower projects, as stated in this comment. The recommendation (Sections 2.1.4 and 5.4.2.1) is to operate the projects in such a way that they would not cause D0 concentrations to fall below 6.5 mg/L, anywhere in the rivers where D0 concentrations are affected by the projects.

- 40. The staff does not believe that collection of one or two years of BD measurements at each dam would describe the so-called existing DD conditions, due to the high variability in DD concentrations and the problems and errors that can be expected in monitoring DD (see response to comment #39). Instead, the staff chose to measure the existing aeration rates at each dam, which were used in models to quantify the effects of hydropower.
- 41. Staff has determined that hydropower development at the Hontgomery site would result in a predicted high level of susceptibility to entrainment of fishes traversing the embayment mouth. At other hydropower sites the relative degree of impact is predicted to be minor or moderate, and mitigation plans for fish protection can be established accordingly.

42. Text has been modified to correct these lists.

Pagends (1922) Weller 1.1. (d. the list only connected that exclusion states are a concerned about certain species, warms federally protected openies are a contarp of will the states and simple is so acted. The federally listed precise aver

ndi a j	C DYCE J	E-
Raio 4	119. m	a ket
Auras	ne fal-	(*),T

Hydras socialas lampilis abrupts Falce p regrisos

In addition, the following species have been decounted as cardidate species and classified Category 2. Reference our connects on page 3-20, paragraph ... Those execute from the list include:

EPHALOS S WHEN	Thryceaner kestekii
Distorn wordraf	Nootoria magister
New England cottorstail	Sylvilaous trensitionalis
Salamander shell	Singlemanae arbigae

Page 3-67, Pisheries. This section chould include a discussion of potential 1 43 water quality [i.e., Di) degradation on shellfish. In Inday's (1971) experiments, all species examined required 6 ppm of 60 for normal growth.

Section 4. Environmental Usereguences. Palagraphs 4.1.2, 4.2.7, and 4.3.2 should all countain an assessment of the penential impact of DC change on shellfish.

Page 4-4, personally 4. DO depletion is probably the most significant issue 44 dealt with in the Wife. The (a model was designed to predict both existing and per-project 20 concentrations. Considerable effort was obviously expended in its development and the subsequent analyzis. The baseline data are, therefore, of relamount intertance. As such, the unroblished data upon which the model was based should be provided in the revised draft. The value of the model and predictions next to opertioned if they were based on limited field data, such as the number of observations reported on pape 5-7, Cable P-1.

Face 4.1. Assessment Methods. 30 measurements were determined at a number of 45 different flow rates and temperatures. Subsequently, dam-acraticn constants (b) and dom-seration coefficients (11) were developed via statistical analysis. The DELS does not, however, indicate how many measurements were made at each flow level and/tr whether the data used are representative of existing conditions. The D0 data analyzed should be provided in table form so reviewers can take a reasoned decinity as to its applicability. The scientific validity of model predictions of future conditions with or without the proposed developments should be verified by water quality experts in state and federal environmental apendies.

Figure 8-8 shows there were no data to calibrate the DO model between Olio 46 River miles 140 and 250 and that only two date points exist below Ohio Piver aile 250. Without reliable and adequate calibration data we further question

43. See response to comment #38.

44. The dam aeration measurements used in the model are presented in Appendix 8 of the FEIS. Staff believes the field measurements used to develop the agration models are sufficient. The uncertainty analysis which is presented in Appendix B shows that the model uncertainty resulting from variability in the dam agration measurements is less than the uncertainty in other parameters controlling DO concentrations, and that uncertainty is least in reaches where DD is controlled by dam aeration. These results indicate that the dam aeration models developed from the field data are of acceptable precision.

Ż

- 45. The dam aeration measurements used in the model are presented in Appendix B of the FEIS. The modeling methods were presented to water quality experts from the Corps and ORSANCO at a meeting in Cincinnati, Ohio, on January 21, 1988 and have also been reviewed by state agencies as part of their review of the DEIS.
- 46. The model was calibrated to the most appropriate data available. Staff does not believe that the lack of calibration data below RM 140 is a serious problem. Below RM 140, the proposed hydropower projects occur at submerged-discharge gated dams that provide little aeration and where hydropower projects would have little effect on DO concentrations; this fact is adequately demonstrated by the field data.

the model's vehicity, particularly for all points below this kiver mile 140. Further, the prefertion that hodes and does below this Piver mile 140 do not provide significant accarbox must be surpect; maxim limited field data exist.

It is unclear which of the compating applications were used to develop the D. 47 model. As significant variations in size and operating arde corur, we must presume the present's impacts on DD world vary. If coursic hydrogener proposals were used in the rated to be well, then it would seem a single ratter to include all hydrogener tites in the latin in the usual style assessment.

Page 4-20, forcer the 1. No information is presently available commiting the direction or turning of find movements in the three Pover and its eager tributaries. Reason, available literature indicates that recalest migratury species move doomstroar frequently. In fast, downstroar migrations of wallage from Yuant has are more likely responsible for the establishment of wallage populations in the opper reactions of the Mannachela Store.

Page 4-25, paragraphs 3 and 4. The staff states that existing entrainment field studies are incomplete and inconclusive and ruch are development work much be completed before fish plotective or quidance devices can be selected. As such, licensing of any projects within the Ghie basin, in alwance of the completion of studies measuring to determine irracts to fish and wildlife resources, world violate providences of the Federal Power Act (Yakina et al. v. FERC).

Page 4-25, Vulnershilly to Entrainment. We information is available which indicates fish would select a non-hazardous downstroam route instead of a inizardous cone. Fish normally follow market patterns unders halitat and cover orbibit significant influences. In the case of lock and dam hydropower development, halitat and cover and current petterns will all increase the likelihood of entransmit. While passage alternatives may be available, the majority of downstroam migranus are expected to pass through the units unless the greatest processor of river flow is through the dam gates.

Fage 4-28, reray at). While the statement is very qualitative, the Service has not seen any data regarding gamefull movements through locks and dans. The analysis done by Samaans (1987) indextes significant downstream movements for contrarching and catfishes. Given the staff conclusion that the studies at Racine are highly delatable and inconclusive, the pertulation that inkely insignificant is questionable.

Page 4-10, paragraph 6. The Service agrees will the staff conclusion that damage to upmedich from turbine passage will be greated that ten percent and that the loss could be nignificant.

Page 4-32, Fish Diversion and Fundertics Devices. Staff has determined there is "... hit is experience with species must affected by entrainment ... ": that "No applicant has corried design plane to sufficient detail to evaluate their proposal fully ... ": and that "... none work at developing and

7

- 47. In the model calibration and analyses, the competing applicants that would have the most impact on 80 concentrations (i.e. the lowest proposed spill flow) were included.
- 48. Although no information on direction and timing of fish movements has been collected on the Ghio River or major tributaries, the DEIS made extensive use of similar information gathered for relevant species in the Hississippi River (Holland et al. 1984; Normandeau Associates, Inc. 1986; Sections 3.2.3 and 4.1.2.3.1). Interpol movements of walleye and channel catfish in those reports have been highlighted in the EIS. The amount of movements of other species was summarized. Downstream movements of walleye at Tygart Dam that are important for populating the downstream river reaches are discussed in Section 3.4.3.2.
- 49. Staff does not agree that its extensive treatment of the entrainment issue violates provisions of the Federal Power Act as interpreted specifically by the Yakima et al. v. FERC decision and the Hational Environmental Policy Act as interpreted by the LaFlamme v. FERC decision.

Staff has considered the issue of entrainment at length in Section 4.1.2.3. It has followed Council on Environmental Quality (CEQ) guidelines (40 CFR Parts i500-1508, par. 1502.22) concerning incomplete or unavailable information. A workshop of agency staffs and noted experts was held to bring to light all pertinent information that is currently available. Relevant studies nationwide and in the Ohio River basin were summarized in the DEIS. Discrete aspects of the entrainment issue were treated, including summaries of culterability of fishes to entrainment (Section 4.1.2.3.1), damages to entrained fish (Section 4.1.2.3.2), and fish diversion and protection devices available for mitigating damages (Section 4.1.2.3.3). Staff analyzed what is known and identified in the DEIS the deficiencies in available information. Staff judged that all information that would be desirable for making a reasoned choice among alternatives cannot be obtained at reasonable cost.

Staff used the extensive, but still incomplete, body of "existing credible scientific evidence which is relevant to evaluating the reasonable forseeable significant adverse impacts on the human environment," to reach its conclusions. Conclusions were presented in the DEIS regarding the severity of risk to fish and what might be done to reduce and/or compensate for that risk. Because experience has shown entrainment to be highly site specific, staff has recommended a sequential mitigation process that included some activities after licensing: (1) monitoring of entrainment (i.e., measuring the actual entraimment after operation has begun), (2) compensation to the resource agency for measured losses until appropriate site-specific mitigation can be designed and installed, (3) a basin-wide cooperative effort to develop and test at selected operating sites a series of prototype bioengineering facilities for fish protection and/or guidance that have not yet been shown to be effective in the Ohio River basin, and (4) reevaluation of alternative mitigation approaches that could include installation of fish protection or guidance devices demonstrated to be effective, continued compensation, or other appropriate mitigation schemes.

Participation of licensees in this sequential mitigation process, to be overseen by FERC and coordinated with resource agencies, would be assured by license article. Aspects of this analysis and mitigation framework in the DEIS that may have been misinterpreted or are in need of clarification have been addressed in the FEIS.

- 50. Comment noted. Section 4.1.2.3 of the DEIS indicated that the route of passage may approximate the division of flows between turbine and gates. Additional clarification regarding cover and flow effects has been added to Section 4.1.2.3.1 on vulnerability to entrainment.
- 51. Section 3.2.3 of the DEIS discussed movements of specific gamefish species at navigation dams on the Mississippi River as summarized by Holland et al. (1984) and Normandeau Associates. Inc. (1986). Wallaye and channel catfish were frequently observed moving between pools. There was no movement of smallmouth bass or largemouth bass. About 20 percent of tagged sauger showed interpool movements. These are the same species found in the Ohio River basin. Text has been modified to remove the conclusion regarding significance of Racine gamefish entrainment from this section.

.

52. Comment noted.

.

.

53. See response to comment #49.

evaluating experimental prototypes will be necessary.". We rust, therefore, conclude that licensing of basin projects would be at odds with provisions of the Federal Power Act (Yakuma et al. v. FFRC, and, LeFlamme v. FFRC).

Page 4-13, remarking J. Compensation is the least preferred approach to mitigate adverse impacts such as fish passage mortability. Compensation means full replacement of project-induced leases with resources of equivalent biological value. If used, compensation must remain the responsibility of the licensee to implement and maintain during the project life. The Service will only enderst a compensation plan that (1) demonstrates compliance with the sequential religation process; (2) provides for full replacement; (3) will be incompensate as an enforceable license combine; and (4) ansures that mitigative measures are the responsibility of the licensee.

As turbine induced mortality has not yet been defined, a properly designed and conducted study is essential prior to developing a mitigation plan. Without an adequate import evaluation, the adequacy and effectiveness of mitigation will remain universamped.

Page 4-47. Section 4.1.3.3. Loss of recreational fishing opportunities during 55 concurrent construction of havin projects is identified as an impact for which mitigation has not yet been resolved. Resolution of this issue should be achieved prior to licensing and after consultation with the Service and state wildlife agencies.

Face 4-49. Section 4.1.3.6. This section identified numerous significant adverse impacts to water quality and finfish should projects be licensed as originally proposed. Hesed on the data previously submitted, the Service concurs that these impacts must be avoided; clearly, licensing as originally proposed would be environmentally unacceptable.

Fage 4-62, Section 4.1.6.2. Endangered/Threatened Species, paragraph 1. It should be added that any babitat disturbance below Willow Island Lock and Dam, belleville Lock and Dam, and Sallipolis Lock and Dam will also require additional consultation with the Service in accordance with Section. 7 of the Endangered Species Act. Reference our latter dated September 28, 1987 and the DEIS, pages 3-19 and 3-20.

<u>Proce 4-71, Section 4.1.6.8</u>. A response to the previous Departmental and state swildlife agency recommendations to utilize clean rock spoil to construct fishery habitat structures should be provided. Implementation of that recommendation would offset project impacts to fishery habitat and recreational use while likely reducing developers spoil-disposal costs.

Page 4-87, Section 4.7. Departmental comments provided pursuant to the National Environmental Policy Act do not represent comments pursuant to the Fish and Wildlife Coordination Act (48 Stat. 401, as amended; 16 U.S.C. 661, et seq.) or Section 7 of the Endangered Species Act of 1973, as amended (16 U.S.C. 153) et seq.). Commants under Section 7(b) of the Fish and Wildlife Coordination Act will be provided when draft licenses, including site-specific articles, are provided for each project.

- 54. The development of compensation glans is recommended by staff for inclusion in any licenses issued for the projects. Staff recognizes that there is no universal agreement among resource agencies that compensation is "the least preferred approach" to mitigate adverse impacts of fish passage mortality. The NVDNR has "contended for some time that adequate monetary compensation of state resource agencies may be the only equitable solution to entrainment mortality problems" (see response to comment f154). Staff outlines a sequential mitigation process (see response to comment 449) in which compensation plays a role in two ways: (1) as interim mitigation while prototypes of engineering schemes for protecting fish from entrainment damages are tested, and (2) as an option for consideration should fish protection devices be shown to be infeasible and other alternative mitigation measures must be chosen. It will be necessary for the licensee and the resource agencies to develop appropriate compensation that amounts to "full replacement for measured losses." This agreement is recommended as part of the FERC license and would be a responsibility of the licensee. A further elaboration of the compensation alternative has been added to Section 4.1.2.3.4.
- 55. The provision of temporary fishing access facilities during construction is identified as a means of mitigating the loss of recreational fishing during concurrent construction. Recommendation 5 of Section 5.4.2.3 specifies that developers file a plan for providing temporary fishing access facilities with the Commission after consulting with the appropriate state and federal agencies.
- 56. Comment noted.

56

59

- 57. Text has been modified to include all of the sites referenced.
- 58. Staff agrees that the utilization of clean rock spoil to construct fishery habitat structures could offset project impacts to recreational use. The provision of reefs and/or other fishery habitat structures is specified as part of the standard level of recreational development in recommendation 1 of Section 5.4.2.3.
- 59. Comment noted. The statement concerning agency comments to be provided when draft licenses are provided for each project is unclear. The Commission does not issue draft licenses.

6

23

Hage 4-b7, Sourise 4.7, Relationships to Levis and Policies, principle 4. A support the Hall's does not constitute consultation under Section 7 of the Hadaqueted Species Act. The consultation process is separate and has not yet been recovered by the Commission. Reference can General Connests.

Further, the PHIS lacks adequate information (site-specific location, design measures to minimize harm) for a full understanding of how the Section 10 and 404 permits form Carps of Degineers or possible Section 402 NPDES permits, affect fich and wildlife resources. Accordinally, our comments do not previous separate exclusion and expectes by the reperiment when reviewing the permit applications. The Department may course, with our without stipulations, or recommend design depending on officers to fish and wildlife resources.

When appropriate site-specific information is available, the Service would be phonse to convinate with the Connassion and/or licenses to proclude delay and to insure that any perfut slipslations or conditions are uncerstood and included in the issued draft and final resteness.

<u>Date 5-6, telegraph</u> 2. While mitigation could reduce impacts of monourrent construction to regulational fishing, site specific mitigative measures are not identified. The National Environmental Pulsey Act requires documentation to explain exactly now measures will mitigate project impacts (Jones v. Gordon).

<u>Page 5-16</u>, Section 5.1.4. Given that "fish mentality concerns" are not received by licensing under any of the eltensitives considered, license inchance prior to resolution of those issues would appear to violate provisions of the Federal Rever Art. (Yakina et al.v. FER') and National Environmental Folicy Jet (LaPlanne v. FERC).

Page in 23, paramont 2. The Fervice feels that there are significant
63 differences between computing license applications and refers staff to project specific lepse transfer letters regarding the proposed "ygart, New Comberland, Willow Irland, and Cellipelis projects.

Page 5-25, Basin-wide Recommendations. Prior Departmental and Service recommendations requested that the Service le identified as a coordinating agency in Diverse articles derling with fish, wildlife, and public uses thereof, en well as with speil disposal, water quality, and mitigation of proper velated effects. In accordance with provincings of the National Divingmental Policy Act, Fish and Wildlife Coordination Act, Federal Power Act, and the Electric Considers Protection Act, Bederal Power state wildlife anomcies rost be specifically identified in each such article. The Council Felacy Act Dist the Service as having jurisdiction by law and/or special opperties on these issues.

Fage 1-27, Item 3. Although conitoring fish reputations may provide sure insight into knowtone impacts, an indirect approach is expected to be tach loss surproful than a direct one. Numercus variables affect the results of even direct sampling and analysis methodologies. Indirect approaches would add or confound many variables, such as catch, hervest, recruitment rates,

- 9

60. Comment noted. See response to comment #38 regarding endangered species consultation. The text has been amended to include a description of Section 10 and 404 permits required by the Corps pertaining to the placement of dredge or fill materials in waters of the United States. Elemeses, for any license issued, would be required to obtain these permits prior to the start of hydropower construction. Staff does not know what the specific stipulations or conditions attached to these permits are and, therefore, cannot include them in the FEIS.

- Recommendation 5 of Section 5.4.2.3 discusses specific measures for mitigating the impacts of concurrent construction to recreational fishing.
- 52. See response to comment #49.

62

64

65

- 63. See response to comment #34.
- 64. Comment noted. The U.S. fish and Wildlife Service has been identified as a coordinating agency, as appropriate, in many of the recommendations identified in Section 5. The text has been changed to also include the U.S. fish and Wildlife Service as a coordinating agency in the following recommendations: Section 5.4.2.1, items 4, 5, and 8; Section 5.4.2.5, item 6.
- 65. Direct measurement of fish entrained and damaged during entrainment at representative sites is identified under recommendation 4 of Section 5.4.2.2. Staff believes that such measurement could be accomplished sufficiently accurately at a representative set of sites rather than at each site. Staff notes that confusion could arise between two forms of "monitoring" discussed in the DEIS: {1} measurement of entrainment rates and damages (recommendation 4), and {2} measurement of combined impacts from all sources on populations (recommendation 3). Staff concurs that the variables mentioned in the comment are sufficient to make direct comparisons of population success with entrainment nearly impossible. The broader form of monitoring, however, is useful for assessment of the overall state of fish populations and can be the trigger for implementing more specific monitoring studies if problems are shown.

worther, river condition, water quality, habitat, emicrotics, indigratics, etc. A direct approach must be developed if successful turbide mortality studies are to be conducted.

Fage 5.77. Item 6, and Page 5-26. Item 5. The proposals for the agencies and some To applicants to develop joint plans to study fish entranment, turbuteinduced normality, and to develop a licengineering test facility, are converdable. However, the Service is concerned that coordination, add upberministration of such large scale undertakings, with an yet undefined financing and inscience of the scale undertakings, with an yet undefined financing and inscience of the exponent of result in an unworkable proposal. If the confiction elects to parame this endoward, then the party to represent all applicants, should be appended and the commutation should assess equivable from the contrarted through an independent organization. But there is no part - negacing rudies to determine the impacts and to develop adequite adjustion requires appears to violate the Rational Environmental holicy bet, which requires account of a function of environmental inpacts before licensing decisions are ande (LaFlance v. 1980).

Fage 5-25, Section 5.4.2.5, 1. Endancement and Threatened Provides. Levelopers 67 at Willow Hilard Took and Bam most also consult with the Service and the state resource populates. Reference our letter dated September 28, 1967 and the DELS, pages 3-19 and 3-20.

Page F-2, Section B.7.1.1.1. Date and figures showing the number of DU samples taken at each river flow, by location, date, time, temporature and climatic conditions are needed to enable reviewers to place the model into perspective with existing conditions.

Fage <u>B-3</u>, paramaph 6 and Table E-1. The statement that dans with lower annation coefficients "N" and more requirive values of "b" arrate Letter, is inconsistent with text in Section 4 and Table 4.1.1.1.

Eage P-16, Figure 1.6, and associated test. The Df18 should explain why the model predicts D0 ieductions of 0.7 mg/2 between this first mule 160 and 240 but no reduction for the 40 miles below Racine Eau; and a E0 increase below Gallipoles Dam, despite the fast that Gallipoles is reported to be a potr scratter. Further, since both actual data and model predictions (see Figure 5-8, 4.1.1-4, and 4.1.2-3) show a 50 sump at Gellipoles, why is Gallipoles said to be a roor actator?

It should also be nored that Racine apparently adds no significant DO surge. (As a licensed project, Racine was not included in the cumulative analysis.) Inclusion of Pacine may have meant recommendations for spill flows which would similificantly include (0 in the lower raver.

<u>Face E-27, preserveb 2. See our General Convents regarding inclusion of all</u> beau project: in the curulative assessment.

16

66. Staff recognizes the need for detailed arrangements for cooperative efforts by developers. The proper time for establishing such details is in the postlicensing consultation process. The suggestions of the USFWS are appreciated. See response to comment #49 in reference to EaFlamme v. FERC.

67. Comment noted. Text modified.

68

69

-71

- 68. Text has been modified in Appendix B to include tables showing the dam aeration data used in the model, including flow and temperature (as available).
- 59. The dam aeration equation in Appendix B is slightly different than the one in Section 4.1.1.1; the sign of the constant b has been changed in Section 4.1.1.1 so that the value of b reflects the amount of supersaturation at a dam. The equations and parameter values are consistent and correct within each section.
- 70. Below Pike Island dam, dam aeration has little effect on DO because the remaining dams are poor aerators. Without dam aeration, DO is provided to the river through aeration at the water surface, which increases as the DO concentration decreases. What the model is predicting below RM 240 is that DO concentrations have decreased to the point where the amount of oxygen added by water surface aeration equals the amount removed by 800 decay.

The field data show that Gallipolis dam is a slightly better aerator than Racine, Belleville, Willow Island, or Hannibal dams. However, the amount of DO predicted to be provided by Gallipolis is minor. There may be some confusion caused by figure B-8 (showing calibration of the Ohio River model), which indicates an increase in measured DO concentrations between the two stations (ORSANCO electronic monitors) at river miles 260 and 279. These two measured data points are above and below the mouth of the Kanawha River, not above and below Gallipolis dam, so they do not measure aeration at Gallipolis.

71. In all the model analyses, Racine and the other licensed hydropower projects were assumed to be in operation. Racine, having no required spill flow, was assumed to provide no aeration. Racine dam is structurally very similar to Belleville dam, with a deeply submerged discharge. Spill flows at Racine would not significantly increase DO concentrations.

Summary Comments

72

73

74

75

76

following issues which were identified in the scoping process as issues of primary concern: water quality, endangered species, hydraulics, altered-flow patterns, flow modifications, turbine-induced fish mortality, recreational fishing and dredge spoil disposal.

The Service does not believe the DEIS provides adequate onverse of the

The DEIS does not provide adequate recommendations or articles to protect fishand wildlife resources of the public use thereof. Yet, staff proposes that our comments on the DEIS will be the rost current set of comments under the Fish and Wildlife Coordination Act and that their response in the FEIS will constitute issue resolution under 10(j) of the Electric Consumers Protection Act.

No indication is given as to which competing applications will be licensed. Conservation and load management was not adequately addressed as a project alternative. Also, the Commission claims to have complied with the Section 7 consultation process of the Erdangered Species Act, by providing the Service with a review copy of the DEIS. This does not constitute consultation. Further, the Service does not concur with staff recommendations to postpone resolution of the issues regarding fish turbine mortality and fish passage until the post-licensing phase.

Because of these concerns and the exclusion of numerous basin projects from the analysis, the Service requests that the DEIS be revised to address the fish and wildlife related problems previously discussed and then recirculated for review and comment.

Unless these issues are adequately resolved, we may request that the Department refer this project to the Council on Environmental Quality under Section 1504 of the Council's Regulations for Implementing the Procedural Provisions of the National Environmental Policy Act. However, we wish to coordinate fully at the earliest possible time because a solution to our concerns can be implemented with a minimum of delay and could preclude the necessity for referral. Coordination can be initiated by contacting me at this address.

Thank you for the opportunity to comment.

Sincerely, Charles J. Kolo

Supervisor

11

Enclosures

72. Your opinion is noted. Staff believes the EIS does assess the environmental impacts of the proposed projects for these issues, determines the significance of the impacts, and where possible, makes recommendations that would reduce significant adverse impacts to an acceptable level in the study area. Staff assessed all available information and used their best professional judgement to resolve any areas where there was conflict in existing information.

73. See responses to comments #31 and #72.

74. See responses to comments #34, #35, #38, and #49,

75. Your opinion is noted.

76. Your opinion is noted.

25

Literature Citro

- Imlay, K. J. 1971. Howevery tests with measure. In Jorgensen and Chatp $(3921,\ p\ 21-512),$
- Simments, L. L. 1987. Fish critaineut studies for the Nillville Hydro Station, FESC Project # 2343, report from Prenny and Dwinnmeental Wanagement, Lne., to the Potomec Edison Cruptory.
- U.S. Department of Dyrgy. 1980. Draft. A study of altributions to the Lavie (Rest Virginia) purped storage project. Weltment 1, 2nd 211. Fibitel under contract by ICF, Luc., and ESFI, Juc. U.S. Govennent Frintung Office, Washington, D. C.

AUGUITIANA PROFITE WE DE INCLUED IN THE OPERATIVE INDRESS AUDUINARY OF THE OPERATION FATHER

Allecheny River Sublasin (Pennsylvania)

Allecheny River (rock and Lam Nr. 5 Project (FDF No. 3671)* Allecheny River Lock and Dan No. 6 Project (FDC No. 3694) Allecheny River Lock and Dan No. 8 Project (FDC No. 3021) Allecheny River Lock and Dan No. 9 Project (FDC No. 3021) Finely River Lock and Lam Re. 9 Project (FDC No. 3021) Fast Branch Fregert/Clarien Eiver (FDC No. 309) Fast Branch Fregert/Clarien Eiver (FDC No. 3292) Loyalhanna Project/Tickimizetas River (FDRC No. 3207) Kinzas Fregert/Dickimizetas River (FDRC No. 3207) Kinzas Fregert/Rickimizetas River (FDRC No. 3207)

Monomabela River Salkanis (Muryland, West Virginiz and Ferns) Lynnia).

Pionongabela River Icck and Exam No. 2 Freeject (FERC No. 3973 ct al.) Honongabela River Icck and Ixm No. 3 Project (FERC No. 3752 et al.) Horganizam Icck and Dam Project (FERC No. 9949) Inke Lynn Project/Chest River (FERC No. 2459) Deep Creek Lake Project/YoughLondeny Fiver (FERC No. 2370) YoughLondeny Dam Project/YoughLondeny (FERC No. 3623)

Benver River Sublasin (Obic and Peursylvania)

Townsent Project/Beaver Fiver (ITEC No. 3451) Patterson Project/Beaver River (ITEC No. 3519) Nativale Project/Beaver River (FIEC No. 3519) Patterson Project/Shemango River (FIEC No. 3450) Shemango Project/Shemango River (FIEC No. 4142 et al.) Pymatuning Project/Shemango Fiver (FIEC No. 9190 et al.) Take Arthur/Makky Greak Filwood City Project/Counsequencing Greek (HEPC No. 7253) Mosquito Like Project/Resquito Greek/Crise Nerlin Take Project/Mahoning River/Chio Milton Take Project/Mahoning River/Chio Milton Take Project/Mahoning River/Chio Milton Take Project/Mahoning River/Chio

Muslingue River Subhaum (Chic)

.

Devola Lock and Dam No. 2 (FERC No. 775)) Markingun Lock and Dam No. 3 (FERC No. 6998) Beverly Lock and Dam No. 4 (FERC No. 7754) Like Chuts Lock and Dam No. 5 (FERC No. 7261) Stockpart Lock and Dam No. 6 (FERC No. 7259) McCornelsville Lock and Dam No. 7 (FERC No. 7262) Rockeby Lock and Lon No. 8 (FERC No. 7222) Philo Lock and Dam No. 9 (FERC No. 7221) Markingup Lock and Dam No. 10 (FERC No. 6563) Llis Look and Lor Do. 11 (HDR: No. 1220) Billon Loke (FLS: No. PCB) Millon Low (FDR: No. 9972) Floorant Hill (HTR: No. 8270) Mills Creek (FDR: No. 8259)

Recenter River Sulfamin (Virginia and West Virginia)

Winfindd Look and Dan Project (FDC, No. 1260)
Namet Look and Dan Project (FDC Loc. 1175)
Loudon Look and Dan Project (FDC Loc. 1175)
Steten project T1k Every (FDC Loc. 9344)
Summersville Project, (FDC Couley Eiver (FDC No. 16226)
Chapter Lake Project, (FDC 749)

Simesance Kiver Subaris (Indiana)

lake Shaeifer Propert lake Freens Project

Chin Eiver Hainsten (Fennsylvania, thio, West Virginia, Indiane, and Illinois

ż

Hanniba) Locks and Dum Project (FIRC Mo. 5296) Racine Locks and Lam Project (FIRC Mo. 2570) Greenup Lock and Dum Project (FIRC Mo. 2570) Gapt. Anthony McRadd Locks and Dum Project (FIRC Mo. 2614) Capt. Anthony McRadd Locks and Dum Project (FIRC Mo. 2779) Markland Locks and Dum Project (FIRC Mo. 2211) McSalpine Focks and Dum Project (FIRC Mo. 229) Cannelton Locks and Dum Project (FIRC Mo. 229) Cannelton Locks and Dum Project Uniontown Locks and Dum Project Uniontown Locks and Dum Project Smithland Locks and Dum Project (FIRS) Do. 6641) Locks and Lam Mc. 52 Project Locks and Dum Mo. 52 Project

Where known the FIRT NO. has been included.

J-29



United States Department of the Interior OFFICE ELECTRONIC MERIDIA REVIEW

WASHINGTON, D.C., 20240

ABS 1 869

77

ER 08/419

Ms. Lois D. Content Acting Secretary Federal Energy Regulatory Contonnation 825 North Capitol Street, N.E. Washagton, D.C., 24426

Dear Ms. Cashelly

The Department of the Interior has completed its review of your draft environmental impact statement for Hydroelectric Development in the Dipier Unio River Resai, Ohio, Pennsylvania, and west Virginia. We have the following comments and recommendations on the document. Specific comments are enclosed as a separate document.

FISH AND WILDLIFE COORDANATION ACT AND SECTION 19() ISSUES

It is our understanding that FERC intends to have the final environmental statement serve as a licensing document. In this context, FERC has stated (on page 4-87) that Departmental comments filed pursuant to the National Environmental Policy Act (NEPA) will be considered by FERC to be the most current reconstantiations filed pursuant to the Fish and Wildlife Coordination Act (FWCA). Departmental comments on a particular impact statement are provided pursuant to NEPA and do not represent comments pursuant to FWCA or Section 7 of the Endangered Species Act (ESA). Comments under Section 2(b) of FWCA are provided when disk applications are made available for each project. Comments under Section 7 of ESA are provided when FERC initiates the consultation process.

FERC also states on page 4-87 that its response to our comments on this environmental statement will be used as a mechanism to comply with Section 10(j) of the Federal power Act (FFA). In our opinion, a response to comments on a general environmental statement covering materous projects does not constitute issue resolution under Section 10(f) prime there is insufficient information to determine whether the staff's recommendations are consistent with those of the FWS. Furthermore, this does not reflect the spirit of the FPA since FERC's action may limit future opportunities to resolve issues on individual licensing actions.

We believe that all conducates made during the licensing process of any project should be part of the administrative record for that project and there should be unalogue between parties at any time if conflicts arise. If EEC intenes to use these environmental documents as part of the Section 18(j) process, then 18(j) discussions should be presented in a separate section that contains specific recommendations for each project, a detailed discussion of any disagreements with the FWS, and adequate documentation of the FERC staff's position. 77. See response to comment #31.

ġ

CUMULATIVE IMPACE ANALYSIS

On June 29, 1987, the FNS commented on FERC's notice of intent to prepare an environmental impact statement on the notential cumulative impacts of hydrochestric projects in the Ohio River Resin. At their time, FWS concurred with | ERC's devision to prepare such an impact statement; however, concern was expressed about which projects were to be included. Since that time FWS has maintained the position that FERC's proposal to examine the potential consulative impacts of 34 projects (with pending prelicense action at 19 sitest does not and cannot adequately address the potential complative adverse huperts of hydronower development throughout the basis. All proposed, constructed, and reasonably potential (regardless of herming status) hydroelectric projects should be included in the cumulative impact analysis. The is consistent with the definition of cumulative impacts contained in the l'ERC regulations. implementing NEPA.

2

78

Under this definition "cumulative impact" is the effect on the environment that results 1 79 from the incremental impact of the action when added to past, present, and luture actions regardless of the agency or person undertaking such actions. In our opinion, the Onio River cumulative impact study is not being conducted in accordance with this definition. FERC argues that "... increasing the size and scope of the study area. (additional licensed or exempted projects) ...would introduce sufficient additional uncertainty to make the results of the impact statement of limited value ... " (August 10, 1987, Scoping Document II, page 15). Contrary to that argument, we believe that the draft, as written, procludes meaningful analysis of potential or real adverse cumulative impacts on fish and wildlife resources in the Ohio River Basin from existing and proposed hydroelectric projects because it one not include such projects. Therefore, we believe that FERC should prepare and circulate for comment a revised draft statement including all licensed, exempted, and proposed projects in the Oldo River Basin. We have enclosed a list of additional projects which we believe must be included in any revised. environmental document.

The draft statement does not indicate which competing application would be licensed at [80 Tygert Hem, Montgomery Locks and Dam, New Cumberland Locks and Dam, Willow Island Locks and Dr.o, or Gallipolis Locks and Drin, although the proposed action is to license hydroelocing projects at these sites. The draft states that there are no major differences among these projects; therefore, no preference is indicated. In our opinion, major differences do exist. For example, one of the applicants for the Gallipolis project plans to install four turbines, two on each side of the river. The competing application only plans to install two turbines at one shutment of the dam. The effects of these projects on water currents and patterns would differ significantly, as would proposed fishing access and aquatic habitat requirements. Furthermore, recreational areas proposed at several sites on government-owned lands are of insufficient size to allow the development. FWS previously described major concerns with several of these projects during review of individual license applications. Unfortunately, FERC has not responded to those concerns. Therefore, the draft should be revised to show which competing application is likely to be licensed and what mitigation measures would be required.

78. See response to commont #32.

79. See response to comment #13.

80. See response to comment #34.

Ļ.

ALTERNATIVES ANALYSIS

Conservation and load management is briefly discused as a principal non-generating alternative to the proposed projects. However, the concept was dismissed by the conclusion that implementation of such measures has been adopted in many cases. Unit assumption is contrary to the conclusions drawn and recommendations made in the Department of linergy's (DOE) study of alternatives to the Davis Power Project extended that study concluded that implementation of any of the three conservation and load management scenarios would result in egnificantly lower costs than building a proposed 1,000 MW pumped storage facility. The DOE report also showed that implementation of conservation and head management could, in and of itself, preclude the need for 1,000 MW of peaking power. As such, a detailed analysis of conservation and load management is appropriate for the present study and by recommended by TWS. Further, the conservation and load management alternatives presently being used should be discussed in concert with all economically available alternatives.

3

81

82

83

Fin draft states that hydropower alternatives were developed to "...give equal consideration to power generation and environmental quality values." Contery to this statement, the data projecting dissolved oxygen (DO) content (Section 4 and Appendix B) clearly show degradation of water quality values with each of the four proposed hydropower alternatives. Specifically, the DO model for Alternative 3 (Project Operation to Meet Antidegradation Criterion) and Alternative 4 (Project Solveted to Minimize Impacts to All Target Resources) predicts DO reductions below Willow Island Locks and Dem for approximately 180 river miles (see Figures 4.4.1-1, 4.4.1-2, 4.3.1-3, etc.). FERC's second decision (Commission Order 464) to waive Section 401 water quality certification for 11 projects does not reflect equal consideration of power generation and environmental quality.

ENDANGERED SPECIES

The data contained in the draft are insufficient to demonstrate that adverse impacts will not occur to the pink mucket pearly mussel (Lompvills orbiculate), a histed endangered species. In fact, FWS believes adverse impacts to that species would occur. Our concerns are hased on the potential depressed DO levels resulting from the proposed hydropower projects. Additionally, FWS believes a significant reduction in numbers of finfish, as indicated, use also adversely affect the pink mucket pearly mussel since fish are an integral part of their life cycle requirements.

The endangered pink market pearly mussel occurs in the area better Gallipolis Lock and Lam and may also occur below Willow Island and Belleville Lacks and Dams. Section 7(c) of ESA requires the Federal agency proposing a major construction entivity significantly affecting the quality of the bunan environment to conduct and submit an essessment to determine the effects of the proposal on listed and proposed species. The biological assessment must be completed before any construction contracts are let. We const believe that we can adequately assess the efforts of proposed actions to the pink mocket pearly mussel without a complete assessment. When conducting a biological assessment, the following may be considered for inclusion:

1. The results of an on-site inspection of the area to be affected by the action to determine if fisted or proposed species are present or occur seriously.

81. See response to comment #35.

82. See response to comment #36.

83. See response to comment #38.

MA LOS D. Ceshell

The views of recognized experts on the species at issue.

A review of the literature and other information.

 An analysis of the effects of the action on the species hebitat, including consideration of the curvidative effects and the results of any related studies.

 $S_{\rm c}$ As subjects of ulternate actions considered by the Federatingency for the proposition.

After LERU has completed and reviewed the assessment, it has the responsibility to determine if the proposed action "uses affect" the pink mucket partly massel. If it determines that the project "may affect" a listed species, FEMU must request (orma) consultation from the appropriate field office of the FWS. When FEMU provides the biological assessment to the FWS. If should melude any other relevant information used. In reacting the requiring Section 7(d) of ESA inductioners the requirement that the Federat Agency or the applicant shall not make any irreversiste or irretrievable commitment of resources during the consultation period which in effect, would deny the formulation or implementation of reusonable and primerial laternatives regarding their actions on any endangered or interfered species.

MINERAL RESOURCES

On page 3-37 we helieve the document is deficient in its listing of mineral resources and gas resource-based industries. The list should include: creak oil and gas, steel, forcealloys, resource-based industries. The list should include: creak oil and gas, steel, forcealloys, including the survest from this proper contist of land impliment and utilization of mine quarry and pit strees for the proper contist of land impliment and utilization of mine quarry and pit strees that use (spoil) strees. Mineral deposits are apparently viewed only as convertent dutip sizes that include. Mineral deposits are apparently viewed only as convertent dutip sizes that include and petroleum product pipelines that may be affected by the vertous proposed construction and distribution liney.

We recommond that the draft include a southor caratineral recourses. The mineral orporats and or runneral-related to stations, including pipelines, partiment to each site or facility and its ancillary components should be asseriated, the impacts of the projects on them should be discussed, and mittrating measures should be described.

SUNLARY.

The draft does not privide internance coverance of the following issues which were identified in the woulding process as truct of primary noncerns water quality, and append species liver affect affect from patterns. How modifications turbing managed ish martably, recreational fight, and oreged spail alsopsail. The draft does not provide adequate recommendations in articles to which computing applications will be theread. Conservation for flow from any and indequately induces on the public defendance.

84. The intent of the discussion of resource-based Industries in Section 3.1.9.1 was to provide a general overview of the regional landscape. The issue of potential project impacts to mineral deposits and/or mineral-related facilities was not identified during the scoping process and staff, therefore, believes a detailed discussion of mineral resources is not within the scope of the document.

85. See response to comment #72.

5

Ms. Lois D. Cashelf.

I ERC claims to have complied with the Section 7 consultation process of 4.8A by providing FWS with a review copy of the daft statement. This does not constitute consultation which has yet to be initiated by FERC. Further, the FWS does not concurr with staff recommendations to postpone resolution of the previously identified issues regarding fish turbine mortality and fish passage until the post-heersing phase. Because of these concerns, and the exclusion of numerous basin projects from the analysis, we recommend that the draft be revised and recorculated. A final statement should not be released with the issues are resolved.

Unless these issues are adequately resolved, the FWS may recommend that this project be referred to the Council on Environmental Quality (CEQ) under Section 1504 of its regulations. The referral would be based on the precedent-setting nature of this type of committive impact study and the severity of potential water quality impacts. However, we wish to coordinate fully at the earliest possible time because a solution to our concerns can be implemented with a minimum of delay and could produce the measury for referral. Coordination can be initiated by contacting the Field Supervisor, U.S. 1 is hand wildlife Service, Suite 322, 315 South Alten Street, State College, Pennsylvania 156801, (814) 234-4800.

Sincercly,

Frue Manchard, Surector

ŝ.

86

ec: Mr. Dean Shumway (Rm. 2041(B) 86. See responses to comments #32, 33, 38, and 49.

.

87. See response to comment ₹76.

Attachment

fish and Wildlife Service's Specific Comments on the Upper Ohio River Draft Environmental Impact Statement

Section 2.1. pages 2.1 thru 2.27. Hydropower Generaling Alternatives. Commission staff has evaluated three dissolved oxygen criteria: (1) those proposed by the applicants. (2) State standards (5 milligrams per liter), and (3) an antidegradation criterion (6.5 milligrams per liter). Had the Service had the opportunity to participate in the scoping of dissolved oxygen alternatives, we would have requested that evaluation of pre-project dissolved oxygen concentrations as an alternative. We do not believe that staff's recommended alternative of maintaining 6.5 milligrams per liter of dissolved oxygen immediately downstream of any existing or proposed hydruelectric project will ensure protection and maintenance of downstream fish and wildlife resources. Therefore, the Service requests that the evaluation of pre-project dissolved exygen concentrations be included in a revised Draft Statement.

Page 2-25. Given that there is insufficient information to allow definition of existing dissolved oxygen conditions, the approach several resource agencies recommended during the scoping process should have been utilized. That approach, filling the dissolved oxygen data gaps prior to licensing, would eliminate many unknowns regarding basin-wide hydropower development.

<u>Page 2-30.</u> The Service does not agree that potential fish passage mortality can be considered a moderate or minor impact. In section 4.1.2.3 the Commission staff states that: (1) the results of existing turbine mortality studies are highly varied; (2) all entrainment studies completed in the basin are incomplete and inconclusive: (3) a firm basis for making mortality estimates awaits better results; and (4) much more work must be done before a fish protective device can be selected. An analysis of existing data does not lead to the conclusion of minor or moderate impact.

<u>Page 3-20. paragraph 2.</u> In addition to federally listed species, one fish species is being studied for possible future listing. The blue sucker (<u>iccleptus plongatus</u>), known to occur in the mainstem of the Ohio River in the immediate study area, is classified by the Service as Category 2. Category 2 comprises taxa for which information now in possession of the Service indicates that proposing to list as endangered or threatened is possibly appropriate, but for which conclusive data on biological vulnerability and threat are not currently available to support proposed rules. (Reference <u>federal Register</u>, Vol. 51, No. 106, Rules and Regulations, page 1994), third column, 4th paragraph.) The Commission is not required to confer with the Service on this species. However, the Commission is encouraged to confer informally when deemed appropriate to avoid jeopardy and to avoid potential economic loss through project modification if the species is later listed.

<u>Page 3-20, 3,1.6,2. Paragraph 3.</u> The smallmouth buffalo (<u>[ctiobus bubalus</u>] and the black rednorse (<u>Maxostoma duquesnei</u>) have been removed from the special concern category.

88

88. See responses to comments 439-71.

L

à

Page 3-20, 3.1.6.2. Paragraph 6. The builerfly (Plagipla lingulata) should be added.

Page 3-21/22. <u>Tables 3.1.6.1.</u> Federally protected species should be so noted. The federally listed species are:

Indiana bat	Huntis endalie
(hin {pink} mucket pearly musse)	lamosilis orbiculata
Peregrine falcon	alco hereorious
Eastern small-footed bat	Myotis subulatus leibi

In addition, the following species have been designated as candidate species and classified as (ategory 2,

These species from the list include:

Appalachian Bewick's wren	<u>Thryonanes bewickij altas</u>
Lastern woodrat	Beotoma floridana mauister
Yew England cottontail	Sylvilagus Trarsitionalis
Salamander mussel	Sumusonalas ambiaua
A CHARGEL EN2261	Sympsphalas ambigue

Page 3-67. Fisheries. This section should include a discussion of potential water quality (i.e., dissolved oxygen) degradation on shellfish. In Imlay's (1971) experiments, all species examined required 5 parts per million of dissolved oxygen for normal growth.

Section 4. Paragraphs 4.1.2, 4.2.2, and 4.3.2 should all contain an assessment of the potential impact of dissolved oxygen changes on shellfish.

Page 4-4. paragraph 4. Dissolved oragen depletion is probably the most significant issue dealt with in the Draft Statement. The dissolved oxygen model was designed to predict both existing and post-project dissolved oxygen concentrations. Considerable effort was obviously expended in its development and the subsequent analysis. The baseline data are, therefore, of paramount importance. As such, the empublished data upon which the model was based should be provided in a revised draft statement. The value of the model and predictions must be questioned if they were based on limited field data, such as the number of observations reported on page 8-7, Table B-1.

<u>Page 4-1. Assessment Methods</u>. Dissolved oxygen measurements were determined at a number of different flow rates and temperatures. Subsequently, damaeration constants "b" and dam-aeration coefficients "M" were developed by statistical analysis. The Draft Statement does not, however, indicate how many measurements were made at each flow level and/or whether the data used are representative of existing conditions. The dissolved oxygen data analyzed should be provided in table form so reviewers can make a reasoned decision as to its applicability. The scientific validity of model predictions of future conditions with or without the proposed developments should be reviewed by water quality experts in State and Federal agencies. figure 8-8 shows there were no data to calibrate the dissolved oxygen model between Ohio River miles 140 and 250 and that only two data points exist below (hio River mile 250. Without reliable and adequate ralibration data we further question the model's validity, particularly for all points below Ohio River mile 140.

It is unclear which of the connecting applications were used to develop the dissolved exygen model. As significant variations in size and operating mode occur, we presume the project's impacts on dissolved oxygen would vary. If peneric hydropower proposals were used in the model, then it would seem a simple matter to include all hydropower sites in the basin in the cumulative assessment.

<u>Page 4-20, paragraph 4.</u> No information is presently available regarding the direction or timing of fish movements in the Ohio River and its major tributaries. However, available literature indicates that some species mave downstream frequently. In fact, downstream regardings of walleye from Tygart Dam are most likely responsible for the establishment of walleye populations in the upper portions of the Monongahela River.

<u>Page 4:25, paragraphs 3 and 4.</u> The Commission staff states that existing entrainment field studies are incomplete and inconclusive and much more development work must be completed before fish protective or guidance devices can be selected. As such, licensing of any projects within the Ohio basin, in advance of the completion of studies necessary to determine impacts to fish and wildlife resources, would violate provisions of the Federal Power Act and would be contary to several court decisions.

<u>Page 4.25</u>, <u>Yulngrability</u> to <u>futrainment</u>. There is no information available to support the contention that fish would select a nonbazardous downstream route instead of a hazardous one. Fish normally follow current patterns unless habitat and cover exert significant influences. In the case of lock and dam hydropower development, habitat, cover, and current patterns will all increase the likelihood of entrainment. While passage alternatives nay be available, the majority of downstream migrants are expected to pass through the units unless the greatest percentage of river flow is through the dam gates.

<u>Page 4-78, paragraph 1.</u> While the Draft Statement is very qualitative, the Service has not seen any data regarding gamefish movements through locks and dans. The analysis done by Simmons (1987) indicates significant downstream movements for centrarchids and catfishes. Given the staff conclusion that the studies at Racine Dam are highly debatable and inconclusive, the postulation that only small numbers of gamefish are entrained and that those numbers are most likely insignificant, is questionable and should be addressed in a revised document.

<u>Page 4-39, paragraph 6.</u> The Service agrees with the staff conclusion that damage to gamefish from turbine passage will be greater than 10 percent and that the loss could be significant.

J-3)

Page 4-32. Fish Diversion and Protection Devices. Commission staff has determined that there is "... little experience with species most affected by entrainment ... "; that "No applicant has carried design plans to sufficient detail to evaluate their propriat fully ... "; and that "... more work at developing and evaluating experimental prototypes will be necessary.". ticensing of basin projects without adequate studies and mitigation in or opinion would violate provisions of the Federal Power Act.

<u>Page 4-33, paragraph 3.</u> Compensation is used to miligate adverse impacts such as fish passage mortality only after adverse impacts are avoided and/or minimized. Compensation means full replacement of project induced losses with resources of equivalent biological value. If used, compensation must remain the responsibility of the licensee to implement and maintain during the project life. The Service will only endorse a compensation plan after adverse impacts are avoided and/or minimized and one that: (1) provides for full replacement; (2) will be incorporated as an enformable license condition; and, (3) assures that mitigative measures are the responsibility of the licensee.

As turbine induced mortality has not jet been defined, a properly designed and conducted study is essential prior to developing a mitigation plan. Without an adequate impact evaluation, the adequacy and effectiveness of mitigation will remain undetermined.

<u>Page 4-47. Section 4.1.3.3.</u> Loss of recreational fishing opportunities during concurrent construction of basin projects is identified as an impact for which mitigation has not yet been resolved. Resolution of this issue should be achieved prior to licensing and after consultation with the Service and State wildlife agencies.

<u>Page 4.49. Section 4.1.3.6.</u> This section identified numerous significant adverse impacts to water quality and finfish should projects be licensed as originally proposed. Based on the data previously submitted, the Service concurs that these impacts must be avoided if possible; clearly, licensing as originally proposed would be environmentally unacceptable.

<u>Page 4-52. Section 4, 1.6.2.</u> Endangered/Threatened Species, paragraph 1. It should be added that any habitat disturbance below Wallow Island Lock and Dam, Belleville Lock and Dam, and Gallipolis Lock and Dam will also require consultation with the Service in accordance with section 7 of the Endangered Species Act. Reference our letter to the Commission dated September 25, 1987, and the Graft Statement, pages 3-19 and 3-20.

<u>Page 4-71. Section 5.1.5 8.</u> A response to the previous Service and State wildlife agency recommendations to utilize clean rock spoil to construct fishery habitat structures should be provided. Irplementation of that recommendation would offset project impacts to fishery habitat and recreational use while likely reducing developers' spoil-disposal costs. 40

5

Page 4-67, Section 4.7. Departmental comments are previded pursuant to the National Environmental Policy Art and do not represent comments pursuant to the Fish and Wildlife Coordination Act (48 Stat. 40), as amended: 16 U.S.C. 66i, et seq.) or section 7 of the Endangured Species Act of 1973, as amended (16 U.S.C. 1531 et seq.). Comments under section 2(b) of the Fish and Wildlife Coordination Act are provided when draft licenses are provided for each project. Comments under section 7 of the Endangered Species Act will be provided when the Commission initiates the consultation process.

<u>Page 4-87. Section 4.7. paragraph 4.</u> A copy of the Braft Statement does not constitute consultation under section 7 of the Endangered Species Act. The consultation process is separate and has not yet been initiated by the Commission.

The Draft Statement lacks adequate information (site specific location, design measures to minimize harm) to evaluate section 10 and 404 permits from the Corps of Engineers or possible section 402 National Pollution Discharge Elimination System permits. Accordingly, our comments do not preclude separate evaluation and comments by the Service when reviewing the permit applications. The Service may concur, with or without stipulations, or recommend denial dopending on effects to fish and wildlife resources.

When appropriate site-specific information is available, the Service would be pleased to coordinate with the Commission and/or licensees to preclude delay and to insure that any permit stipulations or conditions are understood and included in revised draft and final statement.

<u>Page 5.8.</u> paragraph Z_{\pm} While mitigation could reduce impacts of concurrent construction to recreational fishing, site-specific ritigative measures are not identified. The National Environmental Policy Act requires documentation to explain exactly how measures will mitigate project impacts.

Page 5-16, Section 5.1.4. Given that fish mortality concerns are not resolved by licensing under any of the alternatives considered. License issuance prior to resolution of those issues would violate provisions of the Federal Power Act and the National Environmental Policy Act.

Page 5-23, paragraph 2. The Service feels that there are significant differences between competing license applications and refers staff to project-specific Departmental letters resarding the proposed Tygart, New Eumberland, Willow Island, and Gallipolis projects.

Page 5-25. Basin-wide Recommendations. Prior Departmental and Service recommendations requested that the Service be identified as a coordinating agency in license articles dealing with fish, widlife, and public uses thereof, and spoil disposal, water quality, and mitigation of project-related effects. In accordance with provisions of the National Invironmental Policy Act, the Fish and Wildlife Coordination Act, the

federal Power Act and the Electric Consumers Protection Act, the Service and appropriate State fish and wildlife agencies must be specifically identified in each such article. The Council on Environmental Quality Regulations for implementing the National Environmental Policy Act list the Service as having jurisdiction by law and/or special expertise on these issues.

Page 5.77, Item 3. Although monitoring fish populations may provide some insight into long-term impacts, an indirect approach is expected to be much less successful than a direct one. Numerous variables affect the results of even direct sampling and analysis methodologies. Indirect approaches involve many variables, such as catch, harvest, recruitment rates, weather, river condition, water quality, habitat, emigration, immigration, etc. A direct approach of sampling and analysis must be developed if successful turbine mortality studies are to be completed.

Page 5-22. Item 4. and Page 5-28, item 5. The proposals for the agencies and some 15 applicants to develop joint plans to study fish entrainment and turbine-induced mortality, and to develop a bioengineering test facility, are commendable. However, the Service is concerned that coordination, administration, and implementation of such large scale undertakings, with as yet undefined financing and lead responsibility, would be difficult. If the Commission elects to pursue this endeavor, then one party to represent all applicants should be appointed and the Commission should assess equilable fees to each developer to fund the effort. The Service also recommends the work be contracted through an independent organization.

Further, reliance on post-licensing studies to determine the impacts and to develop adequate mitigation measures appears to violate the National Environmental Phicy Act, which requires consideration of environmental impacts hefore licensing decisions are made.

Page 5-29, Section 5.4.2.5, 1. Endangered and Threatened Species. Developers at Willow Island Lock and Dam must also consult with the Service through the Commission and the State resource agencies. Reference our letter dated September 28, 1987, and the Draft Statement, pages 3-19 and 3-20.

<u>Page B.2. Section B.2.1.3.1.</u> Bata and figures showing the number of dissolved oxygen ramples taken at each river flow, by location, date, time. temperature and climatic conditions are needed to enable reviewers to place the model into perspective with existing conditions.

<u>Fage 5.3. paragraph 5 and Table 5.1.</u> The statement that dams with Towar aeration coefficients "H" and nore negative values of "b" aerate better is inconsistent with text in Section 4 and Table 4.1.1.1.

<u>Page P-16. Figure R-R, and associated text.</u> The Draft Statement should explain why the movel predicts dissolved oxygen reductions of 0.7 milligrams between Obio River riles 160 and 240 but no reduction for the 40 miles below Racine Car; and a dissolved oxygen increase below Gallipulis Dam, despite

the fact that Gallipolis is reported to be a poer arrator. Further, since both actual data and model predictions (see Figure B-B, 4.1.1.4, and 4.1.2.3) show a dissolved maygen surge at Gallipolis, it is not consistent that Gallipolis is considered a poor aorator.

~

•

It should also be noted that Racine apparently adds no significant dissolved exygen surge. (As a licensed project, Racine was not included in the complative analysis). Inclusion of Racine may have resulted in recommendations for spill flows which would significantly improve dissolved exygen in the lower river. Pagu B.27, jaragraph 2. See our General Converts regarding inclusion of all basin projects in the curvelative assessment.

.

Literature Cited

.

•

- JmRay, M.J. 1971. Bibastay tests with rayady. In Jorganson and Sharp (1971, pp 38 41).
- Stamons, L.L. 1987. Fish entrainment studies for the Millville Hydro Station, FEPC Project # 2343, report from finergy and Environmental Nanagement, Inc., to the Potomac Edison Company.
- U.S. Department of Energy, 1980. Draft. A study of alternatives to the Davis (Vest Virginia) pumped storate project. Volumes 1, 11, and 111. Printed under contract by ICF, Inc., and ESEL. Inc. U.S. Government Printing Office, Waskington, D.C.

.

FISH ASD WILERING STRATE F Recommon anot

ACCUTIONAL FRODECTS TO BE INTERFAD IN THE CURULATIVE IMPACT ASSESSMENT OF THE ONED REVER BASEN

Allegheny River Subhasin (Penusylvania)

Allegheny River Lock and Dam No. 5 Project (FERC No. 3671)* Allegheny River Lock and Dam No. 6 Project (FERC No. 3494) Allegheny River Lock and Dam No. 8 Project (FERC No. 3021) Allegheny River Lock and Dam No. 9 Project (FERC No. 3021) Pinely Project/Clarion River (FERC No. 309) East Branch Project/Clarion River (FERC No. 3292) Loyalhanna Project/Clarion River (FERC No. 3292) Conomaugh Project/Kiskiminetas River (FERC No. 3203) Kinzua Project/Kiskiminetas River (FERC No. 3203) Kinzua Project/Allegheny River (FERC No. 3203)

Monongahela River Subbasin (Haryland, West Virginia and Pennsvivania)

Monongahela River Lock and Dam No. 2 Project (FERC No. 3973 et al.) Monongahela River Lock and Dam No. 3 Project (FERC No. 3752 et al.) Morgantown Lock and Dam Project (FERC No. 9949) Lake Lynn Project/Cheat River (FERC No. 2459) Deep Creek Lake Project/Youghloghony River (FERC No. 2370) Youghlogheny Dam Project/Youghloghony (FERC No. 3623)

Beaver River Subbasin (Ohio and Pennsylvania)

Townsend Project/Beaver River (FERC No. 3451) Patterson Project/Beaver River (FERC No. 3510) Eastwale Project/Beaver River (FERC No. 3510) Shonango Project/Shenango River (FERC No. 4462 et al.) Pymatuning Project/Shenango River (FERC No. 4462 et al.) Lake Arthur/Huddy Creek Ellwood City Project/Compoundesing Creek (FERC No. 7293) Mosquite Lake Project/Mosquite Creek (FERC No. 7293) Mosquite Lake Project/Mahaning River/Ohio Berlin Lake Project/Mahaning River/Ohio M.J. Klewan Project/Mahaning River/Ohio (FERC No. 9972) Meander Reservoir Project/Mapander Creek/Ohio

Muskingum River Subbasin (Ohio)

Devola Lock and Dam No. 2 (F[RC No. 7751) Muskingun Lock and Dam No. 3 (F[RC No. 7751) Beverly Lock and Dam No. 4 (FRC No. 7234) Luke Chute Lock and Dam No. 5 (FERC No. 7261) Stockport Lock and Dam No. 6 (FERC No. 7262) McConnelsville Lock and Dam No. 7 (FERC No. 7262) Rockeby Lock and Dam No. 8 (FERC No. 7222) Philo Lock and Dam No. 9 (FERC No. 7221) Huskingum Lock and Dam No. 10 (FERC No. 6553) Efficiency and Dar No. 11 (FLFC No. 7220) Diffic Lake (FRC No. 8268) Million Cam (FERC No. 9972) Pheasant Hill (FERL No. 8270) Wills Creek (FERC No. 8269)

Ranawha River Subbasin (Virginia and West Virginia)

Winfield Lock and Dam Project (FERC No. 1290) Harmot Lock and Dam Project (FERC No. 1175) London Lock and Dam Project (FERC No. 1175) Sutton Project/Elk River/ (FERC No. 3344) Summersville Fregeri/Gauley Byver (FERC No. 10226) Claytor Lake Project, (FEPC 739)

Tappecanne River Subbasin (Indiana)

Lake Shaeffer Project Take Engeman Project

Oblo River Mainstem (Pennsylvania, Obio, West Virginia, Indiana, and Illinois

Hannibal Locks and Dat Project (FERC No. 3206) Racine Locks and Dat Project (FERC No. 2570) Greenup Lock and Dat Project (FERC No. 2614) Capt. Anthony Meloshi Locks and Dat Project (FERC No. 2739) Markland (ocks and Dat Project (FERC No. 2211) McAlpine Locks and Dat Project (FERC No. 289) Cannelton Locks and Dat Project Newburgh Locks and Dat Project Uniontown Locks and Dat Project Smithland Locks and Dat Project Locks and Dat Project (FERC No. 6641) Locks and Dat No. 52 Project Locks and Dat No. 52 Project

"Where known the FEPC No. has been included.

.

.

.

	ł
11	HR. TALLJ: If not, at this time 1
1 2	would like to ask if anybody has an oral statement
13	that they would like to provide to us. Does
14	anybody have an oral statement?
15	MR. CLOWER: 1 30.
15	MR. TAYLOR; Your?
17	MR. CLOHER: I'm Chris Clower,
• 18	C-1-c-w-e-r, with the U. S. Fish and Wildlife
19	SATV 100.
20	The Service appreciates the opportunity to
21	comment on the draft EIS and express our interest.
2 2	Chese comments are generic is nature and do not
23	represent the Service or Interior's official

.

· ·

1 response. The Service departmental comments are 2 forthcoming.

The Federal Energy Regulatory Commission ÷ has indicated that the final EIS will serve as the 4 licensing document, and the first response of the 5 Service and state wildlife agencies will constitute 6 issue resolution under Section 10-J of ECWA. I do 7 not believe this approach reflects the spirit or 3 intent of the Federal Power Act as smended by ECWA. 9 National Environmental Policy Act, official 10 11 Wildlife Coordination Act, and other federal statutes and policies. I feel that all previous 12 and future comments must be considered and that 13 additional dialogue between the agencies is 14 necessary if conflicts arise. 15

I remain concerned with the Commission's 16 osission of numerous ossin projects. As stated in 17 the document, the value of the objective function 13 of any particular site is a function of the values 19 of the decision variables at all sites upstream. I 20 believe a cumulative system-wide mulling unalysis 21 is required. Further, since insufficient 22 information exists to find existing DD conditions, 23

89. See response to comment #31.

89

90

90. Seeresponses to comments #32, 33, and 39.

• •

1	t believe bu data gaps should be filled prior to			
2	ticensing.	ŧ		
۲	I believe the Commission's analysis of the	91	ç	H. See ri
4	principal and non-generating alternative should be			
5	greatly expanded. Conservation and load management			
6	was briefly discussed but summarily dismissed.			
7	reportently parameers whereastron of and measures			
đ	had been adopted in many cases. Those cases,			
9	however, were not identified.	٢		
10	The Department of Energy's own study of			
11	alternatives to the Davis Power Project in 1990			
1 2	includes the opposite and, in fact, recommended			
lj	conservation of load management scenarios be			
14	implemented. The staff acknowledged that the			
15	results of previous terminal mortality studies are			
16	inconclusive and that a firm basis for making			
17	mortality astimutes awaits batter results.			
19	I cannot agree that postponing studies	92	92	, See re
15	until after construction is the post solution to			
20	develop mitigating measures. Project impact should			
21	be identified prior to licensing and unavoidable			
22	impacts mitigated during construction.	3		
23	i do not believe the fish population			

1. See response to comment #35.

.

- .
- 92. See response to comments #49 and 72

.

.

1	monitoring as described is the document will	93
2	provide much insight into long-term impacts on the	
3	fish populations. Too many variables affect the	
4	remults of indirect analysis approaches. A direct	
5	sample methodology must be developed if terminal	
ե	mortality studies are to be successful,	ĺ
7	In addition to these issues, I do not	94
6	believe the documents provided adequate coverage to	
9	water quality of endangered apecies, altered flow	ĺ
10	patterns and regreational fishing opportunities.	
11	I also believe if the EIS is to serve as a	1 95
12	licensing document of the project, specific	
13	articles and recornendations should be included to	
14	protect fish and wildlife resources,	
15	Thank you.	
16	AR. TAYLOR: Thank you, Mr. Clower.	
1 7	Does anybody size have an oral statement	
មេ	that they would like to make at this time?	
19	(No response.)	
20	AR. TAYLOR: Are there any questions	
21	you would like to ask at chis time?	
22	(No response.)	
23	NR. TAYLOR: Well, before we finish	

93. See responses of comments #207, 212, and 234.

• • •

94. Your comment is noted.

95. Staff has provided specific recommendation in the FEIS to protect fish and wildlife resources. See the rcommendations in Sections 5.4.2 and 5.4.3 on these protection measures.

J-48

up here this morning. I would like to remind you 1 that we will break here and coat pack after iunch 2 in the event that anyoody has anything they would 3 like to say it that time. 4 5 So with that, I would like to conclude this public desting for the graft environmental 6 1 ispact statement on hydroelectric development in the Upper unio River Basin. Thank you very much 8 for your attendance. • 10 Like I said, the idea was to have the 11 meeting from ten until noon and break for lunch and 12 come back at one of clock and we will run from one p'clock to three o'clock. 13 14 dR. JICH: I would like you to explain why you are having a public meeting with a 15 iυ court reporter. You're definitely not -- that is, you have a choice, you didn't have to do this. 17 łв MR. TAYLOR: Well, we felt like we 19 needed to have a public meeting to allow the public. 20 agencies and individuals, an opportunity to provide comments. We certainly think that these comments 21 at this time on the draft environmental impact 22 statement should become part of the public record 23

J-49

L	and, therefore, we have a stanographer have to
2	provide us with a transcript, which we will also
3	use those contents and information in preparing the
4	final environmental impact statement.
5	MR. ABRAMS: So we don't have to take
6	s lot of notes.
7	MR. TAYLOR: 30 with that, are there
-1	any further questions7
9	(No response.)
10	MR. TAYLOR: No further questions, we
1 1	will break were and we back at one of clock. Thank
12	you again.
	1

.
249 TANK UNITED STATES ENVIRONMENTAL PROTECTION AGENCY



REGION III 841 Chastnut Building Philedelphia: Panayiyania 19107

JUL 2 1 1998

Ms. Lois D. Cashell, Acting Secretary Federal Energy Regulatory Localistica 825 No. Captrol. St., N.E. Washingron, D.C. (2022)

Dear Mey Cashells Rev FERC Dacker No. EL85-19-114

In aboutd with NEFA and Section 309 of the Clean Air Act, EFA has down 98 pleted its review of the Graft ELS, entitled, Hydroelectric Development in the Opper Ohio River Basin. The number of proposed projects is very large and the Federal Energy Regulatory Consission (EERC) has grouped several proposals under one usbrells Draft ELS.

Our review has concentrated upon analyses of alternatives and the destription of the affected environment. We have identified some areas where we believe the EIS could be substantially improved and others where we feel FERC should have included additional infurnation and adalyses. These press are examples of where we believe FERC revid substantially improve the providence:

- * Alternatives and alternatives analyses
- ² Weilands and righting description.
- * Hodeling for dissolved oxygen
- ⁵ Impacts to figh from impingement 4 entrainment.
- 1. Toxics in water and sediments

For reasons spelled out in the attached tochnical commute, we believe FERC should rensider either a supplemental Draft EIS to rover additional information or tiering as each of the proposed projects is considered for design. Based upon these findings, FPA is rating the Draft EIS E0-2 (Environmental Objections: Insuificient information). We believe FLRC should investigate additional alternative which, based upon our analysis, may only not to be less maximum cally harmful and, perhaps, less couly.

If you have any questions, please contact Bob Davis on FIS/Comm. 215: 597-8327.

Sincetely,

Defirey B. Alper, Chier NEPA Compliance Section

PERCILICANSING AUCT ISOU



 Your comment has been holed. Responses to your specific concerns are addressed in response to comments #97-105.

Techniscul Chrynnis

Introduct Sent

Efficienties at the fraft Lis us strentified versal areas where the furformation presented is wither the abreviated to draw purchastons of minimal isymens an winder can be developed arreading there, up additional minematives in our optimient, the information may be action additional minematives an our optimient, the information may be action observation, it should be second disated by detailed and site it is of use as grood observation. It should be suppresented by detailed and site specific information.

In the matter of additional alternatives, the infinteation prevented in the document was the specify for the alternative so be developed to its (black. It would be appropriate for FFRU to available the compositions but w below and to meet with 078 start are necessary to unity our available. the review also showed that the disactived experimiteration should be further developed. The modeling has been based upon frantfichers basedier haforcation and upon two many assumptions to assume and pressul in this case. Significantly joint additional information and pressul in either in a supplemental fract FTS or an information and the fiber in a for review as the prepased projects are brought into specific design phase.

The comprise are organized in the fullowing order:

- Alternatives and alternatives analysis.
- * Matlands and riparian description,
- Dissolved oxygen and nodelany.
- 🕺 impacts to lish itom impicgement à entrainzent;
- Trustes in water and subjective.

Atternutives and atternetives analysis:

Based on our ovelet, we question some of the information in the FIS in tanded for one in developing additional mircrookives. This futuration is found in Section 3, affected Saviruncent, and is used in ronjourtion with the Alternatives of Section 2 and the Staff LowClustens of Section 5 to develop the new listes of Transforms. The four alterparives investigated in the fraft SIS are: 1] operation proposed by the applicant, 2) operation to pathead disperied usygen (3.0.) at the 5.0 mg/l seneration: 3) operation of the fidtoefectul plants of maintain a 2.0 at 5.5 mg/l; and 4) operation to cancest a 0.0. of 0.5 mg/l dis well as to provide proterison of other terperiments. We tocommend the plants to least two other sidenines on he romaidmend: 1) operation of the plants to destrue othering disculved copentenders of 0) operation to the plants to destrue other interived copententions so that the reaches of the plants to river where 0.0, concentrations greater than 6.5 mg/l would be maintained. 2) retrieval or power meets from other sources that are already in exterence.

The first suggestion would involve some additional modeling work to predict operational modes that would neer both the power as well as the environmental largets. However, if appears to us that the needs could be not brough specifying where the needs are prographically located, allocation of power tion concentionally available sources, and operating the proposed power stations selectively so that objectives of power preduction and environ remain quality are not. These would have to be matilized with those reaches where 0.5, is eighter than the 5.5 opt 0.6. Then the whole power couply sys fracmates and order Salament to arhieve both power needs and environmental quality objectives.

97

98

The second suggestion would requite some additional testarch on the part of the applicant. One passible approach is described below for consideration in completing the NCPA process. On page 3.3, curteen land uses are described as follows: 61% forested, 20% approximations, 7% orban, and 2% charge on page 3.22 ff, socio/completesources, it is explained that subregion HI, the Alleghnon County, Dennsylvania area, represents the greatest population boncentration (Taula 3.1.7-1). All economic activities in the subregion are also greater than in other regions (Table 5.1.7-1). It is assumed that the greatest power demand would coincide with these activities. Section 2, however, does not factor these into the alternative analysis, heading to our opinion that doother alternative is possible. Bouweer, additional information is required to complete the section.

On pro 2-27if the Draft EIS status, without sufficient backup, that it is preferrable to use hydropower to ruplace "... generation from ... less efficient steam units." The document does not like these units or their power-production life expectator, although it appears that roughly 30 are in operation. These should be listed. In addition, it should specify where the greatest dwand occurs to assure that the proposed stations scattered throughout the study area would be sufficient to fulfill the dasgod.

On page 2-13ff and f-10ff, it is clear that FERC does not believe that afforw or couply can be supplied from the BC plants cureatly in operation. bowever, the Draft EIS should have included incremental additional loads from the several plants in operation. For example, if only 20 mm were to supplied by 20 of the most efficient plants in operation the domain would be satisfied. While the additional loads would require more coal and other resources than are cureatly required, it seems to around that economies of scale would at least partially affect the loss.

This same approads would be used to develop a subsitematize, namely, the re-distribution of power from within the service area grid, including power acts outside the rervice area (i.e., the Northeast grid). This, bus, is distinged with very little supporting data, yet deserves to depth study which could be based open information available from FERU and other sources. If that supply is proved or over peak, then SERU should include the information in its domains then.

- See response to comment #39. Staff believes that maintenance of "preproject" D0 concentrations would preclude development of projects at dams that aerate well.
- 98. The FEIS states that it would be in the public interest and useful to conserve non-renewable fossil fuels and to reduce atmospheric pollution; displacing steam generation with hydropower will accomplish both. The displacement of inefficient steam generation by hydroelectric generation improves the cost-effectiveness of the displacement.

A review of the April 1988 DOE Code 1E-411 Reports submitted by ECAR, MAAC and by SERE for its VACAR sub region (see Sect. 1.2 for definitions of abbreviations) shows that, as of January 1, 1988, the existing generating resources of utilities in the MAAC Council Area included 36,211 megawatts of fossil-fueled capacity; existing resources in the ECAR Council Area included 86,328 megawatts of fossil-fueled capacity; and the VACAR Sub-region had 12,700 megawatts of existing fossil-fueled capacity.

With more than 135,000 megawatts of existing fossil-fueled generating capacity in these three Reliability Council Areas, the identification of specific steam plants which can be cost-effectively off-loaded by 400 megawatts of hydroelectric capacity is unwarranted.

In the third of the four paragraphs being addressed, it is apparently assumed that FERC staff wishes to establish a need for the 400 megawatts of hydropower, over and above the existing and planned additions of generating capacity. This is not the case. The 1988 IE-411 Reports prepared by ECAR, MAAC and by SERC for its VACAR Sub-Region indicate that existing generating resources and projected capacity additions are sufficient to meet projected load growths and maintain acceptable capacity reserve margins. The objective of staff is to establish the substantial usefulness of 400 megawatts of hydropower by reducing the consumption of non-renewable fossil fuels; by reducing atmospheric pollution, in a nation deeply concerned about acid raim and the greenhouse effect; and by possible long-term reductions in energy production costs.

The above-cited IE-411 reports show that utilities in the ECAR Council Area plan to add 4,813 megawatts of additional fossil-fueled generating capacity during the next ten years to meet projected load growths. Utilities in the MAAC Council Area plan to add 3,950 megawatts of fossil-fueled capacity during the same period and for the same purpose. The VACAR Sub-region of SERC plans to add 993 megawatts of fossil-fueled capacity. The 400 megawatts of proposed hydroelectric capacity can obviate the addition of a small fraction of these planned fossil-fueled ditions and effect accompanying reductions in fossil fuel consumption. Possible long-term reductions in electric emergy production cests should also not be ignored.

The development of possible "subalternatives", such as those proposed in the fourth paragraph being addressed, are being continuously considered by utility planning engineers. Economy purchases and transfers of capacity and energy are implemented daily and by longterm contracts to avoid the necessity of adding new generating facilities. The impacts of such "subalternatives" are accounted for in the IE-411 reports. Such procedures are also of great value during system emergencies, as well.

Wetlands and Riporian Swseriptions:

On page 3-46, a description is presented of his information on setlands and siparian areas was parhered. Apparently, observations from a bost servused as the sole source of riparian vegetation information. While such taken mation is usually elequate for renomalissance investigations, it should not replace detailed studies for specific planning and decision-making purposes. It is further stated on the same page shat detailed surveys were made at 15 sites, but the citation in Table 3.3.51 ("Vegetation surveys wire made at 15 sites, but the citation in Table 3.3.51 ("Vegetation survey sites ...") is unpublished information. A call to the Pittsburgh COE indicates that shake casual observations were made mainly for informational purposes on davigational charts. Detailed studies were made at only 14 pites while at least a dozen more power sites are under study. In addition, it does not appear that the observations led to detailed analyses at other sites where power plauts are not under consideration but purposes not our appear.

Evaluations appear to be based upon this reconnaissance information: on page 4.90 it is started that "(there would be unavoidable loss of rightly and werland segnation ...] at three locations that FERC, justifiably, believes should be excluded from construction plans (pp 5.23 5.24). While we agree with this, it is important to acknowledge that extending such recensissance information to other sites may be inappropriate and should be supplemented by detailed studies prior to dusign. The extrapolation of this information may not cover bone ateas that may be prove to damage from construction and operation.

We are gratified that watland and riparian areas are to be monitored for the first 5 years of the uperation of all proposed facilities. However, we believe decisions to build should be held to abeyonce until full and detailed information regarding these valuable resources is collected.

Modeling and Dissolved Oxygen.

Appendix 3 describes the modeling effort that is to be used as a predictive tool in the decisions regarding the proposed projects and their impacts upon D.C. The dam actuation effort appears to be good and therefore useful in predicting D.C. increases due to the daws. However, the modeling effort for areas upstorem of the daws is based upon assumptions that we believe limit the usefulness of the model as \bullet predictive tool.

The pirameters of importance in D.O. depression and maintenance are blochemical oxygen demand (CBOD), sediment oxygen demand (SOD), and altrogenous oxygen demand (NOD). From our request, it appears that our rate of decay has been applied to all three parameters. In our opinion, SOD and NOD decand rates vary greatly from that for CBOD. Furthermore, SOD is suspected to be of much greater significance in D.O. depression upstream from the date than is represented by the codels and therefore should be considered separately. The decay rates (E_1) are usually based upon actual scream chasuremonts, but we were usable to see where the decay and repression rates were determined. As a combined tate, especially with SOD, the normal procedure for estimating a decay rate would be most difficult to use.

99. Text has been modified. Comments noted.

00

100

Staff used the best available information to assess the impacts of the proposed projects on wetlands. Very limited wetlands research and detailed surveys have been conducted in this region. A recombendation for detailed monitoring is made (Section 5.4.2.4). Staff is concerned with the impacts caused by the loss of wetlands on recreational use, fish and wildlife habitat, and the potential changes or destruction of the wetland habitat itself. Wetlands are scarce in this region and as a valuable resource should be preserved and maintaimed.

100. The BOD loads included in the model at reaches without wastewater dischargers simulate the following: (1) loads from major wastewater dischargers that are not required by the states to monitor efficient BOD concentrations, so no data are available; (2) loads from small, unpermitted, or filegal dischargers, which often have high BOD concentrations; (3) non-point sources of BOD; and (4) sediment BOD. ORSANCO monitoring data show that summer BOD concentrations typically range between 1 and 3 mg/L of ultimate BOD (including five-day BOD and nitrogenous BOD).

Staff conducted a search of National Pollytant Discharge Elimination System files, and EPA's Permit Compliance System was used to quantify 800 loads from dischargers required to menitor 800 and mitrogen. A simple mass balance shows, however, that the major dischargers where effluent BOD concentrations are monitored do not contribute enough BOD to account for the measured in-stream concentrations. For example, a typical discharge from ALCOSAN, the largest municipal discharger in the system, raises BOD concentrations only about 0.6 mg/L when there is 20,000 cfs in the Ohio River. The states in the study area have not conducted sufficient numbers of waste load allocation studies to quantify 800 sources, loadings, and the related rate constants. As far as staff was able to determine, the BOD decay rate for ALCOSAN has never been measured. In the absence of data on these other sources of BOD, calibration of model input BOD loads and decay rates to approximate measured DO and BOD concentrations is necessary. The reaeration rate was determined using the O'Connor-Dobbins equation (Appendix B, Section B.2.1.2).

J-54

The Draft EIS (p B-9) corrected to the orderal to the accuracy of the assessment to identify or quartify individual sources of BDD. This cay be true, however, it is important to the accuracy of the model rationation and inait rate determination. Artificially adjusting loading rates to achieve calibration with the DD data results in questionable rates. Loadings were apparently adjusted to account for unknown sources of β DD and SDD. It apprears that rates were not adjusted to conditions. This process yields questionable rates for production runs. It is our opinion that the model has not here properly calibrated and should not be noted for predictive purposes without further cata analysis and proper calibration exercises.

In our (rview, we also noted (p B-4) that atgal uptake of EC was not in , claded in the PO calculations. The justification is based on the usuaption that algei knowle is minimal during the leve flow worths when DO levels are lowest. To our opinion, algol imparts vith he greatest during the period of interest, i.e., low flow conductors connections with high resperatures. If the preparents have information to the contrury, is should be included in the EIS. Algal growths, to our knowledge, are greatest during the warm worths of the year. They are suspected to have a great impact on EC maintenance and should be factored into the model as such. If insufficient data are available to model algal impacts, then a program should be developed to collect such information.

101

In further regard to flows and their relationship to DC, we note that an incendence flow of 70% is used for calibration. We believe that this is two high for accurate comparison with the 7016 and would prefer to see an exceedence flow for calibration is the range of 10 to 15% greater than the 7016 flow. This would lead to greater confidence in the nodel as a predictive tool during low flow design periods than we currently can share with the preparer.

Impacts to Fish From Incingement & Entrainment:

On page 4 P9 of the Draft EXS, FERC states that some of the fish passing [103 through the turbines would experience injury or mertality. The Draft states:

There are insufficient data to quantify extent of losses, and technology is insufficiently developed for Ohio River basin applications to require installation of directive devices for excluding entraioment. (post linense) donitoring is recommended to quantify the extent of these losses and to develop mitigation."

We support the recommendation that fish losses be considered and appropriate mitigation be developed. We do not support a considering and mitigation plan that would not be developed and implemented until after the installation of hydropower, as suggested on page 3-27. Fish workality/injury studies, as well as mitigation plans, should be designed and approved prior to the issuance of a hydropower lifense. The fish studies should be undertaken before the installation of hydropower to determine pre-project conditions. The studies should

- 101. Historic DO data in the Ohio River basin typically show little diurnal variation in concentrations, indicating liftle effect of algal production on DO. There has been noticeable algal growth in some parts of the basin during the summers of 1987 and 1988. Data collected by the hydropower applicants in the summer of 1987 were analyzed using the single transect method of Odum (Odum H.T. 1956. Primary production in flowing waters. 11mmol. Ocanog. 1:103-117) to determine If DO production and consumption by algae was significant. The analysis showed that bacterial respiration (BOD decay) far outweighed algae as a factor controlling DO concentrations. Because this analysis was unable to quantify significant algal production, algal effects were not included in the model. Data collected by the Corps at their electronic monitors during the summer of 1988 did show diurnal variation in DO concentrations, but the extremely low flows and high temperatures occurring when these data were collected are very atypical and cannot be used to represent the system.
- 102. The calibration data set from 1983 was selected because it was the most complete set of field measurements made during a recent low flow, low DG event. The flows that occurred during that period are higher than the 7010 flows, but 7010 flows are not used as a design flow in the FIS because many of the proposed hydropower projects would not operate at 7010 flows (as a result of minimum flows required for generation and the recommended spill flows). The calibration flows are reasonably close to the design flows used for most of the modeling analyses (referred to as summer moderate flows in Section 4). At the mouth of the Allegheny River the calibration flow was 5,600 cfs and the design flow was 2,200 cfs and the design flow was 5,800; at Gallipolis dam the calibration flow was 23,000 cfs and the design flow was 24,000.
- 103. See response to comment #49. Staff agrees that the sequential mitigation plan for fish protection should include those aspects that can reasonably be known prior to the measurement of entrainment and testing of prototype fish protection devices at specific, representative sites. Staff agrees that local fish population monitoring would need to commence before project construction and continue after operation. Staff considered trash racks as a potential fish protection device. Staff is familiar with that study and is aware of its limitations as well as its strengths.

be continued after hydrogoner generation has begun, if any adverse fishertes inpacts are detected, the already approved mitigation plans should be put for graction.

In developing the mitigation plane, we recommend that EERC consider the relative curits of a mother of mitigation tochologons. One bechuiges that we suppose is the use of trach cacks to reduce (ist passage through turblack. Siense (ind enclosed a copy of an article by Nettles and Gluss (1987) on the subject.

Texacs In Water and Sedimental

As stated on page 4-10 of the braft ELS, "failthough concentrations of oriatile compounds are relatively low in the Onio River system, any increase in concentration would be of significant concern because of the high texisity of these compounds." Biscussion of this statement is inadequate. We ask that the full-owing information be provided: a listing of these compounds found in the Onio River system, the pre-project concentrations of these compounds after installation of hydropower; the anticipated effect of these compounds after installation of hydropower; the anticipated effect of these compounds after installation of hydropower; the anticipated effect of these compounds after installation of hydropower; the anticipated effect of

1.04

Un page 5-31 of the Draft FIS, FERC recommends that hydropower developers [105 "... supple river notisents and bank suits that will be disturbed during construction, or by erosion during operation, to determine the presence of chearcal contamination." We reiterate the need for sodiment sampling and testing. If the sodiment is found to be understely or nearly pulluted, texts or hazardons, or if the naterial is to be disposed of its solitand area, the developer must obtain the proper predict prior to deadly and disposal. Sufficient rime should be allowed for the developer of plan and conduct the sampling and testing properly and for the test of proper disposal. This figurable should be incorporated into the FEC lineage. 104. Concentrations of toxic compounds in the Ohio River basin are monitored by ORSANCO. The discussion of ORSANCO monitoring results has been expanded in Section 3.1.3.

The only field data on volatilization of organic compounds at Ohio River Basin dams are from the Ashland oil spill in early 1988. The few data collected above and below Monongahela L&Ds 2 and 3 and New Cumberland dam were inconclusive. Chloroform may be a special concern because (1) it is the compound for which the most criteria exceedances occur in the Ohio River and (2) it is highly volatile and therefore likely to be removed by volatilization at the dams. Text has been modified to include a reference to chloroform in Section 4.1.12.

105. Text has been modified. As stated in recommendation 6 of Section 5.4.2.4, the developer must file a sediment and spoil testing and disposal plan 90 days prior to the scheduled start of any land disturbance activities. The plan must contain a timetable for testing and safe disposal of all contaminated material. This plan is to be developed in consultation with appropriate federal, state, and local agencies.

RATING SYSTEM CRIILRIA.

a. Bacing the Environmental Impact of the Action.

(1) In (tack of Objections). The review has not identified any potential environmental impacts reducting substantive changes to the preferred alternative. The review may have disclosed opportunities for application of mitigation measures that could be accomplished with no core than rinor changes to the proposed action.

(1) EC (Environmental Longerow). The review has identified environmental implies that should be evolution after to fully statest the environment. Corrective measures may require changes to the preferred altertmative or application of mitigation measures that can reduce the environmental impact.

(3) 50 (Environmental Objections). The review has identified signific. c environmental appaces that should be avoided in order to adequately protect the environment. Corrective peasures may require substantial changes.

to the preferred alterdative or gensideration of some other project alterbative (including the op action alternative or a new alternative). The basis for environmental objections can include situations:

 (a) Where an action has the potential to violate a matienal environmental standard;

(b) Where the Enders' againsy violates its own substantive environmental requirements that relate to FFA's areas of jurisdiction or expertise;

(c) Where there is a violation of in ZFA policy declaration;

(d) Where there are no applicable standards of where applicable standards will not be violated but there is potential for significant environmental degradation that could be corrected by project modification or other feasible alternatives; or

(e) Where proceeding with the proposed action would set a proceedent for future actions that collectively could result in significant epwironmental impacts.

(4) <u>IU (Environmentally Unsatisfactory)</u>. The review has identified adverse environmental impacts that are of sufficient depitude that the proposed action must not proceed as proposed. The basis for an environmentally unsatisfactory determination consists of identification of envirronmentally objectionable impacts as defined above and one or note of the following conditions:

(a) It is highly probable that a violation of national environmental soundards will occur;

(b) There are no applicable standards but the severity, duration, or geographical stope of the impacts associated with the proposed aption warrant special attention; pr

(c) The potential environmental impacts resulting from the proposed action are of national jupottane because of the threat to national environmental resources or polities or for some usher reason.

b. Adequacy of the Lanact Stargent.

(1) "(1. Kacoruate). The draft FIS afequately rely forth the environmental impaction of the preferred alternative and those of the alternatives reaconally available to the preprior of action. No fostiler subtries or dota collection is necessary, but the revision ray suggest the addition of clarifying language of information. (2) "2" (Insufficient Information). The dtaft ISS fore not contain sufficient Information to fully assess environmental impacts that should be avoided in order to fully protect the environment, or the reviews has identified new reasonably available alrentatives that are within the spectrum of alternatives analyzed in the draft IIS with could reduce the environmental impacts of the action. The inder under contained analyses, or discussion should be included in the Inst IIS.

(3) "9" (Inadequate). The drafe E13 does not adequately assess the potentially significant traventerial impacts of the action, or the reviewer has identified may reasonably available, alternatives that are cutside of the spectrum of alternativer analyzed in the infaft E15 which should be analyzed in order to reduce the potentiality significant evolutions are of such analyzed in inadequate information, data, analyzes, or discussions are of such analyzed in the they require full public review as a draft stages. This rating constittuces a finding that the draft fill available, review as not weet the purposes of MPA and/or the Section 30' deview, and thus cuts the formerly reviewed and made available for public comments in a supplemental or reviewed draft stafes.





Alexandra Say care References and a say

65 JUL 20

1945 6, 1949

Ms. Lot: 5. Students, Anting Segretary Federal Energy Regulation devices For-825 Murth Capiton St., 199 Washington, 199, Sagit

PER TILES THERE'S MODELERS THEILS

Dear Ma Cashells

The Dhie Department of Natural Potontoes (1200) has reviewed the Fyderfelectric [165] Development in the Super Oble Fiver Basin Brait Environmental Insert Statement and Dave the following corrects:

Their commands were generated by an inter-disciplinary review by consultation with the livision of Wildlin and other Divisions of the Department. The arcomments have been proposed under the authority of the Fish and Wildlife Coordination Art (14 U.S.C. 66) et reg.), the National Environmental Publicy Act and other applicable bass and regulations.

The ODER contours with the Federal Energy & milatery Derrespondents selection of Alternative 4 or the preferred elternative. Under this alternative, four priority, Allegheny L 6 D But 2. Allegheny 3, 6 D But 3, Montgumer, and Markingan 5, 8 D But 3 would be removed from the list of project and by remove from the list of provide any hydrogener, and the sense by Bit 3 and the removed from the list of provide any hydrogener. Alternative 4 would avoid all magnitude to the target remove data the sense of both habitat, represented and wetlands, rise The Private the target from the target to prove the sense of the habitat, represented to them and wetlands, rise the removing 15 projects should insure the enhancement of restrational facilities that would provide around to there and to there and provide around to the target to the target to the sense of restrational facilities that would provide around to the target to the target to the sense of the theorem of restrational facilities that would provide around to there the sense to the target the target to the

Bolland Royanse in Species

106. Comment noted. Based on information filed by the applicant at Allegheny L&D No. 3, staff's analysis in the FIIS concludes that development of hydropower at this site can occur in an acceptable environmental manner. Revisions to the text for this project has been made and the conclusions for staff's preferred alternative has been changed.

.

Mar wrai bri Gwranu 1 2039 6, Laek 1896 - Pri

Those you for the operatory to provide these comments. Showle you have any queribries plants prime and Mr. Dave bergham, having momental keriew Section of this office at (134) 265 date

Cincersie,

What had Charle to boom

And a start of the

dui linte Clearinghoose (Legter Lake, Chart Covbrues of Mildige Covbrues of Nildige John Parshall, Divierce of Wildige John Parshall, Divierce of Wildige Thirters Stock, Chart Name Campor, Chart Name Schere, Stock Londe Administrator Date of Courter Recreation Services Office of Courter Recreation Service College (Look, Thio FRA



Rate of Ohio Environmental Protection Agency

P O. Box 1049, 1800 Write/Mark Dr Columbus, Ohio 43266-0149

Richard F. Celeste

50 JZ -7 1<u>4 (+ 07</u>

· 11.

July 1, 1988

Ms. Lots D. Cashell Acting Director Federal Lnergy Regulatory Commission 825 North Cepitol Street, Ni Mashington, D.C. 20026

RE: FERC Docket Number EL85-19-114

Dear Ms. Cashell:

The Ohio EPA has received and reviewed the Braft Environmental Impact Statement (DELS) entitled Hydroelectric Development in the Upper Dhio River Gasin. These comments are limited to the water quality issues discussed in the DEIS.

The alternatives investigated in the DEES include (1) operating the hydroelectric plants as proposed by the license applicants, (2) operating the hydroelectric plants to maintain a minimum instream dissolved baygen (DO) concentration of 5.0 mg/l (3) operating the hydroelectric plants to maintain a minimum 00 concentration of 6.5 mg/l and (4) operating the hydroelectric plants to maintain a minimum DD concentration of 6.5 mg/l and provide protection for the other target resources identified in the DEES. The FERC staff recommends selection of alternative 4.

The Ohio EPA believes that a fifth alternative should be investigated; the operation of the hydroelectric projects to maintain existing OD concentrations which currently exceed the 6.5 mg/l minimum value identified in alternatives 3 and 4. We understand that insufficient data are currently available to adequately assess the OD systematide, but believe an investigation of this type it essential to the alternatives analysis and should be included in the ELS. To collect sufficient data to conduct the analysis, we recommend that it conduct pre-construction DD and temperature monitoring both upstream and downstream of proposed hydroelectric sites. Data collection should continue for a time period sufficiently long to define pre-development conditions. The duration of the data collection will vary depending on the climatic and hydrologic conditions which occur during the data collection under moderate to low-flow conditions.

The water qualify model designed for the DELS can not accurately predict post-construction instream DD concentrations because the model was not adequately calibrated to cristing conditions. More detailed comments on this topic are provided on the attached page.

-89071900>5

FERG - DOCKETED JUL 7 1968

108

107 107. See reponse to comment #39.

108. See responses to comments #112-121.

Ns. Lots Cashell July 3, 1988 Page - 2

The Dato EPA indicated in a letter sent to FERC on July 5, 1987 that the pussibility of increased concentrations of volatile compounds due to hydroelectric dev lopment was of concern to this agency. While the DEIS does acknowledge the significance of any increase in the concentrations of the compounds in the Data River, no significant discussion is included. Additional information, analysis and discussion should be added including estimates of volatilization rates, and predicted impacts on instream concentrations.

The Ohio EPA agrees with the assessment that mechanical and/or artificial aeration not be considered as an alternative to maintain instream BO concentrations. The technology has not been proven and it would be unwise to rely on such unproven methods to maintain water quality in the Ohio River at this time.

In summary, the Dhio EPA agrees with the target resources identified for assessment in the LES. However, insufficient data prevent the detailed analysis which is needed to accurately assess and predict impacts to water quality in the Dhio River caused by the construction of hydroelectric power plants. Additional data is needed for use in the water quality model calibration and the assessment of impacts on the valatilization of organic compounds prior to the selection of the operating guidelines for the hydroelectric power plants.

We appreciate the opportunity to provide these comments. If you have any questions regarding our comments please contact Ms. Colleen Grook et (614) 644-2856.

Sincerely.

Contra Tumer

Andrew Turner, Ph.B., P.E. Chief Division Mater Quality Numitoring and Assessment

AT/CSC

cc: Randy Bournique, Ohio EPA-DMQMA William Franz, USEPA-Region V Dave Gergman, Ohio DMR-Gutdoor Recreation Services John Marshall, Ohio DNR-Givision of Wildlife Peter Tennant, OBSANCO Paul Will, West Virginia DNR Carl Wilhelm, OEPA-Planning 109. See response to comment #104.

110. Comment noted.

109

111. Staff believes that the model used for the EIS is appropriate for analysis of impacts of chauges in dam aeration resulting from hydropower development. See responses to comments #112 through #121.

.

1

DETAILED COMMENTS ON THE WATER QUALITY MODELING PRESENTED IN DRAFT ENVIRONMENTAL IMPACT STATEMENT

- the model used assumes complete mixing occurs in the river. This is an unrealistic assumption for the Ohio River. Even under the low flow conditions s whated, the amount of background flow relative to discharger flow is extremely large and would preclude, under most conditions, complete mixing.
- 2) It is unclear how much data were incorporated into the above-dam deficit [13] versus below-dam deficit relationshing. Page 4.1 states that this information was collected primarily under the summer law flow conditions. Does the observed linear correlation hold under the moderate flow levels simulated?
- 3) Page 6-1 states "Benthic paygen demand was not modeled separately from other 80D sources, but the effects were included as part of the overall 800." It is not clear what this statement means or how the benthic demand was incorporated into the analysis. This discussion should be expanded.
- 4) It is indicated on page 8-3 that there were significant differences observed between the predicted and the observed 0.0. saturation values. This can have a significant effect on the calibration of the model and any impacts projected by the model.
- 5) There is no support for the statement on page 8-9 which says that algal 116 production is low when instream 0.0. concentrations are low. The reverse may be true which would indicate that algal productivity is an important factor which must be considered.
- 6) Page 8-13 indicates that the model calibration was accomplished by including "...the addition of substantial amounts of 800....". Unless the BOU inputs can be batter quantified, the inclusion of the additional loads for curve fitting purposes is inappropriate.
- 7) The discussion of model calibration on page 8-33 reports the use of B0D decay rates as low as 0.1 day⁻¹. The lowest B0D decay rate recommended in the USEPA Simplified Method Guidance is 0.2 day⁻¹. This rate is recommended for rivers downstream of factilities with advanced wastewater treatment which is generally not the case for the Ohio River. A higher rate would seem more appropriate.
- 8) The explanation of the differences between the predicted and observed [19] aeration at several of the simulated dams (page 8-13) is not adequate. If the measurements were made at the low flows as stated, then the equations should be able to simulate instream conditions without further adjustment. It is assumed that part of the data which were used in the dam aration many not be appropriate.

- 112. The water quality model calculates a cross-sectionally averaged (onedimensional) value of 00 and assumes that waste loads are uniformly mixed across the river instantly. While this assumption is known not to be completely true, it is reasonable for the analyses done for the EIS. Modeling of mixing zones below major wastewater discharges may show some differences in DO and BOD concentration across the channel, but the fact that most of the BOD in the model is from non-point sources (see response to comment #100) indicates that such differences are not expected to be large enough to be of biological concern.
- 113. Appendix B has been modified to include tables showing the dam aeration data used in the model (including DO concentrations and deficits, flow, and temperature, as available). The range of flows under which the data were collected generally overlaps the range of flows modeled.
- 114. Appendix 8 (Section 8.2.1.1) has been modified to clarify how benchic BOD is treated.
- 115. Differences between actual and modeled D0 saturation concentrations were discussed as a possible explanation of apparent supersaturation at some dams. There is no conclusive evidence that actual D0 saturation concentrations in the Ohio River basin do or do not vary from standard values used in modeling. The methods used for estimating D0 saturation concentrations are widely used in water quality models.
- 116. See response to comment #101.
- 117. See response to comment #100.
- 118. The BGO decay rates [k]) used in the model range between 0.1 and 0.2 days⁻¹. Neasured values of BGO decay rates that include benthic BOO for deep rivers are presented by USEPA (1985; reference in Appendix B). These values range from less than 0.1 to about 0.5. A low value of k1 is appropriate for this model because the major sources of BUO are not wastewater discharges.
- 119. The differences between measured and modeled dam aeration rates shown in Figure B-B are a result of the natural variability in aeration. Because the models for dam aeration were statistically based on all the available data, they cannot be expected to exactly reproduce aeration rates for any individual measurement. An analysis of the uncertainty in the statistical dam aeration models has been added to Appendix B. This uncertainty analysis indicates that the uncertainty resulting from the dam aeration models is low compared to other uncertainties inherent in water quality modeling.

Detailed Comments Page -2-

- 9) Information provided in Section 3.5.2 indicates that approximately 58 municipalities and over 100 industrial entities discharge wastewater to the Ohio River between Pittsburgh and Gallipolis. Only 11 dischargers were include: in the water quality modeling. This seems inadequate. Inclusion of some of these dischargers in the model may preclude the need to 'invent' substantial additions of 800.
- (Figure B-8 illustrates that no DO data were included in the calibration of the water quelity model from approximately RM 34D to RM 255. The model can not be declared calibrated with such a paucity of data.

Based on the comments above, there does not appear to be sufficient data available to realistically model the Ohio River. Conclusions based solely on the modeling analysis presented should not be used. The inclusion of a sensitivity analysis would improve the report, but likely not to the point where the modeling results would present a representative picture of what is actually going on in the Ohio River. 320. All major dischargers (dischargers included in EPA's Permit Compliance System) that are required by state permits to monitor effluent BOD concentrations were considered for inclusion in the model. Some major dischargers may have been dropped because their BOD loadings were negligible. There are many major dischargers, however, that the states do not require to monitor BOD discharges. Hinor dischargers were not included in the model because their individual contributions to BOD are generally negligible.

121. See response to comment #70.

122. The modeling analysis used does not purport to simulate all the detailed processes and rates that affect D0 concentrations in the study area. The purpose of the model was to simulate effects of changes in one component of the D0 budget, which is dam aeration. Although it would be preferable to have a model that accurately simulates all components of the D0 budget, there is not sufficient data available to justify such an approach. Because the model accurately simulates changes in dam aeration resulting from hydropower, and demonstrates how these changes affect downstream D0 concentrations, its results are appropriate for determining hydropower impacts on D0.

	COMMONWEATH OF PENNSYLVANIA APTMENT OF ENVIRONMENTAL RESOURC Post Office Briz 2063 Hamsburg, Pennsylvanis 17120	ES	
	∃szan 21, 1988		
Surrow of Waker Quality Manaperies			17 787-2466
Exis P. Cashell, Acturg Societary Endered Freegy Regulatory Coron 525 Gordt Castrol Struct, NJ Vesturgton, D.C. 2002 Res DER File No. 32312	isse		

Devel May Costentia

We have reviewed the May 19, 1968 Braft F. airon neural Impact Statement (DEB). . unfalled "Hydrowlenthic Develop deat in the Tigger Olde River Basily" FERC Decket ins 13,85, 19, 114, enverying Olin, Pennselvanin, and West Virginia.

protects do not result in violation of criteria which would adversely impact on designated water uses. The major parameter of concern in this regard is dissolved oxygen. Per mylicity dissolved evigen criteria for the streams affected by these projects are minimum daily average 5.6 mg/f and at Value less than 4.6 n.gh (25 PA Code, Chipter 93, Water Seality Standards).

We appreciate the opportunity to comment on this discument and anticipate review of the Fatel Environments, hispart Statements

Spowerels, 1 Sourcels. For Store Prove Science Degree Darnet B. Drawbough Article Director B

Ponossiveral State Wester Deality Standards regular shareper stoned transproprised | 123 | 123 | 123 | Comment noted. The recommended alternative is designed to maintain DO concentrations of 5.5 mg/L anywhere in the rivers where DO concentrations would be affected by the proposed projects.

1

.

.

COMMONWEALTH OF PENNSTLVANIA

ŝ

124

125

126

127

July 15, 1988

- SUBJECT: Comments o, DEIS for Hydroelectric Development in the Upper Onio River Basin, FERC Docket No. EL 85-19-114.
- YC: George Taylor Federal Energy Regulatory Commission
- FROM: Sleve Repler, Fisherics Biologist day det Bivision of Environmental Services Fennsylvania Fish Commission

The subject document has been reviewed by the PFC staff and the following comments should be included in the final EIS.

General Comments:

As was discussed in the numerous scoping meetings the exclusion of numerous projects from the EIS makes they document of little use in evaluating cumulative impacts for the basin.

The use of existing water quality data in developing models gives question as to the reliability of the model. The lack of additional water quality parameters and the effect the dans have on volatilization of chemicals by the dams have not been udequately addressed.

Fishing impacts were considered throughout the DEIS as being moderate or minimal. If every hydropower facility on the river system were to have "moderate or minimal" fishery impacts, then the overall "cumulative" effects would be substantial.

Recreational use for the Ohio River system is a part of the study which needs to be studied more extensively. Use of existing data for the Pennsylvania section, does not give a true picture of the existing recreational trends.

Specific Comments Water Quality - Page 3.33

Each of the navigation dams in the study area aerates [128] differently. The effects of changing aeration (hydropower development) at different dams in the system are olearly comulative and interactive.

Differences in DC caused by changes in aeration at one dan affect not only the pool immediately downstream of that dan but also the peration rate of the pext downstream div, etc. These statements, contained in

124. See responses to comments #32 and 33.

- 125. Compared to most rivers, there is a large base of existing water quality data for the Ohio River Basin. These data were used to parameterize and calibrate the water quality model (see Appendix B). Additional data were collected by the hydropower applicants in the summer and fall of 1987 and used to model dam aeration. The responses to comments #104 and #140 discuss effects of projects on concentrations of toxic compounds. Because mechanisms by which the proposed projects could affect other water quality parameters are unknown, no impacts to other parameters are expected.
- 126. See response to comment #236. Staff assumes that the regulation and management of impacts of the recreational fishery catches on gamefish populations will be handled effectively by the responsible resource agencies, with appropriate measures taken to reduce angler harvest, if degmed necessary.
- 127. Staff utilized the best available information for summarizing the existing recreational use in the Pennsylvania portion of the Ohio River system under study. Oata sources included recreation use statistics from the Corps, a summary of recent fishing license sales and boat registrations, and a 1980 PEC fishing and Boating Inventory.

128. See response to comment 137.

FERC Fage 2 July 15, 1988

> the DEJS show the need to include all the potentia' hydropower sites for the basin. This cumulative effect could also cause reductions in volitelization of other cherical components of the watershed. Without more intensive water quality monitoring at all the sites, the FFC is concerned about potential problems.

All dissolved oxygen data is presented as average concentrations. These values do not give the needed information to draw proper conclusions. Beta which was used for the modeling needs to be included to nelp determine the models validity. More minimum DO concentration data is needed to show what the overall effect will be on the firh and aquatic life of the system. The PFC has recommended that for hydropower projects to be licensed there should be no water quality degradation. 129

Turbine aerstion was mentioned as a possible method to maintain existing DO conditions, yet this procedure has not been proven on the low-head dams on the Ohio liver system.

Page 5-10

The proposed operations at fixed creat dama would decrease the upstream pool elevations when operating. These reductions in pool elevations would cause increases in river velocities of up to 40 percent. Was this taken into consideration for cumulative D0 effects when modeling? Additionally, without consideration of Allegheny L&D's 5,6,8, and 9 how can overall effects be evaluated?

Fish Entrainment/Nortality

Sufficient data for fish mortality caused by hydropower development is not available. Page 5-6, using Racine Hydropower Finut data on fish mortality, states that "cocasional individuals are entrained", which is approximately 39 gamefish/hour or 529,500 gamefish per year. This number of fish 1f comparable at other sites would cause serious reductions in the populations in the river system.

Entrainment of fish is another situation for which no data is available. Each of the dam sites are different; therefore studies will be different. Angled trash rackoffish screens have questionable value, especially with expected intake velocities at some of the proposed sites. With the "migratory" garefish populations in the river (ie. walleye, sauge-), seasonal fish entrainment/impingement could be a real problem. Pre - and post operation fishery studies will have to be mendatory to evaluate the above situations.

- 129. The results of the DO model are cross-sectionally averaged concentrations. To determine the recommended spill flows, the model was used to simulate conditions (high temperatures, low flows, no primary productivity) that would result in near-worst case DO concentrations. The spill flows are designed to prevent impacts to fish populations under these conditions. See response to comment #39 regarding anti-degradation of DO concentrations and evaluation of preproject DO concentrations.
- 130. Comment noted. If turbine aeration can be shown feasible, it is recommended as a way to increase power production without degrading water quality (see recommendation 8, Section 5.4.2.1).
- 131. The model indicates that the predicted increases in water velocities cause little change in DO concentrations. The model analyses for the proposed projects assumed that the licensed projects (Allegheny L&Ds 3, 8, 6, and 5, Racine, and Hannibal) were operating with either spill flows required in their licenses or required by the Corps.
- 132. Staff believes that although such mortality extrapolations are speculative at this time, they do indicate that fish protection devices may be needed. Staff recommends a plan for measuring the entrainment rates and for testing fish protection devices (see response to comment #49).
- 333. See response to comment 449 concerning the entrainment data and its analysis. Pre- and post-operation fishery studies are included in recommendation 3, Section 5.4.2.2.

FERC Page 3 July 15, 1988

Recreational Use/Loss

Again insufficient information is available for assessing the recreational use/loss which would be caused by development. Creek surveys should be conducted before as well as after construction of any project. In addition, facilities should be constructed to enhance the recreational use of the area. 1.134

135

Other Environmental/Physical Consequences

Wetland and fish habitat loss due to pool elevation changes will probably be greater than that predicted by the bEIS. At fixed crest dams, pool elevations could change up to three feet causing damage to wetland areas and the shallower fish habitat. All of these factors need to be addressed in more detail.

srh

- cc: PFC + Arway PFC - Young FWB - Kulp
 - COE Biller

- 134. Recommendation 6 of Section 5.4.2.3 specifies that developers should monitor recreation use at their project sites in order to assess the adequacy of their recreational facilities. Creef studies are specifically recommended before and after project construction as part of this monitoring effort.
- 135. Staff agrees that pool elevation levels at fixed-crest dams could possibly change as much as 3 to 4 feet. The best available data were used to calculate elevation changes for each site where wellands occurred and could be affected. The calculated elevation changes are for specific sites. Monitoring of wellands for 5 years is recommended in Section 5.4.2.4, with mitigative measures subject to approval of appropriate federal and state agencies.

9-68

DEPARTMENT OF CULTURE AND HISTORY
STATE OF WEST VIRGINIA ARCH A. MOORE, JR., GOVERNOR
NORMAN & FAGAN COMMISSIONER

Tane 101, 1988

Martine I. Contell Action Sectodian Has Ras Signal September 1999 Martinester, D.J. 200400

D.C. M.C. By Browdy (Mag. Development) PRMC (1997) 240-244 (2017)

Br. r. Bel, Chielen (1):

(b) effler has recenced the dealer bit for Bellowlectric Alendric programs on the Apper three River Learn (File share) rec. PLRS Dealer).

In Nest Simplers we have seen dealers with each paramet applicant at specific entry modeling attention or concerns about contents modeling attention about cultural resonances have been bet on with to private project implementations.

THE CULTURAL CENTER / CAPITOL COMPLEX / CHARLESTON, WEST VIRGINIA / 253L5 / 364+348+0240

Those you for the repeaturity to prove the element.

Samerately. Unitian le rance

Villing C. Harray Newstr State Process Freemannes STER 62

AND A MEETING.

136. No response required.

.

12 (C H9 3- JUL 88

.

1 136

ARCH & MUORE JA Generati

July 21, 1988

RONALD P POTESTA Director RODERT K. PARSONS Deputy Director

He. Unis D. Cashell Acting Secretary Federal Energy Regulatory Commission 875 North Capitol Street, N.E. Washington, D.C. 20026

BE:	Hydroelestric	Developmen	e in the	t Upper Ohio
	River Batton;	FERC Dock	et tie.	EL85-19-114;
	Draft Environ	arotal imps	ice States	nent (DETS)

Dear Ha. Casheil:

~

The West Virginia Department of Natural Resources has completed a review of the above referenced document insued May 10, 1988 by the tederal Emergy Regulatory Commission (FERC). By formal request dated June 22, 1988, the Department received an extension of the comment period to July 25, 1988. Today's comments are a result of our final analysis of the draft.

The FERC staff evaluated and made recommendations on licensing 24 proposed hydroelectric projects at 19 sites in the Upper Ohio River basin. All of the projects are at existing dama on the Allegheny, Monongahels, Tygert, Muskingua and Ohio rivers. The DEIS presents the following five alternative actions by FERC concerning licensing of hydroelectric projects:

- 1. Licensing projects at all 19 sites as proposed by developers:
- Licensing projects at Ail 19 sites with restrictions requiring operational modifications to maintain downstream dissolved oxygen (DO) levels at 5 mg/l.;
- licensing projects at all 19 sites with restrictions requiring operational modifications to maintain downstream DO levels at 5.5 mg/l.;
- licensing projects at 15 mines with restrictions requiring operational modifications to maintain downstream DO levels at or above 6.5 mg/l; and
- no action: (1.#., not casuing licenses at any of the 19 sites inder consideration);

8808030552

-3/11 FERC - DOCKETED -JUL 2 7 1988

Ne. Lois D. Cithell Page 2 July 21, 1988

The Department is not in complete agreement with any of alternatives presented. Alternative four is the most accepteble, but requires considerable modification. The following as general and specific comments regarding the subject document.

- General
- 137 1. The Ohio and Monongahels rivers are listed by the Department of Natural Resources as high quelicy streams and as such are regulated by the State's anti-degradation standard. With respect to hydropower facilities, the anti-degradation standard requires that pre-existing, pre-project dissolved oxygen (DO) conditions be maintained after the project becomes operational. The Department currently recommendathat hydropower applicants conduct DO and temperature monitoring above and below the project prior to operation and for a minimum one year period. The data gathered is used to establish existing conditions which are not to be descaded by hydrogeneration. After operation, monitoring above and below the project is continued in order to austain pre-existing conditions. The maintenance of DD at or above concentrations of 6.5 mg/1, as per the recommended action, will violate the State's anti-degradation standard and in addition, bring a reduction in the current standards for resource protection.
- 2. The operation of hydropover facilities should in no case result in the degradation of pre-project DG conditions as established by DG studies. The Department will require that the developer use the means necessary (air injection, siteration of spill flow or porject shutdown) to maintain DO concentrations. If a project continuously degrades DO, the Department has the option to withdraw State 401 Certification.
- 3. The DETS fuiled to evaluate the potential impacts of reduced [139 wolatilization of contaminants and the increased re-so rension of contaminated assiments. The impacts of both whould be further assessed prior to the final EIS.
- 4. Reductions in pool levels at fixed crest dats, which is [40] likely to occur when the hydropower facility is in operation, may influence groundwater flow from adjecent aquifers. The reduction is hydroatstic pressure resulting from lower pool elevation may increase the groundwater movement into surface

137. See response to comment #39.

138. See response to comment #39.

- 139. The response to comment #104 discusses volatilization of contaminants. The potential for impacts of resuspension of contaminated sediments is highly site-specific; as stated in Section 4.1.1.3, it is expected that most sites will not contain contaminated sediments. Sitespecific analyses for sediment contamination have not been included in the FEIS because project developers will be required to comply with Eorps dredging permit requirements and to file a sediment testing and disposal plan (recommendation 6, Section 5.4.2.5) with FERC and the states. Compliance with these requirements should prevent significant impacks from sediment contamination.
- 140. A discussion of water quality effects of increased groundwater inflows at pools above fixed-crest dams has been added to Section 4.1.1.2. Since this issue was not raised during scoping, detailed analyses have not been conducted.

-71

Mm. Loie D. Cambell Page 1 July 21, 1480

> waters. If the adjacent aquifer is contaminated, the jud contaminated groundwater could be introduced into surface waters. Further consideration should be given to those pools which will be affected by lower elevations with emphasis placed on those where there is or has been heavy industrialization.

- 3. Disablyed oxygen conditions should be seessed throughout the [14] basin rather than just at proposed project locations.
- 6. The impects associated with the changes in river hydraulica 147 during hydropower generation (i.e. alteration of flow patterns and the effect on aquatic habitst and drinking water intakes) were not thoroughly evaluated in the DEIS. Further consideration and evaluation should be given to these potential impacts.
- 7. The end result of the final EIS should not be a simultaneous 143 licensing of all projects and the subsequent concurrant construction of all the hydroelectric facilities. If all projects are licensed, construction of the facilities should be staggered.

The next general comments deal with basin-wide recommendations (3.4.2) made by FERC staff. These recommendations are addressed in the general comments due to the far-reaching nature of their impacts.

1. Basin-vide Recommendations

5.4.2.1 Recommendations on Water Resources

- Current West Virginia regulations concerning State 401 Gertification require maintenance of pre-project DO conditions on those streams considered High Quality. The Ohio and Monongaheis rivers are classified as High Quality streams and FERC action permitting hydropower operations to degrade DO levels to 6.5 mg/l. will violate this regulation.
 - The Department does not concur that all staff's [145 recommendations concerning spillage were based on accurate or adequate data. The Department objects to the "no spillage" provisions included for projects at Opekiske Lock and Dam (L and D), willow Lefand L and D, Belleville L and D, and Gallipping L and D. The Department recommends that the

- 141. Dissolved oxygen concentrations were simulated throughout the pools between dams in the study area. Although the graphs showing model output plot only concentrations at the beginning and end of each reach, the model calculates the minimum BD concentration in each reach. This minimum concentration in each reach was used to evaluate impacts and assure compliance with DD criteria.
- 142. Section 4.1.5.1 has been modified to include a discussion of effects of changes in hydraulics on water intakes. Development of hydraulic models during design of other hydroelectric projects has successfully maintained acceptable flow conditions during hydropower generation.

The effects of changes in tailwater hydraulics on Fish habitat was extensively discussed in the Section 4.1.2.2 of the OEIS. Entrainment of fish from hydraulic changes at the Intake was discussed in Section 4.1.2.3.

 \gtrsim

- 143. The suggestion that construction of licensed projects should be staggered would extend project construction over a longer period than necessary and increase costs due to inflation. Staggered construction of hydroelectric projects in the basin is, therefore, not a feasible option for mitigating the recreational impacts of concurrent construction. The large number of projects as well as the extended construction period make this suggestion probibilive. Furthermore, it is highly unlikely that all of the licensed projects would be built during the exact same time frame, as project financing and final approvals of project plans will likely transpire during different time periods. The provision of temporary fishing access facilities is the best means of mitigating recreational lapacts during construction.
- 144. See response to comment #39 concerning maintenance of preproject 80 concentrations.
- 145. Staff believes that sufficient field data have been obtained to quantify aeration rates at the dams. The field observations used for each dam aeration model have been added to Appendix B [Table 8-1]. The recommendation for no spillage at Opekiska L&D was made because the submerged discharge at Opekiska, during summer, discharges cold water from the bottom of the stratified Opekiska pool into the bottom of the Hildebrand pool, causing stratification in the Hildebrand pool (Section 4.1.1.1; see discussion of assessment results for Case 2). The hydropower plant at Opekiska would withdraw from all elevations of the Opekiska pool and prevent stratification in the Hildebrand pool; the site-specific recommendations for Opexiska in Section 5.4.3 have been amended to include withdrawal from the entire water column of the Opekiska pool. Because staff believes that this benefit of hydropower at Opekiska, where thermal stratification is a more common and severe problem than in other pools, is especially important at low summer flows, no spill flow is recommended.

Ha. Loie D. Cochell Page 4 July 21, 1948

> following spillage requirements be included when licensing the above projects in order that seration of the dem be further studied: Opekisks L and D, 500 cfs; Willow 1stand L and D, 5,000 cfs; believisle L and D, 5,000 cfs; and Gallipolis L and D, 6,000 cfs. After such s study on spillage flow (2 years seams appropriate), final spillage recommendations may be determined.

- The Department should be included as a regulatory agency which would notify developers to increase or decrease apillage.
- J. In addition to the Corps and ORSANCO, the appropriate State Fish and Wildlife and Water Resource agencies should be notified of any emergency operational modifications by hydropower facilities. Any other operational modifications are subject to review and entual agreement between the developer, the Corps and the State agencies.
- 4. The Department contends that developers should monitor 148 pre-project D0 conditions for at least one year prior to operation to document pre-project conditions. Absence of pre-project monitoring requirements is in conflict with the State anti-degradation palicy (see general comment shows).
- The Department recommends that the developers be required to consult with it when developing water quality monitoring plane.
- 5. The Department should be consulted by the developer when 150 designing streamflow gages to weasure flow through turbines and hypass channels. In addition, sli flow data should be made svailable to the Department upon request.
- 7. A basin-vide water quality management group should include 151 the Water Resources agencies form each state in the basin. The water quality group should represent the interests of both state and federal agencies and participation should not be limited to a single group or agency.
- Developers should be required to consult with the appropriate 152 State Fish and Wildlife and Uster Resource agencies in addition to ORBANCO and the Corps when developing

The water quality analyses clearly indicate that spillage at the deeply-submerged discharge gated dams on the Chio River (Willow Island, Belleville, and Galipolis) provides no more aeration than can be expected from hydropower turbines. For example a DO deficit of Z mg/L is predicted to be reduced by only 0.2 mg/L at Willow Island and Belleville, and by 0.4 mg/L at Galipolis, with no hydropower generation. The spill flows recommended in this comment would have even smaller effects on DO concentrations, and the effects would not be distinguishable from natural variability if studied.

- 146. Staff agrees. The text for this recommendation has been modified to include appropriate state water quality management agencies.
- 147. Staff agrees. The text of this recommendation has been modified to include appropriate state fish and wildlife agencies.
- 148. See response to comment #39.
- 149. Staff agrees. The text for this recommendation has been modified to include USFWS and appropriate state water quality management agencies.

긢

- 150. Staff disagrees. The text for this recommendation has been modified, however, to allow state water quality management agencies to request flow data.
- 15), Staff agrees. The text for this recommendation has been modified to include state water quality management agencies.
- 152. Staff agrees. The text for this recommendation has been modified to include USEWS and appropriate state fish and wildlife and water quality management agencies.

Ma. Loik D. Cistell Page 5 July 21, 1988

recommendations for changes in spiff flow to include artificial accution.

9. No consent.

5.4.2.2 Recommendations on Aquatic Ecology and Fisheries

- The Department contends that intake designs should include provisions for installation of devices to measure fish personge (see below).
- 2. The Department remains concerned regarding entrainment 154 mortality (EM). Entrainment studies conducted in the Ohio River basis to date are not conclusive and many statements in the DEIS concerning EM are not substantiated. Spatial and temporal fish passage studies have not been conducted in the fall, winter and early apring. Conclusions and assumptions concerning qualitative and guantitative fish passage at projects at maving in dama are unsubstantiated.

The Department contends that studies which adequately estimate turbing passage (upatial and temporal) are needed to determine the magnitude of the mortality problem. Some techniques to estimate passage (r.g., that used at the Greenup facility; No. 2614) show promise of providing adequate fish passage estimates. Obviously, new projects should be designed to accommodate sampling equipment. The Department believes that effective fish guidance and/or protection devices in the basin will be extremely costly to design, install and test. The Department believes that adequate passage estimates are needed to determine if installation of fish protection or deterrent devices are cost effective. The Department has contended for some time that adequate mometary compensation of state resource asencies way be the only equitable solution to EM problems. The Department has submitted a plan for the use of such funds for resource enhancement on the Ohio River. Similar plans for other affected rivers could be developed by the appropriate state vildlife agencies. The Department believes that the "reinvestment" of such funds in the form of fish stocking, habitat protection, habitat development, access improvement, and other activities in the affected waters can repuit in effective EM mitigation for the resource and its users. The rivers in question are public resources. Development of the

- 153. Staff included a recommendation in the DEIS that entrainment rates be monitored (Section 5.4.2.2, recommendation 4). The text of recommendation 1 stating that intakes be designed for future fish protection devices has been modified to include the additional provision for installation of devices for measuring fish passage.
- 154. See response to comment #49 concerning substantiation of staff's conclusions regarding entrainment. Deficiencies in the entrainment studies conducted in the Dhio River Basim were highlighted, including the lack of data from seasons other than summer. Recommendation 4 of Section 5.4.2.2 specifically addresses measurement of entrainment rates. Staff recognizes that there was confusion between the two forms of "monitoring" recommended (see response to comment #65). See response to comment #54 for discussion of the conflicting views of compensation and the staff's approach. Staff believes that its recommendations for measuring entrainment are in accord with this comment.

98. 2018 D. ((Aler)) Page 6 Page 8

revers for corporate profet is acceptable provided that any adverse impacts to the resource are prevented or adequately mitigated.

It is therefore recommended by the Pepartment that primary initial emphasis be placed on obtaining adequate estimates of ennual fish passage at the proposed facilities. A mortality rate may thro be assumed and the impact of EM assessed should it be deened cost effective at that time, a bioengineria e facility(s) could then be established. In the interim, state time and wildlife agencies should receive annual monetary compensation for resource enhancement to compensate for EM (as recommended in the DEIS).

3. FERC staff concludes that the high reproductive potential of 155 gizzerd shud and freshuater drum dictates that losses from EM will not impair populations of these species, Ohio River basin streams considered in the DEIS are large, complex squatic ecosystems. Hany natural and men-induced factors affect them. These include (but are not limited to) compercial navigation, industrial water uses and discharges. elimetic extremes, long term water quality trends, and domestic and industrial pollution. In any given year, there factors individually, cumulatively, and/or synargistically influence populations of aquatic life. The degrae of impact resulting from the addition of another mortality factor (in this case, EH) is impossible to separate from other factors. In view of what is currently unknown regarding the above factors. Of could very well adversely impact gizzerd shad or freshvæter drum populations in any given year.

Staff concludes that a plan for monitoring fish populations be implemented to establish the effectiveness of compensatory and mitigative measures. It is the opinion of the Department that while studies to monitor fish can be designed and effected, attributing population variations to a single impact will be cost prohibitive if not impossible. Attempting to determine the effect of a specific mitigative measure by monitoring factors in addition to the "target" factor in question) is unrealistic. To be effective, such studies must consider the impact of all factors influencing the fish populations (which are such studies must consider the impact of all factors influencing the fish populations in question may be identified. The 155. Staff recognizes the complex nature of the aquatic ecosystems of the upper Ohio River basin and the multiple influences on fish population success that the comment relates. Entrainment mortality will be one more source of potential biological damage. Nonetheless, there is an abundance of scientific evidence that attests to the ability of fish species with high reproductive potential to compensate for relatively high mortality rates in many diverse aquatic systems. The DEIS cited one prominent symposium volume as an example reference. This is not to say that there cannot be year-to-year fluctuations in numbers, Most "forage" species exhibit such fluctuations in the absence of entrainment mortality. When mortality rates become excessive (e.g., during extremely cold winters), it may take several years for the population to rebound. In the meantime, predator populations may have reduced food consumption or switch to alternative prey. Staff concludes, however, that entrainment mortality of shad and drum is unlikely over the long term to be a deciding factor in their ponulation numbers.

Staff did not intend to imply that studies monitoring field populations of fish could be devised and implemented to attribute population variations to a single impact. However, the cumulative impacts of hydropower which this EIS addresses should be reflected in population declines if the collective impacts are significant. Monitoring field populations provides an indicator of problems, the cause(s) of which must be established through effect-specific studies. Staff reiterates that the cumulative effects of several sources of potential biological damage to fish populations from hydropower requires an integrative monitoring program as well as effects-specific studies and mitigations. Such integrative information need not be in the form of massive ecosystem-wide studies, but can include common indicators shown useful in fishery management such as creel census (see recommendation 3, Section 5.4.2.2). My, Lnia D. Cimbell Page 7 July 21, 1968

> Department contends that monitoring fish populations to evaluate mitigative measures will provide inaccurate, minleading results while being extremely expensive. Fish population monitoring studies to determine the impact of entrainment should be eliminated as a basin-wide t recommendation.

Staff also states that moderate numbers of (ish killed or 156 injured by entrainment would contribute to predation by gamefish in tailwaters and thus sustain highly productive productor populations. Navigation daws without hydropower projects currently sustain highly productive predator populations. It has not been demonstrated that the addition of EM to the existing tailwater habitate will significantly increase or "sustain" predator populations in navigation daw tailwaters.

- 4. Monitoring EM should not be initiated until adequate 157 estimates of annual fish passage are obtained. Due to the nature of the sites, monitoring of entrainment mortality has proven to be expensive with reliable results difficult to obtain.
- 5. As mentioned earlier, the Department considers adequate estimates of annual fish passage a prerequisite for cascing/developing lish bypass or protection systems. The Department recommends that at least two years of operation at all facilities be dedicated to obtaining such estimates. Effective development of a bioengineering test facility could then proceed based on spatial and temporal fish passage information. It is recommended that the plan referred to here not be submitted co the Commission until the sbove-mentioned acudies are completed.

Staff is extremely vague regarding the bioengineering test [159 facility. It is not stated if the facility will be located at one of the sites considered in the DEIS. No reference is made regarding funding of its development, operation and maintenance.

5.4.2.3 Recommendations for Recreational Resources

1. The Department besically concurs with this section. The 160 Department recommends that the reference to lighting be

- 156. Staff believes that injured fish will most certainly be consumed by predators and scavengers in the tailwaters. If there is evidence to the contrary (other than a few recorded cases in which the numbers of injured fish temporarily overwhelmed downstream consumers), then staff would be happy to review it. This food supply will contribute to sustenence of those predator/scavenger populations (food eaten could hardly do otherwise). Staff did not assert that this extra food supply would significantly increase predator populations in navigation dam tailwaters as the comment implied. There is evidence, however, that certain predator populations are increased below hydroelectric facilities, a fact attributed to entrainment mortality (e.g., squawfish below dams on the Columbia River; Northwest Power Planning Council, Portland, Oregon, annual reports).
- 157, Staff agrees with this comment and modified the text in Sections 4.1.2.3.4 and 5.4.2.2} accordingly.
- 15B. Staff agrees with this comment and modified the text in Sections 4.1.2.3.4 and 5.4.2.2) accordingly.

J-76

159. See response to comment #49.

160. The text has been modified to incorporate these recommendations.

Ma. Lora D. Cashell Page R July 21, 1988

> clatified in order that developers understand that it is to be sufficient to permit hight fishing. This will increase the use opportunity or these areas. Access paths from the top of the bank to the multi-level ground/geved wellways parallel to the shoreline should be included in the required developments. The Department contends such paths should be cs. 100 feet space (st most projects). This allow anglers to muve freely to the area they wish to fish and aliminates conffict with those already fishing. Such paths are currently in place at Willow Laland L and D tailwaters and have proven effective.

- A major concern on several projects is that insufficient 161 lands for recreational development and mitigation (during construction) are designated as project area. Until physical andelong is completed, the location of recreational developments cannot be determined. Often, lands owned by the U.S. Army Corps of Engineers (COE) on the abutment side of navigation dama are not of sufficient extent to support the recreational developments mentioned in 5.4.2.3.1. The Department recommends that project areas of all proposed projects be reviewed by staff with this issue in mind. The Department contends that this problem is particularly evident st the Hildebrand, Morgantown, New Cusherland, Gallipolis and Tygart projects and that their project boundaries should be increased sufficiently to ensure adequate recreational development and mitigation.
- Appropriate state and federal fish and wildlife agencies [152 should be granted consultation rights (in addition to "local" [resource agencies).
- 2. The Department concurs with this requirement.

163

- 3. The Department contends that all permanent and temporary [64 in-river fisherman access developments and bypass systems should be included in hydraulic modeling required by the COE. The Department has routinely recommended this action for three basic sesons:
 - B) To ensure that the location of the developed finhermen facilities would be proximal to currents and eddys (both from the powerplant and the bypass system) that will concentrate (ish at locations

- 161. In the event that sufficient lands are not available for construction of the recreational developments, a recreational compensation plan would need to be filed with the Commission. Off-site lands for recreation compensation are unlikely to have any significant impact on land use provided they conform to local zoning regulations. Compensation for impacts to recreation from project development could include the provision of off-site recreational facilities and/or the upgrading of existing access facilities. The compensation plan would be developed in consultation with the appropriate state and federal agencies and filed with the Commission for approval.
- 162. Text has been modified to incorporate the recommended changes.

163. Comment noted.

164. Recommendation 3 of Section 5.4.2.3 addresses physical hydraulic modeling of all permanent and temporary in-river fishing access facilities. Developers will be required to file a report with the Commission which discusses the design and results of the hydraulic modeling and documents consultations with the appropriate state and federal agencies.

Staff agrees that all facilities affecting hydrodynamics of the site, including the fisheries facilities, should be included in the modeling done for the Corps.

3-77

Ha. Louis D. Caxhell Page 9 July 21, 1988

> accessable to anglers using the developed facilities (r.g., a fishing pier should be focated where fish can be caught). It is for this reason that the Department recommended that WRO biologiste participace in the physical modeling (at developer expense), and that aurways correlating fishing success with flows be conducted at the undeveloped site prior to worksling;

- b) To ensure that the developed facilities do not interfere with navigation and are acceptable to the COC; and
- c) To ensure that temporary fisherwan developments (during construction) do not interfere with newigation and are deceptable to the EDE.
- 4. The Department contends that bypassing 500 cfs at flows 5,000 [155 cfs and below will be insufficient at most facilities, and recommends that 1,000 cfs be seleased below 10,000 cfs.
- 5. The Department basically concurs with this requirement, but 166 does not feel it provides sufficient detail for enalysis.
- 5. The Department rencurs with this requirement in principle, 167 but feels that the filing of reports to the Commission should occur at specific intervals during the license life, i.s., every 3 to 5 years. Due to the length of the license period, predicting wise, number, type, etc. of facilities necessary to meet angler demand is difficult, and reassessment of facility adequacy will periodically be required. The requirement should also dictate that appropriate state and faderal fish and wildlife agencies' comments shall be included in reports filed with the Commission.

5.4.2.4

1. The Department conturn with this requirement.

5.4.2.5

1. The Department concurs with this requirement.

- 165. Recommendation 4 of Section 5.4.2.3 has been revised due to the site-specific issues that would need to be resolved during physical hydraulic modeling. The final specification of flow velocities as well as the plan for maintaining these velocities in the tailrace fishing areas would be determined during the physical modeling and consultations with the appropriate state and federal agencies.
- 166. Text has been modified to incorporate additional details to recommendation 5 of Section 5.4.2.3, such as the need to file a plan with the Commission regarding the provision of temporary fishing access facilities during construction.
- 167. Text has been modified to revise recommendation 6 of Section 5.4.2.3. The recommendation specifies that developers should file a report every five years with the Commission regarding the monitoring of recreational use at project locations. The monitoring effort and report preparation should be done in consultation with the appropriate state and federal agencies.

168. Comments noted.

168

No. Lyży D. Casheli Page 10 Juży 21, 1988

- 2. The Department concurs with this requirement.
- 3. The Department concurs with this requirement.
- 4. The Department concurs with this requirement.
- 5. The Department concurs with this requirement.
- 6. The recommendation for contaminated sediment testing and [169 disposals plans is estisfactory. Rowever, the developer should also be required to consult with the Department concerning testing procedures and results and the disposal plan.

370

- 7. The Department concurs with this requirement.
- 8. The Department recommends that the developer consult with [7] resource egencies regarding the suitability of disposal sites. If deemed necessary (due to endangated species, excellent existing habitat), alternate site designation of a habitat evaluation may be required. This will insure that spoil disposal will not result in destruction of valuable or irreplaceable wildlife or fisherise habitat.

The Department has consistently recommended that clean rock 172 spoil be utilized to construct in-river developments (e.g., plers, underwater reefs and burs for fish cover). The Department recommends the staff permit such flaxibility in spoil disposed provided it is recommended and approved by appropriate state and federal resource agencies.

A plan for revegeration should be developed in combination # 173 with Item 9; Control of erosion, dust and slope stability.

- 9. Sediment control plans currently submitted to the Department [174 require vegetation and maintenance plans. If itam 8. and 9. are not combined then the revegetation and waintenance plan required by Itam 8 should be submitted to Water Resource agencies in those agencies listed.
- II. Cumulative Effects on Fish The staff sesumes positive: 175 impacts on recreational fishing opportunities from hydroelectric development. This may or may not be the case at every site. Excellent fishing now exists at many lock(a);

- 169. Comment noted. Text has been modified.
- 170. Comment noted,
- 171. See response to comment #105.
- 172. Staff agrees with the comment. See response to comment #58.
- 173. Comment noted. Staff recommended that the revegetation plan be developed in close coordination with the erosion and dust control and slope stability plan.

ج

51.

- 174. Comment noted.
- 175. Staff discusses potential impacts to sport fishing from changes in fish habitat quality in Sections 4.1.3.6 and 5.1.1.3. Although the new public fishing access facilities could greatly increase the potential fishing use in the basin, staff acknowledges that projectinduced impacts to fish habitat quality, which would occur with the projects as proposed, would need to be mitigated to maintain the quality of recreational fishing in the basin.

Ma. Loss D. Costeli Pige 11 July 21, 1968

> and dam tailwaters. This is reflected by rapidly increasing use and harvest. Temporary fishing facilities will not compensate for all tishing apportunity lost during project construction and it is not currently known if fishing will be as good, better, or woras after the projects are constructed.

- Should fishing recreation and/or success improve as hoped, recreasional fish harvest vill increase. The Department is charged with management of the fish populations in Nest Virginis waters, and will take action to reduce recreational harvest if necessary. A more desirable solution, as stated by FEKC staff (page 5-7: Per, 1) would be enhancement of water and hebitet quality and the reduction of other sources of mortality. As stated earlier, the Department considers fish habitet and population enhancement as integral components of resource mitigation programs to be funded by monterry compensation FM.
- 112. The Department does not concur with staff that plans to 177 provide Eisbing access during construction will reduce this impact to acceptable levels. When concurrent construction of besin projects is considered, the problem is magnified. The Department recommends that construction be "staggered" at adjacent projects to reduce this impact.
- EV. The Department contends that spillage recommendations ware 178 constinues based on insufficient data. For this reason, spillage (590 of at Opekiska, 5,000 of at Willow Island and Selleville, and 6,000 of at Galipolis) is recommended until a definitive study demonstrates the spillage amounts, if any, that are needed at these sites.

As a matter of record, the Department objects to the action 179 taken by FERC under Commission Order 464 which usived State 401 Certification for the listed projects in West Virginis.

V. The Department is autromely concerned regarding statements in [160 the DEIS which contend that comments supplied by fish and wildlife agencies on the DEIS will be considered the most curtent recommendations applicable under the Tish and Wildlife Coordination Act (FVCA). The Department has submitted detailed comments on each project application within its jurisdiction. Initial FVCA comments when file on such project as well as additional comments when file on such project as well as additional comments when

176. Staff agrees with the comment. Recreational fish harvest at hydropower projects could certainly increase as a result of improved angler access at fish habitat-enhancing features. Staff anticipates that the fishery would require management attention by the WUDMS and other appropriate state and federal agencies. Staff believes its recommended actions would enhance water and habitat quality compard to project developments as proposed and would lead to minimization of entrainment mortality; actions depend on recommended testing of fish protection and guidance devices. Monetary compensation is recommended by staff as an interim mitigative measure and as a possible long-term mitigative alternative for consideration following testing and evaluation of fish protection and guidance devices (Section 5.4.2.2). See response to comment #54.

177. See response to comment #143.

178. See response to comment #145.

179. Your objection is noted. For discussion of the status of water quality certification of the projects in the study area see Section 4.7 of the FEIS.

180. See response to comment #31.

44. Ente B. Carhell Page 12 July 21, 1948 warranted. The Hendrigent contends that all comments (including thoury yet to be submitted on issued licenses) should be considered as comments under the Facta, All of the above referenced comments should be considered in the Section 10(j) (federal Pover Act) resolution process.

,

- W1. Staff concludes the there are no significant difference between competing applications at "Sygart Lake, New Cumberland, will by lahad, and Gallippis (Land D. The Department does not concur. For example, one applicant at Callippits propose to construct two powerhouses (one at the abutent and one in an unused lock chamber), while the other propose not. Gonaiderable differences in plans for tecreational developments are also avident. This poses at a signific problem to reviewers because it is not known which projects will be licensed at sizes where competing applications exist.
- VII. One developer proposes that "scout holes" be extered in the 182 tailraces of several projects. These developments were not recommended by the Department and are not viewed as withgative memoures.
- VIII. Site Specific Recommendations (Section 5,4,3) -- The Deputrment has provided site specific recommendations for seth project vithin its juridiction. The following commence concerning individual sites should not be construed as replacing kindercomments.
- 1. Opekiska L and D (5.4.).4) The Department does not concur 183 with the "zero epill flow" recommended here, since this contradicts current minium requirements exishished by the repartment. A minium of 500 cfs should be spilled until thorough etudies are completed.
- 2. Hildebrand L and D (5.4.3.5) --- Project lands are]84 insufficient for recreational development of this site. The Department recommends that the project area be increased to eccomponents all formeable potential recreational developments.
- J. Morgantown L and D (5,4.3,5) -- Project landa are 105 indufficient for recreational development at this sits. The Department recommends that the project area be increased to

181. See response to comment #34.

182. Staff has recommended bank undulations and reafs to enhance fish habitat and will depend on the resource agencies to advise on the design of specific mitigation measures: such consultation is included in recommendation 1 in Section 5.4.2.3.

183. See response to comment #145.

184. See response to comment #161.

185. See response to comment #161.

Ms. Lois D. Cashell Page 11 July 21, 1988

accommodate all forseeable potential recreational developments.

- 4. Willow usiond L and D ().-.3.14) -- for Department dues not concur with the "tero spill flow" recommended here, and contends a minimum of 5,000 cfs should be spilled until thorough scudies are completed. Zero or insdequate spillage may result in reduced DO concentration and violation of anti-degradation standards.
- 5. Belleville 1, and 0 (5,4,3,15) -- The Department does not concur with the "zero spill flow" recommended here, and contends a minimum of 5,000 cfs should be epilled until thorough atudies are completed. Zero or insdequate spillage may result in reduced DO concentration and violation of entit-degradation atundards.
- 6. Gallipolis L and D (5.4.3.16) -- The Department dows not concur with the "kerp spill flow" recommended here, and contends a minimum of 6,000 cfs should be apilled until thorough studies are completed. Zero or inadequate spillsge may result in reduced DO roncentration and violation of senti-degradation standards. In addition, the lack of spill flow at the three facilities above may reduce the volatilization of organics and contaminant waterials that pocentially occurs at these facilities.
- IX. The issue of hydropower impacts on lockage of recreational 189 venetis was not addressed by the DE15. Should adverse impacts occur, they would adversely affect recreation on the rivers included in the analysis.

Specific Gomments

- Page 2-25; Par. 3 -- The Department maintains that the 190 applicants should be required to install a blank piping system during initial construction for later installation of sic/oxygen injection systems should they be required, deemad necessary or desirable.
- Page 3-19; 3.1.6.1 -- Several species which may occur in the 191 project area have been classified by the U.S. Fish and Wildlife Service as Category 2. Category 2 is comprised of taxa for which proposing to list as endangered of threatened may be appropriate, but for which conclusive data as to biological vulnerability and threat

186. See response to comment #145.

- 187. See response to comment #145.
- 188. See response to comment #145. The response to comment #104 addresses project effects on volatilization of contaminants. However, dams that aerate very little are also expected to have negligible effects on the concentrations of volatile chemicals. Aeration at Opekiska, Willow Island, Belleville, and Gallipolis dams is low because little air gets entrained in the water as it passes the dam, even though high turbulence occurs. The lack of air entrainment also restricts the transfer of volatile chemicals from the water to the air, so hydropower at these four dams is septed to have negligible effects on volatiles. This point has been clarified in Section 4.1.2.
- 189. Section 4.1.3.5 addresses potential recreational boating impacts from the proposed projects. The issue of adverse hydropower impacts on the Tockage of recreational boats was not specifically raised during the scoping process. Physical hydraulic modeling required by the Corps should ensure that the projects will not significantly interfere with recreational navigation and/or lockage.
- 190. The recommendations on water quality (Section 5.4.2.1, recommendation B) have been amended to recommend projects be built in such a way that installation of mechanical aeration systems is not procluded.

191. Text has been modified.

Mm. Loin D. Cashell Page 14 July 21, 1985

are not currently available. These species are: blue suckar (<u>Cycleptus elongatus</u>), Bawich's wren (<u>Thryomanes bawickii</u>), eastern woodret (<u>Meotoma floridana magister</u>), morthern kong-eared bat (<u>Myotis septentrionalis</u>), small-(noted myotis (<u>Myotis leibii</u>), New England cottontail (<u>Sylvilague transitionalis</u>), and salamander shell (<u>Simpsonaias anabique</u>).

Page 3-21; Table 3.1.6.1 -- This table should be amended to include Rafineque's Big-cared bat <u>Pierotys rafinesqueril</u> found in West Virginius and with atstus (.5G); the Golden mouse <u>Orchrotomus</u> <u>mutalit</u> also found in West Virginia and with f.BCJ status.

Page 3-31; figure 3.1.9-3 -- This photograph does not depict Rev | 193 Cumberland L and D.

Page 3-34: Par. 2 -- The Department does not completely concur with these statements because spatial and temporal fish turbide passage and mortality have not been adequately investigated.

Page 3-351 3.2.3.13 -- It is statud here that extensive 195 inter-mavigation pool movement has been documented for valleys. The Department contends that such movement indicates high estrainment potential for this species.

Page 3-35; 3.2.3.9 -- Not all sauger spawning habitate have been dentified. The Department concends that sauger, particularly as aggs and larvae, may be highly vulnerable to EM.

Fage 3-36; 3.2.4 <u>Encreation</u>; Par. 4 -- The Department contands that fishing recreation will be edvarsely impacted at each site during construction in mpite of temporary fishing facilities. Temporary facilities must be constructed downstream of the powerhouse location. The area most impacted by construction is also where most fishing taken place at an undeveloped tailwater. Concurrent construction compounds this impact.

Page 3-57: 3.5.3.1 -- The impact of potenzial DO degradation on 198 freshwater muaset populations should be addressed.

Page 4-9; Pur. 6 -- The Department contends that a blank piping eystem that will permit later installation of air/oxygen injection systems should be a component of each powerplant. 192. Text has been modified.

- 193. The captions for Figures 3.1.9-2 and 3.1.9-3 were inadvertently reversed in the DELS; these have been corrected.
- 194. Text has been modified in Section 3.2.3.1 to include minimization of mortality (e.g., entminment mortality) as a protection in addition to habitat and water quality.
- 195. Staff agrees with the comment.
- 196. Staff does not agree that sauger eggs and Tarwae are highly vulnerable to entrainment mortality. Extensive studies of egg and Tarwał drift at steam electric stations (FSE 1987) in the study area have not identified sauger eggs and Tarwae as abundant components. Sauger spawning is generally in the tailwaters rather than in more quiet pools above dams where drift would be into turbines. Spawning is early in the year during relatively high flows, much of which would not mass through turbines.
- 197. See response to comment #143.
- 198. See response to comment #38. The potential impact of DO degradation on freshwater mussels has been addressed and has been included in the text as Section 4.1.2.5, Assessing the impact of Dissolved Oxygen Change on Freshwater Mussels. A more complete discussion of the topic is included in Appendix i of the FEIS.

199. See response to comment #190.

Mer, drove by Crosberts Priger 15 Unity 25, 1988

Page 4-11; 4.1.7.1.1 State Standards filter 1) -- Meat Virginia's 200 regulations operally that existing water quality on this River basin streams considered in this DEES not be degraded.

Page 4-24; Par. 4 -- The Department contends that elimination 203 of spillage flows at Opexisks, Wittow Island, Belleville and Callipolis will result in elimination of important summer and fall hebitsts at these teliwatets. The Following spillage flows are recommended:

a) Opekisks -- 590 cfs;
b) Willow Island -- 5,000 cfs;
c) Bellewille -- 5,000 cfs; and
d) Callipolis -- 6,000 cfs;

Page 4-25: Par. 3 -- The Department recommends that initial studies be primarily concerned with obtaining adequate estimates of annual fish passage at \$11 units to determine the magnitude of the problem.

Page 4-25; Par. 6; Sent. 5 -- The Department feels Staff is 203 presumptuous in stating entrainment damages are local. This has not been documented.

Fage 4-25; Par. 3; Sent. 1 -- The Department concurs with 204 this sentence and feels it substantists the need for determining fish passage tstes throughout the year. Furthermore, the location of powerbourds near the shormline may influence a disproportionate number of tish to peak through the powerboune rather the steed.

Page 4-28; Par. 1; Sent. 4 -- The Racine Study referred to have dealt with fish movement during a few months of the year. It is attacted elsewhere (page 4-30; par. 1) in the DELS that this study is inconclusive and. therefore, quantitative conclusions cannot be drawn.

Page 4-28; Par. 6 -- The Department does not concur with the summary presented here. Neither larvel fish passage nor sutrainment mortality have been estimated for Ohio River basin facilities. Due to the inherent fragility of many species of larvel fish, turbine mortality could be high.

Page 4	-28;	Par.	7	Stai	If conc	luder 1	bere t	hat	aince	207
well-defined	Cish	разва	e	informet	ion is	ø ba eni	t for	t K İ	uting	1
facilities,	monitori	ing ac	0.0 V	projeci	s is ne	cessary.	. А во	re lo	gicsl	1

200. See response to commont #39.

201. See response to comment #145 concerning effects of spill flows on D0 at these projects. Complete elimination of spill flows will, indeed, affect downstream fish habitat. The flows recommended in the comment might improve habitat, although no justification is given for why these flows are recommended. The text has been modified to include comment on spill flows for fish habitat. However staff believes that for these dams, the turbine discharge would provide adequate habitat in the same general area (Section 4.1.2.2.3).

202. See response to comment #157.

- 203. Entrainment damages are clearly local in the turbines of each plant and not distributed through the river as are impacts of lowered DO concentrations. Staff has stated in the paragraph referenced by the commentor that these local damages can be cumulative in the entire river system.
- 204. The shoreline effect is discussed in Section 4.1.2.3.2.
- 205. The sentence in the text regarding species entrained has been amplified to include the points made in the comment.
- 206. The summary referred to in the comment dealt with vulnerability to entrainment and not mortality. Staff has summarized the pertinent literature and has based its summary on it. The review in Section 4.1.2.3.2 indicates that an abundant scientific literature attests to the resistance, not vulnerability, of egg and larval stages of fish to physical stresses of entrainment.
- 207. The commentor has repeated the staff's recommendation -- monitoring will be necessary to determine the rates of fish entrainment, i.e., "fish passage rates".

J-84

Nu. Loie D. Cashell Page 16 July 21, 1988

conclusion is that since no well defined fish passage rates are evailable, fish passage studies should be conducted.

Page 4-30; Par. 4 -- The Department does not concur that damages to early planktonic life stages of fishes are of no concern (based upon studies at stream electric stations). The Department contends hydropower developers should conduct studies to estimate agg and larval passage through curbines and setimate turbine caused mortality using in situ or laboratory tests.

Page 4-30; Per, 5 n- Conclusions in this paragraph are based 209 upon acudies Staff contend are inconclusive. The conclusion that prolific spawners can companyets for turbing caused mortality when other mortality factors affecting the population are not known is preaumptuous.

Page 4-32; Par. 1; Last Sent. -- Studies to dete are not 210 conclusive and the Fish species wost frequently entrained on an annual basis have not been determined.

Page 4-33: Par. 3 -- This paragraph states that regional [2]] resources will be mershalled "... to select, construct, task, and evaluate engineering procetypes of fish guidance systems ..." The Department contends that clarification of this seatence is required for evaluation.

It is then stated that prototypes "that work" will be installed in full scale on newly constructed (operating) projects. This sammes the above referenced projects can be designed and constructed to accommodate suprem of currently unknown design. Although paragraph 5 of this pags states that plants will be designed to accommodate various possible bypass/deterrent devices, the Department questions if projects can be effectively designed to permit retrafitting for unknown eventualities.

Sentence four of the paragraph states that entrainment losses may be determined to be low (or presumably high) by monitoring. As stated marlier, the Department does not feel that fish population monitoring can successfully identify or quantify EN. The Department emphasizes that fish passage studies, rather than fish population monitoring, should be conducted. Staff does not identify a possible mitigation alternative should EH be determined to be high and if no systems to eliminate EM are proven effective. 208. The text has been modified to include the desirability of on-site studies of damages to early life stages of fishes in turbines even though staff has summarized the scientific literature that provides evidence different from the commentor's view.

209. See response to comment #155.

- 210. Comment noted; text has been modified to include mention of limited sampling.
- 211. See response to comments #5 and 66. The feasibility of fish protection technology and retrofitting to a completed facility is uncertain and will remain so until attempted.

212. The commentor has incorrectly interpreted "monitoring" to mean monitoring of fish populations in the field. Staff referred to monitoring of fish passage and survival in this instance. Staff also clearly identified compensation as the alternative in the same sentence.

.

Me. Loss D. Cashell Page 17 July 21, 1985

Page 4-31: Par. 4 -- Staff does not state if the binengineering 213 rest facility mentioned here is to be located at one of the proposed projects mentioned in this DELS, at a yet to be planned facility, or at an existing facility. Funding of facility construction, operation and maintemance are not discussed.

Page 4-33: Par. 5 -- See earlier comments (Page 4-3; Par. 3) regarding the proposed bioengineering test facility.

Page 4-33; Par, 7 -- The Department recommends that fish screene be installed on the Tygert project to reduce walleye DM as well as that of other fishes. Scaff's conclusion that only minor mortality of tish through turbines is expected during spring, summer and fall is unsubstantiated and not based on any information known to the Department.

Page 4-34: Par. 2 -- Recreational fishing will be advarably 215 impacted at the lock side of Gallipolis L and D due to hydropower development.

Page 4-34; Par. 3 -- See above comment.

Page 4-34; Par. 4 -- See comment concerning additional minimum recreational developments (5.4.2.3.1.).

Page 4-40; Tygart 7307 and Tygart 7399 -- The Department is not actisfied with recreational developments proposed by either applicant at this site. Both projects will require revision of their plans. The staff recognizes Department concerns regarding insufficient lands for recreational developments, but does not recommend a course of action.

Page 4-47; Pat. 3; 4.1.3.2 -- See earlier comments (5.4.2.3.3) regarding bypasted flows.

Page 4-47; 4.1.3.3; Sents. 5-8 -- The Department does not concur with these statements in view of other statements in the document concluding that mitigative measures for recreation lost during construction and due to concurrent construction are adequately mitigated. The Department is not sware of alternative measures that would "... in some manner be beneficial to recrestional fishing in the region ..." This recommended mitigative measures is not sufficiently described by staff to permit evaluation by the Department. Staff does not define the term "local resource agency."

- 213. Staff stated in the beginning of the referenced paragraph that the bioengineering test facility would be "located at one project (or at most, a few)." It is premature at this stage to determine the site(s) most technically suited to the needs of such a facility. All new and existing sites could, in principle, be considered candidates although special design features could be accomodated most easily at a new site rather than an existing one. It is also premature to assign funding responsibilities.
- 214. Staff relied on local agency biologists for the opinion that walleye at Tygart are not likely to be entrained at times other than when they are now flushed downstream intentionally. The text has been modified to reflect the commentor's view. Staff recommends measurement of fish passage and testing of screening devices or other measures to minimize entrainment of fish at Tygart (Section 5.4.3) as a special case of the general recommendations on aquatic ecology and fisheries (Section 5.4.2.2).
- 215. Recreational fishing is more prevalent on the abutment side of the dam than it is on the lock side of the dam. Potential benefits to recreational fishing could occur from the proposed project by introducing tailrace flows on the lock side of the dam. Short term construction impacts to recreational use could occur, but such impacts would probably not be serious due to the large amount of recreational land acreage on the lock side of the Gallipolis tock and Dam.

216. See response to comment #161.

217. At project sites having small areas available for construction, it may not be possible to provide access during construction because of the limited land area. Compensation for impacts to recreation during construction at these sites could include the provision of off-site recreational developments and/or the upgrading of existing access facilities. Recreation plans would need to be amended accordingly after consultation with the appropriate state and federal agencies. 85
No. Lois D. Cashell Page 18 July 21, 1986

This should include the Department and other appropriate federal and scate fish and wildlife agencies.

Page 4-30: Por. 2 -- The Department does not concur with staff [2]B conclusions regarding EN presented hars. See comments concerning 5.4.7.2.2.

Page 4-53: Par. 5 -- It is stated here that hydropower flows [2]9 may impact commercial navigation by alowing lockage treating unsafe conditions. The contends that a greater potential exists to adversely impact lockage of noncommercial (recreational) craft.

Page 4-71: 4.1.6.8 -- The Department contends that in-river [720 disposal of clean rock fill to construct pieze, dikes, fish cover, [reefs, etc. could be a beneficial mitigative measure.

Habitat evaluation procedures may be necessary at cartain upland [221 disposal attes. See comment concerning 5.4.2,3.8.

Page 4-71; 4.1.6.9 -- The Department recommends that developers 222 be required to plant transmission line rights-of-way to low-growing shrubs beneficial to wildlife. Staff merely concludes that this would reduce impacts. Transmission lines should be classed and maintained by mechanical rather than chemical means.

Page 4-71: 4.1.6.10 -- The Believille L and D project (No. 6939) will also require new road and railroad crossing construction. The required this to eliminate existing conflicts between residents of Belleville and current users of the tailwater area. The existing road through Belleville is privately owned.

Page 4-73; 4.2.1 -- The Ohio River is designated by the State of West Virginia as a high quality stream. In this imstance, water quality standards including dissolved oxygen, are supermaded by the State's anti-degradation stendard applied to high quality streams.

Page 4-82; Case 1 -- Gamerating on the recommended flows would 225 violate Weat Virginis water quality regulations. The Department recommends a chorough water quality study at Opekiska, Willow Island, Bellevilla, and Gallipolis after the plants are operating. Appropriate flow should be spilled at these sites until the study is completed and final recommendations are filed with the Commission. 218. Comment noted.

219. Text has been modified to mention potential impacts to recreational boaters. No direct impacts to recreational boaters from changes in flow patterns is expected because recreational boats are generally much more maneuverable than barge tows. See response to comment #189.

220. See response to compent #58.

- 221. Revegetation and erosion control plans al upland spoil disposal sites developed in consultation with appropriate state and Federal agencies are recommended (Section 5.4.2.5). The necessity for habitat evaluations will be determined on a site-specific basis. Thirteen of the projects include proposals to use existing commercial tandfills or abandoned strip mines as disposal sites (Section 4.1.6.1). With appropriate planning, wildlife habitat could be enhanced.
- 222. Recommendation 8 in Section 5.4.2.5 specifically requires revegetation in transmission line rights-of-way using low-growing shrubs and trees native to the area. Mechanical, rather than chemical, means of clearing for transmission line rights-of-way construction and maintenance is also required if at all feasible.
- 223. Text has been modified in Sections 4.1.5.3 and 5.4.3 to incorporate the recommended revision in the project access route.

the recommended new 450 foot long road crosses the slough at Belleville. This slough is a feeding and resting area for migratory birds and is a feeding and spawning area for fish. The developer should consult with ODNR, FWS, and Corps before final development of plans for the road to determine the final placement of the road through the slough (recommendation 15, Section 5.4.3). Any necessary alterations in the alignment of the road and other site specific mitigation to minimize impacts should be discussed with these agencies during consultation. The developer should obtain all required permits before initiating construction.

- 224. See response to comment #39. West Virginia regulations do not necessarily preclude minor degradation of water quality.
- 225. See response to comment #145. Staff believes that effects of West Virginia's recommended spill flows will be undetectable at these four projects.

He. Lois D. Coshell Page 19 July 21, 1988

. .

Page 4-87;	рег.	6 ++ The	Department	strongly ob	jects to the	E 226
last sentence in	this	peragraph	Fish and	Wildlife Coo	dinution Act	1
commence from the	State	of Here	Virginia sh	all include	all commente	
unde in reference	to 44	ch project	t and the re	solution proc	res dictated	
by Section 10 (j)) of th	10 Federul	Power Act a	hall include	all coments	1.
submitted by the	Depart	ment conci	traing each	project.	See general	1.
comment 9.			-	• • • • • • •	•	

Page 4-89; Par. 5 -- The Department does not concur with 227 steff's conclusions caparding entrainment. See general comment 1 (5.4.2.2).

Fage 4-89; Far. 9 -- Staff has not recommended specific habitat 228 management procedures to mitigate this impact.

Page 4-90; Par. 6 -- The Department contends that recreation 229 impacts due to concurrent construction will be considerable. Hurh of this impact could be mitigated by staggering license issuence to prevent concurrent construction.

Page 5-3; Table 5.1.1-1 -- The Department does not concur with 230 aswers! values in this table for the various categories (as described in these comments).

Page 5-3; Par. 1 -- The elimination of spillage at Opakisks, Willow Island, Belleville, and Gallipolis will sggrewate this situation, particularly in the three adjacent lower river projects.

Page 5-6; Par. 3 -- The quantitative and qualitative extent of Lurbine fish passage in the basin is unknown. Staff's conclusion that they do not anticipate demonstrable damages to inthyoplankton is unsubtractizated. See general comment 1 (5.4.2.2).

Page 5-6, Par. 4) Sunt. 3 -- No conclusive evidence is presented to substantiate the contention that high reproductive potential of girzard what and freshwater drum wake significant EN unlikely. Say general comment [(5.4.2.2.3).

Page 5-6; Per. 5; Bent. 2 -- Applicants have proposed entrainment studies after initiation of operation at resource agency insistence. The Department's intent in requiring such studies was to determine the nature and extent of DN. The Department contands that fish passage studies should be conducted at each project. See general comment 1 (5.4.2.7.3). 226. See response to comment #31.

227. Comment noted.

228. Contrary to the comment, staff has recommended mitigation for this impact -- turbine bypass flows to assure a constant movement of water in the turbine tallrace area during periods of shutdown. Such bypasses could be accomplished by special bypass structures in the powerhouse or by opening nearby gates, depending on site-specific characteristics.

229. See response to comment # 143.

230. Comment noted.

- 231. See response to comment #145.
- See responses to comments #206 and 208 on the subject of susceptibility of ichthyoplankton to entrainment damage.

233. See response to comment #155.

234. Staff appreciates the commentor's concurrence with staff's recommendation. However, it may not be necessary to quantify antrainment mortality (as opposed to rates of entrainment) at each site with similar hydropower facilities. Representative sites may suffice to establish the degree of survival for purposes of developing appropriate compensation or establishing the need for fish protection. ģ

Mg, Lois D. Castell Page 20 July 21, 1988

Page 5-6; Par. 3; Last Sent. -- Staff is not specific 235 regarding this facility, and it would be preseture to proceed with it until fish passage studies are completed at all projects. See general comment 1 (5.4.2.2.5).

Page 5-7: Pur. 1: Sent, 3: (4) -- The Department 18 responsible for fish management in West Virginia waters and will take appropriate measures to reduce angler harvant if deemed necessary. See general comment 11.

Page 5-7; Last. Par. -- Staff's conclusion that impacts to 237 represtional Lishing due to DM are not expected to be serious is speculation. Fish passage studies at all facilities are needed. See general comment 4 (5.4.2.2).

Page 5-8; Par. 2 -- Mitigative measures (fisherman access) during construction will provide some fishing recreation. It is not comparable to that now available because the access to be affected by construction are currently the bees fishing locations. Projects with limited federal lunds intensify this problem. Project areas must be expanded to accompodate both fishing during construction and recreational developments after construction. See general comment Iff.

Page 5-8; Par. 3 -- Several festures should be added to 239 basin-wide requirements for recreational development. See general comment 1 (5.4.2.3.1).

Page 5-23; Par. 1; Sent. 4 -- The Department will initiate 240 management measures to reduce any stress on fish populations caused by angler harvest at developed sites should it be deamed necessary. See

Page 5-73; Par. 2 and 3 -- The Department contends there are a 241 major differences between competing projects at the sites mentioned. See general comment VI.

- 235. Staff also believes it is premature to determine specifics of the bioengineering test facility at this time. The need for better information on entrainment rates at operating facilities is one set of information needed for definition of the goals of the test facility. However, it seems undesirable to wait until all projects are completed before proceeding to test mitigating devices, as the commentor suggested, for some projects may never be built and some may be delayed significantly for reasons outside the licensing process. Putting a test facility into operation soon could usefully test prototype fish protection devices for the Ohio River Basin projects on the basis of entrainment information now available.
- 236. Comment noted. Staff assumed that fish management would be the responsibility of the WVDNR.
- 237. Comment noted. Staff agrees that fish passage studies are needed; mitigation for fish entrainment and turbine-induced mortality is recommended in Section 5.4.2.2. Text has been modified in Section 5.1.1.3 to read "impacts to recreational fishing from turbine-induced mortality are not expected to be unmitigable, except where an embayment is located immediately upstream of the proposed project."

ç,

8

- 238. See responses to comments #161 and 217.
- 239. See response to comment #160.
- 240. See response to comment #236.

241. See response to comment #34.

Ma. Ecie D. Caeheli 7ege 2i July 21, 1988

Thank you for granting our request for an extension of the comment period so that we could adequately review this extensive document and have our comments made pert of the record. If you bhould the any questions conserting our recommendations, please contact Dr. Paul L. Nilf of my staff at (Jo4) 348-3761.

.

Gincerely yours,

Renald 1. Parente Director

BRP/mrw/b

- cci U.S. Environmental Protection Agency U.S. Fish and Wildlife Service Ohio Environmental Protection Agency ONSA-Division of Water Resources WVDRR-Division of Water Resources WVDRB-Office of Environmental and Regulatory Affairs

.

Service Corporation 1 Styles and Social Constitute Styles

	OFFER OF THE TOPE MAY		
	88 JUN 30 AM IO: 29	ł	
	FICULA: 47 FICTURESTON FUBLIC REFERENCE ROOM	g	
Me. 1 Acti	Lois D. Cashell ng Secretary	ŝ	
Føderal Energy Régulatory Commission 825 North Capitol Street, N.8. Washington, D.C. 20426			
June	29, L980	1:3	
Rei	Docket No. EL05-19-114 Rydroelectric Development in the Upper Ohio River	~	

Hydroelectric Development in the Upper Ohio River Basin Draft Environmental Impact Statement -- May 1988

Dear Ma, Cashelli

.

4194

d Engineering

These comments on the "Hydroelectric Development in the Upper Ohio River Basin Draft Environmental Impact Statement" dated May 1968 are provided by American Electric Power Service Corporation on behalf of the electric utilities of the AEP System. The AEP electric utilities serve customers in soven states including many in Ohio and West Virginia in the area effected by this report. Many of the residents and businesses of the upper Ohio River Valley are our customers and AEP has always had a strong interest in the prosperity and quality of life in the upper Chio River Valley. Many of the powerplants which serve the region are built along the river and its tributaries. Power produced efficiently on the Ohio River ultimately benefits customers in most of the states in the Ohio River Basin. The river serves both as a source of water for the generation process and as a highway for shipment of coal and other materials used by the power industry and other industries. One of the AEP divisions operates a river transportation barge to supply system powerplants with coal.

That the Chio River is one of the key resources in the region has long been recognized. It served as an early transportation artery. Navigation improvements over the

J-9)

¹Appalachian Power Company, Columbus Southern Power Company, Indiana Michigan Power Company, Kentucky Power Company, Kingsport Power Company, Michigan Power Company, Obio Power Company and Wheeling Power Company,

Ms. Lois D. Cashell June 29, 1988 Page 2

years have culminated in the present system of high-level dams which are the subject of applications for power additions and the subject of this draft Environmental Impact Statement (DEIS). Since the dams were built primarily for navigation, navigation takes primary importance, although other uses are accommodated and must be considered in the type of balancing that FERC does in licensing decisions. Flows in the river are largely controlled by releases at the various navigation dams by Corps lock masters in coordination with headwater releases from Corps flood control storage projects. The water available for hydropower generation is water which would otherwise be spilled. It is the question of evaluating the benefits from addition of power generation under certain regulatory condltions and the management of spill flows through or over dams which are the key considerations in the upper river hasin which this DEIS addresses.

The recommendations of the draft EIS would effectively cause eignificant alterations of the water quality criteria levels by attempting to maintain dissolved oxygen lovels of 6.5 mg/l or more by requiring spills at most dams during the low flow warm weather period of July through October.

We have serious reservations about whether the recommanded alternative goals are justified by the information and analysis developed in the DSIS. The rationale for the staff recommendation requiring spill flows at specified dama to maintain dissolved oxygen levels at 6.5 mg/l or above is apparently based on two lines of reasoning.

The first justification apparently is based on an attempt to take figures for dissolved oxygen levels derived by the U.S. Environmental Protection Agency (EPA) in their document "Ambient Water Quality for Dissolved Oxygen -April 1986" USEPA (1986) and apply the 6.5 level to achieve "no production impairment." The table presented on page 4-12 of the DEIS is taken from page 31 of USEPA (1986). However, table 8 on page 34 of USEPA (1986) contains the EPA recommended National Criteria for Dissolved Oxygen. The recommended warmwater criteria are 6.0 as a 7-day mean and 5.0 as a one day minimum to protect early life stages. Early life stages are defined as "all embryonic and larval stages and all juvenile forms to thirty days following hatching." The criteria for other life stages are set at 5.5 as a 30-day mean and 3.0 as a one day minimum. The figures in Table 1 reflect judgment levels set for national guidance for use by individual states in setting water quality standards to protect fish populations. The EPA criteria are also set out in Table 1 of the Dissolved Oxygen chapter in Quality Criteria for

242

242. Staff is familiar with the documents on DD criteria cited in the comment and had consulted with the author of each of the documents prior to preparation of the DEIS. The commentor raises valid questions about (1) the reliability of data that are used to establish criteria and (2) the varied recommendations that can be found in different documents and in different places in a single document (especially when a variety of durations is considered, such as instantaneous low concentration, 7-day mean, etc.). It is difficult and imprecise to define a "no effect" level on the basis of the existing literature. Because staff felt that it was beyond the scope of the EIS to undertake a reanalysis of the entire literature on DO effects, it relied on the most recent scientific analyses. Staff also took several different approaches to protecting against impacts of low DO: existing state standards, bipenergetics modeling to show cumulative and integrated effects over a year of growth, and a generalized "no effect" level. The "no effect" level that was selected is justified by the best available information and is believed by staff to be protective of all species.

MS. Lois D. Cashell June 29, 1988 Page 3

Water - 1986 published by SPA and referred to as the Goid Book. The criteria selected include allowance for worst case conditions and were judged to be adequate. However, where epacific conditions would likely lead to production impairment, higher levels were presented in Table 2 of the Gold Book chapter on dissolved oxygen which recommands dissolved oxygen levels for early life atages and other life at levels 0.5 higher.

Staff should perform a critical review of studies cited in USEPA (1985) for acceptability for criteria derivation. This criterion document does not contain such an evaluation, as other U.S. BPA criteria documents do (see for example the document for the metal copper). Without such a review, our experience has shown that criteria can be based on faulty data. Apparently the DEIS salects 5.5 the higher of the numbers listed on page 31 of USEPA (1986), as being justified for no production impairment and then applies it as a goal in recommended alternative 3. We view the application of the 6.5 criteria directly as a goal in these circumstances to be a misapplication of water quality criteria, particularly when, according to USEPA (1986), the production impairment levels are subjective. We note, for instance, that data cited (Raible 1975 and Andrews et al. 1973) in USEPA (1986) for channel catfish show that the existing criterion of 5 mg/l could very well protect this species.

The scheme for meeting alternative 3 goals would apparently 243 require spills from many proposed hydro power facilities to avoid modeled predictions of dissolved oxygen declines during what are perceived to be the critical months Juno through October. The "critical months" appear to be based on flow and temperature conditions during the time that the combination of lower flows, critical lower dissolved oxygen concentrations and higher temperatures usually occur but there is no statistical evaluation of the actual frequency of occurrence coincident of these three factors. Are large power losses through spillage necessary if such a frequency analysis shows that the conditions will occur, say, for a few days of a year? Spillage when critical conditions actually occur might be appropriate. Spillage apparently is recommended for the entire June through October period regardless of what actual dissolved oxygen levels are measured in the river.

The second line of reasoning is based on the predicted results of three-tiered modeling. First, predicted dissolved oxygen concentrations of the river were compared with state water quality standards. A second tier compared predicted dissolved oxygen levels with data from USEPA (1986). The question of the appropriateness of the

244

243. The critical season of July through October, when higher spill flows are recommended to protect water quality, was selected by reviewing historic dissolved oxygen data collected by ORSANCO.

The recommended mitigation requirements are designed to prevent the proposed projects from causing DO concentrations to go below 6.5 mg/L. Successful DO mitigation for the upper Ohio River basin projects must (1) protect DD concentrations over a wide variety of conditions (flow rates, temperatures, water quality, etc.), {2} be reliable enough that problems such as failure of DO monitors or water quality events do not allow projects to degrade 00, and (3) account for the cumulative effects of the multiple projects on DO. Staff proposes two alternative mitigation methods that meet the above three criteria. The mitigation method given most consideration is the requirement for spill flows during the season when DO concentrations are typically lowest. The spill flows recommended in Alternatives 3 and 4 of the EIS (Table 2.1.3-1) are designed to prevent DO concentrations from failing below 6.5 mg/L under a wide range of temperatures and flows. The spills were developed using a model simulating conditions when (1) river flows were approximately those at which the hydropower projects would have the greatest impact; (2) water temperatures are those exceeded only 10% of the time, as measured at the ORSANCO monitors: and (3) BOD concentrations are approximately those that occurred during a period in 1983 when DO concentrations were low. The spills were purposely determined using these conditions so that they should protect 00 adequately over 90 percent of the range of water temperatures measured at the ORSANCO monitors, over all flows, and whenever BOD loads are not extremely high. The spill flows are a reliable mitigation system because spill flows can be measured fairly

accurately and easily. The reliability of mitigation is increased by the recommendations that project operators provide real-time D0 monitoring data to water quality agencies (recommendation 4, Section 5.4.2.1) and that the Corps, ORSANCO, and the appropriate state water quality agencies be given authority to increase spill flows when necessary (recommendation 2, Section 5.4.2.1). The recommended spill flows account for the cumulative nature of D0 impacts because they were determined using a system-wide optimization model that considers the impacts of all upstream projects on the D0 at any point, and determines the combinations of spill flows that produces the most power in the basin (Section 2.1.3). The recommended spill flows are a simple, reliable, and easily enforced mitigation system that would protect D0 concentrations over a wide range of conditions.

The EIS allows an alternative to the recommended spill flows that also meets the three criteria for adequate mitigation. Recommendation 7 in Section 5.4.2.1 encourages development of a modeling system to determine instantaneous spill flow requirements based on real-time conditions, considering cumulative impacts of all the projects. The reliability of this system would result from the accumulation and analysis of information from throughout the upper Dhio River basin. Staff would be required to continually operate and interpret such a model. The model would intrinsically include the cumulative effects of hydropower on B0 concentrations at each point and could determine the most effective combination of spill flows. This mitigation system, if implemented, would result in lower spill flows than those recommended for Alternatives 3 and 4 under many conditions, but would

Several project applicants have simply proposed monitoring DD concentrations upstream of their projects and increasing spill flows whenever concentrations are low. Basing spill requirements only on DD concentrations monitored at a site does not meet the mitigation criteria. This mitigation method would not be reliable because spill flows at any site would be determined from readings from one or two DD monitors, which could be out of calibration or inoperative. Under some conditions, the lowest DD concentrations could occur at some point between hydropower projects, and therefore may not be detected by monitors. Basing spill requirements at a site on the DD concentration at that site also does not account for the cumulative impacts of upstream projects on DO. For example, low spill flows at several consecutive dams could cause high spills at a project downstream, when the most cost-effective solution may be to require moderate spill flows at all the projects.

J-94

244. Staff believes that actual growth rates predicted by the bioenergetics model are unlikely to be precise representations of what will actually occur in the river nor especially relevant to the analysis of hydropower impacts. It is the estimated changes in growth between the existing conditions and the conditions with hydropower that are meaningful. These changes can be estimated by using the same assumptions in model runs for both cases, even though the assumptions may not exactly match reality in the river. It is not possible to predict what a reduction in growth rate means for fish populations in the river, and this analysis was not attempted. The text and appendix have been expanded to help clarify the analysis and the significance of the results. Ms. Lois D. Cashell June 29, 1988 Page 4

use of that data was discussed above. A third tier involving a bioenergetics model compared relative fish growth rate predictions with different dissolved oxygen levels. The model was apparently based on a model developed from catfish pond studies with some adjustments made to attempt to simulate what is believed might occur in the Ohio River. The model predicts a relative decline in fish growth rates in certain Ohio River pools on a comparative basis without reporting the actual numbers predicted. The analysis stops short of addressing the question of what predicted growth rate decline would mean for populations. Insufficient information is presented in the DEIS to enable any meaningful evaluation of the modeling results to be made. However, based on the results from the modeling the DEIS predicts certain percentage declines in fish growth for certain fish species, including channel catfish and sauger and walleye based on data substituted in a modified model. The discussion of the modeling and the results on pages 4-15 through 4-18 is incomplete and confusing.

The DEIS proposes three alternative schemes in addition to those proposed by the dovelopers. Alternative 3 would require spills to be made at many proposed hydro power installations during the periods June through October, in order to avoid fish growth rate declines predicted to occur due to hydropower operations. The power loss for this alternative was calculated by the staff to be \$13 million per year. Neither the total amount of the predicted fish production decline nor its value was estimated in the DEIS.

In summary, we question the appropriateness of the criteria levels of dissolved oxygen selected, the appropriateness of the time frames to which they were applied, and the validity of the conclusions derived from the fish production modeling study.

As noted above, AEP has a vital interest in the upper Ohio River Valley and through the years has developed information from studies which it has conducted or supported. Several of these studies were provided to the Commission staff for use in this report, some of them in early publication form. The Bacine Project was licensed to Ohio Power Company and is operated by Appalachian Power Company, both AEP System companies. Data from the fishery study at the Racine Project, conducted jointly with the City of New Martinsville, was relied on to a great extent in evaluating fishery effects of installing bulb turbine hydroelectric units on the dams under consideration in this draft EIS. Although AEP has no applications pending before the Commission in this proceeding and does not plan

245

245. Without quantitative information on the fish biomass, age structure, and current productivity in the river, it is not possible to calculate more than a relative productivity change, as staff has done. Even that calculation is an advancement over anything altempted to date. To attempt to ascribe a monetary value to fish changes comparable to the monetary value of the hydropower forfeited is beyond the state of understanding. There is no requirement to assign mometary values to environmental resources. No. Lois D. Cashell June 29, 1988 Page 5

new hydroelectric development in the region, it has provided data to the Commission and has reviewed the draft BIS for information relating to reports it furnished. Our detailed comments highlight some of the items of specific concern. Our failure to comment further on additional parts of the report should not be taken as concurrence.

From the general concerns expressed above, we now turn to some specific comments. The following comments are offered in order to clarify or amplify certain parts of the report in the hopes that future use of the document may be enhanced.

Page 1-1, next-to-last full paragraphi

Delete words "combustion turbine" from "coal-fired combustion turbine capacity"; change "oil-fired combustion turbine capacity" to "oil/gas-fired combustion turbine capacity."

Page 1-4, Table 1.2-11

Add Toledo Edison Co., change Columbus and Southern Ohio Electric Co. to Columbus Southern Power Co. (AEP), and add (AEP) to Ohio Power Co., in the list of Utilities Serving Ohio Markets;

Add Indiana Nichigan Power Co. (AEP) and, if appropriate, Hoosier Energy Rural Electric Cooperative to list of Utilities Serving Indiana Markets;

Change Michigan Gas and Electric Co. (AEP) to Michigan Power Co. (AEP), and change Indiana and Michigan Electric Co. (AEP) to Indiana Michigan Power Co. (AEP), in list of Utilities Serving Michigan Markets;

If appropriate, add Big Rivers Electric Corporation to list of Utilities Serving Kentucky Markets;

Change affiliation of Penneylvania Power Co. (Ohio Edison Co.) from MAAC to BCAR in list of Utilitles Serving Western Penneylvania Markets;

Change Wheeling Electric Co. to Wheeling Power Co. (AEP) in list of Utilities Serving West Virginia Markets;

Page 1-5, second full paragraph:

246. Text has been modified.

247. The list of power producers has been revised.

Ms. Lois O. Cashell June 29, 1980 Page 5

The phrase "base load capacity from the proposed Basin projects" might cause some confusion. The hydroelectric generation in question will almost certainly be among the first energy resources dispatched on recipient utility systems, and is "base-load capacity" in that sense. However, the term "base-load capacity" also generally denotes resources which are dispatched at or near their maximum capacity, with little variation, for long periods of time. This would not be characteristic of the proposed run-ofriver hydro projects. We suggest replacing the paragraph with the following:

"Between 400 and 400 MW of capacity from the proposed Basin projects would be useful to regional utilities because the low-cost energy produced by these projects would displace energy produced by their most costly, least efficient load-following units (i.e., those highest in the loading order)."

Page 3-3 through 3-16

This section discusses long-term water quality and aquatic ecology trends in the Ohio River Basin. New technical publications provide important documentation on more recent conditions. Baclosed are copies of these recent publications which may be useful. Two papers authored by American Electric Power biologists (with others) (Van Hassel et al. 1988, Attachment 1; Reash and Van Hassel 1988, Attachment 2) provide statistical trends of historic water quality data and fisherias data. Analysis of water quality data, for instance, indicated that pH increased in the upper and middle Ohio River whereas fecal coliform counts and concentrations of ammonia, lead, and zinc decreased from mid-1970's to mid-1980's. Other publications onclosed provide information on navigation impacts to Ohio River fishery resources (Nielsen et al. 1986, Attachment 3) and mussel populations of the upper Ohio River (Zeto, et pl. 1987, Attachment 4).

Page 3-3

The third paragraph of section 3.1.3, particularly the first three sentences dealing with water temperatures, is misleading. Though powerplants discharge heated water, the entire river cross section is not affected, but only a limited area. Thus, increased river temperatures are limited spatially, and cooling is typically rapid longitudinally. The sentence "Wigh water temperatures reduce DO concentrations and inhibit growth of some fish species" is correct in itself, but not accurate if pertaining to power

248

c [249

£ 250

248. The text has been modified to incorporate the suggested revision.

249. Text has been modified to reference the papers by Van Hassel et al. and Reash and Van Hassel in Section 3.1.3.

250. Staff believes there is sufficient evidence that power plants significantly increase water temperatures. Even though individual plants on the Ohio River may not cause measurable decreases in DO concentration, the cumulative effects of the many power plants does increase temperature and reduce the saturation concentration of oxygen. There are individual power plants on tributaries whose impacts on DO concentrations are significant. Ms. Lois D. Cashell June 29, 1988 Page 7

plant effects. No powerplant on the Ohio River adds sufficient hoat to the river to cause any deseration. Results of site-specific fisheries monitoring studies sponsored by AEP and other electric utility companies (The Ohio River Ecological Research Program) have indicated that catch rates of fishes at thermai-influenced stations are typically greater or not significantly different than catch rates upstream of power plants. The most recent results of the study (1987 Ohio River Ecological Research Program) will be sent to FERC shortly.)

Page 3-20

Under the state lists of threatened and endangered species [25] [page 3-20], the channel darter <u>(Percina copelandi</u>) should be added as this species is protected in Ohio. The silver chub should be deleted from the list. The species is the most common chub in the upper Ohio River and the Ohio DNR has removed this species from endangered status. One reference on page 3-20 (WAPORA, Inc., 1986) is not listed in the references section.

Page 3-29 states:

"When hydropowar facilities are added to a dam much of the river flow is routed through a turbine and is no longer spilled over the creat or through the gates of the dam. The result can be a net loss of oxygen input to the river because hydropower turbines provide little seration (AEP, 1969; 1987)."

Page 4-1 states:

"Studies on the Ohio River have shown that little if any meration takes place at existing hydropower plants when river flows are diverted through the powerhouse (AEP: 1969: 1987)."

AEPSC believes that these statements are misleading with respect to the results obtained during AEPSC's DO study conducted at the Racine Hydroelectric Project in 1987. The study at Racine found that the turbine discharges did increase downstream DO levels in the river by as much as 0.2 to 0.6 mg/l. It was also found that the turbine discharges provided increases in downstream DO levels equal to and greater than the dam discharges alone. Although the DO increases found at Racine may not be considered large increases, we believe that it should not be assumed that hydroelectric plant discharges provide no 251. The list has been corrected.

252. The BO model simulates hydropower operation at a dam by assuming that the flow through the turbine receives no aeration. This assumption is based on (1) information in the references cited in Section 4.1.1.1, and (2) the observation that in the kinds of hydropower turbines proposed, although high turbulence occurs there is no source of bubbles sufficient to aerate well. Minor increases in DO concentration may actually occur at hydropower turbines, and these increases may be comparable to the amount of aeration occurring at some of the submerged discharge gated dams that also provide little aeration. However, the amount of aeration that may occur at hydropower turbines is minor compared to other sources of 80, so the assumption that no aeration occurs in the turbines is reasonable, and conservative because any deviation from the assumption would lead to higher, not lower, water quality. Staff avoids conclusions based on the assumption that the dams with deeply submerged outflows (such as Opekiska, Willow Island, Belleville, and Callipolis) provide more aeration than hydropower plants would.

ę,

ŝ

He. Lois D. Cashell June 29, 1988 Page 8

increases in downstream DO levels. Potential DO increases from hydroelectric discharges should be given consideration in the DO modeling effort and any flow restrictions recommended for a specific project.

Section 4.1.2.2.1--Estimating Flow Regimes of Tailwaters

Staff should use caution applying mathematical models to tailwater areas to predict velocity changes in the tailwater. Figure 4.1.2-4 without powerhouse flows are not representative of flow patterns at Racine and may not be for other Ohio River sites. Based on hydraulic modeling by Alden Research Raboratory (Massachusetts) for the Racine Project, flow patterns will be similar from gated dam releases and the powerhouse at the same river flow (CFS).

253

254

Exhibits 1 and 2 show modeled tailwater flow patterns from the Recine dam and powerhouse at 30,600 CPS. This river flow is close to the 30,000 CPS flow used in Figure 4.1.2-4. The flow patterns in Exhibits 1 and 2 are similar because at the 30,600 CFS flow, only two of the dam gates are open. These modeling results have been field-verified by the Corps and AEP staff.

Section 4.1.2.2.3--Habitat Losses

Staff is correct that river flow is discharged through only one or a few dam gates now at low river flow. The statement needs clarification that gated flows are shifted operationally in contrast to the turbine discharge at a single location. The Corps now maintains a standard schedule of gate openings to maintain pool elevations at various river flows. Gates are sequentially opened for specific amounts of openings. For instance, at a low river flow, the center gate is open one foot. At higher river flows, a second gate may be opened one foot.

Habitat downstream of the dam gates will always receive the same gated release for a given river flow. At Racine dam, where there are eight Tainter gates, for river flows which would pass through the hydro project the Corps would open gate 5 up to 5 feet followed by gates 3 and 7. Fell river width skilt water habitat is therefore not changed with hydro project operation at gated dams.

Section 4.1.2.3--Entrainment and Turbine-Induced Fish Mortality

- 253. Staff is aware that physical model studies will be necessary to accurately predict downstream conditions, and a recommendation for such model studies has been included in the EiS for purposes of habitat and recreation management (see Sect. 5.4). Staff used existing models and experience at other sites to estimate tailwater velocity distributions. The commentor raises an important point -- that the "without hydro" flow regime depends greatly on the number and location of gates that are opened under current operating conditions. The Corps has a gate opening schedule for each dam that accommendates all flows. When there is but one or a few gate(s) open, the downstream velocity regime may resemble the single-point discharge of a hydropower turbine. This was noted in the Section 4.1.2.2.3, but not in the section 4.1.2.2.i to reflect this fact.
- 254. See response to comment #253. This section has been clarified to reflect the Corps' preset schedule of gate openings.

5

Ms. Lois D. Cashell June 29, 1980 Page 9

We are pleased Staff agreed with many of the concepts contained in the fish survival report for Racine. The following comments should be used to clarify Staff's statement.

Staff may want to replace Figure 4.1.2-6 with Exhibit 3 to Improve the presentation of fish passage through wicket gates and runners. The Exhibit 3 was more useful at the Staff's fish passage workshop in Charleston as well as at other presentations we have made.

255

256

257

258

259

260

Staff should not use fish survival results from Dadawell et al. 1985 at the Annapolis Tidal Project to predict fish survival at Ohio River projects. Ohio River and Annapolis projects will use horizontal Rapian turbines, but the similarities and there. Both wicket gates and runner blades will probably be adjustable in Ohio River projects; at the Annapolis only the wicket gates are adjustable because the blades on the rim generator are fixed for structural reasons. Secause of these design differences, turbine efficiency is less for less operating time for the Annapolis Project compared to the turbine efficiency of Ohio River projects. Exhibit 4 111ustrates these differences due to design. Higher turbine efficiency for longer operating times will yield higher fish survival. Turbine efficiency at Annapolis is further reduced because the head, changing on tides, is only at the one point for maximum efficiency for a relative short time. Because the head changes relatively slowly on the Corps-regulated Ohio River, efficiency is maintained longer.

We are not certain of Staff's basis for the 10% mortality estimate for shad and drum. The Racine report shows only 6% mortality. Staff's statement regarding more than 10% of game fish damaged should be clarified so that the percentage is not a mortality estimate. Fish which are struck by the blades at Racine can and do survive.

We believe that quantitative data from the 19-year Ohio River Ecological Research Program (cited by Staff as ESE 1987) support Staff's qualitative predictions that losses of gizzard shad and freshwater drum at Racine will not affect their populations near Racine. Based on the 9-year pre- and 5-year post start-up lisherles data downstream of Racine, the project has not detectably affected populations of the fish community.

We are pleased that Staff recognizes that any fish protective device needs to be biologically effective (1.e., yields survival rates greater than Survival rates of entrained fish) before it should be considered for installation. Staff is also correct that the engineering 255. Staff agrees that the figure provided by AEP is superior to the one used in the DEIS, and it is being used as Figure 4.1.2-6 in the FEIS.

256. Staff recognizes the differences between the Annapolis (Nova Scotia) tidal power and Ractne turbine efficiencies. Text has been modified in Section 4.1.2.3.2 to explain the differences.

257. Staff attempted to generalize from the Racine results to indicate that a range of mortalities from about 10 percent to none could be expected from such projects. The sentence about gamefish has been clarified.

258, Comment noted.

259. Comment noted.

250. Comment noted.

0-100

Ms. Lois D. Cashell June 29, 1988 Page 10

practicality and costs of anch devices needs to be evaluated and considered.

Page 4-55

As suggested in the top paragraph on pages 4-55, each 261 licensee should be required to have physical model studies to examine flow patterns and determine the structure design that would avoid undesirable flow patterns and so assure that the project does not cause significant impact to barga lockage. This should take the form of an additional Basin-wide Recommendation under Section 5.4.2.5, perhaps condition 10 dealing with a requirement to furnish a flow model atudy.

We thank you for the opportunity to provide comments and hope that you will find these additional materials useful,

Sincerely 20ES -

Robert W. Reeves

RWR/mac Attachments 251. The Corps is the agency principally responsible for maintaining navigation. Because the Corps has indicated to FERC staff that they will specify and require the modeling studies to avoid navigation impacts of the proposed projects, no additional recommendation for such studies has been included in the EIS.

Four scientific papers provided as attachments by American Electric Power Company are not reproduced here.

UNITED STATES OF AMERICA FEDERAL ENERGY RECULATORY COMMISSION

In re Hydroelectric Development in the Upper Ohio Fiver Basin) }	Docket No. EL 85-19-114. Project Nos. 7914-D03, 7909-D02, 4474-D03, 4017-D02, 7307-000, 7399-D01, 8990-000, 8652-001, 7660-000, 8900-D00, 8675-000, 7041-001, 7566-001, 2971-002, 3490-003, 8901-D01
		10332-000, 3218-001, 6902-003, 9999-000, 6939-001, 9042-000, 10098-000, 698-001

MOTION TO INTERVENE OF AMERICAN RIVERS, INC. AND FRIENDS OF THE EARTH AND COMMENTS ON DBAFT ENVIRONMENTAL IMPACT STATEMENT

American Rivers, Inc. and Friende of the Earth (Intervences) hereby move pursuant to 18 CFR 385.214 and 18 CFR 380.10 for leave to intervene in Doc at Ho. EL85-19-114 and in each of the project proceedings listed above, and are submit the following elements on the Commission's Draft Environmental Impatt Statement (DEIS), "Hydroelectric Development in the Upper Chiu River Bases," issued May 1988. In support of this motion, intervenors state as following

STATEMENTS OF INTEREST

J-102

American Sivers, Inc. 15 a not-for-profit corporation organized under the Laws of the District of Columbia with offices located at 801 Fennsylvania Avenue, SE, in Machington DC. The mission of American Rivern is to preserve the nation's putstanding rivers and their landscapes. With approximately 10,000 members merods the country, American Rivers is the mation's principal river conservation organization. Approximately 700 of American Fivers' sembers reside in the States of Fennsylvania, Ohio and West Virginia. Friends of the Earth is a not-for-profit corporation incorporated under the laws of the State of New York, and has its principal offices at F30 Seventh Street, 5E, in Washington DC. Friends of the Earth has approximately 15,000 members in the United States and is affiliated with conservation organizations in thirty-one different countries. Friends of the Earth has taken an active interest in hydroelectric licensing issues for many years, and has a particular interest in ensuring the proper implementation of the Electric Consumers Protection Act of 1986.

Intervenors are entitled to intervene in these propeedings because their members use and enjoy the upper Ohio Elver and immediately adjacent lands for fixhing and other forms of represented. The decision whether or not to permit construction of the projects that are the subject of the DEIS, and the terms and conditions under which such projects might be licensed, will directly affect the interests of Intervenors and their members. In periodiar, the proposed projects could have a significant adverse effect on fish and other aquatic and riparian wildlife, and Intervenors' members therefore with to ensure that noise of these projects is approved until the mandate of the National Environmental Folicy Act has been satisfied and other applicable environmental requirements are net.

Intervenors also are entitled to intervene in these proceedings because their participation in these proceedings is in the public interest. As national creationations, intervenous' participation will ensure that the value of the upper Ohio River and the importance of the issues raised by this USIS are properly reflected in these proceedings. In addition, because Intervenous have renegatized exercise in the issue and policies relating to hydroelectric development, intervenous will bring valuable experience to the issues raised by

-- 2 --

these proceedings. The Completion already has recognized on numerous occasions that the participation of Intervenues in Completion proceedings serves the public interest.

Finally, Intervenors are entitled to intervene in these proceedings because they are expressly authorized to do so in 18 CFR 380.10.

COMMENTS 1. The PEIS proposes to license projects in violation of the Federal

262

Power Act. In <u>Confederated Tribes and Bands of the Takima Indian Nation v.</u> FERM, 746 5.24 466, 570 (9th Cir. 1984), the Court of Appeals for the Ninth Circuit addressed the question whether or not the Commission could satisfy its obligations under the Federal Power Act "by deferring consideration and implementation of fishery protection measures until after licensing." The Court answered the question by stating: "FERC must consider fishery insues before, not after, issuance of a license." Id. at 371 (Emphasis in original.)

The DEIN indicates that the Commission intends to violate the holding in <u>Tekina</u> by proceeding with the licensing and construction of projects before fundamental fisheries issues have been resolved. For example, the DEIN - recognizes that entrainment and turbine-induced fish mortality is a potentially significant adverse effect of constructing projects in the upper Ohio River, and that no adequate data are currently available to assess these impacts. (IENS at 4-24 to 4-34.) Father than proposing to develop a plan for resolving these questions prior to licensing, however, staff proposes that the Commission license the projects, and that possible changes in project structures and operations to considered after extensive study of the actual impacts of the projects and the establishment of a bioengineering test facility for fish 262. See the response to comment 449.

-- 3 --

bypass systems For the Ohio River. (DEIS at 5-27 to 5-28.) This "power fiint/fith last" approach is flatly inconsistent with <u>Yakira. See also</u> House Report No. 99-507, 59th Cong. 2d Seas. 17-23 (1986).

Furthermore, the plan of action proposed in the DEIS violates the Commission's mandate under meetion %(e) of the Federal Power Act, as amended by the Electric Consumers Protection Act, to give "equal consideration" to power generation and to figheries, 16 U.S.C. 797(e), and violates section 10(j)(1) or the Act, which requires each license to "include conditions for the protection, mitigation, and enhancement" in order to "adequately and equitably protect, miligation, and enhance figh and wildlife (including related spawning groupiz and habitat)." 16 U.S.C. 803(j)(1).

263

264

265

On the one hand, the DSIS gives practically no consideration to whether or not the proposed projects are needed to fill an actual physical demand for power, or to the environmental consequences of displacing other generating capacity or possibly avoiding the construction of alternative generating sources. So far as one can determine from the Commission's surcery analysis, the projects apparently would displace other. Functioning generating equipment, and not fill an actual need for power. (Sectoring, DEIS at 1-4 to 1-5, 2-27 to 2-28.)

On the other hand, the Complexion chooses to ignore the substantial, lorgely unanswered questions about the effects the projects would have on fish and other aquatic species. In short, the Complexion is undistakably placing a higher value on power than fish and other natural values, and thereby violating the Federal Fower Act's "equal consideration" mandate.

2. The DELT viblates the National Environmental Policy Act. For several

-- - --

263. The analysis in the FEIS provides information concerning significant trade-offs between hydropower development and environmental quality among the alternatives considered by staff. Non-hydroelactric and non-generating alternatives and the no-action alternative were also analyzed. Staff has provided recommendations in the FEIS on environmental protection of resources, including fish and wildlife, related to hydropower development in the study area. The information presented in the FEIS will be part of the record from which the Commission will make its decision. The Commission, before making a decision on issuing licenses for the projects, will take into account all concerns relevant to the public interest, to include giving equal consideration to power generation and fish and wildlife resources.

Though the technical feasibility of proposed devices for fish protection is undemonstrated and questionable for the Ohio River Basin conditions, staff believes it is reasonable to require that these mitigation devices be tested in prototype at operating facilities before they are mandated for installation at all sites. Such a phased implementation approach is standard practice for introducing any new technology into service. Staff believes that investment by developers in such a test facility, with commitment to consider effective devices for later installation if needed, is consistent with "equal consideration" for power generation and fisheries.

264. The DtiS states that it is in the public interest and useful to conserve non-renewable fossil fuels and to reduce atmospheric pollution; displacing steam generation with hydropower will accomplish both. The displacement of inefficient steam generation by hydroelectric generation improves the cost-effectiveness of the displacement. The addition of 400 megawatts of hydropower would reduce the consumption of non-renewable fossil fuels, reduce atmospheric pollution in a nation deeply concerned about acid rain and the greenhouse effect, and possibly reduce long-term energy production costs.

265. Comment noted. See response to comment #49 and 263.

reasons, the FEIS is inadequate to satisfy the requirements of the National Environmental Policy Act.

First, the DELS does not analyze potential cumulative impacts in accordance with the guidelines of the Council on Environmental Quality. The Council's guidelines, which are binding on the Commission, are 18 CFR 360.1. define "cumulative impact" as an impact which results from the impact of a proposed action "when added to other part, present, and reasonably foreseeable future actions." 40 CFR 1508.7. Undar this definition, the Commission, in considering the effects of projects for which license applications are perding, has a legal duty also to take into account the effects of already constructed obstructions in the river and of projects for which preliminary permit applications have been filed. <u>Compare National Wildlife Federation v. FERC</u>, Bon F.2d 1505 (9th Cir. 1986) (Commission has duty at proliminary permit stage either to prepare comprehensive plan or initiate environmental studies). The Commission has rejected out of hard performing a cumulative impact analysis of the type required. (DEIS at 1-5 to 1-6.)

Second, the BEIS is in violation of NEPA because the Commission proposes [267] to license the projects and then turn to an evaluation of the environmental consequences of that decision after the projects are constructed. As discussed above, the Commission's approach of "power first/fish last" violates the Federal Power Act. It also violates the National Environmental folicy Act. As the Court of Appeals for the Ninth Circuit has stated, reliance on postlicensing studies violates the requirement of NEPA "that consideration of the environmental impacts of proposed projects take place before any licensing decision is made." LaFlamme v. FERC. No. 65-7571 (9th Eir. Issued March 18, 1988)

-- 5 ---

266. See responses to comments #32 and 33.

266

267. See response to comment #49.

J-106

Finally, the DEIS violates NEPA because the Combission has not gathered sufficient information to intelligently assess the environmental consequences of the proposed projects. In particular, the DEIS is based on inadequate information about current dissolved oxygen levels in the river and the likely consequences for dissolved oxygen levels of building the proposed projects. If the Commission issues licenses for a number of the proposed projects, significant fish Rills and losses of other aquatic life could result through the increased stress from decreases in dissolved oxygen at low flow periods. (See DEIS, at 4-5 to -6, 4-8 to -10, 5-2 to -5). The Commission must seek additional information regarding current and projected dissolved oxygen levels and the effectiveness of mitigation measures prior to project licensing. (See DEIS, at 4-9 to -10, 5-2.) In the absence of this information, the Commission is simply not in a position to make a reasoned judgment.

268

269

3. The DEIS Cannot Substitute for Formal Consultations Under the Endengered Species Act. The portion of the Ohio river being studied provides habitat for the pink mucket, <u>Legosilis ubrupts</u>, which is included on the Federal endangered species list. The DEIS states (at 4-87) that the DEIS is being provided to the U.S. Fish and Wildlife Service to comply with the consultation requirements of Section 7 of the Endangered Species Act. While it is appropriate in principle to integrate the NEFA process with the endangered species consultation process, simply distributing the DEIS for comment is insufficient to comply with the detailed procedural and outstantive requirements of the Endangered Species Act, 16 U.S.C. 1531 <u>et seq</u>., and the Act's implementing regulations, <u>ase</u> 50 CFR 402.01 <u>et seq</u>.

4. The DEIS and Comments on the DEIS Cannot Satisfy the Federal Power Act Section 10(1) Process. The DEIS states (at 4-87) that the Commission staff

-- 8 --

268. Staif does not believe that collection of additional information on 00 is required to dosign mitigation to avoid impacts to aquatic life (see response to comment #39). The spill flows recommended by staff would assure that aeration would be maintained during extremely low flows; few of the projects at dams that aerate well would operate at all during very low flows (compare the spill flow recommendations in Table 2.1.3-1 and the minimum turbine flows in Table 2.3.1-1 to the monthly mean flows in Table 3.3.1-1; the spill flow plus the minimum turbine flow must exceed the flow in the river before generation can occur). The response to comment f243 discusses the effectiveness of the proped in the during means of the means.

269. See response to comment #38. Additional information on <u>Lampsilis</u> <u>abrupts</u>, the federally listed enclangered freshwater mussel, is included in Appendix 1 of the FEIS and has been submitted to the USFWS in compliance with Section 7(c) of the Endangered Species Act.

- will use its opportunity to respond to comments submitted on the DEIS by state 270 and federal fisheries agencies "as the mechanism by which to comply with Section 10(j) of the FPA." Intervenors object to this proposed procedure because the DEIS does not contain sufficiently detailed information, including proposed license terms and conditions, for the fish and wildlife agencies to recommend appropriate terms and conditions is accordance with the purposes of section 10(j). Furthermore, should the Commission identify any inconsistency pursuant to section 16(j)(2), the EIS process would be inadequate to comply with the EDPA resolution requirements and would fail to give the necessary opportunity for fish and wildlife agencies and intervenors to participate. The formission should therefore keep the section 16(j) process open until such time as the gaps in the DEIS are filled in and more detailed, project specific information has been developed.
- 270. See response to comment #31. Staff's recommendations have been developed after considering all comments and recommendations filed to date on the 24 license applications, including comments and recommendations received on the DEIS. Staff has provided both basinwide and site-specific recommendations, based on information available in the geographical study area and from project-specific information.

CONCLOSION

For the foregoing reasons, the Commission should grant intervenors' motion to intervene in these proceedings and consider the foregoing comments. In addition, the Commission should proceed to prepare a revised or supplemental DEIS that responds to the foregoing concerns.

Respectfully submitted,

John D. Echeverria, Esquire Azérican Rivers, Inc. 803 Fennsylvania, S.E. Vashington, D.C. 20003 (202) 547-6300

David R. Friends of the Earth

Friends of the Earth 530 Seventh Street, S.E. Washington, D.C. 20003 (202) 543-4312

July 5, 1958.

CERTIFICATE OF SERVICE

I hereby certify that I have this 5th day of July 1988, served a copy of the foregoing document by first-class mail, postage prepaid on each of the persons listed on the official service lists maintained by the Secretary in each of the proceedings listed in the caption of this document.

AVG U Schererie John B. Faheverrie

-- 9 ---

.

Durson L. Johnson

Allowing Ar Law

P. C. Roy 14927 Margindown, West Vagroom 40505 July 22, 1988

- Islaphane 304 296 3393

Lois D. Cashell Acting Secretary United States of America Federal Energy Regulatory Comm. Washington, D.C. 20425

> Re: Hydroelectric Development Docket No. EL65-19 114 Round Data

Dear Ms. Cashell:

We are the owners of the ROUND BOTTOM track, which comprises 410 271 acres and three miles of river frontage and uncompasses the Hilderbrand Lock and Dom on the east bonk of the Monorganela River at the 106, 107, and 108 mile markers. You sent us a notice of a public hearing in Pittsburgh on July 15, 1988 at 10:00 a.m. The problem, however, is that we did not receive the notice until July 18.

We do, at this time, make a formal objection of any hydroelectric development on the Hilderbrand Lock and Dam in that our property abuts against said dam. The highest end best use of our property is for a high tech residential area. We have designed and commenced construction of roads for the development of said high tech residential area, and the hydroelectric development would be severely adverse to our development. Additionally, electric power is overabundent in West Virginia, and in fact, most of the electricity produced by West Penn Power and other utilities is sold out of our state.

Enclosed is a recent article from our local newspaper which reflects that the Governor has granted us an industrial roadway for the construction of our high tech residential area.

Please advise us timely as to any future public meetings so that we can go on record with the court reporter.

Please advise us accordingly.

Thank you kindly.

Respectfully yours.

Darwin F. Johnson

JUL 2 7 WAR

2808030550

271. This comment notes an individual's objection to the proposed Hildebrand project because of impacts on a "high tech residential area" under construction in the vicinity. The comment does not specify the nature of the adverse impacts anticipated, nor does it include any details that would permit staff to assess the impacts. Staff attempted to obtain additional information through a telephone call on August 30, 1988. The commenter declined to provide sufficient information on the basis of a telephone contact and indicated that he might supply it if officially requested by a letter. Unfortunately, the schedule for preparing the FEIS did not permit Staff to pursue this possible method of ubtaining the needed information.

Sections 4.1.6.1 and 4.1.6.3 have been revised to include the potential impacts of the Hildebrand project on proposed residential development in the vicinity. The potential impacts include possible encroachment of some project facilities on land proposed for residential development and possible exposure of the residential area to noise, dust, and traffic during project construction. Recommendation 6 in Section 5.4.3 has been revised to call for compensation for deterioration of privately owned roads used by construction traffic at this project.

A newspaper article provided by Mr. Johnson is not reproduced here.



OHIO RIVER VALLEY WATER SANITATION COMMISSION

AB CAST FOURTH BTREET, CINCINNATE ONIO ASSOR

ALAN N. VICORT JR., P.E.

July 6, 1968

Hereitante Alexandre Annotation (Complexity) Hereitante Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426

RE: FERC Docket No. EL83-19-114

Dear Ms. Cashell:

The Draft Environmental Impact Statement (DEIS) on Hydroelectric Development in the Upper Ohio River Basin has been reviewed by staff of the Ohio River Valley Water Sanitation Commission (ORSANCO). The DEIS considers the cumulative impacts of 24 proposed hydroelectric projects at 19 sites on the Allegheny, Monongahela, Muskingum, and Ohio Rivers and evaluates several alternatives for mitigation of identified impacts. The recommended alternative would permit hydropover development at 19 of the 19 sites.

Personnel from Oak Ridge National Laboratory who prepared the DEIS consulted frequencly with ORSANCO staff, and made several presentations before this Commission's Technical Committee, which consists of representatives of eight states, including the six states along the Ohio River, U.S. Army Corps of Engineers, U.S. Geological Survey and U.S. EPA. This has provided ORSANCO with opportunities to review the work in progress and the mathodologies used. We are pleased to see the degres to which environmental protection is supported in the final recommendations.

At its Hay 26, 1988 meeting, the Commission adopted the following policy concerning dissolved oxygen monitoring at Chio River hydroelectric fatilities:

Operating licenses for Ohio River hydropower facilities should contain provisions to assure that:

- Adequate studies are conducted prior to facility operation to define seration characteristics of the dam;
- Continuous monitoring of discolved oxygen is provided at representative locations above and below the facility as appropriate, with data available to ORSANCO through remote interrogations;
- Frovisions are made in the facility design and operation to allow maintenance of full service potential of the dam during critical conditions.

272. Comment noted.

272

273. Comment noted.

Ms. Lois D. Cashell July 6, 1988 Page 2.

Specific application of this policy to individual hydropower projects shall be on a case-by-case basis.

It is our opinion that the recommendations of the DEIS are fully in concert with this policy.

On June 30, 1987, under the scoping process of the DEIS preparation. this Commission submitted comments which included the request that the study eras be expanded to include the entire wain stem of the Ohio River. That request was based on our discolved oxygen monitoring results which show that the major problems, and hence the area most sensitive to effects of hydropower on aeration at navigation dams, have been on the middle and lower Ohio River, downstream of the study eres. While the study area was not expanded, several of the recommendations can and should be applied to all Ohio River projects.

Another recommendation in our June 30, 1987 statement was that coordinated control of hydropower operations be investigated. We believe that the seventh recommendation under 5.4.2.1 - Recommendations on Water Quality - begins to address that concern.

One of the problems encountered in preparing the DEIS was that critical dissolved oxygen conditions did not occur during the period when applicants were requested to perform field measurements. In applying the above stated policy of the Commission, it is envisioned that additional data collection would be required to fulfill the first requirement. Given the current drought situation in the Ohio Valley, this summer should provide an excellent opportunity for such data collection.

Given the time frame available for submitting comments on the DEIS, it was detarmined by this Commission's Technical Committee that it would not be practical to attempt to develop a single set of comments on behalf of this Commission and its member states. The states will therefore submit their own individual comments.

We oppreciate the opportunity to comment on the DEIS, and the degree to which George Taylor of your staff as well as the Oak Ridge personnel have maintained communication with ORSANCO during the preparation period.

Harth Visaig .

Copy to: George Taylor

274. See responses to comments #32 and 33.

275. Comment noted.

276. Staff did not believe it was necessary to request project applicants collect additional dam aeration data in the summer of 1988. Staff has reviewed the extensive data collected in 1988 by the Pittsburgh District of the Corps; however, the conditions occurring in the summer of 1988 are so unusual (compared to historic flow rates, water temperatures, etc.) that staff does not believe the data should be used to represent the system. For example, stratification of pools that have rarely before stratified radically changes the apparent aeration rate at some dams.

J-112

COMMENTS OF

ALLEGHENT ELECTRIC COOPERATIVE, INC. LICENSE APPLICANT FOR PERC PROJECT NO. 2971-002 MONTGOMERY HYDROELECTRIC PROJECT

ON

HIDROELECTRIC DEVELOPMENT ON THE UPPER RIVER BASIN FERC DOCKET NO. EL85-19-114

DRAFT ENVIRONMENTAL IMPACT STATEMENT

JULY 1988

1. OVERVIEW

Allegheny Electric Cooperative has read, with interest, the FERC Draft Environmental Impact Statement (DEIS) on hydroelectric development in the Upper Ohio River Basin. The cumulative impact assessment prepared by Oak Ridge National Laboratories (ORNL) on behalf of the FERC is, as expected, a comprehensive document. The DEIS identifies a battery of potential environmental issues and assesses each issue at seemingly the level of current knowledge.

Whereas the DEIS would lead one to believe that the scope of the DEIS was born of the official scoping meetings, held in Fittsburgh, this does not appear to be the case. It seems evident to Allegheny that some of the real decisions were made after the two "informal" West Virginia meetings, the attendance and importance of which is somewhat of a mystery. It appears that many substantive decisions were made at these meetings, apart from limited scoping, that the Applicants had little input about what direction the study should take and what are the priorities. Obviously, from the beginning, dissolved oxygen (DO) was the real

2

277

277. Staff benefited from various sources of information in the preparation of the FEIS. Information provided in project applications, including documentation of applicant and agency consultations, was an important source. Staff used scoping meetings and public meetings to obtain comments on the scope of the EIS and to determine the issues to be discussed and analyzed. Staff also used the information filed by the applicants in response to staff's additional information requests. The responses provided staff with a standard base of information to conduct its analyses for the proposed projects in the study area. Staff meetings with applicants, agencies, and organizations provided the opportunity to obtain information and for staff to present the analytical procedures used in the EIS. Staff assessed all available information and used its best professional judgement in completing its own independent analysis for the proposed projects.

agency issue, with fisheries and wetlands taking a close second. However, no attempt was made in the DEIS to consider future changes to the river system or the effects of such system by other actions apart from hydroelectric generation.

Principally, and perhaps the greatest deficiency of the L 278 DEIS is, in Allegheny's opinion, the principal assumption in the development of Alternatives that all hydroelectric developments will be developed in accordance with the license application plans. In short, the assessment was strictly limited to the recent bast hydro development environment and the further assumption that all hydro would be developed with no provisions for change. Certainly, this is a marrow and very loose interpretation of the real world and perhaps the fatal study assumption. It is ludicrous for the FERC or any other developmental agency to assume that all litensees, upstream or downstream of a particular site, have the capability, financing, and power purchase market available, to develop a hydroelectric project and that all of these sites will be developed within the exact same time frame predicated by the DEIS. This report acts as merely a snapshot of development in a very unlikely scenario that all license applicants will develop hydro plants as outlined in speculative 'icense applications borne of the tax and FURPA incentive legislation of the late seventies.

Allegheny Electric Cooperative, as an operating, generating utility and licensee of a FERC Project in central Pennsylvania, is well aware of the responsibilities, capabilities and mitigation activities associated with the coordinated devel278. See response to comment #32 and 292. Staff concludes that at least one proposed project at each site would have positive net benefits under the recommended alternative and therefore assumes that all sites would be developed, if licensed. opment of hydro facilities. The rather narrow stance taken by the FERC staff in assessing economics and abilities of the economic capability of all licensees to develop and sell hydroelectric power to the intended market, is a very shallow interpretation. Refinement of the DEIS into a final report should, at a minimum, include a case whereby the likelihood of development is assessed and alternative scenarios based on the capabilities and probabilities of development, which would include the generatiop and development of Montgomery as a viable project with Allegheny, an operating utility, as one developer. More discussion on this is contained later in Section III.

Section II discusses the specific technical issues related to Montgomery. At Mortgomery, significant adverse impacts were identified for fisheries, wetland, and recreation. Fisheries (related to perceived dissolved oxygen (DO) and turbine mortality impacts) and wetlands (Montgomery embayment) were expected to be an issue since as early as our original agency contacts in 1980, these were the identified agency concerns.

Recreation, as a major impact, was something of a 279 surprise in that the proximity of an active railroad and the steepness of the shoreline limits recreational fishing potential at the site currently, and provides safety hazards for access continually. We believe that the recreational fishing issue, as it relates to Hontgomery, may have been overemphasized. The Pennsylvania Fish Commission has made previous statements about Nontgomery being the cest tailrace fishing on the Pennsylvania portion of the Ohio River, however, data in Allegheny's license

Ł

279. Recreational fishing is not only valued in terms of the number of users that can be accommodated at a given site, but also in terms of the quality of the recreational fishing possible at that site. For example, both fishing success rates and species composition influence the quality of recreational fishing at a site. The high quality of recreational fishing at a site. The high quality of recreational fishing at a site. In addition, the physical constraints at Nontgomery do not warrant special consideration as they are not as limiting as those found at other project sites in the study area.

application indicates that many more people fish the Dashield Pool due to relative proximity to the Pennsylvania Fish Commission boat ramp. The physical constraints at Montgomery are mentioned in the DEIS, but never appropriately discounted and related to actual fishing ability. This is just one example of the narrowness of the DEIS on site-specific issues.

5

J-117

COMMENTS OF

ALLEGHENT ELECTRIC COOPERATIVE, INC. LICENSE APPLICANT FOR FERC PROJECT NO. 2971-002 MONTGOMERI HIDROELECTRIC PROJECT

ON

HIDROELECTRIC DEVELOPMENT ON THE UPPER RIVER BASIN PERC DOCKET NO. EL85-19-114

DRAFT ENVIRONMENTAL IMPACT STATEMENT

JULY 1988

II. SPECIFIC COMMENTS ON THE MONTGOMERY HIDROELECTRIC PROJECT

A. Dissolved Oxygen (DO)

According to the DEIS under Alternative Nos. 1 and 280 2, Montgomery has a moderate impact on DO. Alternative No. 3, a 4,000 to 16,000 cfs spillage, allows for development, but at reduced energy output.

The no-hydropower-development scenario (DEIS Figures 4.1.1-7 and 8-16) indicates about 0.5 to 1 mg/1 of DO is added by Montgomery Dam reastration during summer months. Allegheny's extensive field surveys in 1987 generally agree with this data. However, based upon Allegheny's 1987 data, which included a flow less than 7Q10, Ohio River DO was always high throughout the water column even upstream of the dam.

Allegheny questions how the model data showing low Ohio River DO (e.g. less than 5.0 or 6.5 mg/l; DEIS Figures 4.1.22, 4.1.2-3) can justifiably be taken as the benchmark to preclude hydro development when recent, rigorously controlled, DO data collected in 1987, during an adverse DO period, indicate 280. Several state and federal agencies also commented that DD analyses should be based on measured data instead of modeling. As stated in the response to comment #39, field measurements are inadequate for a complete analysis of project impacts because data from only a few years are inadequate to describe the range of potential conditions and DD concentrations. Data collected in 1987 do not necessarily represent conditions under which hydropower would have the greatest impact; simulation of such conditions is required to determine adequate spill flows for mitigation. In addition, field measurements of DD do not provide a means of predicting project impacts. Even though data measured in 1987 show adequate DD concentrations at Nontgomery, staff is confident that had the projects proposed upstream of Nontgomery would have been much lower. The model is required to assess the cumulative impacts of all the projects.

The data collected by the Corps in 1983 was the best available set of data for calibration of the model. Temperatures, flow rates, and DO concentrations were measured at 108 sites (see Appendix B), during a short time period. The data collected by applicants in 1987 were used to model dam aeration rates, as part of the basin-wide DO model (the data are presented in reduced form in Appendix B of the FEIS). Data collected by project applicants in 1987 were not used for calibration of the basin-wide model because (1) measurements were not made at all the dams, since some applicants had previously provided the requested dam aeration data; (2) measurements were not made in the pools between dams, as they were in the 1983 Corps data; and (3) the measurements made by the applicants were not all taken during the same time period.

6

high ambient DD. At a minimum, some discussion of actual 1987 field conditions at all the projects should have been graphically displayed in the main text and Appendix B.

Models are notorious for providing numbers, good or bad. Controlled field studies provide actual, albeit limited, glimpses of the real world. Somehow the real world seems to have been lost amidst all the assumptions inherent in the model and the basis upon which it was calibrated (Corps 1983 data), The field data demonstrate that DO was not a problem in 1987 above or below Montgomery Dam either day or night or at high or low flow. In 1987, hydro development would not have adversely affected DO due to existing high saturation above the Dam.

Allegheny questions use of the Corps 1983 data to calibrate its model. Allegheny has not seen the 1983 data on which much of the DEIS's is based, but included in the Montgomery Application the 1984 to 1982 Corps data. If 1983 data was collected as in previous years, it represents a one upstream and one downstream profile taken irrespective of river flow, temperature, and time of day. Some earlier Corps upstream/downstream data were taken as much as one full day spart in earlier years. Allegheny questions the use of limited Corps data when a more recent and comprehensive data base was directed to be taken and available to FERC. If the model was calibrated based upon a high DO deficit (DEIS page B-10), then model results will reflect that condition. High DO deficits were not observed in 1987 field studies at Kontgomery Dam.

-7

In summary, DO modeling results and 1987 field

data collected at Montgomery Dam present radically different pictures of what hydro development would mean in terms of DO at this site. Allegheny questions the absence of any text or graphics depicting 1987 field DO concentration data which was provided to FERC in a timely and efficient manner.

B. Fisheries

The preceding discussion of DO modeling and, 1987 281 field results obviously has direct bearing upon the assessment of DO effects upon Ohio River fishes. Based upon the tiered analysis presented (DEIS page 4-11 to 4-18), had hydro been in operation at Montgomery in 1987, there would have been no adverse effect upon growth of channel catfish, sauger or walleye.

Entrainment and turbine mortality at Montgomery Dam is more of a potential due to the proximity of the Montgomery embayment. Turbine mortality questions remain unanswered after the Racine and Greenup/Vanceburg studies (and probably will forever and ever if the agencies have to have indisputable evidence). <u>Note</u>: On DEIS page 4-25, it is interesting that the Corps and WVDNR believe there is little mortality in passing through existing gates given they have no data to support that opinion.

In Allegheny's opinion, the entrainment/turbine mortality issue is a no-win situation at all hydro sites, given the fisheries recommendations at DEIS page 5-27, since it implies mortality studies, screening, monitoring, etc. without purpose other than to preclude development. 281. Staff is also concerned that highly certain and site-specific answers to the questions of entrainment impacts (e.g., numbers of fish entrained, degree of damage and mortality, effects on populations, and effectiveness of fish protection devices) will not be forthcoming even after expending large amounts of effort and funds. Management decisions must be made on the basis of incomplete evidence, and cooperatively among developers, resource agencies, and regulators. Until there are better data than at present, however, some data collection and prototype testing of mitigation devices is necessary at representative sites before reasoned management decisions can be made. Stocking is clearly one option for mitigation, but it is not accepted by all resource agencies as appropriate.

J-119

8

In Allegheny's opinion, field studies, no matter how comprehensive, may never answer these questions to the agencies' full satisfaction since there have been numerous studies with no conclusive results to show significant mortality. Stocking, as compensation for fish mortality, is a reasonable recommendation, one which, within limits, benefits all.

C. Recreation

There are conflicting statements and conclusions 282 in this portion of the DEIS. The recreation section stresses both need for angler access and a "desirable concentration of sport fishes in the public fishing areas." Ergo, fish mortality due to angling is OK, but mortality due to turbine passage is unacceptable. Also increase access at Montgomery will increase angling pressure in the form of anglers' boats, motor oil, waves, pollution in the embayment (a FWS Resource Category 1). Allegheny questions whether the FWS and PFC consider this impact acceptable.

The minimum recreational development includes; a 283 fishing pier, paved walkways, socess to riverward coffers, fish attractant structures, parking access paths, restrooms, fish cleaning shelter, provisions for handicapped use, solid waste disposal, lighting and drinking water. Allegheny specifically challenges the siting of these facilities at Montgomery, given the active railFoad and steepness of grade. In the event such facilities cannot be located at the development, the DEIS stipulates that offsite compensatory facilities be proposed. *Is* a license Applicant, Allegheny is amenable to some mitigation for 282. If necessary, state agencies will initiate management measures to reduce any stress on fish populations caused by angler harvest at developed recreational fishing sites. The provision of shoreline fishing access at the project tailrace should have no affect on boat angling pressure in the embayment upstream.

283. FERC regulations require that the developer construct, maintain, and operate a standard level of recreational development (refer to recommendation 1, Section 5.4.2.3). Site-specific constraints prohibiting this level of recreational development would need to be discussed with the appropriate federal and state agencies in order that recreational compensation could be determined and a revised plan filed with the Commission.

9

J-120

the benefit of sport flahing, however, to be required to improve upon a condition at Montgomery which does not, in its current form, support natural access, is extortion.

Additionally, the DEIS calls for compensation 284 during project construction (e.g. temporary facilities). Safety issues such as construction equipment and fishermen using the same access road makes this provision impractical.

Lastly, compensatory flow releases are recommended 285 to be made for the newly-created shoreline fishing during the operational phase when the powerhouse is shutdown for maintenance. A flow of between 500 and 2,000 cfs is required as well as aeration capability and safe fish passage features. Once again, the agencies are overreaching the existing condition.

It may be worth noting that the only way fish routinely bypass the dam at present is over the fixed weir, through the gates or through lockage, all of which have some degree of attendant mortality.

D. <u>Wetlands</u>

Potential impacts to the Montgomery embayment are 286 given on DEIS page 4-52. A course of action which has been considered by Allegheny includes a proposal to conduct a pre-license, physical model study by which protection of the embayment from sediment aggradation or degradation can be demonstrated.

The DEIS seems to state that the Montgomery Project will destroy this area, although DEIS does not provide one thread of evidence to support this view. About all DEIS says is that project impacts would be "unacceptable." This position 285. See response to comment #14.

286. The proximity of the Montgomery Embayment and the associated shoreline wetlands is such that staff believes that at least one acre of wetlands would be affected during construction. Diversion of flows from the embayment area will lead to changes in species composition and areal extent of the wetlands. During construction, a potential also exists for damage and possible destruction from increased erosion, turbidity, and sedimentation. Aerial photographs and navigation charts for the area were used to assess the potential changes in wetlands. Staff calculated that at least 0.5 acres of emergent wetland would be affected directly and loss of about 0.5 acres due to increased velocities, erosion, sedimentation and turbidity is not unreasonable. Because of the high value of this embayment and associated wetlands as a nursery and feeding area for fish, development of a project at this site is not recommended until further studies result in adequate mitigation for these potential impacts.

1-121

10

once again discounts the fact that Allegheny has committed to a program that would avoid or mitigate any impacts to the embayment. DEIS concedes there will be no change in water elevation. This only leaves construction impacts and velocity. Allegheny's plan for the construction area <u>does not approach</u> into the embayment and only affects <u>one-half acre</u> of submerged wetland habitat (not 1 acre as stated on DEIS, page 5-9). A model study may show that velocities are nondetectable.

Allegheny just recently completed construction of 287 a hydro project adjacent to a wetland and is pleased to report whe ducks, beaver, snapping turtles and cattails are doing just fine.

Allegheny is troubled that DBIS automatically 288 assumes that the embayment will be impacted. It is interesting to note that while the DEIS refers to such impacts as "unacceptable" or "unavoidable" DEIS does not consider them as resulting in irreversible and irretrievable commitment of resources (see Section 4.9).

In terms of the wetlands issue, another mitigation 289 plan which potentially would complement, not replace, the above proposal might be as follows. At one of our earlier agendy meetings, someone made the offhand remark that the embayment was preated in the 1930's (?) from spoil excavated during the construction of the dam. Allegheny dannot vouch for the accuracy of that statement. However, assuming that it might be true, it is possible to propose to do the same thing with dredge spoil (assuming the sediment chemistry tests do not indicate polluted 287. Comment noted.

- 288. The loss of 0.5 acre of emergent wetlands due to construction would be irretrievable and irreplaceable. Changes in species composition and areal extent would result in one ecosystem being replaced over time by another ecosystem, thereby changing the character of the site.
- 289. It is perhaps possible to restore or reconstruct new wetland areas using clean spoil material as a mitigative measure. Studies of such potential mitigation are recommended before licensing (Section 5.4.1).
sediment). Rather than study and say No, the FERC and agencies should prioritize mitigation activities and work with Applicante on these obvious problems.

E. Endangered/Threatened Species

There are remarks regarding Montgomery on DEIS 290 page 4-62, last paragraph, but due to a typographical error, the statement makes no sense. If, as expected, the statement refers to the attraction of baid eagles to the tailrace to feed on dead fish and subsequent mortality due to electrocution from the transmission lines. Allegheny challenges the likelihood of such mortality. Certainly hydro and baid eagles coexist in Maine, the Pacific Northwest and Alaska. Have baid eagles been adversely impacted elsewhere? Certainly Allegheny could highlight the minor spans of the transmission line (no river crossings are contemplated) with markers so that it would be avoided by any eagle, osprey or heron that might choose to feed in the tailwater.

F. Mitigation

The DEIS appeared to rely heavily on the Corps' 291 1983 data in which all other data was calibrated as well. As previously stated, Allegheny sees little adjustment for the more comprehensive 1987 data base provided which shows higher DO readings. Because of this, DEIS takes the position that the only way the Montgemery Project would be allowed to operate would be if a spill flow of 16,000 cfs was maintained during the fourmonth period, July through October. It is interesting to note that this 16,000 cfs figure is 12,800 more spillage than requested by the Pennsylvania Fish Commission. It is also 10,300 cfs 290. Staff concerns about the baid eagle being attracted by entrained fish are related to the proximity of the proposed project site to an urban area, which create additional hazards to the endangered raptors. Baid eagles and hydropower plants can co-exist in rural locations away from urban populations. The typographical error has been fixed in Section 4.1.6.2.

291. See the response to comment #280 concerning calibration data. The spill flow recommendations were based on a cumulative analysis of many projects' impacts on DO; the recommendations by fish and wildlife agencies may not have considered all the proposed projects at once. The reasons why staff does not recommend spill flows at a site be based on instantaneous conditions at that site are discussed in the response to comment #243. Staff has included a recommendation (recommendation 7, Section 5.4.2.1) that allows for real-time determination of spill flows based on basin-wide water quality conditions.

. 5

more than requested by the U.S. Fish and Wildlife Service in their review of Allegheny's license application. Staff justifies this high flow as that amount required to maintain DO levels above 6.5 mg/l. It would make more sense to restrict project operations when DO levels dropped below 6.5 mg/l, regardless of flow. Further, the DEIS disallows any benefits that may be derived from mechanical or artificial seration.

On DEIS page 5-24, a spill flow of 16,000 cfs (July-October) and 4,000 cfs (November-June) was recommended if the project.was to be licensed. Allegheny suggests that spillage up to 16,000 cfs should be required <u>after</u> upstream DO dropped to 7.0 mg/1. That is, if as in 1987, the river upstream was near DO saturation, there is no biological or BOD rationale for which spillage is required. The high ambient DO concentrations would merely be transported downstream without the need for apillage.

COMMENTS OF

ALLEGHENY ELECTRIC COOPERATIVE, INC. LICENSE APPLICANT FOR FERC PROJECT NO. 2971-002 MONTGOMERY HIDROELECTRIC PROJECT

ON

HIDROELECTRIC DEVELOPMENT ON THE UPPER RIVER BASIN PERC DOCKET NO. EL85-19-114

DRAFT ENVIRONMENTAL IMPACT STATEMENT

JULY 1988

III. ECONOMIC ANALYSIS

The cursory analysis of economic impact presented in 292 the DEIS accepts the license application data as presented in the license applications for cost purposes, however, inaccurately compares the value of the project against a power output from a "generic coal-fired steam electric plant in the Ohio River Valley." This approach is far from reality. In the instant case, the Montgomery Hydroelectric Project would be developed by Allegheny Electric Cooperative, an <u>operating utility</u> (as distinguished from most of the other applicants in the DEIS) in a nonprofit manner for the benefit of over 170,000 consumers in Fennsylvania and New Jersey. The power generated would be used to offset purchases from private power companies at <u>real</u> not generic or proposed wnolesale power cost rates.

In developing the estimated levelized annual cost of the project, Allegheny includes interest, depreciation, FERC Annual Charges, O&M, Administrative & General, insurance, wheeling and taxes, all escelating at utility trend rates. The levelized annual value of the project is based on offsetting wholesale 292. The economic benefits that would be realized from development of hydropower plants on the Ohio River are shown in Table 5.2-1. For each developer, the financial feasibility would be based on actual contracts for the purchase of power, which will not be available until after the projects are licensed.

14

purchases from our major power supplier (Penelec) and escalates according to our projections of Penelec's wholesale rate to Allegheny. Based on this analysis, and using the FERC criteria of a 9% discount rate. Allegheny computes the lavelized net benefit of the project under Alternative 1 at 30.7 mills/kWh not 7.7 mills/kWh (DEIS Table 2.1.1-3). Under Alternative 3, Allegheny computes the levelized net benefit at 12.8 mills/kWh not (-6.2 mills/kWh (DEIS Table 5.3.1). Clearly the project remains viable in Allegheny's economic analysis, if licensed.

Finally, although FERC staff recognizes that the financial attractiveness of projects is under question, no attempt is made in the DEIS to ascertain the financial viability and capability of a license applicant to accurately develop and construct, operate, and maintain a project. Instead, a most unlikely scenario, utilizing the blanket assumption that all projects will be developed in exactly the same time frame and that the cumulative impacts will occur, is projected without regard to real life financing or the current downward trend in hydro development because of FURPA and loss of tax benefits. No sensitivity runs of a development strategy, where one or more of the projects are not developed, is presented. Allegheny believes that the study compounds the economic impact of no hydro development in the Upper Dhio River Basin with the currently depressed economic condition, and reports those Findings as another equally unlikely scenario.

COMMENTS OF

ALLEGHENY ELECTRIC COOPERATIVE, INC. LICENSE APPLICANT FOR PERC PROJECT NO. 2971-002 Montgomeri Hydroelectric Project

ON

HIDROELECTRIC DEVELOPMENT ON THE UPPER RIVER BASIN PERC DOCKET NO. EL85-19-114

DRAFT ENVIRONMENTAL INPACT STATEMENT

JULY 1988

IV. SUMMARY

In summary, Allegheny does not concur with the conclusions sions and recommendations reached in the DEIS and the conclusions as to the specific Montgomery Hydroelectric project for the following reasons:

A. The DEIS is a comprehensive but extremely narrow 293 snapshot of hydroelectric development in the Upper Ohio River Basin. Some of the major assumptions do not take into account past, current, or future trends in the hydro industry, the environment, or the economic area; slthough the study purports to be expansive.

B. While the environmental data and modeling attempts 294 to predict DO water quality data, it is calibrated on 1983 data and in no way verifies field data taken by Applicants in the critical low flow periods in 1987 which snow that the water quality is not as bad as originally perceived by all.

C. The DEIS is flawed in that it assumes throughout 295 that <u>ell</u> projects will be developed in the <u>same</u> time frame, despite industry trends and the realities of hydro development success, to the contrary.

293. Comment noted.

J-127

294. See response to comment #280.

. 295. The four hydepower alternatives evaluated in the DEIS contain

295. The four hydopower alternatives evaluated in the officient contain different combinations of project development. To assess the impacts of this construction, staff has assumed that the projects will come on line in a reasonable time frame and that unforeseen delays will be held to a minimum. D. The DEIS presents a very simplistic view of the 296 economics of hydro development and in the instant case of Montgomery, a prejudiced and inaccurate one, not taking into account the true benefits to the developer of long-term reliable power for Pennsylvanians.

E. That the recommendations for mitigation, while 297 some are possible, are so comprehensive and in no way attempt to prioritize resources and are so egregious that the future for any hydro development in the region is doubtful and no attempt is made to negotiate reasonable economic solutions to develop clean, long-term, reliable and indigenous power resources. 296. Comments noted.

297. Staff believes that its recommendations allow for the development of hydropower projects in the upper Ohio River Basin in an environmentally acceptable manner. Staff's recommendations are based on the importance of certain environmental resources in the study area that would be cumulatively impacted by hydropower development and the importance of other environmental resources at specific hydropower sites.

.

WHERE US THE SECRETARY

8

ŝ

UNITED STATES OF AMERICA B8 JUL - 5 PM 3: 32 BEFORE THE B6 JUL - 5 PM 3: 32 FEDERAL ENERGY REGULATORY COMMISSION FLORENCE FOR SUM FOLDER FROM SUM

Hydroelectric Development In) The Upper Ohio River Basin) Docket No. E185-19-114

> COMMENTS OF ALLEGHENY HYDROPOWER, INC. ON DRAFT ENVIRONMENTAL IMPACT STATEMENT

Pursuant to the Notice of Availability published in the Federal Register in the above captioned proceeding on MM 20, 1988 (53 Fed. Reg. 18131), Allegheny Hydropower, Inc. ("Allegheny"), project sponsor of the Allegheny River Lock and Dam No. 7 project, Project No. 7914 (L&D No. 7), hereby respectfully submits its comments on the Draft Environmental Impact Statement on Hydroelectric Davelopment in the Upper Ohio River Basin (DEIS).

I.

SUMMARY OF POSITION

As the project sponsor of the Allegheny Lock and Dam No. 7 project (Project No. 7917), Allegheny Hydropower, Inc. has a direct interest in the outcome of this proceeding and in the recommendations made in the final Environmental Impact Statement on Hydroelectric Development in the Upper Ohio River Basin.

Allegheny believes that the DEIS fails to adequately take into consideration virious potential mitigative measures which, when undertaken, would serve to adequately address the concerns for short and long-term effects to the riverine environment expressed in the DEIS. Allegheny further believes that the DEIS overstates L&D No. 7's impact, especially in light of the fact that four adjacent projects (L&D Nes. 5, 6, 8 and 9) which are nearly identical to L&D No. 7 have already been licensed. Finally, Allegheny believes that its project is in fact compatible with the development of other projects in the Upper Ohio River Basin, and may in fact enhance and provide additional recreational resources to the public benefit.

11.

COMMENTS

Following are Allegheny's specific comments pertaining to various statements and recommendations contained in the draft EIS to which Allegheny seeks to respond.

Comment 1

On pages 1-6 and 3-38 the DEIS correctly points out that 298 licenses have been issued for L&D Nos. 5, 6, 8 and 9 and construction has begun on L&D Nos. 5 and 6. However, the DEIS fails to point out that other projects already licensed are nearly identical in design and configuration to L&D No. 7, whose license application is now pending. Furthermore, the license applications for other similar adjacent projects were originally submitted only a few months before Allegheny submitted its application for L&D No. 7. The final EIS needs to address this matter in greater detail, as it goes towards the issue of estab298. Staff is aware that there are other licensed projects and pending projects in the study area with similar design and configuration to LAD No. 7. Staff is also aware that some of these pending projects filled license applications at an approximate time of the LAD No. 7 filling. Staff must assess the environmental impacts of the pending proposals and the environmental impacts of alternatives to the proposals prior to the Commissions's decision on licensing actions. Staff has documented in the FLIS the cumulative Interaction occurring with multiple hydropower projects in the study area and has assessed the environmental impacts attributed to each of the pending projects. The application for Allegheny River LAD No. 7, as a pending project in the study area, warranted inclusion in the Els analysis. J-130

lishing cumulative impact. Failure to address why similar projects applied for within months of each other have been treated disparately brings the EIS' analysis process into question.

Connents 2

The DEIS currently places substantial emphasis on the 299 subject projects' individual and cumulative impact on wetlands and biota habitat. However, the DEIS incorrectly concludes that L&D No. 7 will cause irreparable harm to wetlands and shallowwater habitat because no alternative mitigative measures are capable of alleviating the impact of turbine tailwater discharge and discharge channel dredging as currently proposed (<u>see G.g.</u> DEIS §4.1.2.2.3 at 4-23; §5.1.1.2 at 5-5). These statements need to be clarified in the final EIS and staff must recognize the existence of alternative mitigative measures which will adequately address these issues.

The DEIS expresses concern that the L4D No 7 project as currently proposed will cause direct "significant, unavoidable adverse impacts to the target resources identified as fisheries, recreation and wetlands" on and around the Isle of White, a 14 acre shoal located downstream from L4D No. 7 (DEIS §5.4.1 at 5-23). The concern is that as currently proposed, tailrace dredging during construction, and subsequent turbine tailrace discharge just below the upper tip of the Isle of White, will result in the potential removal of and significant erosion of the

-3-

299. Staff used all known information in conducting its analysis for L&D No. 7. See responses to comments #300 through 316. island with time {DEIS $\{5,1,1,4\}$ at 5-8, 5-9}. Furthermore, the DEIS is concerned that currently proposed project flow rates and patterns would alter existing flow regimes and possibly alter the survival and establishment of certain wetland species. (Id.)

The DEIS' analysis of this potential problem, however, is seriously flawed. Staff, in preparing the DEIS, either was unaware of or chose to ignore the existence of potential alternatives which could result in a plan with minimal cumulative impact best adapted to the comprehensive development of LSD No. 7 for beneficial purposes, and in compliance with the recommendations of the Commission and other resource agencies.

An example of a possible alternative which would result in an "alternative orientation of the turbine discharge...capable of avoiding significant, adverse habitat loss" (DEIS at 54.12.2.3 at 4-23 and 55.1.1.2 at 5-5) would be the installation of a Vertical Kaplan Siphon Turbine.

The Vertical Kaplan Siphon Turbine would avoid or effectively mitigate the direct adverse impacts with which the DEIS is concerned because it would parmit the reorienting of turbine tailrace discharge away from the Isle of White and into the main center channel of the Allegheny River. This reorientation is possible because the Vertical Kaplan Siphon Turbine employs a compound draft tube which can accommodate any horizontal discharge angle. Vertical discharge angle would depend on submergence requirements, and on the design of the hydraulic smoothness of the draft tube.

-4-

300. Alternative designs that deflect flow toward the center of the river would increase velocities in the vicinity of the mavigation lock. Navigation traffic could be adversely affected by increased velocities and cross currents. Disruption of navigation is not an acceptable proposal for this site. Because the sweeping angle of the draft tube could be directed towards the center of the main channel, it would permit dredging further upstream and more riverward compared to that required of a conventionally oriented horizontal bulb turbine configuration. The result would be that tailrace dredging during construction would remove little if any of the upstream section of the Isle of White shoal and associated shallow-water habitat.

Furthermore, a vertical Kaplan Siphon Turbine requires a shorter overall powerhouse length than conventional horizontal turbine installations. This is because water is siphoned into a turbine scroll case via a vacuum pump, and then directed downward onto the turbine runner blades. Discharge is conveyed through the draft tube and into the turbine tailrace. Consequently, the vertical turbine shaft and runner blade location would be much farther upstream and closer to the axis of the dam than would be the case with a conventional horizontal turbine installation. The effect of this is that excavation and associated dredging in order to maintain submergence requirements in the tailrace are significantly lens, and therefore, the impact of medimentation, turbidity and erosion on and around the Isle of White could be effectively mitigated.

In addition, other possible and practical methods exist for avoiding significant, adverse wetland and habitat loss in and around the Tale of White. For example, the placement of a deflector in the lower end of the tailrace would direct turbine tailrace discharge into the main center channel of the river and 301. See response to comment #300.

. .

-5-

away from the Isle of White. The Pennsylvania Fish Commission recognized this alternative mitigative measure in its comments on LSD No. 7 submitted to the Commission in 1985, and stated that it "would help to protect the island and also mix the turbine discharge across the river more quickly." (See Letter from Jack G. Miller, Chief Figheries Environmental Services, Pennsylvania Fish Commission to Kenneth F. Plumb, January 11, 1985).

Accordingly, the DEIS incorrectly concluded that there is 302 "no alternative orientation of the turbine discharge...capable of avoiding significant, adverse habitat loss." The final EIS needs to investigate this issue further in light of the alternatives which Allegheny has presented.

Comment_3

The DEIS states that staff is unawars of any off-site 303 compensation that could mitigate potential adverse environmental impacts. (DEIS §5.4.1 at 5-23). The staff however, appears to ignore the suggestion of the United States Environmental Protection Agency that adversely impacted wetlands be replaced on an acre by acre basis. (Seg Letter from John R. Pomporio, Chief Environmental Impact and Harine Policy Branch, U.S. E.P.A. to Kenneth F. Plumb, January 11, 1985.) The final EIS should address and consider the acre by acre replacement of adversely impacted wetlands as a possible and practical mitigation alternative. 302. See response to comment #300.

303. Staff has recommended that monitoring plans for wetlands be developed (Section 5.4.2.4). Staff recommends that such plans be developed in consultation with appropriate state and federal agencies before licensing at Allegheny L&D No. 7 (Section 5.4.1).

۰.

-6-

Comment 4

220-954 0 - 88 - 2

D

The DEIS finds that flow rates and flow patterns caused by the operation of L&D No. 7 as proposed would alter existing flow regimes and may alter the survival and establishment of some wetland species. This finding however, does not take into consideration the availability of alternative turbine tailrace discharge orientations presented by Allegheny in Comment 2.

304

The ability to direct turbine tailrace discharge flows away from the Isle of White and its associated wetlands and shallowwater habitats, and into the main center channel of the Allegheny River, could effectively mitigate or avoid the short and longterm effects of aquatic habitat loss and erosion on and around the island. Adverse impacts could be further mitigated or avoided by the installation of deflectors such as rip-rap or sheet piles at the downstream, lower end of the tailrace. The final EIS should address and take into consideration these additional mitigative measures.

Comment 5

The ability to direct turbine tailrace discharge into the main center channel of the Allegheny River as presented in Comment 2, would represent a better adopted use for hydroelectric production in an environmentally acceptable manner as it would result in a larger percentage of river water being diverted into the main river channel below L&D No. 7. Because a large

-7-

304. The developer will need to provide a detailed plan for the orientation of the turbines in order to properly assess the possible impacts an wetland areas in the tailrace channel. Staff agrees that, if the turbines can be oriented in such a manner that the island and shoal areas are not impacted, short and long-term impacts due to the discharge velocities and patterns could be mitigated. However, see response to comment #300.

305. See response to comment #300.

percentage of river flow occurs in this channel, and key aquatic species thrive in this particular flow regime, the reorientation of turbine tailrace discharge into the main channel would enhance and benefit the survival of these key aquatic species and therefore, enhance and benefit recreational fishing.

Comment 6

Alleghenv believes that section 4 of the DEIS fails to place 305 sufficient emphasis on the value and use of physical hydraulic models in evaluating the cumulative and site specific impacts of the projects reviewed. Alleghenv proposes to develop a physical hydraulic model of its L4D No. 7 project. Use of such a model would not only address the affects of powerhouse operations, but it will also model different flow rates at the dam, lock and powerhouse relative to the overall effect on flow patterns. velocities and elevations, sedimentation, turbidity and erosion, aquatic habitat, wetlands and navigation. The model would essist in maximizing the best location of the power house, submerged dikas, riverward and landward coffer cells, bypass facilities, and fishing piers. In the event the model studies result in negative impact predictions, Allegheny would propose to develop appropriate mitigative measures for approval by the Corps of Engineers and appropriate resource agencies.

Furthermore, the Commission itself has made the use of a 307 physical hydraulic model a requirement for the development of a post licensing study plan to determine mitigative measures in 306. Staff does not believe that sufficient mitigation to prevent the significant adverse impacts identified at Allegheny LD 7 can be designed with physical hydraulic modeling to recommend licensing of the project at this time. The project's impacts on nearby wetlands and shoals could be modeled only with a movable-bed physical model (simulating sediment transport); such models are expensive and complicated to use. However, the recommended action on this project (Section 5.4.1) allows the applicant to pursue studies that could lead to adequate mitigation designs. See response to comment #2.

307. Staff does not recommend development of a proposed project it believes would have significant impacts that may not be mitigable even if physical modeling is used, unless such mitigation is demonstrated before licensing. prior license proceedings. (<u>see e.g.</u> Article 40 of L&D No. 5's License, 29 FERC § 62,028 at 63,056 (1984)).

Allegheny believes that proper formulation and calibration of the physical model will be key to the success of its LED No. 7 project, especially in light of the availability of an alternative turbine configuration as set forth in Comment 2.

Comment.7

The DEIS correctly acknowledges that L&D No. 7 is a poor 308 dissolved oxygen (DO) aerator (DEIS §3.3.2 at 3-40), but ove.emphasizes L&D No. 7's cumulative impact on DO.

The DEIS finds that reductions in the rate at which desirable water quality constituents such as DO will be added to the river and undesirable toxic constituents removed will be impacted by hydropower generation at LSD No. 7. The DEIS however, fails to adequately address the fact that since LSD No. 7 is, by the DEIS' own admission, the worst DO aerator of all the projects under review, hydropower operations at LED No. 7 will have little if any impact on the maintenance of DO levels and the removal of volatile compounds such as chloroform and ammonia.

Comment 8

The DEIS has determined that the project/pool area of L4D 309 No. 7 has the third lowest level of recreational land area of all the projects under review (DEIS 54.1.3.1 at 4-38), and classifies

-9-

308. Staff has discussed the low aeration rate at Allegheny L&D 7 in Sections 3.3.2 and 4.1.1.1. It cannot be assumed that this project would have little impact on the removal of volatile chemicals because the mechanisms controlling the transfer of volatile chemicals from water to air are not identical to those controlling the transfer of oxygen from air to water. Deep plunging of bubbles, which improves aeration, may reduce removal of volatile chemicals.

309. Although the recreational fishing enhancements proposed at the site would be beneficial to recreational users, the enhancements do not compensate for the potential loss of the Isle of White recreational refuge.

the downstream shoreline area as highly disturbed with little vegetation. (DEIS §3.3.5.1 at 3-47). The DEIS however, does not adequately address the beneficial enhancement to recreational resources that Allegheny has proposed to develop at L&D No. 7.

Furthermore, as Allegheny has established in Comment 5, the use of physical model studies would determine the best recreational enhancement configuration, including, but not limited to, such items as the need for shoreline undulations, underwater deflectors, terraces, rockpiles, or dikes in order to further enhance recreational opportunities while maintaining a high sensitivity to the protection of the existing equation resources on and around the Isle of White and the Allegheny River within the vicinity of L&D No. 7. The foregoing would be conducted with consultation and comment from appropriate resources agencies. The final EIS needs to further develop and consider these beneficial impacts.

<u>Comment 9</u>

Section 4.1.4.2 incorrectly finds that Allegheny will remove 330 excavated construction material by barge and that "constant" barge traffic will contribute to serious damage to vegetation on the Isle of White. This section is incorrect and should be removed from the final EIS.

Allagheny does not propose to use barges to remove excavated material, but in fact proposes an onshore removal system for excavated and dredged material during construction. As Allegheny 310. Text has been modified. However, removal of spoil material on shore would cause additional adverse impacts to the neighborhood in which the project would be placed. J-138

has stated in its application, the excavated material has commercial value. Accordingly, Allegheny proposes to process excavated material onshore and remove it by the use of trucks (See E.41 of Allegheny's L&D No. 7 application). Processing onshore and hauling the material offsite alleviates the need for barges and their associated adverse impacts on the Isle of White and the river within reach of the construction site. Accordingly, the -final EIS needs to reconsider the impacts of construction at L&D No. 7.

Comment 10

Sections 4.1.6.1 and 4.1.6.3 overemphasize the direct impact construction at the site will have on the surrounding neighborhood. Allegheny would propose to mitigate the concerns contained in the DEIS by restricting working hours to weekdays from 8:00 a.m. to 6:00 p.m. Allegheny would further propose to mitigate construction related impacts associated with traffic, residential interference and roadway deterioration through the use of calcium chloride or spraying of water to control dust on roads and storage piles, cover haul trucks, limit construction traffic and access to particular roads, and the encouragement of employee car pools. The final EIS needs to further evaluate the impacts of construction in light of these mitigative proposals.

311

311. The impacts described in sections 4.1.6.1 and 4.1.6.3 are those of the project as proposed in the license application and subsequently filed materials. These sections also recommend mitigation measures (similar to those proposed in the comment) that would reduce, but not eliminate, the identified impacts.

J-139

-11-

Comment 11

Because the DEIS failed to adequately recognize and consider 312 alternative turbine discharge orientations, and other mitigative and enhancement proposals, \$4.8 and \$5 are incorrect in finding that substantial unavoidable adverse impacts to wetland habitats will occur. In fact, if the alternatives proposed by Allegheny in its comments above were adopted, the unavoidable and irretrievable loss of wetlands habitat predicted at L&D No. 7 will not occur, and in fact, the loss of wetland habitat associated with the Isle of White will be avoided. Furthermore, if the alternatives proposed by Allegheny were adopted, the wetland habitat associated with the Isle of White may in fact be enhanced. The physical model study proposed in Comment 5 will ensure this, and should be adopted for this reason. Accordingly, the final EIS should contain a reevaluation of advarse impacts in light of Allegheny's suggestions herein.

Comment 12

The DEIS' recommendations in §5 should be further 313 reavaluated because its review process would appear to be seriously flawed. The DEIS failed to consider alternative orientations of turbine tailrace discharge, incorrectly claims that construction barge traffic will adversely impact wetlands, and fails to adequately consider the benefits of LSD No. 7 to recreational resources and water chemistry. Accordingly, staff 312. Staff used available information to assess the impacts. Staff concurs that physical hydraulic modeling could assist in developing possible alternatives. See response to comments #304, 306, and 307.

313. Staff's evaluation of the hydropower project proposed at L&D No. 7, based on the information submitted, has not changed staff's conclusion that such development would cause significant adverse impact to the target resources identified as fisheries, recreation, and wetlands. Staff would not recommend studies be conducted after licensing, with unknown results, to determine mitigative measures for significant impacts that may not be mitigable.

J-140

+12-

must reconsider L&D No. 7's cumulative and individual impacts in light of Allegheny's comments herein.

Comment 13

Because the DEIS' evaluation of L&D No. 7 failed to adequately evaluate the project, the recommendation should be changed to recommend licensing of the project in conjunction with the post licensing development of a mitigation plan which would address the DEIS' concerns for wetlands, fisheries and recreational resources. As Allegheny has demonstrated through its comments, such a mitigation plan is not only possible, but practical. Development of such a mitigation plan in conjunction with consultation with other resource agencies will result in the development of a project in compliance with Commission recommendations and best adapted to the comprehensive development of the river for beneficial purposes.

Comment 14

Furthermore, because Allegheny has demonstrated that the 314 DEIS wrongly concludes that there are no alternative orientations possible which will protect the Isle of White, the final EIS must address the issue of why Allegheny's L&D No. 7 project continues to be treated inequitably from other similar projects already licensed downstream from L&D No. 7 (see Comment 1). The Commission needs to address in the final EIS why it chose to allow

-13-

314. See response to comment #298, 307, and 313.

these similar projects to be licensed subject to the development of mitigative measures, but treats L&D No. 7 differently.

Comment 15

Failure to grant Allegheny a license for L&B No. 7 subject F 315 to the davelopment of mitigative measures in compliance with Commission and other resource agency recommendations, denies Allegheny the finality of Commission action to which it is entitled. Failure to issue a license for L4D No. 7 prevents the limitation of issues involving this project and leaves Allegheny in a regulatory limbo subject to the concomitant burdens and expense entailed by such uncertainty. As a matter of equity, the Commission must close off the availability of objections, review and comment for LGD No. 7. Because the DEIS has unfairly neglacted to perform a thorough and adequate evaluation of LAD No. 7, and fails to address why L&D No. 7 is being treated differently from other similar projects, the DEIS' recommendation should be changed to recommend licensing of L40 No. 7 subject to the implementation of post-licensing mitigative measures prior to construction.

III.

CONCLUSION

The DEIS fails to recommend L4D No. 7 for licensing and 316 development because it failed to fairly and adequately evaluate the project. Because Allegheny has thoroughly and succinctly

-14-

315. See response to comments #298-314. The Commission will make the final licensing decision on L&D No. 7 in the public's interest.

316. Your opinion is noted. See response to comments #298-315.

demonstrated the DEIS' flaws and inconsistencies in this matter on a point by point basis, the DEIS' recommendation should be changed. Allegheny has clearly demonstrated that L&D No. 7 can be developed in an environmentally and economically feasible manner which will adequately protect target resources and avoid adverse impacts on such target resources, in particular those target resources identified as fisheries wetlands and recreation, on and around the Isle of White.

Respectfully submitted,

ALLEGHENY HYDROPOWER, INC.

Les M. Goodwin, Esq. Johathan W. Gottlieb, Esq. Wickwire, Gavin & Gibbs, P.C. 1133 21st Streat, N.W. Suite 500 Washington, D.C. 20036

Counsel for Allegheny Hydropower, Inc.

Dated: July 5, 1988.

.



EASYERN DOITHICT OFFICE AD WASHINGTON STREET, BTH FLOOP NORWALK, COPINECTION U B.A. ORISI PHONE (192) 425-5300

8703-5673 WP8703 L005

July 14, 1988

Office of Hydropower Licensing Federal Energy Regulatory Commission 825 North Capitol Street NE Washington, D.C. 20426

Att: George H. Taylor RB-209B

- Subject: FERC Docket No. EL85-19-114 Draft Environmental Impact Statement Hydroelectric Development in the Upper Ohio River Basin (Ohio, Pennsylvania, West Virginia)
- Re: Allegheny River L/D No. 3 Hydroelectric Project FERC Project No. 4474-003 - Pennsylvania

Gentlemen:

This filing contains the comments of Allegheny Valley North Council of Governments (AVNCOG) on the subject Draft Environmental Impact Statement (DEIS), which was issued for comment by FERC in May, 1988. AVNCOG is the applicant for a license at Allegheny River Lock and Dam No. J (the Project). The Applicant and its consultants have carefully reviewed the DEIS and have had informal meetings or discussions with the FERC staff, the U.S. Army Corps of Engineers (CDE), and the U.S. Fish and Wildlife Service (USFWS) about the DEIS prior to preparing these comments. We have purposely confined our remarks to the words, we have chosen not to comment on numerous items that would normally be considered significant, including inconsistencies between the analysis of similar issues at different projects, to focus on these malected issues.

MORRISON-KNUDSEN ENGINEERS. NIC.

Federal Energy Regulatory Commission July 14, 1988 Page 2

BACKGROUND

The pending license application was filed nearly four year years ago with FERC in September 1984. The agency consultation process began in 1981 during the preparation of the prefeasibility study, when key agencies were contacted and requested to comment on the Project. These agencies were contacted again in August 1982 during the preparation of the feasibility study. In Hay 1984, the agencies were provided with a preliminary Project description and asked to update their comments during the preparation of Exhibit z of the license application. In July 1984, a draft of the FERC license sepplication was circulated to the agencies for their comments.

After the Application was accepted for filing by FERC in December 1984 the Applicant responded to agency comments received by FERC during the public notice period. During the next year or so there was little direct activity associated with the Project, either by the applicant or FERC. AVNCOG was investigating various mources of funds to continue activities associated with the Project; FERC was deciding how to address the cumulative impacts of hydro development in the Ohio River Basin.

In September 1986 AVNCOG was successful in its efforts to raise funds to proceed with development of the Project and authorized its consultants to proceed with engineering and environmental studies associated with the Project.

Throughout the consultation process, the key issues have been instream releases, dissolved oxygen, protection of wetlands and fish habitat, and the use of flashboards. The Applicant proposed the use of flashboards to mitigate the possible impacts of the Project on upstream wetlands and increase power generation. The COE objected to the use of flashboards because of concerns about the potential impacts on the existing lock and dam structures and navigation. The Applicant has continued to pursue the use of flashboards, because it believes that the Project would be more economically and environmentally acceptable with flashboards. In the DEIS it appears that FERC concurs with our assessment that the use of flashboards would likely have a beneficial impact on upstream wetlands.

From the very beginning of the consultation process the Applicant has strived to identify mitigation plans that are reasonable and acceptable to all parties; that is, we were looking for solutions that both the Applicant and the concerned agency would find satisfactory over the long run. Over the past 10 months, the Applicant has expended substantial resources to collect additional information in an effort to resolve outstanding

MORRISON-KNUDSEN ENGINEERS, INC.

Federal Energy Regulatory Commission July 14, 1968 Page 3

environmental, technical, and institutional issues associated with the Project. These efforts have included extensive consultation with the USFWS, Pennsylvania Fish Commission (PFC), and the COE. Data collection has included a continuous D.O. monitoring program developed in close consultation with the agencies, wetlands surveys, extensive topographic surveying, and the preliminary design of and selection of turbine/generating equipment for the Project. Ongoing studies include physical model testing and subsurface exploration programs developed in consultation with the COE.

The Applicant believes that the results of these activities have been well worth the effort. At a meeting in January of this year, the Applicant reached agreement with USFWS and PFC on mitigation associated with D.O. and recreation. In addition, our efforts have resulted in very productive discussions and imaginative approaches to impact avoidance and mitigation with other agencias. It is possible, however, that our efforts to keep FERC abreast of these developments were not as successful, as it appears that some information used by FERC in the preparation of the DEIS is out of date, incorrect, or superceded by later fillings.

In any event, the DEIS process has been useful in that it has served to again highlight the critical issues. Further, it has resulted in significant progress towards resolving the flashboard issue with the COE, which will be discussed in a later section.

GENERAL CONMENTS ON DEIS

The recommendation proposed in the DEIS is that hydro development not be allowed at four of 19 sites to minimize the cumulative environmental impacts resulting from hydro development in the Upper Ohio River Basin. Allegheny River Lock and Dam No. 3 is one of these four sites. The Project received this designation because of what were assessed by FERC to be major, unavoidable impacts in the categories of fish habitat, wetlands habitat, and recreation that cannot be adequately lessened through modified operation, design, or mitigation. The Applicant strongly beliaves that these impacts can be avoided or adequately lessened through mitigation and design, and in several instances has already reached agreement with the applicable agencies.

We are mystified as to why the analysis of the Project in the DEIS assumed that no flashboards are used, which is contrary to what is proposed in the application, but given this assumption, we are not at all surprised by the conclusions. We believe that the analysis should have been performed on the basis that 317. Comment noted.

Federal Energy Regulatory Commission July 14, 1988 Page 4

318

319

flashboards are used, since that is a Project feature specifically proposed by the Applicant to avoid an adverse impact upon wellands. If FERC has concerns about the Applicants' ability to implement such a Project feature; we believe that it is more appropriately handled by a special license article, similar to the way instream releases or recreational facilities are dealt with in the licensing process. Then, if the Applicant Can not obtain COE approval of flashboards or identify an alternative method to maintain the upper pool level, both PERC and the COE would have the ability to prevent construction of the Project.

The Applicant intends to use the results of the physical model testing to demonstrate avoidance and/or identify specific mitigation measures related to fish habitat and recreation. Developing the information necessary to implement the model testing has been a long and involved process that has included detailed site surveys, preparation of preliminary layouts of the powerplant based on the size and type of generating equipment, and extensive consultation with the COE, including the preparation of model testing specifications by COE personnel. Bids have recently been received and we anticipate that an award will be made shortly. Our schedule calls for preliminary results from the model testing to be available by approximately October 1988.

In the past, detailed model studies for projects located at COE navigation structures have been performed after the license was issued. While we are not suggesting that FERC issue a license before major environmental issues are resolved, we do request that the Applicant not be denied the opportunity to complete such activities as model testing to resolve these issues before making a recommendation that the Project not be developed.

INSTREAM RELEASES

In its 1984 license application, the Applicant proposed a continuous minimum spill flow of 300 cfs with incremental increases based on DO level measurements taken during plant operations. Since that time, the Applicant has consulted extensively with USFWS, PFC, COE, and Pennsylvania Department of Environmental Resources and has collected extensive DO data. At a meeting held on January 25, 1988, the Applicant reached agreement with USFWS and PFC on a proposed mitigation plan to protect water quality (see Enclosure 1). The plan provides for a minimum instream release of 300 cfs when DO levels exceed a specified value (approximately 7.6 ppm, saturation at 85°F). If DO levels measured upstream of the dam drop below this value, 318. Flashboards were not included in staff's analysis in the DEIS because the Corps objected to previous flashboard proposals for Allepheny L&D No. 3. Staff estimated that the applicant would not be able to meet all of the requirements imposed by the Corps. Recent discussions between the Corps and the applicant have shown that significant progress has been mode towards resolving the flashboard issue with the Corps. Text has been modified in Sections 2.1.1.3 and 4.1.5.2 to reflect the possible addition of crest gates.

319. Staff's analysis in the FEIS concludes that, based on information filed by the applicant for Allegheny LAD No. 3 (see response to comment #17), development of hydropower at this site can occur in an acceptable environmental manner. Revisions to the text for this project has been made and the conclusions for staff's preferred alternative has been changed.

Development of a hydraulic model is a useful tool during detailed design of a hydroelectric plant. Results are used to refine details of a preliminary design. Because it is not reasonable to expect or require a model to be built at an early stage of project planning, development of the model should wait until after a license is issued.

320

320. Comment noted.

MORRISON-KNUDSEN ENGINEERS, INC.

Federal Energy Regulatory Commission July 14, 1988 Page 5

releases will be increased and/or natural and mechanical turbine venting systems will be used to bring the downstream DO up to an acceptable level. If these measures are unsuccessful, the Project will shut down.

FERC recommends a continuous minimum release of 1,000 cfs at the Project to protect instream water quality. The applicant is prepared to accept this recommendation, as the COE is currently requesting releases of a similar amount to prevent deterioration to the existing concrete spillway-dam.

RECREATION

A revised recreation plan was proposed by the Applicant in its November 1987 filing in response to FERC's request for additional information dated September 11, 1987. This plan provides for greatly improved access along the right bank with areas of the river bank developed for improved fishing access and picnicking. The plan provides for better fishing access for more individuals than currently exists.

Under current conditions at the site, water tends to stagnate in the back channel during times of low flow. As a result of the proposed Project, flows will increase through the back channel and along the right bank of the river. The proposed recreational facilities are designed to provide access to the moving water where fish tend to congregate, particularly in the spring.

The Applicant intends to look for additional opportunities to improve fishing during the physical model testing program. Such improvements might include habitat manipulation and/or additional modifications to the proposed recreation plan.

The proposed recreation plan has been accepted by the Pennsylvania Fish Commission and the USFWS, who have agreed to review the model with us to determine if additional measures would be beneficial (see Enclosure 1).

The DEIS states that "Disruption of the existing fishing opportunities at Fourteen Mile Island would seriously impact the existing users of the site." The Applicant believes that the ability of fishermen to wade to Fourteen Mile Island is greatly overestimated. Only during very low flows (<5,000 cfs) it is advisable to attempt to wade out to the island. Based on observations during the various on-site environmental and engineering studies, the vast majority of fisherman and other individuals gain access to Fourteen Mile Island via boat.

321

322

321. See response to comment #283.

322. Concern regarding the potential loss of recreational wading to Fourteen Mile Island was expressed by the Corps at a recreation workshop held on November 2, 1987. The provision of new public fishing access facilities, as specified in recommendation #1 of Saction 5.4.2.3, could potentially compensate for the loss of recreational wading to Fourteen Mile Island. The developer should file a revised recreation plan with the Commission after consulting with state and federal agencies regarding any additional recreational compensation measures that may be needed.

NORRBON-KNUDSEN ENGMEERS, INC.

Federal Energy Regulatory Commission July 14, 1988 Page 6

Project construction activities represent an advarse impact on recreation, but the impact would be temporary. Facilities proposed by the Applicant would provide long-term (permanent) improved recreation. As previously mentioned, the Applicant intends to investigate other possible improvements to fisherman access as part of the physical model studies. The applicant is also villing to make 30 parking spaces evaluable to the general public, provided their use can be temporarily restricted to Project-related vehicles and activities during times of major maintenance.

The current recreation plan was developed in consultation with USFWS and PFC and it is the Applicant's understanding that the conceptual plans have been accepted by these agencies. Nonetheless, the Applicant recognizes that it may be necessary to make minor alterations to the plan during the initial years of operation and will cooperate with the agencies in identifying and making these improvements.

The Applicant is confident that the proposed facilities represent 325 a major improvement over existing conditions and will provide excellent recreational opportunities at Allegheny Lock and Dam No. 3.

PISH HABITAT

FERC concludes that "there will be significant, adverse, and honmitigatable affects to the downstream habitat" at the Project. The Applicant believes that it is highly speculative to draw such a conclusion based on the available information. In addition, this conclusion was largely based on the assumption that a 500foot wide tailrace channel would be dredged, while more recent Project layouts show a considerably smaller (it varies between 100 and 300 feet wide) tailrace channel, thus only a portion of the existing gravel-cobble habitat will be disturbed.

Irrespective of the tailrace width, we noted with interest the following statement contained in the DEIS, as we believe it to be a far more accurate assessment of the situation:

"Fish babitat may suffer initially as the channel stabilizes to new conditions. There may actually be more shallow riffle area available around the island complex after flows change to the back channel, although this is not clear." 323. Recommendation 5 in Section 5.4.2.3 specifies measures required.to mitigate potential construction impacts to recreational use.

324. See response to comment #283.

325. See response to comment #283.

326. Text has been revised taking into consideration changes in project layout made since preparation of the DEIS (Section 4.1.2.2.3).

327. The DEIS section quoted in the comment indicated that replacement for any lost habitat might be forthcoming as the new flow and substrate regime stabilizes.

MORRISON-KNUDSEN ENGINEERS, INC.

Federal Energy Regulatory Commission July 14, 1988 Page 7

The Applicant has discussed the proposed tailrace excavation with USFWS and FFC in an effort to formulate a plan to replace any lost habitat. Both agencies have indicated that they did not require such a mitigation plan.

The Applicant plans to use the physical model of the site to assess what, if any, habitat mitigation is appropriate. Based on field observations, the Applicant believes that fish habitat in the back channel will be improved simply by the increased flow through the channel, but is receptive to considering other mitigation opportunities if such proves not to be the case. The physical model testing affords the Applicant an excellent opportunity to work in cooperation with the interested agencies to evaluate various mitigation plans or alternative designs aimed at avoiding impacts. USFWS, PFC, and the COE all have expressed interest and a willingness to cooperate in such a program. We intend to take advantage of this opportunity.

WETLANDS

implemented.

FERC estimated that 2.5 acres of vetlands would be disturbed or removed as a result of the proposed Project. This estimate was comprised of the following components:

- A. 0.2 Acres of riparian habitat along the intake and tailrace channels;
- B. 1.0 acres which would be destroyed by the removal of an island in the downstream pool; and
- C. 1.3 acres which would be disturbed by operational activities, primarily decreasing the pool elevation by 2 feet.

With respect to Component B, the U.S. Department of the Interior

incorrectly concluded that the cofferdam extended all the way to Fourteen Nile Island and that the sandbar would be removed during the excavation of the tailrace. The misunderstanding was pointed out by the Applicant in a letter dated June 19, 1985 responding to DOI's letter dated March 27, 1985. Enclosure 2 shows the location of this small island, referred to as the

C would not apply if the Applicant's proposed design is

We believe that component B is based on a incorrect agency comment that was subsequently clarified. In addition, Component

opportunity.

330

329

- 328. Staff has an independent responsibility to assure mitigation for what staff considers to be probable losses to fishery and ecological resources.
- 329. See response to comment #319.

330. Staff used aerial photographs and navigation charts to determine the potential impacts on wetlands. Staff believes that the increased disturbance to the island, which the Corps has described as an emergent and shrub wetland dominated by water willow (Section 3.3.5.5) and not as a sandbar, is such that damage would be permanent to the aquatic plant species on and around the island.

NORRISON-KNUDSEN ENGINEERS, INC.

Federal Energy Regulatory Commission July 14, 1988 Page 8

sendbar in the application, relative to the proposed Project features.

The Applicant acknowledges that there will undoubtedly be some temporary disturbance to the edge of the island, however, the island (sandbar) will not be removed. Under existing conditions this sandbar changes size and shape in response to river flows, a process that will continue even after the Project is in operation.

With respect to component C, the staff's analysis was based on the assumption that flashboards would not be used at the site. The Applicant has specifically proposed the use of flashboards to provide a more stable pool to protect the wetlands. Additionally, the use of flashboards is highly beneficial to the Applicant because it increases energy generation and more effectively develops the water resources in the public interest.

331

Interestingly enough, while staff acknowledges in the DEIS that impacts on wetlands can be sitigated with the installation of flashboards, they choose not to analyze the Project with flashboards because the COE has reservations about their use.

The Applicant has had extensive discussions with USFWS and PPC about the potential impacts to upstream wetlands that may result from the Project. All parties agree that the use of flashboards will protect the existing upstream wetlands and, as a result of a more stable pool lavel, may enhance wetlands. The Applicant prepared and submitted a detailed report that identified the wetlands in the pool above Lock and Dam No. 3 and assessed the Project's impacts upon these wetlands.

The Corps of Engineers has been concerned about the use of flashboards at the Project because of potential technical problems. During a recent informal meeting between the Applicant and the COE to discuss the flashboard issue, however, the COE indicated that a more sophisticated means of controlling pool alevation, such as a crest gate system, may be acceptable, provided that it can be demonstrated that such a system did not have a negative impact on the stability of the lock and dam or cause downstream scour. The Applicant has requested that the COE provide a letter clarifying its current policy regarding the use of Crest gates on navigation dams. Enclosure 3 is the Applicant's request. The response by the Corps of Engineers will be forwarded as scon as it is available.

The Applicant intends to continue to work with the Corps of Engineers to identify an acceptable crest gate system that can be 331. The use of crest gates at Allegheny L&D No. 3 would provide a more constant pool elevation. The Corps had previously indicated that they would not consider flashboards as a viable alternative and therefore they were not considered as an alternative in staff's analysis. If the Corps approves the use of crest gates at this project site, the disturbance of a potential 1.3 acres would be mitigated and usage of such structures could possibly enhance the wetlands in the pool.

MORRISON-KNUDSEN ENGINEERS, MC.

Federal Energy Regulatory Commission July 14, 1988 Page 9

used to protect and anhance the upstream wetlands. The Applicant has already discussed on a preliminary basis how the physical model might be used to demonstrate what impacts various creat gate systems might have on the existing structures. The Applicant intends to pursue these discussions as additional information becomes available.

CLOSING COMMENTS

FERC has obviously made an extraordinary effort to collect, review, and enalyze a vast amount of information in a short period of time. While Applicant does not agree with staff's recommendation that the hydroelectric project proposed at Allegheny River Lock and Dam No. 3 not be developed, the methodology mnd data used to arrive at the conclusions are clearly stated.

In our response to the DEIS we have tried demonstrate that in in some instances superceded information or a misunderstanding about the Project has resulted in an incorrect conclusion, while in other instances reasonable opportunities exist to modify the Project design to avoid major environmental impacts, and, in some cases, enhance existing resources. Specifically, we have shown the following:

- <u>Recreation</u> The plan proposed by the Applicant represents a major improvement over existing conditions and will provide excellent recreation opportunities at the Project.
- Fish Habitat Reasonable opportunities exist to assess and either svoid or mitigate any potential adverse impacts.
- <u>Wetlands</u> Approximately 0.2 acres of wetlands will be destroyed by construction of the Project. Use of crest gates to stabilize the upstream pool elevation will protect and may improve the upstream wetlands. Crest gates can be stipulated as a license condition.

On this basis we respectfully request that staff's recommendation 334 with respect to the Project be changed in the final EIS.

332

333. Comments noted.

332, Comment noted.

334. See response to comment #319.

MORRISON-KNUDSEN ENGINEERS, INC.

Federal Energy Regulatory Commission July 14, 1988 Page 10

335

335. Comment noted.

Three enclosures provided by Morrison-Knudson are not reproduced here.

The Applicant's actions throughout the licensing process demonstrate its commitment to the Project and its concerns about the resolution of the environmental issues. The Applicant intends to continue its efforts to resolve the outstanding environmental and technical issues. environmental and technical issues.

Very truly yours Hatella

Project Manager

Enclosures

· ...,

co: Celeste Calfe, AVNCOG Carol Cunningham, MKE Roy Slack, TES

	IS, INC/	SY .
EASTERN: DRITECT OFFICE SO WAR-INGTON STREET ATH FLOOR NOTIMULE CONFECTION US & DESA PHONE (2003) 838 3300	68 AUG - 5 1/11:2	0
		87

.....

8703-5673 WP8703 1.008

5

4

P

336

÷

August 3, 1988

Office of Hydropower Licensing Federal Energy Regulatory Commission 825 North Capitol Street NE Washington, D.C. 20426

Att: George H. Taylor RB-2098

- Subject: FERC Docket No. EL85-19-114 Draft Environmental Impact Statement Hydroelectric Development in the Upper Ohio River Basin (Ohio, Pennsylvania, West Virginia)
- Re: Allegheny River L/D No. 3 Hydroelectric Project FERC Project No. 4474-603 - Pennsylvania

Gentlemen:

By letter dated July 14, 1988 the Allegheny Valley North Council of Governments (AVNCOG) provided comments on the subject Draft Environmental Impact Statement. In the discussion regarding the use of creat gates to avoid negative impacts on the upstream wetland, the Applicant indicated that it had requested the U.S. Army Corps of Engineers (COE) to clarify its position with respect to the use of creat gates at Allegheny River Lock and Dam No. 3. Attached hereto is the COE's response to that request.

Very truly yours,

ман 156 X. Project Manager

Enclosure

cc: Celeste Calfe, AVNCOG Carol Cunningham, MKE Roy Slack, TES 336. The letter from the Corps is noted. See response to comment #318.

J-154



DEPARTMENT OF THE ARMY PITTSBURGH DISTRICT, CORPS OF ENGINEERS WILLIAM S. MOORHEAD FEDERAL BUILDING 1000 LISBATT A VENUE, PITTSBURGH, PA 15222-4166

P13 - 1'89

July 26, 1986

Ms. Marsha A. Fickert, F.E. Project Manager Morrison-Knudsen Engineers, Inc. 50 Washington Street, 9th Floor Norwalk, Connecticut 06854

Dear Ms. Fickert:

This is in response to your letter of June 27, 1988 requesting clarification of the Corps of Engineers' position on the use of dam creat gates in connection with development of a hydroelectric project at Lock and Dam 3, Allegheap River.

It is generally believed that the wetlands upstream of the dam would be damaged by lower than normal pool levels from the proposed hydropower operations at the existing fixed creat dam. The Corps of Engineers recognizes the possibility of avoiding these environmental losses by maintaining historic pool levels during hydropower operations. Therefore, we will consider the installation of facilities to control pool levels during periods of low flow provided the following concerns are addressed:

Any device installed on the dam, such as crast gates, must be fully controllable.

b. A detailed study of the wetlands should be conducted to determine the minimum required pool elevation. All possible means of protecting this resource, including dam creat gates, dikes around the wetlands, act., should be evaluated. This must be carefully coordinated with the Corps of Engineers and other appropriate state and Federal agencies.

c. The optimum arrangement for a creat gate system would allow the device and its jet to conform to the dam creat profile when lowered. The hydraulic effects of any modification must be evaluated in a physical model study to ensure equal discharge efficiency and preclude any scour. Any increase in upstream flooding caused by the creat gates will be the responsibility of the Licensee.

d. If the upper pool/lower pool relationship is changed, a stability analysis must be performed for the lock walls and the dam to evaluate the new loading conditions.

oseph & Eyre LTC, En George 1. Miller, Jr. Colonel, Corps of Engineers District Engineer

Coptes Furnished:

Commander, Ohio River Division ATTN: CEORD-ED-T/Mr. Hart

Mr. Martin Knweid Regional Director, KERC New York Regional Office 201 Varick Street Room 644 New York, New York 10014

.

1 A A

.

LEBOEUF, LAMB, LEIBY & MACRAE

A PARTNERSHIP INCLUDING PROFESSIONAL CORPORATIONS

1333 NEW HAMPSHIRE AVENUE, N.W.

WASHINGTON, DC 20036

12021-457-7500

NEW YORK NY SALT LARE CITY, UT BALLINGH, NO BAN FRANCISCO. CA LOS ANGELLS. CA

TELER- HADETA FACSINILE EDE-487-754 CONCENT, LAND, LEIBY & MACHAE (UNI LONDON, ENGLAND LEBORUF, SUNDATRON FERRY CRUSSELS, SELGIUM

BOSTON. MA SOUTHBORT. CT NEWARK, HJ ALBANY, NY UACKSONVILLE. PL

.

<u>ين</u> <u>ر</u>ب

Ξ.

(1.3% - S (1.3)

July 5, 1988

The Hon. Lois D. Cashell Acting Secretary Federal Energy Regulatory Commission 825 North Capital Street, N.E. Washington, DC 20426

Re: Docket No. EL85-19-114

Dear Ms. Cashell:

Enclosed for filing with the Commission on behalf of the City of New Martinsville, West Virginia are an original and 14 copies of a response to the draft Environmental Impact Statement (EIS) for the Upper Ohio River Basin for (1) the Willow Island Hydroelectric Project (Project No. 6902) and S (2) the New Cumberland Project (Project No. 6901). J

> Respectfully submitted, ₽

 $\overline{\mathbf{N}}$ ş oren na Jan / Louis Rosepman

Attorney for City of New Martinsville, West Virginia

Enclosure

cc: Michael Francis, Esq. Mr. James Price

Willow Island Hydroelectric Project - No. 6902 Response to Federal Energy Regulatory Commission Regarding Draft Environmental Impact Statement (EIS) for the Upper Ohio River Basin

4.1.2.3 Entrainment and Turbine Induced Mortality. This section of the 4 337 EIS considers factors that were similarly considered in the report of studies at the Racine project conducted jointly by the project applicant and the Ohio Power Company. The study at the Racine project gave a reasonable indication of the level of mortality to adult game fish and allowed the applicant to establish reasonable estimates of compensation. The approved plan for the study at Racine did not include the question of the effect of turline induced mortality on overall fish populations (game and forage); the question of the effects on icthyoplankton and juvenite mortality; or the effect of mortality of forage species on game fish populations, as all parties (the Licensee, WVONR, ODNR, USF & WS) agreed that such studies involved great expense with a substantial risk that the results would not be useful. We have, to date, seen no study approach that would yield reliable answers to these questions within reasonable time and cost frames. Such studies would need to address three areas: (1) extent of icthyoplankton and juvenile mortality and the resulting effect on population; (2) effect of forage fish mortality on game fish populations; and (3) geographic extent of incal population effects of mortality, including adequate data on migratory habits of some of the species of fish present.

> We do not feel that studies of fish mortality and mortality effects on population are likely to be economical or conclusive and suggest rather that the overall fishery resources be monitored, if necessary, by population sampling techniques as the Commission has recommanded for the New Martinsville Project at Hannibal Dam. If no adverse pattern is revealed by these more general, effect-based studies, then the far more extensive and costly cause-based studies can be deferred. Use of funds poid by licenses as compensation for fish mortality to an "Ohio River Basin Environmental Fund" could be used for such studies, if considered appropriate by the administering agencies.

4.1.2.3.4 Staff Conclusions on Entrainment. We concur with the staff conclusions that no fish diversion devices have been shown effective in the conditions considered. Proposed further investigations of the effectiveness of potential fish diversion structures may be warranted, but only if a need can be demonstrated. Contributions by licensees to an "Ohia River Basin Environmental Fund" are a reasonable sitemative to diversion measures and have been accepted as such by Ohio and West Virginia. Such funds could be used to carry out such studies by the concerned agencies. We do not concur with the staff conclusion that instalistion of fish protection measures (structural devices) 337. Staff believes that both entrainment and population monitoring are needed as post-licensing requirements. Entrainment monitoring for fish passage rates and percentage mortality probably would not be needed at every site. See comment #234. .

J-158

338. Staff generally agrees with the comment. The "Okio River Basin Environmental Fund" could be a practical way for developers to jointly fund studies such as the bioengineering studies of prototype fish diversion devices. Staff has altered the recommendation that includes the assumption that fish protection devices are likely to a suggestion that they may be shown to be practical and effective.

1
is likely and do not believe that installation of bypasses and related features in anticipation of such installation is desirable or economical at present.

We feel that the design concept adopted by the applicant for this project, incorporating a large, low velocity cross section for both the power plant intake and the inlet channel, provides the best protection against high mortalities and is consistent with sound power plant design. We do not believe that any structural system can be designed to economically or effectively divert fish in the environment. Because of the low head at each dam on the Obio River, the flow of water required to generate a kilowatt hour of power is relatively large. Any device attempting to somehow process this flow is necessarily large and costly, and causes a loss of energy. This prublem is compounded by the large trash loading of the river, which places great burdens on any structure instatled in the flow. The associated costs will result in power production costs in excess of those warranted by current or expected future power values and consequently cancellation of projects.

5.1.1.2 Aquatic Ecology and Fisheries (Turbine Induced Fish Mortality)

The comment that "mortality of larger fish would be greater than 10 percent of those entrained' has no basis in the Racine study, date presented in this report, or other data which we have reviewed. This statement is inappropriate unless some basis can be provided. Data provided in the Racine report (Volume I, page 13) shows survival rates of 88 to 100 percent with instantaneous survival of 97.9 percent and latent survival of 94.5 percent for game fish.

5.4.2.2 Recommendations on Aquatic Ecology and Fisheries

Item 1 - We disagree with the suggestion that licensees provide for undefined and unprovam intake protection structures where there is no evidence that such are required and no evidence that any particular technology would be effective. Such installations would involve extensive costs for no indicated gain.

Item 2 - We disagree with this early timing (12 months after the license issue date) for agreement on compensation payments.

Item 3 - We disagree with the requirement that licensees conduct fisheries related studies. Compensation paid to the resource agencies should provide for this effort. The agencies could then properly decide on expenditures for various categories of effort; fish restocking, angler access, fisheries studies, or other uses.

Item 4 - We question the effectiveness of the requested mortality study but feel that any such studies to be conducted should be from compensation funds.

- 339. Staff has generalized from the Racine Studies. Larger fish are known to have a higher degree of damage than smaller fish. Until more studies are carried out, an estimate of 0-10 percent mortality, with closer to 10 percent for larger fish, seems to be a reasonable conclusion. The statement has been clarified in the text.
- 340. Staff believes that there are certain design features that can be built into plans at this time, such as passageways to allow transfer of fish to the tailrace or provisions for anchoring equipment to the powerhouse. Staff did not intend that any fish protection structures themselves be built until after prototype testing (that is considered in a subsequent recommendation).
- 341. Staff contends that the one-year period is in conformance with the timing on similar recommendations for fisheries protection and is therefore appropriate.
- 342. Staff believes that it is premature to determine at this stage whether funds for fish population monitoring, entrainment studies, or bioengineering test facilities are paid directly by developers or through a compensation fund to the resource agencies. Such a compensation fund seems to have merit. Selection of the funding scheme must be determined after licensing, in consultation with all parties.
- 343. See response to comment #342.

Item 5 - Bioengineering studies, if considered appropriate, should be paid for from compensation funds. Any open-anded requirement to install diversion structures should be subject to formel hearings at the commission and subject to economic constraints for each project.

.

-

• •

-

344. See response to comment #342.

344

.

.

m

New Comberland Project - No. 6901 Response to Federal Energy Regulatory Commission Regarding Draft Environmental Impact Statement (EIS) for the Upper Ohio River Basin

WATER QUALITY CONSIDERATIONS

3.5.2 Water Quality

This section discusses existing water quality in the river. The duration curves based on Orsanco data (1980-1986) for two locations, RM 40.2 (New Curnerland Pool) and RM 102 (Hennibel Pool), indicating recurrence of the two criteria level dissolved oxygen (DO) values at these locations for the summer months are as follows:

	<u>> mg/l</u>	6.5 mg/1
RM 48.2 (New Cumberland Roat)	8%	25% (Fig. 3.5.2-2)
RM 102 (Hannibal Pool)	5%	25% (Fig. 3.5.2-3)

.

This data does not appear consistent with data presented in Figure 3.5.2-6 showing a DO profile of the river, where no DO values below 7 mg/i have been recorded since 1974 in the 5 years of data shown for that period. Data collected by the applicant in 1925 and submitted to the Commission shows a minimum upstream DO level of 6.7 mg/i, with values below 7 mg/i for oily 4 days of the 52 days sampled in the period from July through Seytember, 1985. We therefore consider any plan based on the duration curve discussed above to be an effort to correct problems that do not, in fact, occur with the frequency projected in the EIS.

4.1.1.1 Water Quality - DO

Assessment methods. The DO model used in the analysis is based on calibration using 1983 Corps of Engineers (COE) data. The explenation of the model and its calibration provided in Appendix B is not adequate to determine its reliability in this application. Data used in the calibration are not presented in full nor is a description of the method and extent of data collection. It is impossible to judge the adequacy of the analysis without this information.

The model also relies greatly upon correlation between a DO level, atated as a deficit, above and below (downstream from) the dems. Statistical measures of this correlation are given (see Table 4-1.1-1 and Table 8-1); however, the statistical data are not adequate to allow a thorough evaluation of the analysis and of the subsequent water quality modeling. Based on the limited information provided in the report, it appears that the water quality model relies entirely on this correlation analysis for the reagration coefficients assigned for each dam. The discussion in B 2.1.3.1 of problems encountered when b does not equal zero, that is, when the line of best fit 345. The temperature and DO duration curves based on ORSANCO data are presented to provide background information on the frequency with which specific values have occurred. The values presented in Figure 3.5.-6 are from occasional samples taken by the Corps and do not necessarily represent any specific conditions. The DO duration curves were not used directly as a basis for any of the impact analyses. The temperature duration curves were used to determine what water temperatures represent reasonably-forseeable adverse conditions (the temperature exceeded IOX of the time in August was selected as a basis for spill flow recommendations). Electronic temperature measurements are generally highly accurate.

- 346. Appendix B has been expanded to include the data used in modeling dam aeration, and descriptions of sensitivity and uncertainty analyses performed on the model.
- 347. The discussion in Appendix B (Section B.2.1.3.1) of negative values of b includes several possible explanations of this result. The explanation that slight degrees of supersaturation occur is expected from aeration theory (since the saturation concentration of DO is higher at the depths to which bubbles are submerged at some dams). Gas supersaturation at dams is a common phenomenon.

crosses the x-axis at a negative value, gives an indication of the difficulties in developing these coefficients. However, the text provides no adequate rationalization for the methods of use of this inconsistent data. This section also references "RMS errors generally within the range of accuracy that can be expected with DO models" without noting that range and without noting the importance of any inaccuracier. There is no significant discussion of these reaeration coefficients as to the reasonableness of the coefficients or the sensitivity of the water quality model predictions to the range of error in the coefficients. Such a discussion is critical to the evaluation of the water quality modeling process and the modeling results because of the very large value of the resource here at bake.

Also, in Section B 2.1.4, Algae Production, the Importance of this factor is noted as being relatively minor based on referenced personal communication with COE staff. We find this assumption to be an overriding critical error in the analysis. Field data filed by the applicant showed a substantial occurrence of supersaturation in the upstream pool at dama studied in 1985, with the only reasonable explanation be ng algal oxygen production. While such algal-source DO may not always be present, failure to account for its frequent (likely preponderant) presence ignores a major factor governing utilization of this resource.

The modeling effort also appears to ignore the increase in diasolved oxygen (DD) concentrations caused by an operating hydroelectric facility when there is no flow through the dam gates. A report submitted to the Commission by American Electric Power (1998) on the Racine hydroelectric project showed that at the Racine site the hydro units increased DD levels downstream as much as the dam releases and sometimes more. In September 1987 testing at Racine showed a downstream DD increase in the range of 0.4 mg/l caused by the hydro while no significant increase due to passage through the dam gates was noted. This was thought to be due, in part, to the shallower depth water with higher dissolved oxygen concentrations drawn off and discharged by the hydro. The failure of the DEIS to consider this information results in unnecessarily high spillage requirements.

It is unclear whether the optimization procedure described for impact analysis properly evaluates the tradeoff in DD effects at various splits for each project (Section 8.3.2 - Bounded Implicit Enumeration). In particular, the report should show the relative effects of reduced splitage requirements at the projects at the downstream range of those requiring apHiage - Pike Island, New Cumberland, and Montgomery. We question whether the Incremental improvements claimed to be produced by splitage at these projects are justified.

348. See response to comment #101.

349. See response to comment #252. The withdrawal of water from the upper level of the upstream pool as a way of mitigating for aeration loss has been proposed in other river basins. This technique does not actually add any oxygen to the water but only moves the more highly oxygenated water from the pool above the dam to the pool below it. Throughout most of the upper Ohio River basin stratification strong enough to make this technique effective is rare. If a hydropower plant was successful in withdrawing only from the more highly oxygenated surface layer of the above-dam pool, it would promote stratification and lower water quality above the dam.

J-162

350. The optimization model works by finding the combination of spill flows (at dams that are effective aerators) that maximizes the power generation in the basin while maintaining DD concentrations above 6.5 mg/L in all pools where DD was above 6.5 mg/L in all pools where bD was less than 6.5 mg/L without hydropower. In pools where DD was less than 6.5 mg/L without hydropower, only slight decreases (less than 0.5 mg/L) in DD are allowed. Under the design conditions used for the model analyses, the spill flows at Pike Island, New Cumberland, and Montgomery are required to maintain DD concentrations in approximately the first 200 miles of the Obio River. Because dams downstream of Pike Island provide little aeration, aeration at these three dams affects DD far downstream and is required to prevent DD degradation. The sensitivity analysis of the DD model presented in Appendix B indicates that aeration at RM 100.

4.1.2.1 Assessing the Impact of DO Change on Fish (For Hydrodevelopment as Proposed)

This section discusses the three potential levels of DO maintenance 351 with respect to fisheries. It establishes 6.5 mg/l DO as a standard for "zero effect" on fisheries. The need for the highest level considered (6.5 mg/l DO) appears not to be based on the survival data provided (Figure 4.2.1-1), as the data indicate that a level of 5.5 to 6 mg/l DO causes no decrease in survival.

The national criteria for ambient DO concentrations for the protection of freshwater equatic life are presented below for warmwater fisheries;

	Early Life Stages	Other Life Steges
30-day Mean	N/A	5.5
7-dey Mean	6.0	N/A
7-day Mean (misimum)	N/A	4.0
1-dey Minimum	5.0	3.0

Source: U.S. Environmental Protection Agency (U.S. EPA) 1986 "Guality Criteria For Water."

These criteria recommend a 7-day mean DO concentration of 6.0 mg/l and a 1-day minimum of 5.0 mg/l to protect early life stages at warmwater fisheries.

The criteria were derived from production impairment estimates which were based primarily upon growth data and information on temperature, disease, and pollutant stresses. The average DO concentrations selected are 0.5 mg/l above the slight production impairment values and represent values between no production impairment and slight production impairment, therefore, each criterion is an estimate of the threshold concentration below which some detrimental effects may be expected. The criteria represent DO concentrations which the U.S. EPA believes to "provide a reasonable and adequate degree of protection for freshwater aquatic life."

The requirement in the Draft Environmental Impact Statement (EIS) (pages 5 through 25) that developers maintain DO concentrations of 6.5 mg/l throughout the basin during July through October is not reasonable and not supported by the U.S. EPA "Water Guality Criteria" or other research on DO. Reasons for our disagreement with this requirement are listed below:

A. The requirement is based on allowing no production impairment of fish in the Chia River. U.S. EPA, (1986) states that DO levels 0.5 mg/l above slight production impairment provide a "reasonable and sdequate" degree of protection. These levels (5.0 mg/l for early life stages) apply as a 7-day moving mean

3

351. See response to comment #242. The fish growth model is further explained in the text and Appendix E. Staff has sought to place the somewhat ambivalent documents on water quality criteria and the varying time frames for applying criteria into a quantitative framework that integrates effects over time. This seems preferable to matching arbitrarily selected conditions. In the fish growth model. there is an explicit summation of effects of DO on growth rate over the durations of exposure to varying levels of DO as determined by temperature, flow, and other factors included in the model of DO concentrations. The fact that the original formulation was developed for ponds should not invalidate its application to a series of navigation pools, especially when the model and its results are used in a comparative fashion (i.e., to compare relative performances between contrasting Ghio River conditions). The comparisons are not among pands and the Ohio River, but among several conditions of the Ohio River using the same assumptions in each case.

only where and when early life stages are present. We disagree with this approach to the "no production impairment" philosophy and question whether protection of early life stages is needed during the entire July through October period. It is our opinion that fish fry are most abundant during May and June and do not require protection in the fail months of September and October.

B. The draft EIS does not state whether the 6.5 mg/l standard being required will be a 30-day mean, 7-day mean, or 1-day minimum and whether these means will be moving, calendar week, or calendar month. Insufficient data is presented for us to determine which condition was modeled.

U.S. EPA criteria recommend a 7-day moving average to protect early iife stages of fish and a 30-day moving average to protect other life stages of fish. The draft EIS appears to be recommending a 7-day moving average criteria that is (5, mg/l) above the EPA standard for early life stages (6.5 mg/l recommended as compared to 6 mg/l EPA standard) and then attempting to apply it to a 4-month period (July to October). This is clearly a misapplication of the criteria which were not intended to be applied to monthly averages. This misapplication results in an effective increase of at least 1 mg/l in the standard from the 5.5 mg/l EPA standard for the 30 day mean to the 6.5 mg/l level required in the draft EIS for the 123 day period of July through October.

The discussion of the growth model (Section 4.1.2.1.3) involves extrapolation of data for catilish being raised in a densely populated farm altuation to a very different environment, that of the Ohio River pools, and involves extrapolation of the results for catilish to many other species. This is done based on minimal data and with no significant explanation of the process. This is a critical omission in consideration of the final recommendation of the report. The assumption that the growth of fish which must forage for food in the natural environment of the Ohio River will be affected by DC levels as greatly as fish raised in ponds with an abundant food supply is questionable based on published research on the subject.

- 4.2- In these sections, the report discusses the relative merits of
 4.4 meeting the three tiers of uses and DO levels.
 - 4.2 Alternative 2 Meet DO Standards
 - 43 Alternative 3 Meet Antidegradation Standards
 - 4.4 Alternative 4 Minimize impacts to all target resources.

This section builds entirely on the water quality modeling and, in Section 4.4, the fish growth modeling, discussed in Section 4.1 for the "projects as proposed." The applicant's questions regarding the validity of both of these modeling efforts have been noted above. In particular, we invite comparison of Figure 4.3.1.3, the graphic representation of modeled Ohio River DO profiles (meeting the 6.5 352. The D0 model was used to simulate a set of design conditions under which hydropower impacts would be relatively severe; maintenance of 00 concentrations under these design conditions is assumed to protect D0 adequately under most reasonably-expectable conditions. There is πo reason to expect the model to match the Corps data presented in Figure 3.5.2-6, which was collected under a variety of conditions. mg/1 criteria) with the measured DO data of Figure 3.5.2-6. It appears that spillage is being required at the New Cumberland Dam to allow for a DO decline in the Pike Island pool, modeled to be a decline of more than 1 mg/1 DO.

A comparison to Figure 3.5.2-6 shows this pool as having a rising DO level rather than a declining DD level for the calibration year (1983), as well as for subsequent data sets of 1985 to 1985.

For only 2 of the 7 years sampled (1977 and 1978) did the DO levels in the pool decline by an amount in the range of 1 mg/l. Given the substantial reduction in waste loading of the upper Chia River since 1978 (as noted by the COE in past public discussions on this topic), it appears that the modeling results for this pool are especially open to question.

5. STAFF CONCLUSIONS

- 5.1.1 Alternative 1 Projects as Proposed
- 5.1.1.1 Water Quality

and Aquatic Ecology and Fisherles

5.1.1.z

Based on out preceding comments questioning the reliability and the basis for projecting impacts (the DO and fish growth models and supporting date), we consider the extrapolation of the results to the indicated conclusion to be poorly founded. We also note that the discussion of the potential reduction in taxle compound volatilization suggests possible problems but no supporting data is shown or referenced. This speculation is inappropriate unless some basis is provided.

The comment that "mortality of larger fish would be greater than 10 percent of those entrained" has no basis in the Racine study, data presented in this report, or other data which we have reviewed. This atatement is inappropriate unless some basis can be provided. Data provided in the Racine report (Volume 1, page 13) shows survival rates of 88 to 100 percent with instantaneous survival of 97.9 percent and latent survival of 94.5 percent average for game fish.

We feel that the evaluation of all of the alternatives not selected [355 (1 through 3) similarly relies on dats and procedures for which the level of reliability is not edequately documented and attributes adveces impacts without a proper basis.

5.2 Economic Evaluation of Alternatives

The energy production values provided in Table 5.2-1 are higher than 1356 those predicted by the applicant; however, we concur in the estimate of losses in energy production for Alternatives 3 and 4.

- 353. See responses to comments #345-350 and response to comment #104 concerning volatile compounds.
- 354. See response to comment #339.
- 355. The methods and assumptions used in evaluating Alternatives 1-3 are the same as those used for the recommended Alternative 4.
- 356. The applicant's prediction of generation and concurrence of losses between alternatives are noted.

The loss in energy production with alternative 3 or 4 at the New Cumberland project would have no economic justification in fisheries effects. The loss of 40 million kWh/year shown in Table 5.2-1 (178.9 million kWh minus 138.3 million kWh) at a value of 75 mills/kWh tataling \$3,000,000 per year may be compared to the fisheries data presented in Table 5.4 of the New Cumberland License Application. For a total of 13,800 fish kept from the pool and taliwater in 1981, the cost would be \$217 per fish. If the growth model extrepolation of Section 4.1.2.1 is accepted (although we contest its validity), this \$217 would result in a 20 percent increase of 0.465 pound at a cost of \$542 per pound, or \$33.90 per dunce. This seems a very high cost for a projected benefit which is lead to be a fish peculative.

The license application and supplementary information on DO submitted 358 to the Commission by the Applicant (September 30, 1985) addresses the question of potential DO effects of the hydro project. The license application included consideration of incidentes spillage (leakage) of 2,535 cubic feet per second (cfa) and low water shutdown of about 1 percent of the year or 4 days per year. The shutdown periods resulting from low flow would be entirely during the low flow and low DO months of July through October.

The supplemental filing of September 1985 included an analysis of DO data collected in the period of July through September 1985 and evaluated the effect on project operation of measures to assure adequate DO conditions. That analysis on page 17 noted that 16.7 days of shutdown could be provided with 100 percent bypass of the hydro facility as a mitigative measure when DO drops below 6 mg/l. Such DO related shutdowns would be based on DO measurements for each day of operation. These days of shutdown for DO would be expected to coincide with the 4 days of shutdown for IOW would be above. The economic loss due to this extent of shutdown was included in the estimates of energy production and economic evaluation of the project.

if the DO atandard were to be raised to 6.5 mg/l as proposed in the EIS, we would estimate, from Figure 3.5.2-3, a 15 percentage point increase in shutdown periods during the months of July through October (123 days), an additional 18 days of shutdown, or 34.7 days tatel.

The facility will be operating at approximately 8 megawatts during the period. The resulting loss in energy production is:

£

8,000 x Z4 x 18 = 3,456,000 kWb.

357. Environmental protection is not to be determined as a cost-benefit calculation according to numerous legislative and legal precedents.

358. See response to comment #243 concerning why continuous spill flows are recommended as mitigation. 220-954 0 - 88 - 23

This is equivalent to approximately 2.1 percent of energy production and is the additional energy loss not accounted for in the original application, resulting from modification of operation to achieve 6.5 mg/1 on the basis of response when needed rather than the 6 mg/1 as proposed in the itemse application.

Based on the report on DO studies conducted in 1987 at the Recine Hydroelectric Project, we consider the above analysis of shutdowh for DO to be conservative. That report, on page 13, concludes that the hydroelectric power plant increased downstream DO levels by a combination of turbulence and withdrawal levels. The New Cumberland Project license application to DO. Adopting the indicated typical DO improvement through the Racine power plant of 0.4 mg/l, the analysis filled by the Applicant in September of 1985 is essentially correct. The addition of 0.4 mg/l DO by the hydroelectric plant would nearly offset the 0.5 mg/l increase in the standard from 6 mg/l to 6.5 mg/l DO. The resultant energy loss would be very minimal. A 3 percent increase in spillage would result from the 0.1 mg/l DO margin required (0.5 - 0.4 = 0.1) or 4 days of spillage from the 123 day summer period. The resultant energy loss is:

0,000 x 24 x 4 = 768,000 kWh.

This is equivalent to 0.5 percent of ennuel output and represents a minor effect on project economics.

In summary, the relative effects on energy production for the three levels of action are as follows:

- Applicant proposed bypass when Per license epplication needed at 6 mg/i DO.
- Applicant modified for 6.5 mg/l
 Contract anargy loss
 DO ignoring reservation by hydro.
- Applicant modified for 6.5 mg/i
 0.5 percent energy loss
 DO showing for reservation by hydro.
- d. EIS proposed constant 15,000 cfs 22 percent energy loss bypass for 4 months and 4,000 cfs for 8 months.

Based on the above analysis, the imposition of a 6.5 mg/i DO standard, with resulting bypass flows to be required only when justified by actual measurements of downstream DO on a dally basis, would have a minor adverse economic effect on the project. The year-round leakage through the dam of about 2,500 cfs, in conjunction with the relatively rare shutdowns for DO improvement, would provide the benefits considered necessary in the EIS while causing a modest

7

loss in energy production. By comparison, the continuous yest-round release of 4,000 cfs minimum and 15,000 cfs during 4 low flow months would have an extremely advorse affect on project economics.

5.4 Recommended Alternative

We strongly disagree with the staff's recommandation that the New 359 Cumberland project be subject to the epillage requirements of Alternative 4, given the inadequate basis for the assessment of adverse impacts and the disproportionate loss in power production in comparison to the claimed benefits. The staff is, in effect, raising the DO standard for the stream to 6.5 mg/l for hydropower projects, while all other river users are subject to no restriction (nevigation and nonpoint pollution sources) or to the properly errived at standard of S mg/l (for municipal and industrial wastewater discharge). Such a change in the standard should not be based on the limited data and inadequate, unproven analysis provided in this study and should not be applied to only one type of development, hydroelectric power plants. If, however the higher DO standard is to be imposed on hydroelectric projects, spillage should be based on real time measurements, and should not be set as a continuous apillage requirement.

Applicant - Proposed DO Maintenence Procedure. The applicant suggests the procedure outlined below for assuring the quelity of the Ohio River while maximizing energy production for the dams under consideration. This procedure is based on real-time coordination of hydroelectric plant operations with water quality measurements as suggested in the precedure discussion. The express he based on the Memorandum of Agreement (MOA) between the Corps of Engineers (COE) and the City of New Martinsville (Licensee) for the New Martinsville Hydroelectric Project at Hannibal Locks and Dam. This MOA, developed in 1985, addresses project operation in relation to DO levels during the Initial period of operation, before DO dats for the operating plant is swellable. The wording of the MOA is as follows:

"...The Licenses agrees to meintain interim standards that will minimize short-term water quality impacts, provide werning notification to the Corps, and an approved action plan for continued hydropower generation if the dissolved oxygen concentration falls below 6 mg/l, and cease power generation if dissolved oxygen levels drop below 5 mg/l."

We feel that a temporary and possibly permanent operating program based on the above concept would be effective and economic for the New Cumberland hydro site, as well as for other sites for which the study recommends splitage. Splitage would be an option during periods of concern when the DO level fails below 6 mg/l. In order to be effective, this program would need to be applied to, and coordinated with, all of the affected sites (per the report, dams :

- 359. Staff recommends maintenance of 6.5 mg/L of D0 for several reasons. Recent evidence indicates that concentrations as low as 5 mg/L may cause significant effects to some aquatic species. Allowing hydropower projects to reduce D0 concentrations to 5 mg/L would leave no additional D0 available for other uses, such as additional waste loads from industries and municipalities. The benefits of many millions of dollars worth of wastewater treatment should not be degraded by hydropower. Staff agrees that real-time monitoring and management of water quality should be considered as an alternative to continuous spill flows (see response to comment #243 and Recommendation 7, Section 5.4.2.1.
- 360, Recommendation 7 in Section 5.4.2.1 encourages implementation of a basin-wide model to determine instantaneous spill flows. Such a system would increase power production while monitoring and managing water guality.

upstream from and including Pike Island). We feel that such a program, coordinating all hydro projects through the COE as in the case of Hannibal Locks and Dam, would be effective.

Based on data we have gathered and analyzed, action would be required for only a few days in a typical year using the 6 mg/t "warning notification" criteria of the COE unless and until the upstream power plants were built, brought into operation, and were to cause degradation of upstream dissolved axygen. It would be necessary to develop a scheme for ellocation to each project of the responsibility for meintaining DO levels as each project passed through the development process. It should be feasible to develop flexible and fair guidelines for responses by project developers while meeting environmental criteria.

- 5.4.3 Site Specific Recommendations. New Cumberland Locks and Dam The 361 recommended split of at least 4,000 cubic feet per second (cfs) yearround and 15,000 cfs during the summer months (July to October) is unreasonable, and may prevent development of the project and result in an uncconomic decision. The recommended splitage is not justified by the study for the following reasons:
 - 1. The basis of analysis for DO and fish growth is unproven. [352
 - No adequate enalysis of the banefits of the upstream spills without the New Cumberland spill or at a reduced New Cumberland spill is provided.
 - No adequate evaluation is given of the relationship of fisheries benefits provided to the value of fisheries protection measures.
 - 4. No adequate consideration is given to the option of payments to resource agency programs in compensation for estimated adverse impacts. Such payments have been proposed as an appropriate mitigation measure by Ohio and West Virginia for fish mortality.
 - 5. The loss of generation resulting from the high spillage recommended in the EIS may make the project infeasible and result in its not being built. The spillage requirements are unreasonably rigid. No reasonable justification is provided for the 4,000 cfs year-round spill; at the Racine and Greenup projects, zero spillage operation is practiced and the fishing is excellent. The 15,000 cfs summer spillage should not be required based on the study; if actual DO conditions, operating prigrams for spillage based on real-time measurements of DO would allow a much more efficient operating program while protecting fish. This approach has been agreed to es a satisfactory program at an existing site.

- 361. Comment noted.
- 362. See responses to comments #337-350 concerning adequacy of the model analyses. Staff used a model published in the scientific literature and cited in the references as the basis for linking DD concentrations in the Ohio River and fish growth responses. The model was "proven" in the pond environments for which it was developed. The mechanisms incorporated into the model are not unique to ponds and apply as well to the Ohio River. The model was used in a comparative way (see response to comment #351) which yields an indication of the responses of fish to the conditions that change with hydropower.
- 363. Spill flows were determined in such a way that power production in the basim, not at any individual project, is optimized while meeting water quality constraints. This objective is in accordance with Commission responsibilities. Analyses of the benefits of additional upstream spills or reduced New Cumberland spills are not addressed in the discussions of the recommended alternative because such changes would result in either lower overall power production or failure to maintain the desired water quality.
- 364. See response to comment #357.
- 365. Staff recommends compensation as an interim alternative and potential long-term alternative in Section 5.4.2.2, recommendation 2.
- 366. The non-critical season spill flows are recommended to assure water quality and to protect aquatic habitat. See response to comment #243 concerning selection of mitigation measures. The comparison to the Racine and Greenup projects are not meaningful because those dams do not provide the relatively high aeration rates that New Cumberland dam does.

9

6. Loss of the project would result in losses much greater than the action because of the loss in potential employment during project construction, and to a lesser degree, during operation in an area with already high unemployment.

FISH MORTALITY CONSIDERATIONS

368 4.1.2.3 Entrainment and Turbine Induced Mortality. This section of the EIS considers factors that were similarly considered in the report of studies at the Racine project conducted jointly by the project applicant and the Dhip Power Company. The study at the Racine project gave a reasonable indication of the level of mortality to adult game fish and allowed the applicant to establish reasonable estimates of compensation. The opproved plan for the study at Racine did not include the question of the effect of turbine induced mortality on overall fish populations (game and forage); the question of the effects on icthyoplankton and invenile mortality; or the effect of mortality of forage species on game fish populations, as all parties (the Licensee, WVDNR, ODNR, USF & WS) agreed that such studies involved great expense with a substantial risk that the results would not be useful. We have, to date, seen no study approach that would yield reliable enswers to these questions within reasonable time and cost frames. Such studies would need to address three areas: (1) extent of inthycolankton and invenile mortality and the resulting effect on population; (2) effect of forage fish mortality on game fish populations; and (3) gaographic extent of local oppulation effects of mortality, including adequate data on migratory habits of some of the species of fish present.

> We do not feel that studies of fish mortality and mortality effects on population are likely to be economical or conclusive and suggest rather that the overall lishery resources be monitored, if necessary, by population sampling techniques as the Commission has recommended for the New Martinaville Project at Hannibal Dam. If no adverse pattern is revealed by these more general, effect-based studies, then the far more extensive and costly cause-based studies can be deferred. Use of funds paid by licenses as compensation for fish mortality to an "Chio River Bas's Environmental Fund" could be used for such studies, if conside is appropriate by the administering agencies.

4.1.2.3.4 Staff Conclusions on Entrainment. We concur with the staff conclusions that no fish diversion devices have been shown affective in the conditions considered. Proposed further investigations of the effectiveness of potential fait diversion structures may be warranted, but only if a need can be demonstrated. Contributions by licensees to an "Ohio River Basin Environmental Fund" are a reasonable alternative to diversion measures and have been accepted as such by Ohio and West Virginis. Such funds could be used to carry out such

- 367. In addition to the lost benefits from power production, as shown in Table 5.2-1, the beneficial and adverse socioeconomic effects of project construction and operation (identified in Section 4.1.6.3.) would be eliminated.
- 368. Nowhere in the DEIS did staff recommend the extensive studies outlined in this comment. Staff recommended monitoring of entrainment rates and fish mortality at selected, representative sites and general population monitoring in the vicinity of each facility.

J-170

369. See response to comment #338.

10

studies by the concerned agencies. We do not concur with the staff conclusion that installation of fish protection measures (structural devices) is likely and do not believe that installation of bypasse and related features in enticipation of such installation is desirable or economical at present.

We feel that the design concept adopted by the applicant for this project, incorporating a large, low velocity cross section for both the power plant intake and the inlet channel, provides the best protection against high mortalities and is consistent with sound power plant design. We do not believe that any structural system can be designed to economically or effectively divert fish in the Ohio River environment. Because of the low head at each dam on the Ohio River, the flow of water required to generate a kilowatt hour of power is relatively large. Any device attempting to somehow process this flow is necessarily large and costly, and causes a loss of energy. This problem is compounded by the large trash loading of the river, which places great burdens on any structure installed in the flow. The associated costs will result in power production costs in excess of those warranted by current or expected future power values and consequently cencellation of projects.

5.1.1.2 Aquatic Ecology and Fisheries (Turbins Induced Fish Mortality)

The comment that "mortailty of larger fish would be greater than 10 percent of those entrained" has no basis in the Recine study, data presented in this report, or other data which we have raviewed. This statement is inappropriate unless some basis can be provided. Data provided in the Racine report (Volume I, page 13) shows survival rates of 88 to 108 percent with instantaneous survival of 97.9 percent and latent survival of 94.5 percent for oame fish.

5.4.2.2 Recommendations on Aquatic Ecology and Fisheries

Item 1 - We disagree with the suggestion that licensess provide for undefined and unproven intake protection structures where there is no evidence that such are required and no evidence that any particular technology would be effective. Such installations would involve extensive costs for no indicated gein.

Item Z - We disagree with this early timing (12 months after the 1372 license issue date) for agreement on companiation payments.

Item 3 - We disagree with the requirement that licensees conduct fisheries related studies. Compensation paid to the resource agancies should provide for this effort. The agencies could then properly decide on expenditures for various categories of effort; fish restocking, angler access, fisheries studies, or other uses. 370. See response to comment #339.

371. See response to comment #340.

372. See response to comment #341.

373. See response to comment #342.

Item 4 - We question the effectiveness of the requested montailty study but feel that any such studies to be conducted should be from compensation funds.

Item 5 - Bioengineering studies, if considered appropriate, 375 should be paid for from compensation funds. Any open-ended requirement to install diversion structures should be subject to formal hearings at the commission and subject to economic constraints for each project.

.

.

374. See response to comment #342.

375. See response to comment #342.

.

Testimony Submitted at Public Neeting Federal Energy Regulatory Commission Docket No. EL85-19-114 Ohio River Environmental Impact Statement Pittsburgh, Pennsylvania - July 15, 1988

Testimony submitted on behalf of the City of New Martinsville, West Virgiula, Applicant for project Nos. 6901 and 6902.

The City of New Martinsville wishes to record the following comments with respect to the above referenced proceedings.

We have submitted comments on the EIS to the Commission on July 5, within the comment period. Those comments were in summary: (1) that the EIS proceeded from poorly founded analysis of dissolved oxygen (DO) conditions, (2) that recommended studies and provisions for fish diversion structures are not justified or shown to be effective; and (3) that the means proposed in the EIS for meeting the 6.5 milligrams per liter (ang/1) DO standard were wasteful. We proposed an equally effective and much more economical means of meeting the 6.5 mg/1 DO standard, if it is to be applied.

Subsequent to submitting that material, we have considered several additional items which we wish to bring to the attention of the Commission:

- 1. The Commission should not judge the financial feasibility of a project prior to licensing, but should leave the decision to the Applicant/License to make after the license is issued. Receipt of the license may allow the Licensee to negotiate without the license, and may allow the Licensee to negotiate a more supportive contract than would have been possible prior to issuance of the license.
- 2. The draft EIS does not give the equal consideration of environmental issues and power production required by the amendments to the Federal Power Act of 1986 (ECPA). The EIS recommends development to achieve the minimum environmental impact by "maximizing protection of all target (environmental) resources" (page xviii) without regard to the resulting power loss. The EIS proposes a major impact on power production (a reduction of 28 percent from the already restricted output of the facilities as proposed) while minimizing environmental impacts. For instance, the original New Curberland Project application proposed spillage at a cost of 2 percent of output to minimize environmental impacts on water quality. The spillage program proposed in the EIS would result in a cost of more than 20 percent of output. We feel that the magnitude of cost proposed in the license application, 2 percent, represents a mitigative action conforming properly to the equal consideration requirements.
- 3. The Applicant has not been allowed to participate fully or equitably a 378 in the EIS process. When the initial motice of intent to conduct the EIS was circulated, the Applicant submitted information showing the extent and means of mitigating environmental impacts and requested to be held out of the EIS insofar as license issue is concerned. This request was denied by staff as was a further request to appeal the decision of staff to the Commission.

- 376. Your comment is noted that the Commission should not judge the financial feasibility of a project prior to licensing. Although staff realizes that an applicant usually cannot negotiate a power-sales contract until after a project is licensed, staff must make a judgement that a project is feasible to determine if it would be constructed (see response to comment #3).
- 377. Your comment that the EIS does not give equal consideration of environmental issues and power production is noted. Staff has evaluated a wide range of alternatives that generate varying amounts of power. Staff has made an effort to balance the effects on the environment with the loss of generation. At the New Cumberland project, staff agrees that the estimated 20 percent generation loss is substantial, but required to meet the goals established for Alternatives 3 and 4. Staff's analyses (Section 4.1.1.1) indicate that the applicant's proposed spill flows are not adequate to provide the desired degree of mitigation.

378. See responses to comments #32 and 277.

in the process of developing the draft EIS, the Commission and consultant staff conducted meetings with groups of environmental agency representatives which expressly excluded representation of Applicants. We feel that the resulting EIS represents agency opinions to the exclusion of the opinions of the Applicants or the general public. We feel that the study results make recommendations corresponding closely to subjective agency opinions even though the report presents the results as an objective scientific study. We feel that, as agencies were allowed to participate in the study during the analysis stage, their recommendations have been given unduly strong consideration in the draft report, to the exclusion of the potential recommendations of the applicants.

We feel that the Applicant should have been allowed to participate in the consideration of project Nos. 6901 and 6902 during the analysis stage, rather than only at the draft review stage. We feel that the Applicants' comments on the draft report are likely to have much less influence on the results when given at the draft report stage, rather than at the analysis stage as should have been permitted.

In its submittal of July 5, the Applicant provided proposals for mitigation which were not considered in the EIS analysis and which could have been presented and considered in EIS preparation had the proper participation been allowed. Therefore, the recommendation of the Applicant, included in the July 5 submittal, should be fully considered and adopted in the final report.



Ms. Lois Cashell Acting Scretary Foleral Energy Regulatory Commission 425 North Capitol Street, N.E. Mashington, D.C. 20426

Dear Ms. Cashell

FERC Docket No. EL 19-114 DE15

The Allegheny County Lydropower Programs is responding to the subject 88 document through general comments regarding operation aspects of the proposed Ohio River Dashields Bits and Allegheny River LED 4. 늘 4 There are two major categories of comment: РH 1. Water Quality Modeling ω The statement by FERC in the document on page 2-25. "The most 379 fundamental problem with the concept of licensing projects to maintain existing conditions is that insufficient data are available to define existing DO conditions throughout the System." We concur, and therefore unve that the results of the water quality modeling not be adopted inflexibly as conditions of licensing. FERC has concluded the relationships between DO deficits above 380 and below the dams are linear. We are of the opinion the statement on page 8-8. "neither temperature nor flow rate has a consistently significant effect upon aeration." We conclude that the measured data shows that a linear regression model is not acceptable as a model for dam seration.

Even if linear relationships were accepted, those ostablished for Dashields and L6D 4 obviously are highly questionable. The b and M parameters have been developed from only nine field observations at L6D 4, and 13 at Dashields. The correlation

- 379. The statement quoted in this comment is from a discussion of why staff chose to use a water quality model to determine spill flows that maintain 5.5 mg/L of DO, instead of requiring developers to maintain "existing BO conditions". See response to comment #243 concerning selection of mitigation measures.
- 380. A comprehensive review of the literature and analysis of the available data led staff to conclude that the linear aeration model is the best available technique for modeling dam aeration. There are fewer data available at Dashields and Allegheny L&D 4 for estimating the linear model parameters because the applicant at these sites, Allegheny County, chose not to respond to FERC's request for additional information (dam aeration studies) made in July, 1987. See response to comment #243 concerning how mitigation measures were selected. See responses to comments #100-102 and #112-121 concerning calibration and accuracy of the model. The spill flows recommended at Allegheny LED 4 and Dashields are relatively high because these dams aerate well and are located where aeration requirements are high.

Ms. Lois Cashell Page Two Joly], 1988

FERC Docket No. EL 19-114 DELS

roefficients shown for these projects in Table B-1 do not demonstrate statistical reliability; quite the opposite. In short, the fundamental acration capability determined by FERC for the existing dams at these sites can only be characterized as a broad approximation.

The field observations of DO on which the above relationships were established were taken generally during the summer low-flow season, and represent probably the most significant contribution to aeration rendered by the duns in question. Also, the entire river model was calibrated against data collected in the summer of 1983, which had July-October flows of only two-thirds normal for the past twenty years. Nonetheless, the model was then used to establish spill requirements which FERC proposes be applied to all years, wet or dry.

The statement on page B-13, "the addition of substantial amounts of BCD was required to simulate DD deficits as large as Measured deficits." The possibility that inaccurate inputs (point and non-point sources) could skewer the calibration to most any assumption desired, occurs. The conditions of flow and temperature will not be constant for all times and conditions which in itself invalidates the assumed model calibration.

It appears the recommended spills during low-flow periods for Dashields and Alleghany River LED 4 are higher than needed to maintain 6.5 mg/1 DO. The recommended levels seen to place a penalty upon Dashields and LED 4 (and others in the lower Alleghany region) in order to enhance downstream projects DO criteria.

2. Impact on Project Feasibility

To evaluate the impact of the operating constraints proposed by FERC on energy production at Dashields and LGD 4, computerized Operation studies were performed for both projects based on mean dealy recorded flows for the twenty-year period 1967-1986. Earlier flow data were not used since the flow regime was altered in 1967 by upstream storage projects which came on line in that year.

381

381. The existing dams at Allegheny L&D 4 and Dashields are important aerators. The estimated generation loss of more than 20 percent is a tradeoff that must be considered when Alternatives 1, 2, 3, and 4 are evaluated by the Commission. Ms. Lois Cashell Fage Three July 1. 1988

FERC Ducket No. E. 19-114 DEIS

The results of the studies are summaried as follows:

	Average Annual Energy Generation, GWH	
	L4D 4	Dashields
With spills as proposed in County's license applications	77.6	115.0
With spills as proposed in FERC's DE15	61.2	83.9
Percentage reduction in energy generation	21.1	27.0

It can be seen the impact on Dashields and LGD 4 are rather severe when several other Pittsburgh area dams are viewed in relationship to the entire river system.

We therefore suggest the water quality model is tenuous; FERC should develop alternative monthly spill requirements based on wet, everage and dry years, as opposed to using dry year only under all conditions.

Compality MES W. KNOX, Director

co: Mr. A. E. Carl

.

382. See response to comment #243 concerning selection of mitigation measures.

.

.

FR FO DEFICE BY THE LECOLTARY

Mitex, Inc. 88 819 29 AM 9: 47 A Member of the Sithe/Energies Group 91 Newbury Street Boston, Massachusette 02110 Telephone #171 424-1888 Telecopler (617) 287-7687 CECULAR BEEN RAMISSING PUBLIC REFEILINCE ROOM June 28, 1988 Ms. Lois D. Cashell Acting Secretary Federal Energy Regulatory Commission 825 North Capitol Street, NE Washington, DC 20426 8 RE: Gallipolis Hydro Project; Ě FERC Docket No. EL85-19-114; Draft Environmental Impact Statement for Hydroelectric <u>+</u> Development in the Upper Ohio River Basin ₽ Dear Ma. Cashell: -Gallia Hydro Partners ("GHP") is the applicant for license form the proposed hydroelectric development at the Gallipolis Locks and Dam on the Ohio River (FERC Project No. 9042). GMP competes with WV Hydro, Inc. and the City of Pt. Pleasant, West Virginia (FERC Project No. 10098) for the license at this site. Gallipolis Locks and Das is contained in the Commission's cumulative impact assessment for the upper Ohio River and

evaluated in the Draft Environmental Impact Assessment for Hydroelectric Development in the Upper Ohio River Basin ("DEIS"). This letter contains GHP's comments on the DEIS as they pertain to Project Nos. 9042 and 10098.

The following represents the specific comments of GHP regarding Project No. 9042.

On pg. 4-60, Table 4.1.6-1, the DEIS summarizes the land use requirements for transmission facilities. For Project No. 9042, the DEIS summarizes that 29.5 acres would be required, assuming a right-of-way width of 100 feet. Again, on pg. 4-74, Table 4.1.6-10, the DEIS summarizes the estimated area of habitat changed or lost from construction and maintenance of the proposed transmission line corridors. Here, the DEIS estimates that the habitat area changed or lost for new ROM would amount to 36.4 acres of land.

383

383. Text has been modified. Calculations are based on numbers provided in the application of a 10,000 foot new ROW, 3000 feet on existing ROW, and 2630 feet of new ROW. These numbers differ from the information provided in the June 28, 1988 response to the DEIS.

¹⁾ Land Use

> As GHP calculates the land use for transmission purposes, for Project No. 9042, approximately 10,500 feet of right-of-way ("ROW") will be new and approximately 5,730 feet will parallel an existing 100 foot ROW. Assuming a 100 foot ROW width for the new transmission line and only 50 feet additional for the paralleling segment, the "New ROWis:

(10,500 X 100) + (5730 X 50)3 / 43,560 = 30.6 acres

Approximately 1000 feet of the 50 foot ROW is over the river so the 30.6 acres should be reduced by 1.1 acres leaving a "New ROW" of 29.5 acres (as shown in Table 4.1.6-1).

With regard to section 4.1.6.4, GHP would like to note that the Ohio Historic Preservation Officer did grant clearance for all construction at the Lock and Dam area (1986). He did, however, request a survey of the transmission line route.

2) Recreation

Page 4-47. Section 4.1.3.4: It is recommended that a bypass flow system through the powerhouse of all projects be installed to maintain fishing opportunities when powerplants are not generating. GHP believes that this provision is unnecessary at the Gallipolis site because the incremental increase in fishing opportunity would be very small compared to the increase in capital costs and adverse impacts on generation.

Such a bypass system would be useful only when the sum of the turbine minimum operating flow, spill flow, lockage, and leakage is less than the total river flow, when river flows are too high for sufficient operating head, or when the powerplant is shut down for any other reason. For GHP's proposed project, the minimum river flow for operations is approximately 5100 cfs which is exceeded more than 99.99% of the time (see Table B-1, Application for License, FERC Project No. 9042, January 1986; 99.99% exceedence flow is 7924 cfs; shutdowns for low flows will not be a problem at this site. The hydroproject will be shutdown for insufficient head when flows are greater than approximately 120,000 cfs at which point the downstream tailwater elevation is approximately 12 feet above normal and flows will be turbulent across the river; a 2000 cfs bypass would be insignificant. Finally all planned maintenance or

384. Although the Ohio Historic Preservation Office granted a clearance of activity at the lock and dam areas, a conditional finding of no adverse effect was given for the transmission line route. A Phase II cultural resource survey of the transmission line corridor would need to be performed prior to project construction.

385. See response to comment #165.

Mitex, Inc.

> shutdowns would be performed on only one unit at a time when flows allow operation of only one unit. Therefore, planned maintenance will require that only one unit be off-line, thus tailrace flows will be maintained. Unplanned shutdowns for hydroprojects average approximately 3%, or eleven days per year. These shutdowns would thus result in approximately four days of lost fishing opportunity during the four months of highest recreational usage (assuming that fishing conditions are favorable during the period of unplanned outage).

> The cost of such a bypass system at this site would likely manifest itself in terms of decreased installed capacity as well as increased capital costs (a ten foot diameter pipe would likely be required to maintain velocities at 25 feet per second or less). Because of the site constraints at Gallipolis (pier 8 on the riverward side and the steep bank to Ghio Route 7 on the landward side), the ten foot pipe and associated structural elements would decrease the area available for turbine generator units so the 8.2 meter units proposed may not be usable. Smaller units would allow less efficient use of the available water resource and less energy.

In summary, the bypass system at Gallipolis would only mitigate for approximately four days of lost fishing opportunity during the four months of greatest retreational usage and would increase capital costs while decreasing generation. The benefits of the bypass system at the Gallipolis site are not of the same magnitude as those at dams which provide significant aeration and must cease operation during low river flows.

The following represents the specific comments of GNP with respect to Project No. 10098. In Section 5.4.1, "Recommended Alternatives", the staff states that their comparison of the competing hydropower applications at the Gallipolis L&D reveals that the proposed projects do not significantly differ, "...either in environmental acceptability or in power generating capabilities" (pg. 5-23). With respect to relative impact values, GHP believes that significant differences do exist with the second phase of construction proposed by Project No. 10098. GPH believes that consideration should be given to the environmental impacts associated with the Phase II development proposed by Project No. 10098. Specifically:

 section 2.1.1.23 describes the proposed two phase development of Project No. 10098; the second phase consisting of installing two 12.5 MW units in the riverside

386

386. Development of a hydroelectric project at an existing Corps of Engineers' lock and dam will only be acceptable if it does not interfere with the existing navigation project. See response to comment #392.

Mitex, Inc.

lock. The Corps, in a letter dated February 16, 1988 (copy attached) and attached to GHP's Better Adapted Statement filed with the Commission on March 18, 1988, states that:

no consideration will be given to a hydro plant located in an existing lock chamber or on the island created between the old and new locks. [emphasis added]

Therefore, all analyses of Project No. 10098, if "phase 2" is considered, must assume that the second powerhouse will replace a second gate outside the existing lock chamber (requiring a change to Figure 2.1-25 in the DEIS).

2) Section 2.1.1.23 also states that the transmission line would be 1.7 miles long. However, WV Hydro, Inc. and the City of Pt. Pleasant, WV propose two 1.7 mile long transmission lines, one on each side of the river (see Exhibit G, Project No. 10098).

Following is an itemization of the evaluated resources, and a description of the additional impacts associated with the phase II development of Project No. 10098.

- A. Water Quality The short term adverse impacts of increased turbidity and resuspension of sediments due to construction will be doubled with the two distinct phases of construction contemplated by Project No. 10098.
- B. Fish Mortality Project No. 10098 will utilize more water than Project No. 9042. The passage of additional water through the turbine units of Project No. 10098 will result in the increased potential for fish mortality. The location of the phase II station on the opposite side of the river for Project No. 10098 doubles the risk that shoreline species may be impinged or entrained.
- C. Fish Habitat The second phase of Project No. 10098 will cause greater decreases in fish habitat (in the same manner as described for initial project development in Section 4.1.2.2.3).
- B. Wetlands Habitat Phase 2 transmission line construction for Project No. 10098 may impact wetlands habitat associated with Flatfoot Creek which is located downstream on the West Virginia side.
- E. Recreation Project No. 10098 consists of two distinct 391 construction periods which will cause twice the adverse short term impacts to recreation. Furthermore, the second

Mitex. Inc.

- 387. Although this additional impact may occur, it is considered minor.
- 388. Staff agrees that operating powerhouses on both sides of the river will increase the likelihood of entrainment damages to species that move along shorelines. This essentially doubles the risk of a singleshoreline plant, assuming equal numbers of fish move along each shoreline.
- 389. Staff believes that this assumption would need testing with a physical hydraulic model. A second powerhouse could actually spread flow across more of the river and create-better tailwater habitat.
- 390. Correspondence dated May 22, 1987 to FERC from the applicant for Project No. 10098 indicates that the Phase 1 transmission line along the Ohio side has been deleted and Phase 2 is the preferred route. The path of the Phase 2, as shown on the maps, will pass through the Flatfoot Ereek wetlands. A site-specific recommendation (recommendation 17, Section 5.4.3) is made that the applicant schedule a field trip to the site with WDNR, FWS, and Corps staff prior to final pole location determinations. Minor deviations in alignment or other site-specific mitigation measures can also be determined. Prior to construction, the developer should provide a plan for approval by FERC and the appropriate state and federal agencies. Any necessary permits must also be obtained prior to construction.

391. See response to comment #215.

phase of construction will not provide any additional benefit to recreation.

F. Land Use

Transmission Line

392

GHP believes that the transmission line routes proposed by Project No. 10098 (see Exhibit G, Project No. 10098) will cause significant impacts to Land Use and aesthetic appeal in the 1.7 miles downstream of the dam on the Ohio and the West Virginia sides of the river. 138 kV lines will extend parallel to the river very close to the shore on both sides. Furthermore, the transmission line for the first phase of Project No. 10098 traverses a very large amount of residential land as may be seen in the attached Property Map for the Gallipolia Dam area. In this regard, pg. 4-59, last paragraph and Table 4.1.6-2 misrepresents the area through which Project No. 10098 will pass its' transmission line. In its first 5000 feet, the transmission line will traverse approximately 18 parcels of property; ten of which are almost definitely residential with associated structures (see USGS quadrangle, Apple Grove, WV, section attached). GHP's transmission line, on the other hand, crosses Ohio State Route 7 and two structures before entering wooded areas and bypassing all other structures.

In addition, the the 1.7 mile distance for Project No. 10096 in each phase is only the length to existing 138 kV line. It is highly unlikely that the two lines would terminate at the existing 138 kV line as the interconnection facility for each would cost in excess of \$500,000 and probably closer to \$1,000,000 based on interconnection experience with APS at projects under construction. A more likely solution would be to continue both lines, using ROWs parallel to the existing 138 kV line, to the Apple Grove substation as proposed by Project No. 9042. This would extend the required transmission line by approximately 1.4 miles for a total length of 4.8 miles.

Assuming a 100 foot ROW for single lines and 50 foot ROW for parallel lines, the total acreage required for Project No. 10098 is 58.2 acres. Using the same assumptions as for Project No. 9042 above, the "New ROW" for Project No. 10098 would be:

C(2 X 1.7 X 5280 X 100)+(1.4 X 5280 X 50)-(1000 X 50)3/ 43,560

= 48.5 acres.

392. The applicant for project 10098 filed a change in the proposed transmission line route with FERC in a latter dated May 26, 1987. This change eliminated the previously proposed route along the Ohio shore of the river. The impacts of the revised route are as described in the DEIS.

Milex, Inc.

- Ms. Lois D. Cashell
- June 28, 1986

 N
 Ms. Lo

 O
 June 2

 Page 6

 51

0

÷

8

- 24

Access Road

- As described under section 4.1.6.10, on pg. 4-71, an access road similar to that proposed by GHP for Project No. 9042 will be required by Project No. 10098 even if not specifically stated in the application for license for Project No. 10098.
- G. Socioeconomics Froject No. 10098 has a much greater potential for increasing flood elevations due to the additional obstruction provided by its second powerhouse. As described under "Indirect Impacts" on pg. 4-61 there is a high probability of some increase in upstream flood elevations at the Gallipolis L&D. However, the replacement of two of eight gates (Project No. 10098) could provide as much as a 100% increase in flood impacts compared to replacing only one of eight gates (Project No. 9042).

With respect to project economics as presented in Table 2.1.1-3, on pg. 2-4, the cost for Project No. 10098 may be significantly higher than included in the table with the proposed second phase of construction requiring replacement of a gate bay instead of installation in the existing lock chamber (see item 1 (pg. 3) above).

Major sources of increased cost would include the demolition of the existing gate structure, installation of a complete cofferdam, and construction of a powerhouse similar to that proposed for the first phase of construction. In addition, Table 5.2-1, on pg. 5-20 would require revision to reflect any modifications made to Table 2.1.1-3.

Thank you very much for the opportunity to provide these comments on the DEIS.

Sincerely.

Michael G. LaRow

Agent, Gallia Hydro Partners

MGL/cay

- Enclosures: Corps letter dated February 16, 1988 Copy of USGS Apple Grove Quadrangle Property Map for Gallipolis Dam Area
- cc: Jim Richards, Sithe George Taylor, FERC File

- 393. Text has been modified to reflect the need for a short access road in the Gallipolis (No. 10098) project.
- 394. Sections 4.1.6.1 and 4.1.6.3 have been revised to indicate that project 10098, because of the Phase 2 development, has a greater potential for causing long-term increases in upstream flood elevations than does the competing project (9042) at Gallipolis.
- 395. Comments are noted.

Three enclosures provided by Mitex, Inc., are not reproduced here.

Mitex, Inc.

Lois D. Cashell. Secretary Federal Energy Regulatory Commission 825 W. Capitol St., NE Vashington, DC 20426

July 25, 1988

Comments on Draft Obio River Environmental Impact Statement, RL-85-19-114

Dear Ms. Cashell:

This letter responds to the Draft Environmental Impact Statement on the Ohic River Basin issued by the Federal Energy Regulatory Commission recently. Fosh Corp. and a related company, WV Hydro, Inc., are pursuing six hydroelectric projects that are included in the study and two others that may be included. We view the Draft as an excellent effort performed in a short time period on a difficult topic. We wish to provide some comments on points that are important and should be considered in the final EIS.

396

1. The environmental impact of the four siternatives is not accurately assessed, because it does not consider the environmental effects of the alternatives to generation by hydropower. Decreasing hydropower generation as proposed in alternatives 2 through 4 of the DEIS will not result in less electrical demand, but will change the generation from hydropower to fossil-fired capacity. The environmental impact of producing power by a fossil-fueled source rather than by hydropower has not been addressed.

The generation of several hundred GVh of electricity per year by hydropower would avoid the mining of hundreds of tone of coal, which would probably be strip mined, and would avoid air pollution of millions tons of sulfur and airrows oxides.

During the 50 plus year life of the hydro projects in the heavily populated Ohio River Valley, the environmental impact of substituting fossil fired generation for the mome of the hydropower generation proposed in alternative 1 will have a significant impact on the terrestrial shd air quality of the region. There will be some effect on water quality from erosion of mined lands and runoff from ask piles. The benefit of hydroelectricity to the environment far exceeds the detriment and avoids the impost from acid rais and other fossil-fuel pollution.

2. The DEIS ignores the best method for adding DO to the river: air injection. This is a preferable method to the large spills suggested for projects in the upper Ohio and Mon Rivers. The DEIS repeatedly states an error: that injection is not a demonstrated technology.

#oab Corp. & WW Hydro, Inc. 120 Calumet Ct Aiken, 3C 29801 (803) 642-2749 396. The concern for increased environmental impacts is noted. Table 5.1.4-1 and Figure 2.5-2 include the summation of the estimated effects of all of the proposed upper Ohio River Basin projects.

397. The EIS does not ignore mechanical aeration, but states that it must be proven as a viable mitigation measure before it can be recommended or approved (see Recommendation 8, Section 5.4.2.1). Turbina aeration has been used at a number of large high-head hydroelectric plant, but with mixed success. Even with the low flow rates of such plants (compared to the projects proposed in the Ohio River Basin) large blower capacities are required to attain much aeration. There are significant differences between such high-head plants and bulb turbines, so mixed success at high-head plants does not assure success at bulb turbines. Staff recommends developers be allowed to replace spill flows with mechanical aeration only after they have shown it to be effective, and that other benefits of spill flows (e.g., reliability and improvements to aquatic habitat) be taken into consideration in reducing spill flows. This expected increase is consistent with those observed on the Kanawha River and at other hydro plants (ref. 2, 3, 4, 5, 7). The fact that injection has never been used in a bulb turbine on the Ohio River does not mean it will not work there. From successful installations referenced the design of an injection system can be tailored to the particular application. Page 4-9 of the DEIS correctly mentions the success of injection systems in other slightly different applications, then contradicts the success at esveral locations.

injection seems like a more economical siternative to large spills and the resulting lost generation. This is particularly true of the New Cumberland project on the upper Obic (ours -10332) and is true to a lesser extent on the upper Non (Point Marion through Hildsbrand). Injection should be an option available to developers to add DD to the river to overcome any deficit caused by hydropower.

Minimum blower capacity on the upper Mon will equal 3,000 cfm (cubic ft./min.). The number and size of blowers can be increased if required by the measurement of dissolved oxygen during operation. The 3,000 cfm of air injected into the tailrace by the system will increase the DO level by 1.02 mg/l at the minimum discharge of 1200 cfs. This amount of air injection is equal to or greater than any aeration measured in the operating range for the upper Mon projects (ref. 1). Blower capacity at Maw Cumberland is sized according to the flow increases.

3. Ve do not agree that Alternative 4 is the best choice 393 for implementing hydropower in the upper Ohic River. The reason for choosing alternative 4 seems to be that it requires a target value of 0.5 mg/l of DO and allows more capacity than alternative 3 by spilling a significant amount of water in the summer and fall. There are several reasons why we do not agree that alternative 4 is preferable.

a. A target of 6.5 mg/l is an enhancement over present 399 water quality, not maintenance of present conditions; hydropower generation should not be reduced to overcome losses in DO caused by other sources. A DO level of 6.5 mg/l is an unnecessarily restrictive burden for hydropower and is not merited by the potential impact on DO from hydropower. Apparently, equatic life is doing well under the present conditions; we see no reason to improve them at the expanse of hydropower. The quoted benefit on fish species at higher DO is a laboratory study and has not been verified in the Ohio River bared on a single lab experiment.

b. As discussed in point 5, the DO model seems flawed. If (400 6.5 mg/l is desired there is no need to worry at the upper Non projects (Point Karion through Opekiska), the DO is almost never below those values based on measurements we collected. The same is true of our other proposed project at New Gumberland.

Roah Corp. & VV Hydro, Inc. 120 Calumet Ct. Aiken, SC 29801 (902) 410,0200 398. Comment moted. See response to comment #399.

399. Staff does not recommend that hydropower projects enhance D0 concentrations, since spill flows cannot increase D0 above what it would be without hydropower. Staff does not believe hydropower generation should be allowed to negate water quality benefits of the many millions of dollars of public and private money spent on wastewater treatment in the basin.

400. The model was designed to simulate DO under relatively severe conditions, so that recommendations based on it will maintain DO concentrations over the wide range of conditions that occur. Comparison to measured DO concentrations (Figure 3.4.2-2) shows that model results are not unreasonable. DRAFT ENVIRONMENTAL IMPACT STATEMENT - OHIO RIVER HYDRO PROJECTS July 25, 1966 Page 3

The data presented in the license applications for all 5 of these projects (including Korgantown) as well as reference 1 and data measured by us at Morgantown and Point Marios in 1987 shows that DD is seidom below 6.5 mg/l. This is especially true by the operating plan we propose which does not generate at low flow.

Dissolved oxygen at Point Marion is soldow below 7 mg/l and has never been measured below 6.3 mg/l according to existing data. The average increases in DD at the dam is 0.6 mg/l for Corps measurements; our measurements in 1987 showed an average of 1.2 mg/l increase at generating flows.

Dissolved oxygen below Korgantown Dam is seldom below 6 mg/l and has never been measured below 5.7 mg/l according to existing data (see reference i and 1987 measurements). At planned generating flows (> 1050 cfs) the observed DO increase in 0.4 mg/l, which is within the measurement error for field DO measurements.

Dismoived oxygen below Hildebrand Daw is never below 6 mg/l according to the Corps of Engineers and recent data (ref. 1). About 70% of the time an increase in DO was observed (ref. 1) in 1986.

Measurements in 1986 at Opekiska showed a significant increase in DO occurs about 25% of the time, but usually at low flow. The average observed DO increase in reference 1 is slight, 0.2 mg/l. Air injection should not be necessary, because the DO increase does not occur during generating flows. Although a potential problem exists, dismoived oxygen below Opekiska Dam is rarely below 5 mg/l according to the Corps of Engineers, and not during generating flows. Hone of the observations in 1986 (ref. 1) showed DO below 6 mg/l.

Keasurements at New Cumberland showed DO was not below 6.5 mg/l there. The averation effect of the dam could be replaced by air injection of the cepacity suggested in the license application.

c. Ve cannot determine the basis for the 1500 cfs spillsge 401 proposed in the DEIS for the upper NON (15,000 at New Cumberland). Based on the existing DO this spillage seems unnecessary: it is not supported by measurements at the sites. Ve cannot determine how the FERC arrived at a spillage flow because the DO model does not depend on flow. A high spillage means operation will only occur about 10 to 15% of the time in the critical months. We do not find spillage to be an economic alternative to infection: the cost is prohibitive at New Cumberland and probably at other upper Ohio dams.

\$cah Corp. & VV Hvdro, Int. 120 Calumet Ct. Aiken, SC 29801 (803) 542-2749 401. The recommended spill flows were determined by finding the combination of spill flows at all dams that provides the most power generation while maintaining D0 concentrations of 6.5 mg/L (or maintaining preproject D0 concentrations where preproject concentrations are less than 6.5 mg/L). This optimization was performed under the simulated "summer Moderate flow conditions" described in Section 4.1.1.1. The D0 model determines the amount of aeration as a function of (1) the aeration capacity of the dam (i.e. how much D0 the spill flow picks up) and (2) the amount of flow that is used for hydropower vs. spillage. See response to comment #397 concerning mechanical aeration.

DRAFT ENVIRONMENTAL IMPACT STATEMENT - ONIO RIVER HYDRO PROJECTS July 25, 1968 Page 4

d. A spillage like that proposed will mean generation does 402 not occur for a high percentage of the summer and fall months. Ye wonder if FERC considered the minimum generation necessary for the turbine configuration. At the upper Kon sites with a single large unit the minimum generating flow is about 800 - 1000 cfs. if we spill 1500 cfs there is no generation until the river flow is 2300 to 2500 cfs. Based on our proposed generation plan in the upper Kon sod at Haw Cumberland, there is no generation about 70% of the summer and fall.

As a result of installing one large unit at the four dams on the upper Non, the minimum flow used for generation is approximately 1050 cfs. A generating flow of 1050 cfs corresponds to a river flow of 1390 cfs because of the 340 cfs bypass. For 40% of the year the river flow is balow 1390 cfs; all river flow will bypass the powerplant 40% of the time. Not operating for 40% of the time (at low flow) dramatically reduces the potential of an effect on dimeolved oxygen, because DO is usually increased at lower flows and low DO occurs during low flow periods.

The upillage in Table 2.1.1-2 will change because gaskets will be installed with hydropower to reduce leakage under the dam gates.

4. The model used for calculating spillage does not seem to support the data from which it was derived. The estimated DD shown in Figures 4.1.1-3 and 4.1.1-6 does not agree with mesoured data from the sites in the upper Konongahela River, particularly no DO increase of 2 mg/l at moderate flow (or 2.5 at low flow) has been observed at Hildebrand. The low DO values at low flow

The DO model as stated does not include any dependence on flow. How is there a difference at low and moderate flows? We agree that low flow usually, but not always, means low DO. We have found DO increase to be weakly correlated to flow in regression analysis of field data for the few cases we have examined. Because the correlation model is based on deficit the sensitivity to errors becomes greater. The error percentage of the deficit is much larger than the error percentage of the measurement.

Ve are concerned that the model used in the DBIS is indicating an excessive flow should be spilled to provide a certain javel of DO. The Opekisks increase in DG from hydro generation does not appear to be shown in Figure 4.1.1-6. If DO is low upstream of Opekisks and just below Fittsburgh, injection at the turbine might increase DG easily so that high DO would propagate downstream.

No credit is taken for added DO by hydropower generation. A06 although tests have shown that from 0.3 to 0.4 mg/l is added by hydropower (ref. 3).

Roah Corp. & VV Hydro, Inc. 120 Calumet Ct. Aixen, SC 29801 (803) 642-2749 402. The minimum flows required for generation, as proposed by project applicants, were considered in determining the power produced under each alternative. The installation of gaskets to reduce leakage will increase power production, not change the spill flow requirements. The spill flow requirements are in addition to any leakage.

Under Alternative 1, staff estimates that there would be no generation 74 percent of the time at Hildebrand, 71 percent of the time at Opekiska, and 61 percent of the time at New Cumberland (#10332) during the July to October period. The applicant's estimate of no generation approximately 70 percent of the time is accurate.

- 403. The model was used to simulate conditions under which no field measurements have been made, so model results are not directly comparable to field data. However, data in Figure 3.4.2-2 show that DO increases of 2 to 2.5 mg/L do occur when DO above Hildebrand Dam is low.
- 404. The D0 model does take flow into account in determining velocities, reaeration rates, and concentrations resulting from tributary and wastewater inflows (Appendix B). See Appendix 8 for discussions of sensitivity of the model to flow and other parameters, and of uncertainties in the results.
- 405. Staff believes that the aeration models used simulate effects of spill flows on aeration adequately. No aeration was assumed at Opekiska because under many conditions the dam discharges low-DD water from the stratified Opekiska pool into the Hildebrand pool. See response to comment f397 concerning mechanical aeration.

406. See response to comment #252 concerning zeration at turbines.

DRAFT ENVIRONMENTAL IMPACT STATEMENT - OHIO RIVER HYDRO PROJECTS July 25, 1988 Page 5

- 5. In several places the Draft indicates that competitive projects are roughly equivalent in environmental impact; this is stated for our projects 9999, 7399, 10332, and 10098. In each case our proposed project generates substantially more energy from hydroelectric power and proportionately avoids generation by fonail fuels. We feel that a major shortcoming of the Draft is that it does not take credit for the most obvious environmental benefit of hydropower: evoldance of mining and use of fossil fuels, primarily coal, to generate the energy that the Ohio River Basin needs. This was stated above in 1.
- The energy calculation in 5.2-1 is incorrect for New Cumberland (10332), Point Marion (7680) and Tygart (7399). We used flow data in monthly flow duration curves simulating upstream reservoir operation and vendor data on equipment performance. We do not believe there could be such a large discrepancy between our calculation and that of FERC. We plan to submit an additional calculation for Tygart.
- 5. The fish mortality issue is an unanswered question and will likely always be an unanswered question. The best solution is as stated on page 4-26 that pawsage monitoring will be necessary at the new projects. The significant conclusion is that gizzard shad and drum will reproduce to offset any mortality lowses. The only significant impact from mortality should be that associated with larger game fish; steps should be taken to minimize this impact if observations show it is occurring at a particular site.

The merics of mortality mitigators (regional screen study, etc.) suggested on page 4-33 do not answer the mortality question, but spend money unnecessarily. We think the correct obsice is to ignore mortality for the reason stated above unless observations show there is significant loss of game fish. There is no indication that may economic screening will ever be available for the Ohio River projects.

Please call if there are any questions.

Yours truly. James B. Price James B. Price President

Nosh Corp. & VV Hydre, inc. (20 Calumet Ct. Aiken, SC 2980) (803) 642-2749 407. See response to comment #34.

- 408. Staff energy calculations for New Cumberland, Point Marion, Tygart, and all of the other projects were based on flow data from the Corps of Engineers. The same data was consistently applied to both projects at competing sites.
- 409, Comment noted.

LIST OF LITERATURE

- Woah Corp.; "Assessment of Cumulative Impacts on Vater Quality and Fish Mortality", filed May 22, 1987 with the FERC for the Openisks, Hildebrand and Point Marion Hydroelactric Projects.
- D. C. Ransy, "Turbing Appiration for Oxygen Supplementation", Journal of the Environmental Engineering Division, American Society of Civil Engineers, Vol. 103, pp. 341 – 352, April 1977.
- A. R. Sheppard, D. E. Killer, C. L. Buck, "Prediction of Oxygen Uptake in Hydroelectric Draft Tube Aeration Systems", Proceedings of American Society of Civil Engineers National Conference on Environmental Engineering, pp. 544 - 551, July 1951.
- A. R. Sheppard, D. H. Niller, "Dissolved Daygen in Hydro Plant Discharge Increased by Astation", Power Engineering, Vol. 86(10), pp. 62 - 65, October 1982.
- TVA/OFRED/A&VR-&4/27, "Improving Reservoir Releases". Tentessee Valley Authority, Office of Matural Resources and Economic Development, Division of Air and Vater Resources, Knoxville, Tennessee, September 1984 and successive impress.
- 5. Ohio Fower Co., "Report on the Vater Quality Studies at the Markland Dam", filed September, 1969 with the FERC on Racine

Hydroelectric Project - Project 2570, Exhibit S. Appendix 3

- 7. Ohio Power Co., "Report on the Dissolved Dayges Study at the Winfield Dam", filmi September, 1969 with the FERC on Racius Hydroelectric Project - Project 2570, Exhibit 5, Appendix 2
- G. C. Holdren, R. A. Storm, F. R. Bhaskar, "Results of the Dissolved Cxygen Monitoring Program at the Ohio Falls Generating Station," Louisville Gam and Electric Company, February, 1986.

Third Avenue, Sulle 3040	New York, New York 10022	Telephone (212) 230-2100
June 28, 1988	RESTAND LEVELS NOT	1016000107(212)230-2120
Ms. Lois D. Cashel Acting Secretary Federal Energy Reg 825 North Capitol Washington D.C. 20	l ulatory Commission St. N.E. 426	
Re: Upper Chio Riv FERC Docket No Comments of Al	er Basin Draft Environmental . EL-85-19-114 legheny B and 9 Hydro Partner	Impact Statement
Dear Ms. Cashell:		<u> </u>
the Allegheny Rive Projects (FERC No. included within th are nonatheless me is reached that in operations of the have a direct inte	r Locks and Dams 8 and 9 Hydr 3021). Although these fact e decision scope of the refer ntioned, impacts are assumed, cludes proposed modifications L&D 8 and 9 projects. Thus t rest in this proceeding.	roelectric & ilities are not renced DEIS, they , and a conclusion s to the the Partnerships
Our concern is prim the DEIS. The Comm that means to charm to the confluence of the mainstem Ohio.	marily with water quality rel mission has developed a disac acterize the Allegheny River with the Monongahela, and the According to the DEIS (pg. DO measurements taken by the	tated sections of 41 blved oxygen model from LSD 9 down m on down 3-40) the model corps of ita.
Engineers, ORSANCO	, and the State of Pennsylvan	
Engineers, ORSANCO Although we cannot model, we can comma and 5-26) that dam. River, and that hy elimination of most on page 5-26 that should be re-evalue	, and the State of Pennsylvar Comment on the internal work ent on the results. The DEIS s 8 and 9 efficiently aerate dropower installation will re t of this aeration effect. T the interim minimum flows at lated in light of the model's	cings of the i mplies (p. 4-5, the Allegheny esult in The DEIS concludes L&D 8 and 9 results.

410. The data collected at Allegheny L&Ds B and 9 by the Corps are sufficient to show that significant aeration can be provided by these dams (see the linear model parameters for these dams in Appendix B, Section B.2.1.3). The licensed projects at these sites, if operated with low spill flows, would significantly reduce DD concentrations. If the spill flows for these projects are re-evaluated by FERC, a review of all the DD data measured there should be made.

filing a separate response to the DEIS, and we would like to join in and incorporate by reference their comments into this letter.

With specific respect to Allegheny LSD 8 and 9, we note in the DEIS no reference to the DO monitoring work we conducted last summer. This included extensive transect based measurements during the summer low flow. It is our understanding that this data (which has been filed with the Commission), is significantly more detailed than the site specific Corps Data used to calibrate the DEIS Water Quality Model.

These data show that D0 uptake at both Allegheny 8 and 9 is highly variable and difficult to predict. We do not how the Staff can make arguments like that on page 4-8 ("The other licensed projects at L&D Nos. 5 and 6 ... prevent D0 concentrations from recovering from the decreases below L&D No. 8 and 9 ..."), or on page 5-26 ("... initial model results indicate that higher spill flows at Allegheny L&D 8 would reduce downstream D0 degradation and allow more generation at downstream projects.") after having observed these field conditions. Certainly, if the Commission insists on referencing L&D's 8 and 9 in the DEIS, at a minimum the model should be recalibrated using the full amount of actual data gathered at the sites.

We finally take serious issue with the recommendation at page 5-26 of the DEIS that interim minimum flows at Allegheny 8 and 9 be reconsidered in light of "the studies conducted for this DEIS". First, we consider the DEIS model seriously flawed due to its limitation on input data, and its seeming incompatibility with observed field conditions. Second, and perhaps more importantly, the minimum flows at L&D 8 and 9 are being determined based upon extensive site specific study. Under no circumstances should this study be supplanted by regionalized, very general inferences that do not seem to account for true site specific conditions.

411

Thank you for the opportunity to submit these comments.

Sincérel William S. Fowler

William S./Fowler Project Manager

WSF/dc

cc: Jim Richards, Sithe George Taylor. FERC File 411. The DD model clearly shows that the licensed hydropower projects at Allegheny LBDS 5, 6, 8, and 9 have the cumulative potential to significantly reduce DD concentrations. The final paragraph of Section 5.4.2.1 recommends evaluating the tradeoffs between water quality and power generation that would be possible if spill flows were changed at these licensed projects. These projects were licensed without an analysis of cumulative impacts on water quality and power generation in the basin, and more effective management of the basin's resources may be possible. Because the effects of changes in aeration at these hydropower projects are cumulative and reach far downstream, the site-specific studies conducted by the licensees should be incorporated into a basin-wide analyses for impacts to be thoroughly assessed and for the best management decisions for the basin's hydropower resources to be made.

ernor er stat etter Mitex, inc.	A Member of the Skhe/Energies Group
June 29 - 2968 June 1998	Telephone (617) 424-1988 Telecopier (617) 267-7687
Ms. Lois D. Cashell Acting Secretary Federal Energy Regulatory Commission 825 North Capitol St. N.E. Washington D.C. 20425	a
Re: Docket No. EL85-19-114; Comments of Upper Mississippi Water Compan Draft EIS for Upper Ohio Basin	y regarding 1
Dear Ms. Cashell:	3

The Upper Mississippi Water Company ("UNSK" or the "Applicant") is the Applicant for License for the proposed Muskingum LockFand Dam No. 3 Hydroelectric Project. The site is being considered within the Commission's cumulative impact assessment, draft EIS for the Upper Ohio Basin. These comments represent the response of UNSKC to the draft EIS.

On page 5-24 the draft EIS reaches the conclusion that the Muskingum L&D 3 site should not be licensed, because it would "have significant, unavoidable adverse impacts to the target resources identified as fisheries, recreation, and wetlands." The staff considers these impacts as "unavoidable", because their analysis reveals that "... adequate site-specific mitigative measures are not known at this time "(p. 5-23).

UNKC strongly believes these conclusions are erroneous. A detailed review of the DEIS shows many misleading or questionable assumptions and observations about the site. UNMC urges that the Commission reconsider its position based upon corrections and additional information supplied in this letter, and recommend license approval.

Following are specific comments, including statements of general conditions which are pertinent in the evaluation of the project and the impacted target resources.

1) The Islands and Downstream Channel

There are four islands located below the dam at Muskingum. The DEIS states that the navigation charts used to identify the tailwater area "depict the islands as being mostly shallow bars, with the center of the islands having floodplain forest vegetation". In addition, the DEIS implies that wetlands and riparian vegetation occur on the edges of all four small 412. Staff used navigation charts and aerial photographs of the lower Muskingum River to determine the number of islands; the license application did not provide a better set of charts. The commentor notes reports on river depths submitted by the applicant, but even the Commission's request for additional information did not yield the needed details of the river between the dam and the downstream navigation channel. Staff has reviewed all available information and revises its estimate of the extent of shallow rapids to about one mile, extending from the dam to the downstream navigation channel. The text has been changed to reflect this reassessment. The original characterization of the riffle in Section 4.1.2.2.3. as "approximately 1000 ft below the dam" is correct.

islands. These conditions do not characterize the site correctly.

The largest island located downstream (approximately 4 acres) is the only island which maintains any sort of permanent vegetation, that may be characterized as having "floodplain forest vegetation". Only one of the other islands have wetland and riparian type vegetation; the other islands have essentially no vegetation, and are only exposed cobbles at low flow.

Furthermore, the DEIS characterizes the channel for two miles downstream of the dam as "shallow rapids interspersed with small islands" (pgs. 4-23, 5-5). This characterization is not correct. Beyond the initial 1000 feet downstream of the dam, the Muskingum is a pool with relatively constant depth; indeed the opposite of "shallow rapids". This has been documented to the Commission in reports submitted by the Applicant in docket 6998.

Finally, the DEIS states that "EHligh velocities and the large volume of water in the backchannel are likely to cause erosion of Ethe largel island, and staff believes this will extend to others downstream". As noted above, there are no islands downstream of the initial 1000 feet. This being the case, any erosion which may occur to the large island will not impact additional islands downstream.

2) Netlands

The DEIS concludes that "significant, unavoidable adverse impacts" would occur to the wetlands associated with the construction of the facilities at Muskingum Lock and Dam No. 3. The description of those areas lost (pg. 4-52) indicates that for construction purposes 1 acre of riparian vegetation would potentially be lost "..along the main shore between the island and the shore...". The DEIS also states that the project operation may potentially damage about 1 additional acre of wetlands and riparian communities. Although riparian vegetation is often associated with wetlands in the transition zone, it generally comprises the terrestrial vegetation, not submerged by water.

Information filed by the UNMC clearly shows that the substrate in the affected channel is almost entirely silt and mud; not wetland type vegetation. In addition, most riparian vegetation is on the shoreline, above the zone that is usually flooded. Consequently, elimination of one to two acres of riparian vegetation may have a minor effect on terrestrial habitat. It will however, have little to no effect on actual "wetlands". The DEIS states at several points (e.g. p. 5-25) that the loss

Mitex, Inc.

413

The USFWS definition of wetlands was used in our analysis. The project area is classified as riverine, lower perennial, open water, intermittently exposed/permanent wetlands. The perimeters of the islands are classified as unconsolidated shore with sub-classes of cobble-gravel, sand, mud, or vegetated. These classifications do characterize the islands in this complex. The increased velocities and volume of water in the backchannel will cause increased erosion of the large island and the two smaller islands located downstream of the large island (Figure C-18) will be further affected by increased turbidity and sedimentation.

413. The substrate in the channel downstream of the dam is "predominantly medium to coarse gravel" (MAPDRA 1986). The backchannel area also has exposed roots, snags, and submerged trunks of dead trees along the bank and within the channel. See response to comment #412. This area provides spawning and feeding areas for riffle-inhabiting species. The undercut bank and overhanging canopy are also used for cover, feeding, and spawning by game species. The shallow water around the islands may serve as an important nursery for fish and aquatic organisms. Yegetation on the islands, especially the large island, may provide protective cover for small mammals, amphibians, reptiles, and songhirds.

About 1 acre of riparian vegetation would be destroyed by construction of the tailrace channel and about 0.3 acre (wooded mwd bank upstream) may be destroyed by construction of the intake channel. Another 0.5 to 1 acre of wellands along the island perimeter and shoreline in the backchannel would be affected by the increased velocities, turbidity, and sedimentation. The islands complex and shallow substrate below the dam are unique and of ecological importance. Because of the scarcity of these welland complexes in this area, the 1-2 acre loss and/or change in species composition and areal extent is considered to be significant and would impact recreational and aesthetic uses of the area.

Text has been modified in reference to "maintenance dredging."

of 1 to 2 acres of "wetlands" would cause a "major" or "significant" impact. This statement is questionable on two counts. First, although riparian vegetation is often associated with wetlands, the majority of the riparian vegetation along the main shore and the island shoreline is not associated with emergent "wetlands". It is emergent wetlands that are of importance to the fishery. Secondly, UMWC is not certain how the status of "significance" was determined, since there was no reference to how much similar habitat is available locally. The shoreline habitat in the channel is not unique to the area, nor is it a major fraction of the available shoreline in the tailwater area.

Finally, the DEIS states that "maintenance dredging" would be responsible for continued "additional" loss of wetland and riparian vegetation. This statement contradicts information which the UMWC has provided in docket 6998, as well as the estimates provided in the DEIS on spoil disposal requirements (Table 4.1.6-8.). Intake and tailrace channels of this type of project are expected to be self sustaining, without the need for maintenance dredging. Table 4.1.6-8 includes the estimate of cubic yards of spoil material for maintenance dredging as zero. Consequently, any anticipated increases in sedimentation and turbidity due to maintenance dredging should no longer be considered as having additional impacts on wetlands/riparian vegetation.

3. Fish Habitat

On p. 4-24 the DEIS asserts that "aquatic habitat changes would be inevitable .. in the scour pool and riffle extending approximately 1000 feet below the dam which would have little flow (1500 cfs minimum spill proposed by applicant) and be partially dewatered."

This statement indicates a poor understanding of river hydraulics. UNHAC has proposed a minimum spillage flow of 1500 cfs, which is more than <u>double</u> the 7010 flow at the site, and close to 5 times the historic low flow. 1500 cfs is indeed in excess of average river flows for weeks in average water years, and months in dry years. Therefore, UNHAC takes issue the statement that "CMISSION flow periods ... to be maintained ... over fixed crests during low-flow periods ... are unlikely to be sufficient to generate currents in the tailwaters comparable to present conditions" (pg. 4-20).

In addition, 1500 cfs is well in excess of the minimum flows proposed by the DEIS for water quality at the site (600 cfs non-critical; 1000 cfs critical, pg. 2-26). The critical and

415

Mitex, Inc.

414

414. Staff considers a 1500 cfs flow at times when the river has significantly greater flow than this upstream of the dam to represent an important loss of habitat in the riffle below the dam. The commentor suggests that staff does not understand river hydraulics, but the comment suggests that the applicant will be providing a minimum flow when that water does not exist in the river. Such a situation cannot be the case unless the applicant is providing upstream storage.

Staff believes the following flow scenario will apply:

- a) At very low river stages, e.g., in the range of historic low flows and 7010 flows, the turbines will not operate and all flow that is present above the dam will pass over the dam (<1500cfs). The applicant cannot alter river flows to maintain a 1500 cfs minimum flow.
- b) At flows >1500 cfs but less than the minimum needed to operate the turbine, there will still be 100 percent passage of river flow over the dam (>1500 cfs).

-] 94
- c) At river flows that are greater than the sum of 1500 cfs and the minimum turbine flow, the turbine will operate and only 1500 cfs will be spilled over the dam. In this flow range, the downstream habitat would normally have the Full flow spread across the riffle habitat but now it has only 1500 cfs. The difference is lost habitat space, some as dewatered shallows, some as shallower areas that still contain water, some as reduced flow velocity in the lower zones where more pool-like conditions begin. With a suitable physical hydraulic model the areas and durations affected could be calculated.
- d) At river flows that are greater than the sum of 1500 cfs and the maximum turbine flow, a spill greater than 1500 cfs will again occur. The differences in habitat in the riffle below the dan between full river flow and the flow allowed over the dam after subtracting the turbine flow could be calculated if there was sufficient information such as a physical hydraulic model.
- e) The annual loss of habitat in the zone extending primarily about 1000 ft downstream from the dam would be th sum of losses in c) and d) above. Staff did not have sufficient information to conduct such a quantitative analysis and was obliged to make more general statements.
- f) flow differences in reaches downstream of the riffle and islands will be ameliorated as the turbine discharge fans out and fills the channel. The extent to which this fanning will occur upstream of the navigation channel is uncertain.

This scenario is consistent with the staff's statement quoted from the DEIS $p,\ 4\text{-}20$.

415. The recommendations for Muskingum L&D No. 3 have been modified (Section 5.4.1) to include the spill flows recommended by USFWS for protection of mussels, plus any additional spill required for maintenance of DO concentrations. These spill flows would apply only if adequate mitigation was developed to allow the project to be recommended; staff believes that such mitigation has not yet been determined.

non-critical minimum flows identified for the project in the Upper Ohio River Basin were determined to meet (1) a "no effects" DO concentration that, if maintained, would not limit the survival, growth, or distribution of aquatic organisms; and (2) a spillage at dama sufficient to prevent DO concentrations from being below this "no effects" concentration as a result of hydropower generation.

For estimating flow regimes of tailwaters the DEIS uses "...available site-specific hydrographic data obtained from the applicants and the Corps, aerial photographs of the tailwater area, and general knowledge from developed sites" (pg. 4-22). UMHC is not certain how the staff arrives at the conclusion that with the proposed minimum flow "1000 feet below the dam would be partially dewatered." UMHC did not submit any references that may have indicated such a conclusion, nor are we aware of any in the DEIS references. As identified above, the 1500 cfs minimum flow proposed by the applicant exceeds natural flow quite often, and therefore, UMHC argues against the belief that this stretch of the streambed would be "dewatered". Most biologists agree that habitat is usually limited by the minimum flow events, which would not change under the project.

Finally, it appears that the conclusions reached for impacts to the tailwater fishery are based upon the same incorrect or and misrepresented conditions of downstream habitat, as were the conclusions of impacts to wetlands. The DEIS concludes (pg. 4-24) that "significant, adverse modifications of the tailwater habitat between the Muskingum L&D No. 3 and the navigation channel 2 miles downstream is inevitable ... This modification is expected to be detrimental to fish resources ... ". Furthermore, the DEIS concludes on pg. 5-5 that the construction and operation of the proposed project at Muskingum L&D No. 3 "...would result in significant adverse impacts to the regional resource of shallow, tailwater, aquatic habitats and that there are no adequate means for mitigating these impacts." UMAC argues that the DEIS does not adequately evaluate the region of impact, nor does it properly evaluate the reasonable means of mitigation already proposed by the Applicant, specifically a minimum spill flow which usually exceeds natural summer low flow conditions. In addition, as a run-of-river facility, the flow of water as it is discharged below the large island would obliquely fan out and fill the river channel (on pg. 4-18 the DEIS states that this can occur at one length of the dam). There is no chain of downstream islands which may prevent this, and cause dewatering impacts for 2 miles downstream.

Throughout the evaluation, the DEIS ignores the comments of the ODNR, OEPA, and USFWS, all of whom have formally commented (in docket 6998) that the project as proposed will not lead to a

416. See response to comment #414. Within the operating range of the turbine, the recommended minimum flow of 1520 cfs during July through March would be similar to 235 percent of existing 7010 conditions. This lower flow would occur more frequently with hydropower operation than it has in the past.

Staff used bottom depth and velocity profile data supplied by the applicant to conclude that a drop in river elevation in the zone for which data were supplied would result in dewatering at edges of islands and shoals and shallowing of other depths. The analysis could not be quantitative because the applicant's depth data did not extend over the full downstream reach and there was no relationship established between flow over the dam and river elevation in the riffie.

The comment is correct in stating that some habitat features are determined by the minimum flow events, and that the river's minimum flows (<1500 cfs) will not be affected by hydropower (as described in response to comment #414). However, in addition to the instantaneous minimum, one must consider the duration of lower flows. As described in response to comment #414, there will be extended time periods when flows are approximately 1500 cfs with hydropower compared to the condition without hydropower.

In addition, staff considered the impacts of a major transition on the flow regime in this section of river due to hydropower development. A shift of a large percentage of the flow to the right bank would be likely to result in major riverbed reequilibration, including erosion and resedimentation of the riverbed, not just the islands. Mussel populations, some of which are on federal and state lists, may not tolerate such riverbed instability.

417. See response to comments #412, 414, and 416.

418. FERC is obliged by NEPA to conduct an independent analysis which embodies the concerns raised by agencies in the scoping process. Responses to comments #412, 414, 415, and 416 more fully explain staff's reasoning. J-196

416

significant adverse effect on the fisheries. In arguing that the local, independent experts whose explicit mission is to protect the resource, are wrong, we would expect the DEIS to be very well documented, at minimum explaining why these experts are incorrect. No such effort is made.

4) Recreation.

On page 4-46 the DEIS states that the Applicant's Recreation Flan, with minor modifications, is adequate for the site. The DEIS proceeds, however, to conclude (pg 5-25) that the proposed project would have a significant adverse impact on recreation. Absent specific justification for this conclusion, we must assume it is a result of the perceived "major changes in fish habitat quality" (pg. 5-7), extended to mean a reduction in fish available for catching, thus less fishery. If the habitat concerns are alleviated, recreational ones should follow.

UMHC has demonstrated in the license application, supplementary filings, agency comments, and the above discussions that a significant adverse effect should not be anticipated at Muskingum L&D No. 3. As such, there will be no significant impact to recreation.

5) Endangered Species.

The DEIS at numerous points implies or actually states that the proposed project will have a negative impact on the Federally Endangered Pink Mucket Pearly Mussel. Although not directly referenced as a reason to deny the license it is likely such a condition will influence the Commission's ultimate decision. We find this impact conclusion to be incorrect and unsupported by the evidence before the Commission.

On pg. 3-20 the DEIS states that "[I]n recent years, the greatest number of fresh or relatively fresh shells of the pink mucket pearly mussel has been found in the Muskingum L&D No. 3 site at Lowell. Ohio." A close scrutiny of Stanbery's report reveals that in fact only dry or weathered shells have been collected in the vicinity of L&D No. 3. Although shell material was found in this reach of river in a study completed in 1983, the exact locations of where they were found are not described. Also, in other recent studies (Davies, 1963; Stilwell, et. al. 1975; and Rothwell, 1979) no living specimens or fresh shells were encountered. Stansbery's report documents the most recent evidence of pink mucket populations at L&D No. J to be a "badly weathered shell" found in 1980 (likely very old), and a "dry shell" found 25 years ago. 419. The proposed recreational enhancements cannot compensate for the loss of important ecological habitat that could occur from the development of the proposed project. The project site is unique in comparison to the other project sites in the study area due to its natural setting and location on a river managed solely for recreation purposes. The existing state park facilities, the undisturbed setting, and the unique fish and wetland habitats all contribute to the site's high recreational quality. Adequate mitigation for project impacts to the existing recreational quality of the area is not known to staff at this time.

420. See Appendix I for a additional information and analysis on impacts to <u>Lamosilis abrupts</u>, the pink mucket pearly mussel. Staff is obliged to make an independent assessment under NEPA, and it believes its conclusions are valid with available information.

Contrary to the statement on pg. 4-23, the Pink Mucket Pearly Mussel is not found in the tailwaters of L&D J. Given the information presented in Stansbery's report, one can only conclude that Pink Mucket Pearly Mussels <u>may</u> inhabit the tailwaters of L&D J. To state that such a population is even <u>likely</u> is carrying this belief much further than either Stansbery or the US Fish and Wildlife Service is willing. If the Commission is to make this assertion, it should be documented with evidence.

Nonetheless, the Project (and its proposed mitigation) has been designed to operate in a way that minimizes impacts <u>should the</u> <u>mussel exist</u>.

The Commission is apparently unaware that the US Fish and Wildlife Service has issued an opinion letter specifically for Pink Muckets at L&D 3 (copy attached). This opinion, which is the FWS' formal decision document under the Endangered Species Act, specifically states that "ISJhould the above conditions be incorporated into the project, this precludes the need for further action on this project as required by the 1973 Endangered Species Act". The FWS has concluded that hydropower at L&D No.3 will not cause an adverse impact to the Pink Mucket Pearly Mussel. The contrary conclusion in the DEIS is thus highly suspect.

6. Alternatives and Implications of the "Proposed Action".

In the "Recommended Action" section, the DEIS proposes that the license for the Muskingum L&D 3 project be denied. The section goes on to generally discuss the implications of denial on the "target resources". The section fails, however, to recognize the implications of license denial on non "target" resources.

If the project is not completed, then an equivalent amount of power must be generated by an alternate means. According to the Alternatives Section (pg. 4-86) the offsetting generation would come from a coal fired plant. While obviously denial of L&D 3 will not force the construction of a 400 MW thermal power plant, it will force an equivalent amount (7 MW) of capacity and energy to be produced at a plant where it otherwise would not. This must be recognized.

If one ratios energy generation of the Muskingum L&D 3 facility against the Alternative Coal Plant used by the Commission (pg. 4-86), the Muskingum Facility would offset the following emissions: 421. The impact to non-target resources are recognized in the EIS (see Section 4.5). It is reasonable to assume that the Muskingum L&D 3 project would eliminate a percentage of the coal fired-emissions presented in Section 4.5 equal to its percentage (2.6 percent) of the annual generation of the proposed projects.

Mitex, Inc.

- 2.3 tons/year of ash:
- 59 tons/year of Sulphur Oxides;
- 114 tons/year of Nitrogen Oxides;
- 8.9 million gallons of water consumptively used in a cooling tower; plus
 Unquantified effects of cooling water
- intakes, coal transportation, sludge disposal, ash disposal, and CO-2 emission.

These benefits should be specifically weighed against any residual impacts associated with project approval.

In conclusion, we find various statements and positions in the DELS regarding the Muskingum LGD 3 Project to be factually incorrect. Conclusions based upon such incorrect information should be critically reviewed, particularly when they are at odds with the prior comments from the state and federal jurisdictional agencies. Any decision should be carefully weighed against the full implications of license denial, including specific emissions offsets.

It is UMWC's position that, considering the full wealth of evidence, the FERC should reach a conclusion that the Muskingum L&D No. 3 project will not contribute to a significant adverse impact to the environment.

The Commission's decision with respect to this project will affect hundreds of thousands of dollars spent to date by UNAC in good faith reliance on the opinions and official comments of independent biological experts, as well as state and federal agencies. If the final EIS is to conclude, even with the factual corrections made in this letter, that license rejection is still warranted, the decision should be very well supported. To do otherwise would be grossly unfair to the Applicant.

Thank you for the opportunity to submit these comments.

Sincerely

William J. Fowler

Project Manager Agent for the UPPER MISSISSIPPI WATER COMPANY

WSF/dc

Enclosure: FWS letter dated April 6, 1987

cc: Jim Richards, Sithe George Taylor, FERC File

- 422. Staff's evaluation of the hydropower project proposed at Nuskingum L&D No. 3, based on the information submitted, has not changed staff's conclusion that such development would cause significant adverse impact to the target resources identified as fisheries, recreation, and wellands. See responses to comments #412-421 for staff's discussion of this information. The Commission will make the final licensing decision on Muskingum L&D No. 3 in the public's interest.
 - 423. Comment noted. See response to comment #422.

The enclosure provided by Mitex, Inc., is not reproduced here.

Mitex, inc.

NATIONAL RENEWABLE RESOURCES, INC. GULF & WESTERN BUILDING 15 COLUMBUS CIRCLE. SUITE #05 NEW YORK, N.Y. 10023 (2121) 245-2721

June 27, 1988

Ms. Lois Cushell Acting Secretary Federal Energy Regulatory Commission 825 North Capitol Street, N.W. Washington, B.C. 20426

Re: <u>Maxwell Lock and Dam Hydroelectric Project #8908</u>

Dear Ms. Cushell:

I am writing on behalf of the applicants, Pennaylvania Renewable Resources, Inc., Nashington County and the Borough of Brownsville to enclose an original and thirtean copies of our Comments on the Draft Environmental Impact Statement issued on May 20, 1988.

If you have any further questions regarding this matter, please give me a call.

Sincerely,

Jeffing Konsah. Jeffing Konsah

JK/mj

CC: Mr. Bill Sember, Washington County Mr. Estel Knisley, Borough of Brownsville

8807100055

Der 6 2 NUL La

Applicants' Draft EIS Comments Maxwell L & D 8908-000

~

Page 5-3 Itan a		
: - 14.	The recommended spill of 500 cfs 400 cfs would be sufficient.	is excessive.

۰.

444

424

.

.

424. Staff's water quality analyses show that a spill flow of 500 cfs is necessary to assure adequate DD concentrations (see Section 4.3.1).

NATIONAL RENEWABLE RESOURCES, INC. GULF & WESTERN BUILDING 15 COLUMBUS CIRCLE, SUITE 900 NEW YORK, N.Y. 10023 (2191 245-3721

F

C. S. L.

£.....

June 27, 1988

Ms. Lois Cushell Acting Secretary Federal Energy Regulatory Commission 825 North Capitol Streat, N.W. Washington, D.C. 20426

Re: Monongahela L4D Hydroelectric Project 44675

Dear Ms. Cushell:

10. 23

I am writing on bahalf of the applicants, Pannsylvania Renavable Resources, Inc., Weshington County and the Borough of Cherleroi, to enclose an original and thirteen copies of our Comments on the Braft Environmental Impact Statement issued on May 20, 1988.

If you have any further questions regarding this matter, please give ms a call.

Sincerely,

Jeffrey Kossak

JK/mj

CC: Mr. Bill Samber, Washington County Mayor McCluckay, Borough of Charleroi

28071017054



Applicants' Diaft EIS Comments Monongahela L i D No. 4 4675-002

Deuteria (198-03

Sec. 2

- Table 2.1.1-3 Page 2-4 Table 5.2-1 Page 5-20
- The draft EIS significantly underestizates 425 energy production. Based on discussions with FERC staff, the number used in the draft EIS assumes that the turbines would be shut down for net heads of less than 10.3 fest. We have confirmed with our engineers and a major turbine supplier that the units would operate at reasonable afficiencies and without undus cavitation or vibration for heads not lower than 6.9 fest. As a result energy production would be increased approximately 3.5 GWA. In addition, FERC staff and applicants appear to have used different tailwater curves. This accounts for an additional 3GWA. Thus energy should be increased 6.5 GWA. The additional energy production would also increase the levelized benefits in the last column of Table 2.1.1-3.
- Page 5-33 Item 9
- The recommended spillage of 500 cfs is 426 excessive. 400 cfs would be sufficient. This would increase energy production.

425. Comment noted.

.

426. Staff's water quality analyses show that a spill flow of 500 cfs is necessary to assure adequate DO concentrations (see Section 4.3.1).

J-203

NATIUNAL RENEWABLE RESOURCES, INC. GULP & WESTERN BUILDING 15 COLUMBUS GIRCLE, SUITE 505 NEW YORK, N.Y. 10035 (SIE) 245-2751

June 27, 1988

Ns. Lois Cushell Acting Secretary Federal Energy Regulatory Commission 825 Horth Capitol Street, H.W. Washington, D.C. 20426

Rei Tygert Dem Hydroelegtrig Project 17307

Dear Ms. Cushell:

I am writing on behalf of the applicant, the City of Grafton, to enclose an original and thirteen copies of our Comments on the Draft Environmental "spect Statement issued on May 20, 1988.

If you have any further questions regarding this matter, places give we a call.

Sincerely, Jeffrey Kossak

JX/mj

co: Kayor John Murray, City of Grafton

8807100056

JUN 2 9 1989

Applicant's Draft 218 Comments

Engineer to the second

Section 4.1.6.1 Page 4-59

- Applicant 7399-000, because of the size of 427 its proposed project, would infringe upon the size of the city of Grefton/Taylor County water treatment plant, located immediately downstream of Tygarc dam, and would likely require relocation of this plant. The sfeet of the project on the water treatment plant and availability of water to firstfon and Taylor County is a major diverse impact. Weither Grafton nor Taylor County have consented to the relocation of their water treatment plant hor has a proposal been made by applicant for the relocation and cost thereof.
- Section 4.1.6.3 Page 4-65 Same comment as above. Table 5.1.1-1 Page 5-3, Table 5.1.2-1 Project 7399-000 should be stated to have significantly greater advaras land use footnote 3 and impacts than compating project 7307. Table 5.1.3.-1 Page 5-15, footnote 3 and Table 5.1.4-1 Page 5-17, footnote 3 Section 5.1.1 Page 5-11
 - The impact on the Grafton/Taylor County water treatment plant should be noted as a major adverse impact.

427. As currently proposed, project No. 7399 at Tygart Dam would not encroach on the site of the City of Grafton/Taylor County water treatment plant. While the project tailrace would be located quite close to the treatment plant, it would not encroach on the city's property, and there has been no indication that it would intefere with operation of the treatment plant.

Mitex, Inc.		A Member of the Sithe/Energies Group	
91 Newbury Street	Boston, Massachusetts 02116	Telephone (\$17) 424-1888 Telecopiar (\$17) 257-7587	
June 28, 1988 Ms. Lois D. Ca Acting Secreta Federal Energy 825 North Capi	shell ry Regulatory Commission tol St. N.E.	האדוכי כד דוב במכה האז את -5 אדוב הבכיב גועהיץ לעוא	
Washington D.C	. 20426	16:50 17:50	

Re: Upper Ohio River Basin Draft Environmental Impact Statement FERC Docket No. EL-85-19-114

Dear Ms. Cashell:

428

Mitex is the general partner of Allegheny Hydro Partners and Allegheny No. 6 Hydro Partners (collectively "the Partnerships"), which are the respective Licensees for the Allegheny River Locks and Dams 5 and 6 Hydroelectric Projects (FERC No. 3671 and 3494). Although these facilities are not included within the decision scope of the referenced DEIS, they are nonetheless mentioned, impacts are assumed, and a conclusion is reached that includes proposed modifications to the operations of the L&D 5 and 6 projects. Thus the Partnerships have a direct interest in this proceeding.

Our concern is primarily with water quality related sections, of the DEIS. The Commission has developed a dissolved oxygen model that means to characterize the Allegheny River from L&D 9 down to the confluence with the Monongahela, and then on down the mainstem Ohio. According to the DEIS (pg. 3-40) the model is based upon spot DO measurements taken by the Corps of Engineers, ORSANCO, and the State of Pennsylvania.

Although we cannot comment on the internal workings of the model, we can comment on the results. The DEIS implies (p. 4-5) that dams 5 and 6 efficiently agrate the Allegheny River, and that hydropower installation will result in elimination of most of this agration effect. The DEIS concludes on page 5-26 that the interim minimum flows at L&D 5 and 6 should be re-evaluated in light of the model's results.

We find these conclusions to be incorrect, and apparently based on a selective portion and small fraction of the amount of data available. Mitex has conducted extensive dissolved oxygen sampling programs at Lock and Dams 4. 5. and 6 for nearly 3 428. Data from FERC licensees or applicants is not always directly cited in the FEIS; however, Sect. 8.2.1.3.1 mentions that data used were from either hydropower applicants or the Corps and Table B-I mentions that the aeration model developed for Allegheny L&D 5 was developed from data from both the Corps and the applicant (or in this case the licensee). The model for Allegheny L&D 6 was developed from Corps data in the DEIS and has been re-evaluated using transect data from the licensee and additional data collected by the Corps in 1988. The revised equation for aeration at Allegheny L&D 6 is: the DO deficit downstream of the dam = 0.82 times the deficit upstream of the dam. This revision of the dam aeration model will not affect results of the impact analyses in Sect. 4 because the low spill flow at Allegheny L&D 6 makes the effects of aeration there minor.

The hourly monitoring data collected by the licensees were not used to evaluate dam aeration at Allegheny LEDs 5 and 6 because data from continuous monitors are not sufficiently accurate for aeration modeling.

The "analysis" of the Corps DO measurements presented in the table accompanying this comment letter is meaningless. Simply averaging the change in DO concentration for all the individual measurements does not take into account the mixing at the dam (measurements must be cross-sectionally averaged before above-dam can be compared with below-dam) and does not take into account the dependence of aeration on the above-dam DO deficit. The linear model of below-dam vs. abovedam DO deficit used in the EIS accounts for mixing and the dependence of aeration on the above-dam deficit. The linear model of predicts that there are situations when deaeration, no aeration, and positive aeration will occur, depending on DO deficits above the dam. Staff recognizes that variability in aeration rates occurs, but believes the DO model is sufficiently accurate to predict impacts of project operations.

The enclosure provided by Mitex. Inc., is not reproduced here.

years. These studies included two years of transect-based measurements, and one year of continuous recording monitors up and downstream of Dams 5 and 6, and upstream of Dam 4. The results of this data collection effort have been continually filed with the Commission (in the Project 3494 and J671 dockets) and made available to agencies. Much of the information was sent directly to the DEIS preparers. Yet not a single reference to any of this data, or the program, is found in the DEIS.

The COE data that apparently provides the DEIS' aeration assumptions for each dam were taken (we understand) over a one hour period, generally once a year. This is less than 1% of the site specific data gathered by the Fartnerships. ORSANCO data is generally not taken both above and below the dams, and State data is very limited.

We would not be as concerned if relying on the Corps data to calibrate results ended up with an accurate model. We cannot, however, come to this conclusion. First, an analysis of the Corps data in our posession (copy enclosed, covering 1976 to 1983) yields an average amount of oxygen uptake at L&D 5 of 0.02 mg/l. At L&D 6 it is 0.03 mg/l. This is absolutely insignificant. We are very skeptical of any model that uses these figures to yield a conclusion of "efficient aeration" at these L&D's.

The data gathered by the Partnerships directly over the past years shows quite mixed results. At some times the dams aerate; at others they deaerate; and at still others, apparently no change to DO occurs. During the summer low flow, DO uptake varies considerably. Under no circumstances, however, can the dams be characterized as consistently "efficient aerators". Any conclusion based upon such a characterization must be flawed.

We finally take serious issue with the recommendation at page 5-26 of the DEIS that interim minimum flows at Allegheny 5 and 6 be reconsidered in light of "the studies conducted for this DEIS". First, we consider the DEIS model seriously flawed due to its limitation on input data, and its seeming incompatibility with observed field data. Second, and perhaps more importantly, the minimum flows at L&D 5 and 6 are being determined based upon a very extensive, very expensive, site specific study. Under no

Mitex, Inc.

circumstances should this study be supplanted by regionalized, very general inferences that do not seem to account for true site specific conditions.

Thank you for the opportunity to submit these comments.

Sincerely, a O 81444 William S. Fowler

William 8.2 Fowler Project Manager

WSF/dc

Enclosure: COE DO data



Blar Mr. Shumway:

CREED & ROTEF (2000) CLE 2007 FULLED FOR

June 24, 1988 ÷ . -. Dean L Shueway -Acting Director, $\sum_{i=1}^{n}$ Division of Project Review Federal Genergy Regulating Condition 71 W 24 Room 9310 825 North Capitol Street N.E. Washington DC. 20426 · • •

SUBJECT:	Comments to DEIS Montgomery Hydroelestric Project 20, 3499	ອີ້ສູ່ JUL 21
he Commission piesse entai impact Statemen	find one original of the t for Potter Township	4H 9:

Attached for filing with the Commission please find one original of the Comments to Draft Environmental Impact Statement for Potter Township Montgomery Hydroelectric Plant.

In the event you or your staff require more copies or have any questions or regarding this submittal, please feel free to contact us.

95 :01 HV⁻ 47 NNT 08 Very truty yours. CREEN INTERNATIONAL AFFILIATES, I XI. Richard A. Volkin, P.E. Executive Vice President

RAV/nk





Ē

CONSULTING ENGINEERS AND TLAT VERS





MONTGOMERY HYDRCELECTRIC PLANT FERC PROJECT NO. 3490-001

COMMENTS TO DEIS

調査 JUNE 1988 登録



GREEN INTERNATIONAL AFF! LATES INC.

Draft Environmental Impact Statement Hydroelectric Development in the Ohio River Basin FERC Dockst No. EL85-19-0114

Subject: Montgumery Hydroelectric Project

After careful review of the DEIS, the comments and evaluations as they relate to the Montgomery Hydroelectric Project, FERC Project No. 3490 as proposed by Potter Township, are not properly addressed as presente in the "Competing Application in Response to Project 2971-002 Application for License, Montgomery Hydroelectric Project" dates February 5, 1956 and all additional information submitted thereto. 429

Key issues that are contrary to the comments and evaluations in the DEIS are:

- 1. The computing application preserves the wetland areas.
- Aeration is utilized in replacement of D.O.
- 3. Recreation area is improved and expanded.
- 4. Fish protection devices incorporated in the program.

It appears that considerations were given to the FERC Project 2971-002 by Allegheny Electric Cooperative, with little to no regard to FERC Project 429. Staff has reviewed the material provided in Attachment C without finding evidence to change any conclusions or recommendations. Staff still concludes that the effectiveness of mechanical aeration at bulb turbines (specifically, the exygen transfer efficiency, which was not addressed in the material provided) is unknown.

Fish entrainment at Montgomery L&D, including both proposed projects, is discussed in Sections 4.1.2.3.5 and 4.1.2.3.6 of the FEIS. Staff concludes that the potential far significant, adverse impact to fish populations remains with this competing application and that the proposed mitigation is unproven in the Ohio River.

The FEIS includes a drawing of the recreational facilities proposed by the applicant (Figure F-16, Appendix F). Although the proposed facilities would improve recreational fishing access at the site, staff still concludes that specific adverse impacts to game fish from turbine-induced mortality would create unmitigable impacts to recreational fishing.

Staff has addressed the impacts of porous dikes on wetlands (Section 4.1.4.3) and has concluded that the level of protection these devices could provide would not be sufficient to adequately protect the embayment area. Staff recommended that the project not be licensed until mitigation measures proposed and designed by the developer to prevent impacts to the wetlands are approved by appropriate state and federal agencies.

The three attachments provided by Green International Associates are not reproduced here.

No. 3490 by Potter Township.

The wetlands abayment area is preserved and fisneries, juvenile and adult are preserved and enhanced by the porous dike profile incorporated in the design.

No construction vis a vis construction laydown area would be incorporated in this area as depicted in the UCIS and in Allgebeny Electric Cooperative Plan. Refer to Attachment A for backup data as it relates to addressing the Wetlands Abayment area.

With Incorporation of the porous dike and the further incorporation of a fishing platform on the downstream area of Eydropower plant, the recreation areas on both sides of the facilities will be enhanced. See Attachment B for support data that was not taken into consideration during the DEIS.

Flow and D.O. considerations, not considerations in the DEIS are detailed in Attachment $C_{\rm e}$

Therefore, with the enhanced and environmentally safe system that is proposed by Potter Township for the Hontgomery Locks and Dam FERC No. 3490 should be considered and approved for licensing. J-212